Studies of coupled nonlinear Schrödinger equations and nonlinear scalar field equations via variational methods

変分法を用いた非線型連立シュレディンガー 方程式系,非線型スカラー場方程式の研究

February 2011

Waseda University Graduate School of Fundamental Science and Engineering Major in Pure and Applied Mathematics Research on Nonlinear Analysis

Norihisa IKOMA

Contents

1	Intr	roduction	1					
	1.1	Part I: Introduction to (CNLS)	1					
		1.1.1 Summary of Chapter 2	2					
		1.1.2 Summary of Chapter 3	5					
		1.1.3 Summary of Chapter 4	6					
	1.2	Part II: Introduction to (NSF)	9					
		1.2.1 Summary of Chapter 5	12					
		1.2.2 Summary of Chapter 6	13					
-	a							
I	Co	oupled nonlinear Schrödinger equations	17					
2	Exi	stence of standing waves for variable coefficient problems	19					
	2.1	Introduction and main result	19					
	2.2	Preliminaries	22					
		2.2.1 Function spaces and functionals	22					
		2.2.2 Nehari manifold and Nehari type manifold	23					
		2.2.3 $(PS)_c$ sequence $\ldots \ldots \ldots$	30					
	2.3	Semitrivial solutions	33					
	2.4	Achievements of $b_{\mathcal{N}}, b_{\mathcal{N}_{\infty}} \dots \dots$	33					
	2.5	Proof of Theorem 1.1 (ii) (when β is large)	36					
	2.6	Proofs of Theorem 2.1.1(i) and Theorem 2.1.2. (when β is small)	37					
		2.6.1 Proof of Theorem 2.1.1(i). \ldots \ldots \ldots \ldots \ldots	37					
		2.6.2 Proofs of Propositions 2.6.1 and 2.6.3.	39					
		2.6.3 Proof of Theorem 2.1.2	43					
	2.7	Proof of Theorem 2.1.4.	44					
3	Uniqueness of nontrivial positive solutions							
	3.1	Introduction	47					
	3.2	A priori bounds for nontrivial positive solutions	49					
		3.2.1 L^{∞} bound of nontrivial positive solutions	49					
		3.2.2 Uniform exponential decay estimates in $S_{\bar{\beta}}$	50					
	3.3	Proof of Theorems 3.1.1, 3.1.4 and 3.1.6	55					
		3.3.1 Nondegeneracy of solutions when $\beta = 0$	55					
		3.3.2 Proof of Theorems 3.1.1, 3.1 4 and 3.1.6	56					

3.4	Symmetry and monotonicity of nontrivial positive solutions $(N = 1)$	57
Exis	stence of concentration solutions	61
4.1	Introduction	61
4.2	Constant coefficient problems	66
	4.2.1 Preliminaries	66
	4.2.2 Ambrosetti and Colorado's condition	67
	4.2.3 Minimizing property	71
	4.2.4 Some compactness properties	74
4.3	Nehari manifolds and the Palais-Smale condition	77
	4.3.1 A singular perturbation problem	77
	4.3.2 Nehari type manifold $\mathcal{M}_{\varepsilon}$ and a projection $\mathcal{P}_{\varepsilon}: \mathcal{S}_{\varepsilon}^{\delta} \to \mathcal{M}_{\varepsilon}$	80
	4.3.3 The Palais-Smale condition in $\mathcal{S}^{\delta}_{\varepsilon} \cap \mathcal{M}_{\varepsilon}$	82
4.4	An estimate of $I_{\varepsilon}(U)$ on $\mathcal{S}^{\delta}_{\varepsilon} \cap \mathcal{M}_{\varepsilon}$	86
	 3.4 Exi 4.1 4.2 4.3 4.4 	3.4Symmetry and monotonicity of nontrivial positive solutions $(N = 1) \dots$ Existence of concentration solutions4.14.1Introduction4.2Constant coefficient problems4.2.1Preliminaries4.2.2Ambrosetti and Colorado's condition4.2.3Minimizing property4.2.4Some compactness properties4.3Nehari manifolds and the Palais-Smale condition4.3.1A singular perturbation problem4.3.2Nehari type manifold $\mathcal{M}_{\varepsilon}$ and a projection $\mathcal{P}_{\varepsilon}: S_{\varepsilon}^{\delta} \to \mathcal{M}_{\varepsilon}$ 4.4An estimate of $I_{\varepsilon}(U)$ on $S_{\varepsilon}^{\delta} \cap \mathcal{M}_{\varepsilon}$

95

II Nonlinear scalar field equations

5	Exis	stence	of positive and infinitely many solutions: homogeneous case 97
	5.1	Introd	uction
	5.2	Prelim	inaries
		5.2.1	Modification of g
		5.2.2	Fundamental properties of $H^1_r(\mathbf{R}^N)$
		5.2.3	A comparison functional J
	5.3	Minim	ax arguments
	5.4	Functi	onal $\tilde{\tilde{I}}(\theta, u)$
	5.5	Bound	ledness and compactness of (θ_i, u_i)
	5.6	Least of	energy solutions
	5.7	Proof	of Lemma 5.6.1
6	Exis	stence	of positive and infinitely many solutions: inhomogeneous case119
	6.1	Introd	uction \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 119
	6.2	Staten	nent of main results $\ldots \ldots 120$
		6.2.1	Results for the equation $(6.1.1)$
		6.2.2	Results for the equation $(6.1.2)$
	6.3	Prelim	inaries $\ldots \ldots \ldots$
	6.4	Minim	$ax arguments \dots \dots$
	6.5	proofs	of Theorems 6.2.1, 6.2.2 and 6.2.4
		6.5.1	Monotonicity method
		6.5.2	Pohozaev type inequality
		6.5.3	Proof of Theorem 6.2.1
		6.5.4	Outline of proof of Theorem 6.2.2
		6.5.5	Proof of Theorem 6.2.4
	6.6	Proofs	of Proposition 6.4.1(ii) and Lemma 6.3.2, and technical lemma $\ .$ 138
		6.6.1	Proof of Proposition 6.4.1(ii)
		6.6.2	Proof of Lemma 6.3.2

		6.6.3	A technical len	nma			 	 	•	 	•	• •	 •	• •	. 1	43
7	App	endix													1_{4}	45
	7.1	Assum	ptions and mai	n sta	tement	5.	 	 		 					. 1	45
	7.2	Proof	of Theorem 7.1	.1			 	 	•	 	•	• •	 •	• •	. 1	46
Li	st of	origin	al papers												1!	59

Acknowledgement

I would like to express my sincere gratitude to my supervisor Professor Kazunaga Tanaka for leading me to variational problems and continuously discussing with me. I could not have finished writing this present thesis without his kind advice and supports.

I am also deeply grateful to Professor Mitsuharu Otani, Professor Tohru Ozawa and Professor Yoshio Yamada for acting as a referee of my doctoral thesis and their valuable suggestions.

It is my pleasure to appreciate Professor Kazuhiro Kurata and Professor Hitoshi Ishii very much for their helpful comments and thoughtful suggestions about Lemma 3.2.1.

I would like to extend my sincere thanks to Professor Matthias Hieber, Professor Takaaki Nishida, Professor Yoshihiro Shibata and Professor Masao Yamazaki for their valuable suggestions, comments and encouragements. In particular, Professor Shibata and Professor Hieber organized the exchange program supported by JSPS and DFG, and I could spend fruitful time in Germany due to the program. My thanks also go to Waseda University and Technische Universität Darmstadt for their supports.

I have greatly profited from fruitful comments, discussions and encouragements of my colleagues and friends. Especially, I wish to thank Okihiro Sawada for his helpful supports, discussions and suggestions in Germany and Japan. I also wish to thank;

Yuko Enomoto, Takayuki Kubo, Tohru Wakasa, Norikazu Yamaguchi, Yohei Sato, Jun Hirata, Yuka Naito, Junichi Harada, Hiroyoshi Mitake, Kazuhiro Oeda, Kohei Soga, Daizo Ishikawa, Jiro Iwanaga, Masaya Maeda, Yusuke Maeyama, Tsunekazu Saito, Kiwamu Watanabe, Tomoyuki Kawashita and Yuto Imai.

I sincerely appreciate my old friends, Takuo Taninaka and Atsushi Ishibashi. They always encourage and give me heartfelt advice.

Lastly, I would like to thank my family for their supports.

I have been supported by Grant-in-Aid for JSPS Fellows.

Chapter 1 Introduction

A main subject in this thesis is an analysis of coupled nonlinear Schrödinger equations (CNLS) and nonlinear scalar field equations (NSF) by variational methods. More precisely, we consider the following nonlinear partial differential equations:

(CNLS)
$$\begin{cases} -\varepsilon^{2}\Delta u_{1} + V_{1}(x)u_{1} = \mu_{1}u_{1}^{3} + \beta u_{1}u_{2}^{2} & \text{in } \mathbf{R}^{N}, \\ -\varepsilon^{2}\Delta u_{2} + V_{2}(x)u_{2} = \beta u_{1}^{2}u_{2} + \mu_{2}u_{2}^{3} & \text{in } \mathbf{R}^{N}, \\ u_{1}, u_{2} \in H^{1}(\mathbf{R}^{N}) \end{cases}$$

and

(NSF)
$$\begin{cases} -\Delta u = g(|x|, u) & \text{in } \Omega, \\ u \in H^1(\Omega). \end{cases}$$

Here $V_1, V_2 : \mathbf{R}^N \to \mathbf{R}$ are potential functions, $\varepsilon > 0, \mu_1, \mu_2, \beta \in \mathbf{R}$ constants, $1 \le N \le 3$ in (CNLS), $N \ge 2$ in (NSF), $\Omega \subset \mathbf{R}^N$ and we denote the Sobolev space by $H^1(\Omega)$:

$$H^{1}(\Omega) := \{ u : \mathbf{R}^{N} \to \mathbf{R} : \|u\|_{H^{1}(\Omega)}^{2} := \|\nabla u\|_{L^{2}(\Omega)}^{2} + \|u\|_{L^{2}(\Omega)}^{2} < \infty \}$$

This thesis consists of two parts. In Part I, we treat (CNLS) and in Part II, we consider (NSF). Part I includes from Chapters 2 to 4 and Part II from Chapters 5 to 7. In what follows, we give a summary and main results of this thesis.

1.1 Part I: Introduction to (CNLS)

In Part I, we consider (CNLS). The equation (CNLS) appears when we consider the existence of standing waves of the following time-depend coupled nonlinear Schrödinger equations (TCNLS):

$$(\text{TCNLS}) \quad \begin{cases} i\hbar \frac{\partial \psi_1}{\partial t} + \frac{\hbar^2}{2m} \Delta \psi_1 + \tilde{V}_1(x)\psi_1 + (\mu_1|\psi_1|^2 + \beta|\psi_2|^2)\psi_1 = 0 & \text{in } (0,\infty) \times \mathbf{R}^N, \\ i\hbar \frac{\partial \psi_2}{\partial t} + \frac{\hbar^2}{2m} \Delta \psi_2 + \tilde{V}_2(x)\psi_2 + (\beta|\psi_1|^2 + \mu_2|\psi_2|^2)\psi_2 = 0 & \text{in } (0,\infty) \times \mathbf{R}^N \end{cases}$$

where \hbar, m are positive constants, $\tilde{V}_1, \tilde{V}_2 : \mathbf{R}^N \to \mathbf{R}$ given functions and $\psi_1, \psi_2 : (0, \infty) \times \mathbf{R}^N \to \mathbf{C}$ unknown functions. The standing wave solutions of (TCNLS) are solutions of the form $\psi_j(t, x) = \exp(i\lambda_j t/\hbar)u_j(x)$ (j = 1, 2) where $\lambda_j \in \mathbf{R}$ and $u_j(x)$ is a real valued function. Substituting this form into (TCNLS), then we obtain (CNLS) with $\varepsilon^2 = \hbar^2/2m$ and $V_j(x) = \lambda_j - \tilde{V}_j(x)$ (j = 1, 2).

The equations (CNLS) and (TCNLS) appear in nonlinear optics and the theory of Bose-Einstein condensates. Recently, a lot of researchers have studied (CNLS) and (TC-NLS) not only in physics but also in mathematics. For example, see [36, 39, 44, 45, 81, 83, 89, 102, 111] for physical treatments, [9, 27, 58, 59, 101] for numerical treatments, and [3, 4, 10, 12, 13, 26, 29, 30, 31, 37, 43, 46, 47, 49, 50, 51, 65, 66, 67, 68, 69, 70, 71, 73, 74, 75, 76, 77, 79, 80, 82, 85, 86, 87, 94, 100, 105, 106, 107, 108, 109] for mathematical treatments. See also references therein.

In this thesis, we concentrate on (CNLS) and consider the case $\mu_1, \mu_2 > 0$. The constant β in (CNLS) plays an important role. It stands for the strength of interactions between u_1 and u_2 . We call the interaction *repulsive* if $\beta < 0$ and *attractive* if $\beta > 0$. In the articles mentioned above, they consider both cases, namely the repulsive and the attractive case. Through Chapters 2 – 4, we focus on the attractive case, i.e., $\beta > 0$.

We also remark that (CNLS) has a *semitrivial solution*. Here we call solution $u = (u_1, u_2)$ semitrivial solution if u solves (CNLS) and either $u_1 \equiv 0$ or $u_2 \equiv 0$. If $u_2 \equiv 0$, (CNLS) becomes the following scalar nonlinear Schrödinger equation:

(SNLS)
$$-\varepsilon^2 \Delta u_1 + V_1(x)u_1 = \mu_1 u_1^3 \quad \text{in } \mathbf{R}^N$$

Under some suitable conditions for $V_1(x)$, (SNLS) has a nontrivial solution. We refer to [2, 5, 8, 11, 21, 22, 23, 24, 32, 33, 34, 38, 42, 56, 63, 93, 103, 104] for more precise results. Therefore, under some suitable conditions, (CNLS) has a semitrivial solution.

On the other hand, we call $u = (u_1, u_2)$ nontrivial solution if u solves (CNLS) and $u_1, u_2 \not\equiv 0$. Furthermore, if both of u_1 and u_2 are positive function, then we say it as nontrivial positive solution. In this thesis, we are interested in the existence of nontrivial positive solutions of (CNLS).

From now on, we suppose that $\mu_1, \mu_2, \beta > 0$. In the following, we state a summary of from Chapter 2 to Chapter 4.

1.1.1 Summary of Chapter 2

Chapter 2 is based on a work of [50]. In Chapter 2, we study (CNLS) in the following setting: $\varepsilon = 1$. Namely we consider the following equations:

(1.1.1)
$$\begin{cases} -\Delta u_1 + V_1(x)u_1 = \mu_1 u_1^3 + \beta u_1 u_2^2 & \text{in } \mathbf{R}^N, \\ -\Delta u_2 + V_2(x)u_2 = \beta u_1^2 u_2 + \mu_2 u_2^3 & \text{in } \mathbf{R}^N, \\ u_1, u_2 \in H^1(\mathbf{R}^N). \end{cases}$$

First we state known results in this setting. The most simple case is that $V_j(x) \equiv V_j > 0$ (j = 1, 2), namely all coefficients are positive constants. Many researchers have considered this case and we refer to [3, 4, 10, 12, 13, 26, 29, 30, 31, 46, 47, 65, 66, 75, 76, 94, 107]. In particular, Ambrosetti and Colorado [3, 4], Lin and Wei [65], and Sirakov [94] showed that for N = 2, 3, there exist positive constants $0 < \tilde{\beta}_1 \leq \tilde{\beta}_2 < \infty$ such that

- (i) If $0 < \beta < \tilde{\beta}_1$, then (1.1.1) has a nontrivial positive solution.
- (ii) If $\tilde{\beta}_2 < \beta$, then (1.1.1) has a nontrivial positive solution.

Moreover, if $V_1(x) \equiv V_2(x) \equiv V > 0$, then Sirakov [94] showed that $\tilde{\beta}_1 = \min\{\mu_1, \mu_2\}$, $\tilde{\beta}_2 = \max\{\mu_1, \mu_2\}$ and there is no nontrivial positive solution of (1.1.1) if $\tilde{\beta}_1 \leq \beta \leq \tilde{\beta}_2$ and $\tilde{\beta}_1 < \tilde{\beta}_2$. A similar result is also obtained in Bartsch and Wang [12]. In these articles, they used essentially the compactness of the embedding $H_r^1(\mathbf{R}^N) \subset L^p(\mathbf{R}^N)$ for $2 if <math>N \geq 3$ and for 2 if <math>N = 2. Here we denote $H_r^1(\mathbf{R}^N)$ as the space of all radially symmetric functions in $H^1(\mathbf{R}^N)$. This compactness property plays an important role to show the existence of nontrivial positive solutions.

In contrast with constant coefficients case, the existence problem becomes delicate if $V_1(x)$ or $V_2(x)$ is not equal to a positive constant identically. In fact, we will prove the following nonexistence result.

Theorem 1.1.1 (Theorem 2.1.4 in Chapter 2). Let N = 1, 2, 3 and V_1, V_2 satisfy the following conditions:

- (2–V1') For $j = 1, 2, V_j \in C^1(\mathbf{R}^N, \mathbf{R})$ and $\nabla V_j \in L^{\infty}(\mathbf{R}^N, \mathbf{R}^N)$.
- (2-V2') For $j = 1, 2, 0 < \inf_{x \in \mathbf{R}^N} V_j(x) \le \sup_{x \in \mathbf{R}^N} V_j(x) < \infty$.
- (2-V3') There exists a $\nu \in \mathbf{R}^N \setminus \{0\}$ such that $(\partial V_j / \partial \nu)(x) = \nabla V_j(x) \cdot \nu \ge 0$ for j = 1, 2.

(2–V4') It holds that either $\partial V_1/\partial \nu \neq 0$ or $\partial V_2/\partial \nu \neq 0$.

Then (1.1.1) has no nontrivial positive solution for any $\beta > 0$.

We note that if $V_1(x)$ and $V_2(x)$ satisfy the conditions (2-V1')-(2-V4'), then $V_1(\varepsilon x)$ and $V_2(\varepsilon x)$ also satisfy (2-V1')-(2-V4'). Therefore from Theorem 1.1.1, we see that (CNLS) has no nontrivial positive solution for any $\beta > 0$ under (2-V1')-(2-V4').

Theorem 1.1.1 suggests that even if graphs of V_1 and V_2 are close to positive constants, (1.1.1) does not have any nontrivial positive solution. Therefore, if the coefficients in (1.1.1) depend on the space variable $x \in \mathbf{R}^N$, the existence problem may turn out to be delicate.

Next we recall the result of G. Wei [105]. He considered (1.1.1) under the conditions: For j = 1, 2,

(1.1.2)
$$V_j \in C^{\infty}(\mathbf{R}^N), \quad 0 < V_{0,j} \le V_j(x) \quad \text{for all } x \in \mathbf{R}^N, \\ \max\{x \in \mathbf{R}^N : V_j(x) < M\} < \infty \quad \text{for all } M > 0.$$

The condition (1.1.2) appeared in Bartsch and Wang [11] and this is a generalization of the condition in Rabinowitz [93]. Under these conditions, G. Wei [105] proved that for N = 1, 2, 3, there exists a $\tilde{\beta} > 0$ such that if $\tilde{\beta} < \beta$, then (1.1.1) has a nontrivial positive solution. His method also depends on the compactness of the embedding. Indeed, in [105], he worked in the following space instead of the usual $H^1(\mathbf{R}^N)$ space:

$$\mathcal{H} := \left\{ u = (u_1, u_2) \in (H^1(\mathbf{R}^N))^2 : \int_{\mathbf{R}^N} V_1(x) u_1^2(x) dx < \infty, \ \int_{\mathbf{R}^N} V_2(x) u_2^2(x) dx < \infty \right\}.$$

In this case, the embedding $\mathcal{H} \subset (L^p(\mathbf{R}^N))^2$ is compact for $2 \leq p < 2N/(N-2)$ if $N \geq 3$ and $2 \leq p < \infty$ if N = 1, 2. Therefore the situation is similar to ones in the constant coefficient case.

Now we explain our setting in Chapter 2. We will consider (1.1.1) under the following conditions:

(2-V1) For
$$j = 1, 2, V_j(x) \in C^1(\mathbf{R}^N, \mathbf{R}).$$

(2-V2) For $j = 1, 2, 0 < \inf_{x \in \mathbf{R}^N} V_j(x) \le \sup_{x \in \mathbf{R}^N} V_j(x) \equiv V_{\infty,j} < \infty.$

(2–V3) For $j = 1, 2, V_j(x) \to V_{\infty,j}$ as $|x| \to \infty$.

In this setting we shall prove the following existence result.

Theorem 1.1.2 (Theorem 2.1.1 in Chapter 2). Suppose that N = 1, 2, 3 and V_1, V_2 satisfy (2-V1)-(2-V3). Then there exist $0 < \beta_1 \leq \beta_2$ such that

- (i) If $0 < \beta < \beta_1$, then (1.1.1) has a nontrivial positive solution.
- (ii) If $\beta_2 < \beta$, then (1.1.1) has a nontrivial positive solution.

In Chapter 2, we will also show the characterization of solutions found in Theorem 1.1.2. In order to state a result, we need the following notion.

Definition 1.1.3. The solution $u = (u_1, u_2)$ is said a *least energy solution* of (1.1.1) if u satisfies the following equality:

(1.1.3) $I(u) := \inf\{I(v) : v \text{ is a solution of } (1.1.1) \text{ and } v \neq (0,0)\}$

where

$$\begin{split} I(u) := & \frac{1}{2} \int_{\mathbf{R}^N} |\nabla u_1|^2 + V_1(x) u_1^2 dx + |\nabla u_2|^2 + V_2(x) u_2^2 dx \\ & - \frac{1}{4} \int_{\mathbf{R}^N} \mu_1 u_1^4 + 2\beta u_1^2 u_2^2 + \mu_2 u_2^4 dx \in C^1((H^1(\mathbf{R}^N))^2, \mathbf{R}). \end{split}$$

We remark that the functional I corresponds to (1.1.1), which means that solutions of (1.1.1) are equivalent to critical points of I. Therefore if the minimizer of (1.1.3) exists, then the minimizer has the least energy among all solutions have except for the trivial solution (0,0).

The following theorem gives a characterization of solutions found in Theorem 1.1.2.

Theorem 1.1.4 (Theorem 2.1.2 in Chapter 2). Suppose that V_1 and V_2 satisfy (2–V1)–(2–V3).

- (i) There exists a $\beta_3 > 0$ such that if $\beta \in (0, \beta_3)$, then the nontrivial positive solution obtained in Theorem 1.1.2 (i) is not a least energy solution.
- (ii) If $\beta_2 < \beta$, then the nontrivial positive solution obtained in Theorem 1.1.2 (ii) is a least energy solution.

1.1.2 Summary of Chapter 3

Chapter 3 is based on a work of [49]. In Chapter 3, we observe a uniqueness of nontrivial positive solutions of (1.1.1). As mentioned in subsection 1.1.1, the existence of positive solution is well-studied. However, the uniqueness of nontrivial positive solutions of (1.1.1) is not studied very well even for the constant coefficients case. We will split into the following two cases:

- (I) constant coefficients case $(V_j(x) \equiv V_j > 0, j = 1, 2)$ or
- (II) either $V_1(x)$ or $V_2(x)$ depends on $x \in \mathbf{R}^N$.

More precisely, in the case (II), we assume the following conditions:

(3-V1) For
$$j = 1, 2, V_j \in C^2(\mathbf{R}^N), V_j(x) = V_j(|x|)$$
 and $\limsup_{|x|\to\infty} V_j(x) > 0.$

(3–V2) For j = 1, 2,

$$\inf\left\{\int_{\mathbf{R}^N} |\nabla u|^2 + V_j(x)u^2 dx: \|u\|_{H^1(\mathbf{R}^N)} = 1\right\} > 0.$$

- (3–V3) For j = 1, 2 and $r \ge 0, V'_j(r) \ge 0$.
- (3-V4) There exist C > 0 and M > 0 such that $|V_j(r)| \leq C(1+r)^M$ for j = 1, 2 and $r \geq 0$.
- (3–V5) When N = 3, the function

$$H_j(r) := \frac{4}{3}r^2 V_j(r) + r^3 V'(r) - \frac{4}{27}$$

has a unique simple zero in $(0, \infty)$.

First we give a remark about symmetry property of nontrivial positive solutions of (1.1.1).

Remark 1.1.5. (i) The conditions (3-V1)-(3-V5) include the constant coefficient case, namely, $V_1(x) \equiv V_1 > 0$, $V_2(x) \equiv V_2 > 0$ satisfy (3-V1)-(3-V5).

(ii) It is easy to see that $V_j(x) = |x|^{\alpha}$ with $\alpha \ge 2$ satisfies (3–V1)–(3–V5). Therefore, the conditions (3–V1)–(3–V5) include unbounded potentials.

(iii) Under the conditions (3–V1), (3–V3) and $N \ge 2$, by the result of Busca and Sirakov [20], any nontrivial positive solution of (1.1.1) is radially symmetric with respect to some point in \mathbf{R}^N . Furthermore, it is nonincreasing with respect to r = |x|. However, this symmetry property holds even for N = 1. We will prove it at section 3.4 in Chapter 3.

By (ii) in Remark 1.1.5, it is sufficient to consider the uniqueness of nontrivial positive solutions which are radially symmetric.

Before stating main results in Chapter 3, we need the following notions.

Definition 1.1.6.

(i) For V_1 and V_2 satisfying (3–V1)–(3–V5), we define $\mathcal{H}_{V_1,V_2,r} \subset H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N)$ as follows:

$$\mathcal{H}_{V_1,V_2,r} := \left\{ u = (u_1, u_2) \in (H_r^1(\mathbf{R}^N))^2 : \int_{\mathbf{R}^N} V_1(x) u_1^2 dx < \infty, \ \int_{\mathbf{R}^N} V_2(x) u_2^2 dx < \infty \right\}$$

(ii) A solution $\omega = (\omega_1, \omega_2) \in \mathcal{H}_{V_1, V_2, r}$ is said nondegenerate in $\mathcal{H}_{V_1, V_2, r}$ if the linearized equation of (1.1.1) at ω

$$\begin{cases} -\Delta u_1 + V_1(x)u_1 = 3\mu_1\omega_1^2u_1 + \beta\omega_2^2u_1 + 2\beta\omega_1\omega_2u_2 & \text{in } \mathbf{R}^N, \\ -\Delta u_2 + V_2(x)u_2 = 2\beta\omega_1\omega_2u_1 + \beta\omega_1^2u_2 + 3\mu_2\omega_2^2u_2 & \text{in } \mathbf{R}^N, \\ (u_1, u_2) \in \mathcal{H}_{V_1, V_2, r} \end{cases}$$

has only trivial solution $u_1 \equiv u_2 \equiv 0$.

First we consider the case (I) (constant coefficient case).

Theorem 1.1.7 (Theorem 3.1.1 in Chapter 3). Suppose that N = 1, 2, 3 and $V_j(x) \equiv V_j > 0$ for j = 1, 2. Then there exists a $\beta_1 > 0$ such that if $0 < \beta < \beta_1$, then the nontrivial positive solution of (1.1.1) is unique up to translations. Furthermore, the unique nontrivial positive solution of (1.1.1) is nondegenerate in $H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N)$ for $0 < \beta < \beta_1$.

Here we should mention a work of Wei and Yao [107]. In [107], they obtained a similar result to Theorem 1.1.7.

Next we consider the case (II) (variable coefficient case). In this case, a uniqueness result is the following:

Theorem 1.1.8 (Theorem 3.1.4 in Chapter 3). Suppose that N = 2, 3 and $V_1(x), V_2(x)$ satisfy the conditions (3-V1)-(3-V5). Then there exists a $\beta_2 > 0$ such that if $0 < \beta < \beta_2$, then (1.1.1) has a unique nontrivial radially symmetric positive solution in $\mathcal{H}_{V_1,V_2,r}$. Furthermore, the unique nontrivial positive solution of (1.1.1) is nondegenerate in $\mathcal{H}_{V_1,V_2,r}$.

1.1.3 Summary of Chapter 4

Chapter 4 is based on a work of [51]. In Chapter 4, we treat (CNLS), and observe the existence of nontrivial positive solutions and the asymptotic behavior as $\varepsilon \to 0$. Throughout in Chapter 4, we assume N = 2, 3.

In general, there are 4 kinds of asymptotic behaviors of a family of solutions of (CNLS). Let $u_{\varepsilon} = (u_{\varepsilon,1}, u_{\varepsilon,2}) \neq (0,0)$ be a family of solution of (CNLS). Then

- (I) The function $u_{\varepsilon,1}$ concentrates at a critical point $P_1 \in \mathbf{R}^N$ of $V_1(x)$ and $u_{\varepsilon,2}$ converges to 0.
- (II) The function $u_{\varepsilon,2}$ concentrates at a critical point $P_2 \in \mathbf{R}^N$ of $V_2(x)$ and $u_{\varepsilon,1}$ converges to 0.
- (III) The function $u_{\varepsilon,j}$ concentrates at a critical point $P_j \in \mathbf{R}^N$ respectively and $P_1 \neq P_2$.

(IV) Both of components concentrate to the same point.

The aim of Chapter 4 is to show the existence of a family of solutions which is type (IV).

As to (I) or (II), we can easily construct such a family of solutions of (CNLS) from the scalar nonlinear Schrödinger equation (SNLS). Indeed, many researchers have studied (SNLS). We refer to [2, 5, 21, 22, 23, 24, 32, 33, 34, 38, 42, 56, 63, 103, 104] and references therein. In these articles, it is shown that under the suitable assumptions, (SNLS) has a family of solutions $(u_{\varepsilon,j})$ concentrating to a critical point P_j . Setting $u_{\varepsilon} := (u_{\varepsilon,1}, 0)$ or $u_{\varepsilon} := (0, u_{\varepsilon,2})$, we can obtain a family of solutions of (CNLS) which is type (I) or (II).

Before stating known results concerning the existence result of type (III) and (IV), we state a main result of Chapter 4. After the main result, we explain known results.

The aim of Chapter 4 is to show the existence of a family of solutions which is type (IV). In order to state the main result, we prepare some notations.

First, to analyze the asymptotic behavior of (CNLS), it is important to study the following constant coefficient problem: For $P \in \mathbf{R}^N$,

(1.1.4)
$$\begin{cases} -\Delta v_1 + V_1(P)v_1 = \mu_1 v_1^3 + \beta v_1 v_2^2 & \text{in } \mathbf{R}^N, \\ -\Delta v_2 + V_2(P)v_2 = \beta v_1^2 v_2 + \mu_2 v_2^3 & \text{in } \mathbf{R}^N, \\ v_1, v_2 \in H^1(\mathbf{R}^N). \end{cases}$$

We define a functional J_P corresponding to (1.1.4) and the least energy m(P) among nontrivial solutions as follows:

$$J_{V_1(P),V_2(P)}(v) := \frac{1}{2} \int_{\mathbf{R}^N} |\nabla v_1|^2 + V_1(P)v_1^2 + |\nabla v_2|^2 + V_2(P)v_2^2 dx$$

$$- \frac{1}{4} \int_{\mathbf{R}^N} \mu_1 v_1^4 + 2\beta v_1^2 v_2^2 + \mu_2 v_2^4 dx,$$

$$\mathcal{M}(V_1(P), V_2(P)) := \{ v \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N) : v_1, v_2 \neq 0,$$

$$J'_{V_1(P),V_2(P)}(v)(v_1, 0) = J'_{V_1(P),V_2(P)}(v)(0, v_2) = 0 \},$$

$$m(P) := \inf_{v \in \mathcal{M}(V_1(P),V_2(P))} J_{V_1(P),V_2(P)}(v).$$

Next, we state our settings in Chapter 4. First we assume

$$(1.1.5) 0 < \beta < \sqrt{\mu_1 \mu_2}.$$

Secondly, we suppose the following two conditions:

Assumption (4–A1) There exists a set $A = [a_{10}, a_{11}] \times [a_{20}, a_{21}] \subset (0, \infty) \times (0, \infty)$ with the following properties:

(i) For any $(\lambda_1, \lambda_2) \in A$, it holds that operators $-\Delta + \lambda_1 - \beta \hat{\omega}_2^2$ and $-\Delta + \lambda_2 - \beta \hat{\omega}_1^2$ are positive definite on $H_r^1(\mathbf{R}^N)$. This means that

$$\int_{\mathbf{R}^N} |\nabla \varphi|^2 + \lambda_1 \varphi^2 - \beta \hat{\omega}_2^2 \varphi^2 dx, \quad \int_{\mathbf{R}^N} |\nabla \varphi|^2 + \lambda_2 \varphi^2 - \beta \hat{\omega}_1^2 \varphi^2 dx > 0$$

for all $\varphi \in H^1_r(\mathbf{R}^N) \setminus \{0\}$. Here $\hat{\omega}_i \in H^1_r(\mathbf{R}^N)$ is a unique positive solution of $-\Delta \omega_i + \lambda_i \omega_i = \mu_i \omega_i^3$ in \mathbf{R}^N (See Kwong [60]).

(ii) $(V_1(P), V_2(P)) \in A$ for all $P \in \mathbf{R}^N$.

Assumption (4–A2) There exists a bounded open set $\Lambda \subset \mathbb{R}^N$ such that

(1.1.6)
$$\inf_{P \in \Lambda} m(P) < \inf_{P \in \partial \Lambda} m(P).$$

Finally, we define the following value and set:

$$m_0 := \inf_{P \in \Lambda} m(P),$$

$$K := \{P \in \Lambda : m(P) = m_0\}$$

Now we can state our main result.

Theorem 1.1.9 (Theorem 4.1.3 in Chapter 4). Let $V_1, V_2 \in C(\mathbf{R}^N)$ and suppose that (1.1.5) and Assumptions (4–A1), (4–A2) hold. Then there exists an $\varepsilon_0 > 0$ such that (CNLS) has a family of nontrivial positive solutions $(u_{1\varepsilon}(x), u_{2\varepsilon}(x))_{0<\varepsilon<\varepsilon_0}$ satisfying the following properties: after taking a subsequence $\varepsilon_j \to 0$ there exists a sequence $(P_{\varepsilon_j}) \subset \Lambda$ such that

$$(1.1.7) P_{\varepsilon_i} \to P_0 \in K,$$

(1.1.8)
$$(u_{1\varepsilon_j}(\varepsilon_j x + P_{\varepsilon_j}), u_{2\varepsilon_j}(\varepsilon_j x + P_{\varepsilon_j})) \to (w_1(x), w_2(x))$$

strongly in $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$

Here $(w_1(x), w_2(x))$ is a nontrivial radial positive solution of the limit problem:

$$\begin{cases} -\Delta w_1 + V_1(P_0)w_1 = \mu_1 w_1^3 + \beta w_1 w_2^2 & \text{in } \mathbf{R}^N, \\ -\Delta w_2 + V_2(P_0)w_2 = \beta w_1^2 w_2 + \mu_2 w_2^3 & \text{in } \mathbf{R}^N \end{cases}$$

and it satisfies $J_{V_1(P_0),V_2(P_0)}(w_1,w_2) = m(P_0) = m_0$.

Theorem 1.1.9 suggests that if the conditions (1.1.5), (4–A1) and (4–A2) hold, then there exists a family of solutions which is type (IV). About when the condition (4–A2) is satisfied, see Remark 4.1.5 (i). By Remark 4.1.5, we can see that there are many examples in which the condition (4–A2) is satisfied.

Next we compare our results to known results. As to the existence of a family of solutions which is type (III) or (IV), we refer to Lin and Wei [68], Montefusco, Pellacci and Squassina [79], Pomponio [87] and G. Wei [105, 106]. Here we only state the results of Lin and Wei [68].

Lin and Wei [68] studied the existence of least energy solutions among nontrivial solutions and its asymptotic behavior. More precisely, we define

$$\begin{split} I_{\varepsilon}(u) &:= \frac{1}{2} \int_{\mathbf{R}^{N}} \varepsilon^{2} |\nabla u_{1}|^{2} + V_{1}(x)u_{1}^{2} + \varepsilon^{2} |\nabla u_{2}|^{2} + V_{2}(x)u_{2}^{2}dx \\ &\quad - \frac{1}{4} \int_{\mathbf{R}^{N}} \mu_{1}u_{1}^{4} + 2\beta u_{1}^{2}u_{2}^{2} + \mu_{2}u_{2}^{4}dx, \\ \mathcal{H} &:= \left\{ u \in (H^{1}(\mathbf{R}^{N}))^{2} : \int_{\mathbf{R}^{N}} V_{1}(x)u_{1}^{2}(x)dx < \infty, \ \int_{\mathbf{R}^{N}} V_{2}(x)u_{2}^{2}(x)dx < \infty \right\}, \\ \mathcal{M}_{\varepsilon} &:= \left\{ u \in \mathcal{H} : u_{1}, u_{2} \neq 0, \ I_{\varepsilon}'(u)(u_{1}, 0) = I_{\varepsilon}'(u)(0, u_{2}) = 0 \right\}, \\ b_{\varepsilon} &:= \inf_{u \in \mathcal{M}_{\varepsilon}} I_{\varepsilon}(u). \end{split}$$

In [68], they proved that b_{ε} is attained by nontrivial positive solution u_{ε} and studied its behavior under $0 < \beta < \beta_0$ and some conditions concerning behaviors of $V_1(x)$ and $V_2(x)$ as $|x| \to \infty$. Especially, their result about asymptotic behaviors is the following: Let $(u_{\varepsilon}) \subset \mathcal{M}_{\varepsilon}$ be a family of minimizer. Then they proved that

$$I_{\varepsilon}(u_{\varepsilon}) = b_{\varepsilon} \to \min\left\{\inf_{P \in \mathbf{R}^{N}} m(P), \inf_{P_{1} \in \mathbf{R}^{N}} e_{1}(P_{1}) + \inf_{P_{2} \in \mathbf{R}^{N}} e_{2}(P_{2})\right\} \quad \text{as } \varepsilon \to 0.$$

Here $e_i(P_i)$ stands for the least energy value for $-\Delta w_i + V_i(P_i)w_i = \mu_i w_i^3$ in \mathbf{R}^N and it has the following formula:

$$e_i(P_i) = \frac{V_i(P_i)^{(4-N)/2}}{\mu_i}e_0$$

where $e_0 > 0$ is the least energy value for $-\Delta w + w = w^3$ in \mathbb{R}^N . Moreover, Lin and Wei showed that if

$$\inf_{P \in \mathbf{R}^N} m(P) < \inf_{P_1 \in \mathbf{R}^N} e_1(P_1) + \inf_{P_2 \in \mathbf{R}^N} e_2(P_2)$$

then the behavior of u_{ε} is type (IV) and a concentration point of both components of u_{ε} is a global minimum point P_0 of m(P). On the other hand, if the opposite inequality

$$\inf_{P_1 \in \mathbf{R}^N} e_1(P_1) + \inf_{P_2 \in \mathbf{R}^N} e_2(P_2) < \inf_{P \in \mathbf{R}^N} m(P)$$

holds, then (u_{ε}) is type (III), which means that each component concentrates at P_i respectively and $P_1 \neq P_2$. Here P_i is a global minimum point of $e_i(P)$.

We remark that we can construct the following example. The function m(P) has a global minimizer, however, the minimizer on $\mathcal{M}_{\varepsilon}$ is type (III), which means each component of the minimizer concentrates at different point. On the other hand, applying our theorem 1.1.9, we can find a family of solution which concentrates at a global minimum point of m. This is a type (IV) solution.

1.2 Part II: Introduction to (NSF)

In Part II, we treat nonlinear scalar field equations (NSF). The equation (NSF) appears in various research fields.

First we consider the following nonlinear Klein–Gordon equation:

(1.2.1)
$$\psi_{tt} - \Delta \psi + m^2 \psi - f(\psi) = 0 \quad \text{in } \mathbf{R} \times \mathbf{R}^N$$

where m > 0. We assume that f satisfies $f(e^{i\theta}s) = e^{i\theta}f(s)$ for all $\theta, s \in \mathbf{R}$. We look for a standing wave solution, which is a form of $\psi(t,s) = e^{i\omega t}u(x)$ where $\omega \ge 0$ and u is a real valued function. Substitute this form into (1.2.1), then we obtain

$$-\Delta u + (m^2 - \omega^2)u = f(u) \quad \text{in } \mathbf{R}^N$$

This is a special case of (NSF) with $g(r,s) = -(m^2 - \omega^2)s + f(s)$ and $\Omega = \mathbb{R}^N$.

Secondly, we consider a travelling wave solution of (1.2.1). This is a solution which has the following form: $\psi(t, x) = u(x - ct)$ where u is a real valued function and $c \in \mathbf{R}^N$. Then (1.2.1) is reduced to the following equation:

(1.2.2)
$$-\sum_{i,j=1}^{N} (\delta_{ij} - c_i c_j) \frac{\partial^2 u}{\partial x_i \partial x_j} + m^2 u = f(u) \quad \text{in } \mathbf{R}^N.$$

We note that the equation (1.2.2) is elliptic provided |c| < 1. In fact, let $A = (A_{ij})_{ij} = (\delta_{ij} - c_i c_j)_{ij}$, then we have

$$\xi \cdot A\xi = |\xi|^2 - |c \cdot \xi|^2 \ge (1 - |c|^2)|\xi|^2$$
 for all $\xi \in \mathbf{R}^N$.

Thus (1.2.2) is elliptic. Using the change of coordinates, (1.2.2) is equivalent to

$$-\Delta u + \tilde{m}u = \tilde{f}(u) \quad \text{in } \mathbf{R}^N.$$

Thus this equation is also a special case of (NSF).

We can also derive (NSF) from the following nonlinear Schrödinger equation:

(1.2.3)
$$i\frac{\partial\psi}{\partial t} + \Delta\psi + V(|x|)\psi + f(\psi) = 0 \quad \text{in } (0,\infty) \times \mathbf{R}^N$$

where $f(e^{i\theta}s) = e^{i\theta}f(s)$ for all $\theta, s \in \mathbf{R}$. When we look for a standing wave solution of the form $\psi(t,s) = \exp(i\omega t)u(x)$, then (1.2.3) becomes

$$-\Delta u + (\omega - V(|x|))u = f(u) \quad \text{in } \mathbf{R}^N.$$

This is (NSF) with $g(r,s) = -(\omega - V(r))s + f(s)$.

Thus the equation (NSF) is a generalization of the nonlinear Klein–Gordon equations, the nonlinear Schrödinger equations and so on.

The equation (NSF) has been extensively studied by many authors [7, 11, 14, 15, 16, 17, 18, 19, 28, 55, 61, 62, 95].

As to (NSF) with $\Omega = \mathbb{R}^N$ and g(r, s) = g(s), Strauss [95] showed the existence of at least one radial positive solution for $N \ge 2$. He also treated the existence of infinitely many radial possibly sign changing solutions. Colemann, Glaser and Martin [28] studied the existence of a least energy solution for $N \ge 3$. Berestycki, Gallouët and Kavian [14] (N = 2) and Berestycki and Lions [15, 16] $(N \ge 3)$ gave an almost necessary and sufficient conditions for the existence of nontrivial solutions. In [14, 15, 16], they assume the following conditions for g(s):

(5-g0) The function $g \in C(\mathbf{R}, \mathbf{R})$ and g is odd: $g(-\xi) = -g(\xi)$. (5-g1) For $N \ge 3$,

$$\limsup_{\xi \to \infty} \frac{g(\xi)}{\xi^{(N+2)/(N-2)}} \le 0.$$

For N = 2,

$$\limsup_{\xi \to \infty} \frac{g(\xi)}{e^{\alpha \xi^2}} \le 0 \quad \text{for any } \alpha > 0.$$

(5-g2) For $N \ge 3$,

(1.2.4)
$$-\infty < \liminf_{\xi \to 0} \frac{g(\xi)}{\xi} \le \limsup_{\xi \to 0} \frac{g(\xi)}{\xi} < 0.$$

For N = 2,

(1.2.5)
$$-\infty < \lim_{\xi \to 0} \frac{g(\xi)}{\xi} < 0.$$

(5-g3) There exists a $\zeta_0 > 0$ such that $G(\zeta_0) > 0$, where $G(\xi) \equiv \int_0^{\xi} g(\tau) d\tau$.

Under the conditions (5-g1)–(5-g3), they proved the existence of at least one radial positive solution and infinitely many radial possibly sign changing radial solutions.

Brezis and Lieb [18] $(N \ge 2)$ and Brüning [19] (N = 2) considered the existence at least one positive solution in not only scalar case but also vector case.

In [55], Jeanjean and Tanaka studied the relationship between a mountain pass solution and a least energy solution of (NSF). They proved that a mountain pass solution is actually a least energy solution under (5-g1)-(5-g3).

Next we consider the case where g(r, s) depends on r. First, Li [62] studied such a problem. He treated (NSF) with $N \ge 3$ and $\Omega = \mathbb{R}^N$ or $\Omega = \{x \in \mathbb{R}^N : |x| > R\}$. In the case where $\Omega = \{x \in \mathbb{R}^N : |x| > R\}$, he consider (NSF) under the Dirichlet boundary condition. In [62], it is shown that (NSF) has at least one radial positive solution and infinitely many radial possibly sign changing solutions under some conditions of g. In particular, they assumed that g(r, s) is a monotone function with respect to r. In [61], Li and Li treated (NSF) in the case N = 2, and proved the same result.

On the other hand, Azzollini and Pomponio [7] considered the following nonlinear Schrödinger equation:

(1.2.6)
$$-\Delta u + V(|x|)u = \tilde{g}(u) \quad \text{in } \mathbf{R}^{N}$$

where $N \ge 3$ and \tilde{g} satisfies (5-g1)–(5-g3). In [7], they showed the existence of at least one positive solution without the monotonicity condition about V(r), but they assumed

$$\left\| (x \cdot \nabla V(|x|))^+ \right\|_{L^{\frac{N}{2}}(\mathbf{R}^N)} < 2S_N$$

where

$$(x \cdot \nabla V(|x|))^{+} := \max\{0, x \cdot \nabla V(|x|)\} \quad \text{and} \quad S_{N} := \inf_{u \in H^{1}(\mathbf{R}^{N}) \setminus \{0\}} \frac{\|\nabla u\|_{L^{2}(\mathbf{R}^{N})}^{2}}{\|u\|_{L^{2^{*}}(\mathbf{R}^{N})}^{2}}$$

Part II consists of three chapters. In Chapter 5, we treat (NSF) with $\Omega = \mathbf{R}^N$ and g(r,s) = g(s). Chapter 6 is devoted to (NSF) where g(r,s) does depend on r and $\Omega = \mathbf{R}^N$ or $\Omega = \{x \in \mathbf{R}^N : |x| > R\}$. Chapter 7 is an appendix of Chapters 5 and 6, and we prove that a sequence of minimax values defined in Chapters 5 and 6 tend to infinity.

1.2.1 Summary of Chapter 5

Chapter 5 is based on a work of [48]. In Chapter 5, we consider the following equation:

(1.2.7)
$$\begin{cases} -\Delta u = g(u) & \text{in } \mathbf{R}^N \\ u \in H^1(\mathbf{R}^N). \end{cases}$$

Next we state settings in Chapter 5. We suppose the conditions (5-g0), (5-g1) and (5-g3). Moreover, we assume

(5-g2')
$$-\infty < \liminf_{\xi \to 0} \frac{g(\xi)}{\xi} \le \limsup_{\xi \to 0} \frac{g(\xi)}{\xi} < 0.$$

We remark that there is a slight difference between the case $N \ge 3$ and N = 2 in [14, 15, 16] (cf. (1.2.4) and (1.2.5)).

Now we state our main result in Chapter 5.

Theorem 1.2.1 (Theorem 5.1.3 in Chapter 5). Assume $N \ge 2$ and (5-g0), (5-g1), (5-g3) and (5-g2'). Then (1.2.7) has a least energy positive solution and infinitely many radially symmetric (possibly sign changing) solutions, which are characterized by the mountain pass and symmetric mountain pass minimax arguments in $H_r^1(\mathbf{R}^N)$.

As mentioned above, Theorem 1.2.1 is a slight extension of results in [14]. Furthermore, our idea to prove Theorem 1.2.1 is different from ones in [14, 15, 16]. Indeed, in [14, 15, 16], they considered the following problem:

• $(N \ge 3)$ Find critical points in $H^1_r(\mathbf{R}^N)$ of

$$\int_{\mathbf{R}^N} |\nabla u|^2 dx \qquad \text{subject to } \int_{\mathbf{R}^N} G(u) dx = 1$$

or

• (N = 2) Find critical points in $H^1_r(\mathbf{R}^2)$ of

$$\int_{\mathbf{R}^2} |\nabla u|^2 dx \quad \text{subject to } \int_{\mathbf{R}^2} G(u) dx \ge 0 \quad \text{and} \quad \|u\|_{L^2(\mathbf{R}^2)} = 1.$$

If we find a critical point v(x) of the above problems, then there exists a $\lambda > 0$ such that $-\Delta v = \lambda g(v)$ in \mathbb{R}^N . Thus by setting $u(x) = v(x/\sqrt{\lambda})$, u(x) is a solution of (1.2.7).

On the other hand, our approach is the following. We consider an unconstrained functional

$$I(u) := \frac{1}{2} \int_{\mathbf{R}^N} |\nabla u|^2 dx - \int_{\mathbf{R}^N} G(u) dx.$$

Then we shall find critical points of I directly. Our approach is based on symmetric mountain pass arguments in Ambrosetti and Rabinowitz [6] and Rabinowitz [92]. Moreover, in order to obtain bounded Palais–Smale sequences, we use an idea in Jeanjean [53]. Namely, we use the following augmented functional

$$\tilde{I}(\theta, u) := I(u(e^{-\theta}x)) = \frac{e^{(N-2)\theta}}{2} \int_{\mathbf{R}^N} |\nabla u|^2 dx - e^{N\theta} \int_{\mathbf{R}^N} G(u) dx \in C^1(\mathbf{R} \times H^1_r(\mathbf{R}^N), \mathbf{R}).$$

This functional is based on the scale properties and gives us bounded Palais–Smale sequences.

1.2.2 Summary of Chapter 6

Chapter 6 is based on a work of [52] and devoted to study the inhomogeneous case:

(1.2.8)
$$-\Delta u = g(|x|, u) \quad \text{in } \Omega.$$

Here $\Omega = \mathbf{R}^N$ or $\Omega = \{x \in \mathbf{R}^N : |x| > R\}$ and $N \ge 2$. Moreover, if $\Omega = \{x \in \mathbf{R}^N : |x| > R\}$, then we consider (1.2.8) with the Dirichlet or the Neumann boundary condition:

(D)
$$u = 0$$
 on $\partial\Omega$,

(N)
$$\frac{\partial u}{\partial \nu} = 0$$
 on $\partial \Omega$

where ν stands for the outward vector of $\partial \Omega$. Namely, we consider the following equations:

$$\begin{array}{ll} (P_{\mathbf{R}^{N}}) & -\Delta u = g(|x|, u) & \text{in } \mathbf{R}^{N}, & u \in H^{1}(\mathbf{R}^{N}). \\ (P_{D}) & -\Delta u = g(|x|, u) & \text{in } \{|x| > R\}, & u = 0 & \text{on } |x| = R, & u \in H^{1}(\{|x| > R\}). \\ (P_{N}) & -\Delta u = g(|x|, u) & \text{in } \{|x| > R\}, & \frac{\partial u}{\partial \nu} = 0 & \text{on } |x| = R, & u \in H^{1}(\{|x| > R\}). \end{array}$$

For the above problems, we assume the following conditions for g(r, s). In what follows, we regard R = 0 if $\Omega = \mathbf{R}^{N}$.

- (6-g1) $g \in C([R,\infty) \times \mathbf{R}, \mathbf{R})$ and g(r, -s) = -g(r, s) for all $r \ge R$ and $s \in \mathbf{R}$.
- (6-g2) If $R \le r_1 \le r_2 < \infty$ and $s \ge 0$, then $g(r_1, s) \le g(r_2, s)$.
- (6-g3) As $r \to \infty$, $g(r,s) \to g_{\infty}(s)$ in $L^{\infty}_{\text{loc}}(\mathbf{R})$.
- (6-g4) There exists an $m_1 > 0$ such that

$$-\infty < \liminf_{s \to 0} \inf_{r \ge R} \frac{g(r,s)}{s} \le \limsup_{s \to 0} \sup_{r \ge R} \frac{g(r,s)}{s} \le -m_1$$

(6-g5) For $N \ge 3$,

$$\lim_{s \to \infty} \sup_{r \ge R} \frac{|g(r,s)|}{s^{2^*-1}} = 0 \quad \text{where} \quad 2^* = 2N/(N-2).$$

For N = 2,

$$\lim_{s \to \infty} \sup_{r \ge R} \frac{|g(r,s)|}{\exp(\alpha s^2)} = 0 \quad \text{for any } \alpha > 0.$$

(6-g6) There exist $\zeta_0 > 0$ and $R \ge R_0$ such that

$$\inf_{r \ge R_0} G(r, \zeta_0) > 0 \quad \text{where} \quad G(r, s) := \int_0^s g(r, \tau) d\tau$$

For $(P_{\mathbf{R}^N})$ and (P_D) , we do not need the following condition (6-g7), however for the Neumann problem (P_N) , in addition to (6-g1)–(6-g6), we assume

(6-g7) $-\infty < \inf_{s \in \mathbf{R}} G(R, s).$

Now we state one of main results in Chapter 6.

Theorem 1.2.2 (Theorems 6.2.1 and 6.2.2 in Chapter 6). Suppose that $N \ge 2$ and g satisfies (6-g1)–(6-g6). Then

- (i) The problem $(P_{\mathbf{R}^N})$ (resp. (P_D)) has at least one radial positive solution and infinitely many radial possibly sign changing solutions.
- (ii) In addition to (6-g1)–(6-g6), assume (6-g7). Then (P_N) has at least one radial positive solution and infinitely many radial possibly sign changing solutions.

Here we give a remark. Li and Li [61] and Li [62] considered $(P_{\mathbf{R}^N})$ and (P_D) under the similar conditions to (6-g1)–(6-g6). However, they also assume g(r,s) = -s + o(s)as $s \to 0$ uniformly with respect to r. Besides, they did not treat the Neumann problem (P_N) .

In Chapter 6, we also consider the nonlinear Schrödinger equation without the monotonicity condition about g(r, s). Namely, we treat the following equation:

(1.2.9)
$$-\Delta u + V(|x|)u = \tilde{g}(u) \quad \text{in } \Omega.$$

Here $\Omega = \mathbf{R}^N$ or $\Omega = \{x \in \mathbf{R}^N : |x| > R\}$ and $N \ge 3$.

Azzollini and Pomponio [7] studied the case where $\Omega = \mathbf{R}^N$ and showed the existence of at least one radial positive solution. We note that if we set $g(r,s) = -V(r)s + \tilde{g}(s)$, then (NSF) becomes (1.2.9). As in the above, we consider both of the Dirichlet and the Neumann problems when $\Omega = \{x \in \mathbf{R}^N : |x| > R\}$.

For $(P_{\mathbf{R}^N})$, (P_D) and (P_N) with $g(r,s) = -V(r)s + \tilde{g}(s)$, we assume the following conditions:

(6-g8) \tilde{g} satisfies (5-g0)–(5-g3).

(6-g9)
$$-\infty < \inf_{s \in \mathbf{R}} \left(-\frac{1}{2} V(R) s^2 + \tilde{G}(s) \right)$$
 where $\tilde{G}(s) = \int_0^s \tilde{g}(t) dt$.

(6-V1) $V \in C^1([R,\infty))$ and $V(r) \ge 0$ for all $r \ge R$.

(6-V2)
$$\lim_{n \to \infty} V(r) = 0.$$

(6-V3) $||(x \cdot \nabla V(|x|))^+||_{L^{\frac{N}{2}}(|x|>R)} < 2S_N$ where

$$(x \cdot \nabla V(|x|))^+ := \max\{0, x \cdot \nabla V(|x|)\} \quad \text{and} \quad S_N := \inf_{u \in H^1(\mathbf{R}^N) \setminus \{0\}} \frac{\|\nabla u\|_{L^2(\mathbf{R}^N)}^2}{\|u\|_{L^{2^*}(\mathbf{R}^N)}^2}.$$

Now we state a result concerning the nonlinear Schrödinger equations (1.2.9).

Theorem 1.2.3 (Theorem 6.2.4 in Chapter 6). Suppose that $N \ge 3$ and $g(r,s) = -V(r)s + \tilde{g}(s)$ satisfies (6-g8) and (6-V1)-(6-V3). Then the following hold:

- (i) $(P_{\mathbf{R}^N})$ (resp. (P_D)) admits at least one radial positive solution and infinitely many radial possibly sign changing solutions.
- (ii) Assume (6-g9) in addition to (6-g8) and (6-V1)–(6-V3). Then (P_N) admits at least one radial positive solution and infinitely many radial possibly radial sign changing solutions.

In Theorem 1.2.3, we treat both of the Dirichlet problem and the Neumann problem. Furthermore, we establish the existence of infinitely many solutions.

Part I

Coupled nonlinear Schrödinger equations

Chapter 2

Existence of standing waves for variable coefficient problems

2.1 Introduction and main result

In this chapter, we consider the existence of standing waves for (TCNLS) with $\varepsilon = 1$. Namely, we consider

(2.1.1)
$$\begin{cases} i\frac{\partial\psi_1}{\partial t} + \Delta_x\psi_1 + \lambda_1(x)\psi_1 + (\mu_1|\psi_1|^2 + \beta|\psi_2|^2)\psi_1 = 0 & \text{in} \quad (0,\infty) \times \mathbf{R}^N, \\ i\frac{\partial\psi_2}{\partial t} + \Delta_x\psi_2 + \lambda_2(x)\psi_2 + (\beta|\psi_1|^2 + \mu_2|\psi_2|^2)\psi_2 = 0 & \text{in} \quad (0,\infty) \times \mathbf{R}^N, \end{cases}$$

where $\mu_1, \mu_2, \beta > 0$ are constants and the dimension N = 1, 2, 3. The system (2.1.1) appears in many physical problems, especially in the Hartree–Fock theory and nonlinear optics. We refer to [3, 4, 12, 36, 45, 65, 75, 94, 105, 111] and references therein for more physical treatments.

In order to obtain standing waves, we substitute $\psi_j(t,x) = e^{i\tilde{\lambda}_j t} u_j(x)$ into (2.1.1). Then $u_1(x), u_2(x)$ solve

(2.1.2)
$$\begin{cases} -\Delta u_1 + V_1(x)u_1 = \mu_1 u_1^3 + \beta u_1 u_2^2 & \text{in } \mathbf{R}^N, \\ -\Delta u_2 + V_2(x)u_2 = \beta u_1^2 u_2 + \mu_2 u_2^3 & \text{in } \mathbf{R}^N, \\ u_1, u_2 \in H^1(\mathbf{R}^N), \end{cases}$$

where $V_j(x) = \lambda_j - \lambda_j(x)$. In particular we are interested in a nontrivial positive solution of (2.1.2). Here, we say $u = (u_1, u_2)$ is a *nontrivial positive solution* of (2.1.2) if u solves (2.1.2) and both u_1, u_2 are positive in \mathbf{R}^N .

Our aim of this chapter is to study the existence of a nontrivial positive solution for the system with variable coefficients. Our work is motivated by Sirakov [94], and Ambrosetti and Colorado [4]. They consider (2.1.2) in constant coefficient case, which means that $V_j(x) \equiv \text{const.} > 0$. Roughly speaking, they proved that there exist positive constants $\tilde{\beta}_1$ and $\tilde{\beta}_2$ such that if $0 \leq \beta < \tilde{\beta}_1$ or $\tilde{\beta}_2 < \beta$ holds, then (2.1.2) has a nontrivial positive solution. We remark that the existence problem becomes delicate when the coefficient

depends on the space variable x. In Theorem 2.1.4 we give an example even if $V_j(x)$ is very close to constant, (2.1.2) does not have any nontrivial positive solutions.

In this chapter, except for the nonexistence result (Theorem 2.1.4), we assume that $V_i(x)$ satisfies the following conditions:

(2-V1) For
$$j = 1, 2, V_j(x) \in C^1(\mathbf{R}^N, \mathbf{R})$$
.

(2-V2) For
$$j = 1, 2, 0 < \inf_{x \in \mathbf{R}^N} V_j(x) \le \sup_{x \in \mathbf{R}^N} V_j(x) \equiv V_{\infty,j} < \infty.$$

(2–V3) For $j = 1, 2, V_j(x) \to V_{\infty,j}$ as $|x| \to \infty$.

Here we introduce some terminologies. We call $u = (u_1, u_2)$ nontrivial solution if u solves (2.1.2) and $u_1, u_2 \not\equiv 0$. On the other hand, we call u semitrivial solution if u solves (2.1.2) and $u_1 \equiv 0$ or $u_2 \equiv 0$. We remark that if $V_j(x)$ satisfies (2–V1)–(2–V3), then (2.1.2) has a semitrivial solution. Indeed, the equation

$$\begin{cases} -\Delta u_1 + V_1(x)u_1 = \mu_1 u_1^3 & \text{in} \quad \mathbf{R}^N, \\ u_1 \in H^1(\mathbf{R}^N) \end{cases}$$

or

$$\begin{cases} -\Delta u_2 + V_2(x)u_2 = \mu_2 u_2^3 \quad \text{in} \quad \mathbf{R}^N, \\ u_2 \in H^1(\mathbf{R}^N) \end{cases}$$

has a nontrivial solution (for instance, see Willem [110]). Then $u = (u_1, 0)$ or $u = (0, u_2)$ is a semitrivial solution of (2.1.2).

Hereafter, we fix $\mu_1, \mu_2 > 0, V_1(x), V_2(x)$ and consider the range of $\beta > 0$ in which (2.1.2) has a nontrivial positive solution. Here we state the main theorem in this paper.

Theorem 2.1.1. Suppose that N = 1, 2, 3 and $V_j(x)$ satisfies (2-V1)-(2-V3). Then there exist $\beta_1 > 0$ and $\beta_2 > \beta_1$ such that

- (i) If $0 < \beta < \beta_1$, then (2.1.2) has a nontrivial positive solution.
- (ii) If $\beta_2 < \beta$, then (2.1.2) has a nontrivial positive solution.

Next, we consider whether the solutions obtained in Theorem 2.1.1 is a least energy solution or not. We say a solution $u = (u_1, u_2)$ of (2.1.2) is a least energy solution if u satisfies the equality

$$I(u_1, u_2) = \inf \{ I(v_1, v_2) : (v_1, v_2) \not\equiv (0, 0) \text{ solves } (2.1.2) \}.$$

Here, we use notation: for $v = (v_1, v_2) \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$,

$$I(v) = \frac{1}{2} \int_{\mathbf{R}^N} |\nabla v_1|^2 + V_1(x)v_1^2 + |\nabla v_2|^2 + V_2(x)v_2^2 dx - \frac{1}{4} \int_{\mathbf{R}^N} \mu_1 v_1^4 + 2\beta v_1^2 v_2^2 + \mu_2 v_2^4 dx.$$

Theorem 2.1.2. The following hold:

(i) There exists a $\beta_3 \in (0, \beta_2]$ such that if $\beta \in [0, \beta_3)$, then the nontrivial positive solution obtained in Theorem 2.1.1 (i) is not a least energy solution.

(ii) If $\beta > \beta_2$, then the nontrivial positive solution obtained in Theorem 2.1.1 (ii) is a least energy solution. Here β_2 is given in Theorem 2.1.1.

Remark 2.1.3. Ambrosetti and Colorado [4] obtained a nontrivial positive solution of (2.1.2) in the constant coefficient case with the mountain pass argument on the Nehari manifold. When $\beta > 0$ is small, they showed that the nontrivial positive solution of (2.1.2) has a higher energy than the semitrivial positive solutions.

Next, we give the nonexistence result. We assume that $V_j(x)$ satisfies the following conditions:

(2–V1') For j = 1, 2 and $i = 1, ..., N, V_i \in C^1(\mathbf{R}^N, \mathbf{R}), \nabla V_i \in L^{\infty}(\mathbf{R}^N)$.

(2-V2') For $j = 1, 2, 0 < \inf_{x \in \mathbf{R}^N} V_j(x) \le \sup_{x \in \mathbf{R}^N} V_j(x) < \infty$.

(2-V3') There exists a $\nu \in \mathbf{R}^N \setminus \{0\}$ such that $(\partial V_j / \partial \nu)(x) = \nabla V_j(x) \cdot \nu \ge 0$ for j = 1, 2.

(2–V4') It holds that either $\partial V_1/\partial \nu \neq 0$ or $\partial V_2/\partial \nu \neq 0$.

Here we state the nonexistence result.

Theorem 2.1.4. Let $V_j(x)$ satisfy (2–V1')–(2–V4'). Then (2.1.2) has no nontrivial positive solution for any $\beta > 0$.

Remark 2.1.5. There is a function which is close to a constant and satisfies (2-V1')-(2-V4'). For instance, setting $V_j(x) = \varepsilon \arctan(x_1) + \pi$, then $V_j(x)$ satisfies (2-V1')-(2-V4') and (2.1.2) has no nontrivial positive solution for any $\varepsilon \in (0, 2)$. This fact implies that the existence of nontrivial positive solution is a delicate problem and we need some conditions concerning the behavior of $V_j(x)$ at infinity in order to show the existence of .

We prove Theorem 2.1.1 by variational methods. To obtain a nontrivial solution of (2.1.2), we introduce the Nehari manifold \mathcal{N} and the Nehari type manifold \mathcal{M} :

$$\mathcal{N} := \left\{ u \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N) : u \not\equiv (0,0), \ I'(u)u = 0 \right\}, \mathcal{M} := \left\{ u \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N) : u_1, u_2 \not\equiv 0, \ I'(u)(u_1,0) = I'(u)(0,u_2) = 0 \right\}.$$

When $\beta > 0$ is large, which implies the setting of Theorem 2.1.1(ii), a nontrivial solution will be obtained as a minimizer of I on \mathcal{N} (see section 2.5).

When $\beta > 0$ is small, which is dealt in Theorem 2.1.1(i), we will also observe that $\inf_{\mathcal{N}} I$ is also attained. However the minimizer turns out to be a semitrivial function and the Nehari type manifold \mathcal{M} plays a role to find a nontrivial solution. In section 2.2, we will prove that \mathcal{M} is a smooth Hilbert manifold with codimension 2 under the condition $0 < \beta < \sqrt{\mu_1 \mu_2}$ and a nontrivial solution will be obtained as a minimizer of I on \mathcal{M} (see section 2.6).

We remark that for problems with constant coefficients Sirakov [94] introduced manifolds in the space of radially symmetric functions:

$$\mathcal{N}_r := \left\{ u \in H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N) : u \not\equiv (0,0), \ I'(u)u = 0 \right\}, \mathcal{M}_r := \left\{ u \in H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N) : u_1, u_2 \not\equiv 0, \ I'(u)(u_1,0) = I'(u)(0,u_2) = 0 \right\}$$

He obtained a nontrivial solution as a minimizer of I on \mathcal{N}_r (\mathcal{M}_r respectively) when $\beta > 0$ is large ($\beta > 0$ is small respectively). We remark that when $\beta > 0$ is small Ambrosetti and Colorado [4] develops a mountain pass argument in \mathcal{N}_r to find a nontrivial solution. We also remark that in these works, the compactness of the embedding $H^1_r(\mathbf{R}^N) \hookrightarrow L^4(\mathbf{R}^N)$ is very important to get the Palais–Smale condition ((PS) condition).

In our setting, we cannot work in the space of radially symmetric functions and due to noncompactness of the embedding $H^1(\mathbf{R}^N) \hookrightarrow L^4(\mathbf{R}^N)$, the corresponding functional Idoes not satisfy the (PS) condition. To solve this difficulty we will develop a concentration compactness type result and give the estimates of critical value of I.

Finally, we give a mention to a work of Wei [105]. Wei considered (2.1.2) with variable coefficients, but under different conditions of $V_j(x)$ from ours. He considered the case where $V_j(x)$ is smooth, positive and $V_j(x) \to \infty$ as $|x| \to \infty$. The functional I is considered on

$$\mathcal{H} = \left\{ u \in H : \int_{\mathbf{R}^N} V_j(x) u_j^2 \, dx < \infty \quad \text{for} \quad j = 1, 2 \right\}.$$

In this case, the embedding $\mathcal{H} \hookrightarrow L^4(\mathbf{R}^N) \times L^4(\mathbf{R}^N)$ is compact (See Rabinowitz [93], and Bartsch and Wang [11]), which implies that I satisfies the (PS) condition on \mathcal{H} .

This chapter is organized as follows: In sections 2.2 and 2.3, we give some preliminaries: especially we give functional frameworks and introduce our variational settings. In section 2.4, we prove the achievement of $\inf_{\mathcal{N}} I$ for all $\beta > 0$. It is important to determine whether the minimizer is nontrivial or not. In sections 2.5 and 2.6, we give a proof to Theorems 2.1.1 and 2.1.2. In section 2.5, we deal with the case where β is large and it turns out that the minimizer of $\inf_{\mathcal{N}} I$ is a nontrivial solution. In section 2.6, we study the case where β is small. In this case the Nehari type manifold \mathcal{M} plays a role. Moreover we will show that for sufficiently small β , a least energy solution of (2.1.2) is a semitrivial solution. In section 2.7, we prove Theorem 2.1.4.

2.2 Preliminaries

In this section, we prove some preliminary results to prove Theorem 2.1.1.

2.2.1 Function spaces and functionals

We set $H := H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$ and denote elements of H by $u = (u_1, u_2)$. For $u = (u_1, u_2)$ and $v = (v_1, v_2) \in H$, we define inner products and norms in $H^1(\mathbf{R}^N)$ and H as follows:

$$\begin{split} \langle u_{j}, v_{j} \rangle_{j} &:= \int_{\mathbf{R}^{N}} \nabla u_{j} \cdot \nabla v_{j} + V_{j}(x) u_{j} v_{j} \, dx \quad (j = 1, 2), \\ \langle u_{j}, v_{j} \rangle_{\infty, j} &:= \int_{\mathbf{R}^{N}} \nabla u_{j} \cdot \nabla v_{j} + V_{\infty, j} u_{j} v_{j} \, dx \quad (j = 1, 2), \\ \langle u, v \rangle &:= \langle u_{1}, v_{1} \rangle_{1} + \langle u_{2}, v_{2} \rangle_{2}, \qquad \langle u, v \rangle_{\infty} &:= \langle u_{1}, v_{1} \rangle_{\infty, 1} + \langle u_{2}, v_{2} \rangle_{\infty, 2}, \\ \|u_{j}\|_{j}^{2} &:= \langle u_{j}, u_{j} \rangle_{j}, \qquad \|u_{j}\|_{\infty, j}^{2} &:= \langle u_{j}, u_{j} \rangle_{\infty, j} \quad (j = 1, 2), \\ \|u\|^{2} &:= \|u_{1}\|_{1}^{2} + \|u_{2}\|_{2}^{2}, \qquad \|u\|^{2}_{\infty, 2} &:= \|u_{1}\|_{\infty, 1}^{2} + \|u_{2}\|_{\infty, 2}^{2}. \end{split}$$

We remark that $\|\cdot\|_{j}$, $\|\cdot\|_{\infty,j}$ are equivalent to the standard $H^{1}(\mathbf{R}^{N})$ norm under the conditions (2–V1)–(2–V2). We define the functional $I: H \to \mathbf{R}$ as follows:

$$I(u) := \frac{1}{2} \|u\|^2 - \frac{1}{4} \int_{\mathbf{R}^N} \mu_1 u_1^4 + 2\beta u_1^2 u_2^2 + \mu_2 u_2^4 dx.$$

Differentiating I, we have

$$I'(u)v = \langle u, v \rangle - \int_{\mathbf{R}^N} \mu_1 u_1^3 v_1 + \beta u_1 u_2^2 v_1 + \beta u_1^2 u_2 v_2 + \mu_2 u_2^3 v_2 \, dx \quad \text{for all } v \in H.$$

It is easily seen that any critical point of I is a solution of (2.1.2). We also use a notation $\nabla I(u) \in H$, where $\nabla I(u)$ is a unique element satisfying $I'(u)v = \langle \nabla I(u), v \rangle$ for all $v \in H$. We also define the functional $I_{\infty} : H \to \mathbf{R}^N$ as follows:

$$I_{\infty}(u) := \frac{1}{2} \|u\|_{\infty}^{2} - \frac{1}{4} \int_{\mathbf{R}^{N}} \mu_{1} u_{1}^{4} + 2\beta u_{1}^{2} u_{2}^{2} + \mu_{2} u_{2}^{4} dx.$$

The functional I_{∞} corresponds to the problem 'at infinity':

(2.2.1)
$$\begin{cases} -\Delta u_1 + V_{\infty,1}u_1 = \mu_1 u_1^3 + \beta u_1 u_2^2 & \text{in } \mathbf{R}^N, \\ -\Delta u_2 + V_{\infty,2}u_2 = \beta u_1^2 u_2 + \mu_2 u_2^3 & \text{in } \mathbf{R}^N, \\ u_1, u_2 \in H^1(\mathbf{R}^N). \end{cases}$$

Any critical point of I_{∞} is also a solution of (2.2.1).

It is easily seen that the following equalities hold:

$$I'(u)u = ||u||^2 - \mu_1 ||u_1||_{L^4}^4 - 2\beta ||u_1u_2||_{L^2}^2 - \mu_2 ||u_2||_{L^4}^4,$$

$$I'(u)(u_1, 0) = ||u_1||_1^2 - \mu_1 ||u_1||_{L^4}^4 - \beta ||u_1u_2||_{L^2}^2,$$

$$I'(u)(0, u_2) = ||u_2||_2^2 - \beta ||u_1u_2||_{L^2}^2 - \mu_2 ||u_2||_{L^4}^4.$$

2.2.2 Nehari manifold and Nehari type manifold

In this subsection we introduce the Nehari manifold \mathcal{N} and the Nehari type manifold \mathcal{M} and state some properties of \mathcal{N} and \mathcal{M} .

We define $J, J_1, J_2: H \to \mathbf{R}$ as follows:

$$J(u) := I'(u)u, \quad J_1(u) := I'(u)(u_1, 0), \quad J_2(u) := I'(u)(0, u_2).$$

Definition 2.2.1. We define the Nehari manifold \mathcal{N} and the Nehari type manifold \mathcal{M} as follows: $\mathcal{N} := \{u \in H : u \neq (0, 0) \mid J(u) = 0\}$

$$\mathcal{N} := \{ u \in H : u \not\equiv (0,0), \ J(u) = 0 \},\$$
$$\mathcal{M} := \{ u \in H : u_1 \not\equiv 0, \ u_2 \not\equiv 0, \ J_1(u) = J_2(u) = 0 \}$$

We also define \mathcal{N}_{∞} and \mathcal{M}_{∞} corresponding to (2.2.1):

$$\mathcal{N}_{\infty} := \{ u \in H : u \neq (0,0), \ J_{\infty}(u) = 0 \}, \mathcal{M}_{\infty} := \{ u \in H : u_1 \neq 0, \ u_2 \neq 0, \ J_{\infty,1}(u) = J_{\infty,2}(u) = 0 \}.$$

Remark 2.2.2. (i) $\mathcal{M} \subset \mathcal{N}$ and $\mathcal{M}_{\infty} \subset \mathcal{N}_{\infty}$.

- (ii) Except for (0,0), any solution of (2.1.2) belongs to \mathcal{N} .
- (iii) If u is a nontrivial solution of (2.1.2), then $u \in \mathcal{M}$.

Remark 2.2.3. We set $|u| := (|u_1|, |u_2|)$, then the following hold:

- (i) If $u \in \mathcal{N}$, then $|u| \in \mathcal{N}$.
- (ii) If $u \in \mathcal{M}$, then $|u| \in \mathcal{M}$.

Next, we state the fundamental properties of \mathcal{N} and \mathcal{N}_{∞} .

Proposition 2.2.4. The following properties hold:

(i) For each $u \in H$ with $u \not\equiv (0,0)$, there exist unique $\theta_0 > 0$ and $\theta_{\infty,0} > 0$ such that $\theta_0 u \in \mathcal{N}, \ \theta_{\infty,0} u \in \mathcal{N}_{\infty}.$

(ii)
$$I(u) = \frac{1}{4} ||u||^2$$
 on \mathcal{N} , $I_{\infty}(u) = \frac{1}{4} ||u||_{\infty}^2$ on \mathcal{N}_{∞} .

(iii) There exist $\delta_0 > 0$ and $\delta_\infty > 0$ such that

$$||u|| \ge \delta_0 \quad \text{for all } u \in \mathcal{N}, \quad ||v||_{\infty} \ge \delta_{\infty} \quad \text{for all } v \in \mathcal{N}_{\infty}.$$

Proof. We only prove for \mathcal{N} . (i) Suppose that $u \in H, u \not\equiv (0,0)$ and set

$$f(\theta) := I(\theta u) = \frac{\theta^2}{2} \|u\|^2 - \frac{\theta^4}{4} \int_{\mathbf{R}^N} \mu_1 u_1^4 + 2\beta u_1^2 u_2^2 + \mu_2 u_2^4 dx.$$

Then we see

$$f'(\theta) = I'(\theta u)u = \theta \left\{ \|u\|^2 - \theta^2 (\mu_1 \|u_1\|_{L^4}^4 + 2\beta \|u_1 u_2\|_{L^2}^2 + \mu_2 \|u_2\|_{L^4}^4) \right\}.$$

Thus $f'(\theta) = 0$ holds if and only if $\theta = \theta_0$, where

$$\theta_0 := \frac{\|u\|}{\sqrt{\mu_1 \|u_1\|_{L^4}^4 + 2\beta \|u_1 u_2\|_{L^2}^2 + \mu_2 \|u_2\|_{L^4}^4}} > 0.$$

(ii) Let $u \in \mathcal{N}$. Then it follows that

$$||u||^{2} = \mu_{1}||u_{1}||_{L^{4}}^{4} + 2\beta ||u_{1}u_{2}||_{L^{2}}^{2} + \mu_{2}||u_{2}||_{L^{4}}^{4}.$$

From the above equality, we obtain

$$I(u) = \frac{\|u\|^2}{2} - \frac{\|u\|^2}{4} = \frac{\|u\|^2}{4}$$

(iii) Let $u \in \mathcal{N}$. By using Hölder's inequality and Sobolev's embedding, we have

$$\begin{aligned} \|u\|^{2} &= \mu_{1} \|u_{1}\|_{L^{4}}^{4} + 2\beta \|u_{1}u_{2}\|_{L^{2}}^{2} + \mu_{2} \|u_{2}\|_{L^{4}}^{4} \\ &\leq \mu_{1} \|u_{1}\|_{L^{4}}^{4} + 2\beta \|u_{1}\|_{L^{4}}^{2} \|u_{2}\|_{L^{4}}^{2} + \mu_{2} \|u_{2}\|_{L^{4}}^{4} \\ &\leq C(\mu_{1}\|u_{1}\|_{1}^{4} + 2\beta \|u_{1}\|_{1}^{2} \|u_{2}\|_{2}^{2} + \mu_{2} \|u_{2}\|_{2}^{4}) \\ &\leq C(\|u_{1}\|_{1}^{4} + \|u_{2}\|_{2}^{2})^{2} = C \|u\|^{4}. \end{aligned}$$

Therefore it follows that

$$\frac{1}{C} \le \|u\|^2.$$

г		
L		
L		
L		_

Next, we prove that \mathcal{N} and \mathcal{M} are smooth Hilbert manifolds.

Lemma 2.2.5. It holds that

- (i) For each $\beta > 0$, \mathcal{N} and \mathcal{N}_{∞} are smooth Hilbert manifolds with codimension 1.
- (ii) If $0 < \beta < \sqrt{\mu_1 \mu_2}$, then \mathcal{M} and \mathcal{M}_{∞} are smooth Hilbert manifolds with codimension 2.

(iii)
$$T_u \mathcal{N} = \{ v \in H : J'(u)v = 0 \}.$$

(iv)
$$T_u \mathcal{M} = \{ v \in H : J'_1(u)v = J'_2(u)v = 0 \}$$

The above lemma will be derived from the following well-known lemma. For example, see Ambrosetti and Malchiodi [5].

Lemma 2.2.6. Let $O \subset H$ be an open set. Suppose $G, G_1, G_2 \in C^m(O, \mathbf{R})$ and set $M := G^{-1}(0), \tilde{M} := G_1^{-1}(0) \cap G_2^{-1}(0)$. Then the following hold:

- (i) If $G'(p) \neq 0$ for each $p \in M$, then M is a C^m Hilbert manifold with codimension 1.
- (ii) If $G'_1(p)$ and $G'_2(p)$ are linearly independent for each $p \in \tilde{M}$, then \tilde{M} is a C^m Hilbert manifold with codimension 2.
- (iii) $T_p M = \{ q \in H : G'(p)q = 0 \}.$
- (iv) $T_p \tilde{M} = \{q \in H : G'_1(p)q = G'_2(p)q = 0\}.$

We prove Lemma 2.2.5 with the aid of Lemma 2.2.6.

Proof of Lemma 2.2.5. We only prove (i) and (ii) since (iii) and (iv) are directly derived from Lemma 2.2.6.

(i) For $u \in \mathcal{N}$, we have

$$J'(u)u = 2\|u\|^2 - 4(\mu_1\|u_1\|_{L^4}^4 + 2\beta\|u_1u_2\|_{L^2}^2 + \mu_2\|u_2\|_{L^4}^4) = -2\|u\|^2 < 0.$$

In particular, we have $J'(u) \neq 0$ for any $u \in \mathcal{N}$. Thus applying Lemma 2.2.6 to $J : H \setminus \{0\} \to \mathbf{R}$, we have (i) of Lemma 2.2.5.

(ii) Next we apply (ii) of Lemma 2.2.6 to $J_1, J_2 : H \setminus \{u_1 = 0 \text{ or } u_2 = 0\} \to \mathbf{R}$. For $u \in \mathcal{M}$, we have

$$J_1'(u)(u_1,0) = -2\mu_1 ||u_1||_{L^4}^4, \quad J_2'(u)(0,u_2) = -2\mu_2 ||u_2||_{L^4}^4, J_1'(u)(0,u_2) = J_2'(u)(u_1,0) = -2\beta ||u_1u_2||_{L^2}^2.$$

Define A(u) by

$$A(u) := \begin{pmatrix} J_1'(u)(u_1, 0) & J_1'(u)(0, u_2) \\ J_2'(u)(u_1, 0) & J_2'(u)(0, u_2) \end{pmatrix} = \begin{pmatrix} -2\mu_1 \|u_1\|_{L^4}^4 & -2\beta \|u_1 u_2\|_{L^2}^2 \\ -2\beta \|u_1 u_2\|_{L^2}^2 & -2\mu_2 \|u_2\|_{L^4}^4 \end{pmatrix},$$

and we see

$$\det A(u) = 4(\mu_1 \mu_2 \| u_1 \|_{L^4}^4 \| u_2 \|_{L^4}^4 - \beta^2 \| u_1 u_2 \|_{L^2}^4) \ge 4(\mu_1 \mu_2 - \beta^2) \| u_1 \|_{L^4}^4 \| u_2 \|_{L^4}^4 > 0.$$

The above inequality implies that $J'_1(u)$ and $J'_2(u)$ are linearly independent. Thus by Lemma 2.2.6 we infer that \mathcal{M} is a smooth Hilbert manifold with codimension 2.

Lastly we state some properties of the level sets of \mathcal{N} and \mathcal{M} . For each $\alpha > 0$, we define \mathcal{N}^{α} and \mathcal{M}^{α} as follows:

$$\mathcal{N}^{\alpha} := \{ u \in \mathcal{N} : I(u) \le \alpha \}, \quad \mathcal{M}^{\alpha} := \{ u \in \mathcal{M} : I(u) \le \alpha \}.$$

Proposition 2.2.7 (Properties of \mathcal{N}). The following properties hold:

(i) The set \mathcal{N} is a closed subset of H and \mathcal{N}^{α} is a bounded closed subset of H. In particular,

 $0 < \delta_0 \le ||u|| \le 2\sqrt{\alpha}$ for all $u \in \mathcal{N}^{\alpha}$,

where δ_0 is given in Proposition 2.2.4.

(ii) For each $\alpha > 0$, it holds

 $0 < 2\delta_0 \le \|\nabla J(u)\| \le c_1(\alpha)$ for all $u \in \mathcal{N}^{\alpha}$,

where $c_1(\alpha)$ depends on α but not on $u \in \mathcal{N}^{\alpha}$.

Proof. (i) It is clear from Proposition 2.2.4 (ii) and (iii).

(ii) Since $J'(u)u = -2||u||^2$ and $||u|| \ge \delta_0$, we have $2\delta_0 \le ||J'(u)||$. On the other hand, since $J': H \to H^*$ maps bounded sets to bounded sets and \mathcal{N}^{α} is bounded, we infer the conclusion of Proposition 2.2.7.

We define $T_u \mathcal{N}^{\perp}$ and $T_u \mathcal{M}^{\perp}$ as the orthonormal complement of $T_u \mathcal{N}$ and $T_u \mathcal{M}$, respectively:

$$T_{u}\mathcal{N}^{\perp} := \{ v \in H : \langle v, h \rangle = 0 \text{ for } h \in T_{u}\mathcal{N} \},\$$
$$T_{u}\mathcal{M}^{\perp} := \{ v \in H : \langle v, h \rangle = 0 \text{ for } h \in T_{u}\mathcal{M} \}.$$

We also define $P_{T_u \mathcal{N}^{\perp}}$ and $P_{T_u \mathcal{M}^{\perp}}$ as the projections from H to $T_u \mathcal{N}^{\perp}$ and $T_u \mathcal{M}^{\perp}$, respectively:

$$P_{T_u\mathcal{N}^{\perp}}: H \to T_u\mathcal{N}^{\perp}, \quad P_{T_u\mathcal{M}^{\perp}}: H \to T_u\mathcal{M}^{\perp}.$$

By Lemma 2.2.5, we have $T_u \mathcal{N}^{\perp} = \text{span} \{ \nabla J(u) \}$. Thus

$$P_{T_u \mathcal{N}^{\perp}} u = \left\langle \frac{\nabla J(u)}{\|\nabla J(u)\|}, u \right\rangle \frac{\nabla J(u)}{\|\nabla J(u)\|}.$$

By Lemma 2.2.5 and Proposition 2.2.7, we have the following corollary.

Corollary 2.2.8. For each $\alpha > 0$, there holds

$$0 < c_1(\alpha) \le \|P_{T_u \mathcal{N}^\perp} u\| \le c_2(\alpha)$$
 for all $u \in \mathcal{N}^\alpha$,

where $c_1(\alpha)$, $c_2(\alpha)$ are positive constants and depend on α .

Next we state the properties of \mathcal{M} .

Proposition 2.2.9 (Properties of \mathcal{M}). Let $\alpha > 0$.

(i) There exist $\beta_1(\alpha) \in (0, \sqrt{\mu_1 \mu_2}), c_1(\alpha), c_2(\alpha) > 0$ such that for each $\beta \in (0, \beta_1(\alpha))$ and $u \in \mathcal{M}^{\alpha}$,

$$c_1(\alpha) \le ||u_j||_{L^4} \le c_2(\alpha), \ c_1(\alpha) \le ||u_j||_j \le c_2(\alpha), c_1(\alpha) \le ||\nabla J_j(u)|| \le c_2(\alpha) \quad (j = 1, 2).$$

- (ii) If $\beta \in (0, \beta_1(\alpha))$, then \mathcal{M}^{α} is a closed subset of H.
- (iii) There exists an $\varepsilon_1(\alpha) > 0$ such that for each $u \in \mathcal{M}^{\alpha}$ and $\beta \in (0, \beta_1(\alpha))$,

$$|\langle \nabla J_1(u), \nabla J_2(u) \rangle| \le (1 - \varepsilon_1(\alpha)) \|\nabla J_1(u)\| \|\nabla J_2(u)\|.$$

(iv) There exist $c_3(\alpha) > 0$ and $c_4(\alpha) > 0$ such that for each $\beta \in (0, \beta_1(\alpha))$ and $u = (u_1, u_2) \in \mathcal{M}^{\alpha}$,

$$0 < c_3(\alpha) \le ||P_{T_u \mathcal{M}^{\perp}} U_j|| \le c_4(\alpha) \quad (j = 1, 2),$$

where $U_1 = (u_1, 0)$ and $U_2 = (0, u_2)$. Moreover, there exists an $\varepsilon_2(\alpha) > 0$ such that

$$|\langle P_{T_u\mathcal{M}^{\perp}}U_1, P_{T_u\mathcal{M}^{\perp}}U_2\rangle| \le (1 - \varepsilon_2(\alpha)) \|P_{T_u\mathcal{M}^{\perp}}U_1\| \|P_{T_u\mathcal{M}^{\perp}}U_2\|$$

for all $u \in \mathcal{M}^{\alpha}$.

Proof. (i) Since $\mathcal{M}^{\alpha} \subset \mathcal{N}^{\alpha}$, \mathcal{N}^{α} is a bounded set in H and J'_{j} maps bounded sets to bounded sets, it is sufficient to show that

$$0 < c_1(\alpha) \le \|u_j\|_{L^4}, \ 0 < c_1(\alpha) \le \|u_j\|_j, \ 0 < c_1(\alpha) \le \|\nabla J_j(u)\|$$

for each $u \in \mathcal{M}^{\alpha}$. We only show the statements for u_1 and ∇J_1 since the same argument is valid for u_2 and ∇J_2 .

Since

$$||u_1||_1^2 = \mu_1 ||u_1||_{L^4}^4 + \beta ||u_1u_2||_{L^2}^2,$$

using Hölder's inequality and Sobolev's embedding theorem, it follows that

$$||u_1||_{L^4}^2 \le C||u_1||_1^2 \le C(\mu_1||u_1||_{L^4}^4 + \beta ||u_1||_{L^4}^2 ||u_2||_{L^4}^2).$$

This implies that

$$\frac{1}{C} - \beta \|u_2\|_{L^4}^2 \le \mu_1 \|u_1\|_{L^4}^2.$$

Since $||u_j||_j$ are bounded, there exists a $\beta(\alpha) > 0$ such that if $\beta \in (0, \beta(\alpha))$, then

$$0 < c_1(\alpha) \le ||u_1||_{L^4}.$$

By Sobolev's embedding, we have

$$c_1(\alpha) \le ||u_1||_{L^4} \le C ||u_1||_1.$$

Since $J'_1(u)(u_1, 0) = -2\mu_1 ||u_1||^4_{L^4}$, we have $c_1(\alpha) \leq ||\nabla J_1(u)||$. (ii) By (i) and the continuity of $J_j(u)$, it is easy to check that (ii) holds. (iii) Let $u \in \mathcal{M}^{\alpha}$ and set

$$\xi_1 := \frac{\nabla J_1(u)}{\|\nabla J_1(u)\|}, \quad \xi_2 := \frac{\nabla J_2(u)}{\|\nabla J_2(u)\|}, \quad \tilde{\xi}_2 := \xi_2 - \langle \xi_1, \xi_2 \rangle \xi_1, \quad \xi_3 := \frac{\tilde{\xi}_2}{\|\tilde{\xi}_2\|}.$$

Since \mathcal{M}^{α} is bounded and ∇J_1 , ∇J_2 map bounded sets into bounded sets, we only prove that there exists a $c(\alpha) = c > 0$ such that

(2.2.2)
$$0 < c \le \|\tilde{\xi}_2\|^2 \quad \text{for all } u \in \mathcal{M}^{\alpha}.$$

Indeed, since

$$\|\tilde{\xi}_2\|^2 = 1 - \langle \xi_1, \xi_2 \rangle^2 = \frac{\|\nabla J_1(u)\|^2 \|\nabla J_2(u)\|^2 - \langle \nabla J_1(u), \nabla J_2(u) \rangle^2}{\|\nabla J_1(u)\|^2 \|\nabla J_2(u)\|^2}$$

(iii) follows from (2.2.2).

Set $U_1 := (u_1, 0), U_2 := (0, u_2)$ and define A(u) as follows:

$$A(u) := \begin{pmatrix} \langle U_1, \xi_1 \rangle & \langle U_1, \xi_3 \rangle \\ \langle U_2, \xi_1 \rangle & \langle U_2, \xi_3 \rangle \end{pmatrix}.$$
Since

$$\det A(u) = \frac{1}{\|\tilde{\xi}_2\|} \det \begin{pmatrix} \langle U_1, \xi_1 \rangle & \langle U_1, \tilde{\xi}_2 \rangle \\ \langle U_2, \xi_1 \rangle & \langle U_2, \tilde{\xi}_2 \rangle \end{pmatrix} = \frac{1}{\|\tilde{\xi}_2\|} \det \begin{pmatrix} \langle U_1, \xi_1 \rangle & \langle U_1, \xi_2 - \langle \xi_1, \xi_2 \rangle \xi_1 \rangle \\ \langle U_2, \xi_1 \rangle & \langle U_2, \xi_2 - \langle \xi_1, \xi_2 \rangle \xi_1 \rangle \end{pmatrix}$$
$$= \frac{1}{\|\tilde{\xi}_2\|} \det \begin{pmatrix} \langle U_1, \xi_1 \rangle & \langle U_1, \xi_2 \rangle \\ \langle U_2, \xi_1 \rangle & \langle U_2, \xi_2 \rangle \end{pmatrix}$$

and

we have

$$\det A(u) = \frac{4(\mu_1\mu_2 \|u_1\|_{L^4}^4 \|u_2\|_{L^4}^4 - \beta^2 \|u_1u_2\|_{L^2}^4)}{\|\tilde{\xi}_2\| \|\nabla J_1(u)\| \|\nabla J_2(u)\|} \ge \frac{4(\mu_1\mu_2 - \beta^2) \|u_1\|_{L^4}^4 \|u_2\|_{L^4}^4}{\|\tilde{\xi}_2\| \|\nabla J_1(u)\| \|\nabla J_2(u)\|}.$$

By (i) and the assumption of β ,

(2.2.3)
$$\det A(u) \ge \frac{C(\alpha)}{\|\tilde{\xi}_2\|} \quad \text{for all } u \in \mathcal{M}^{\alpha}.$$

On the other hand, the components of A(u) are bounded, which implies that there exists a $C_1 = C_1(\alpha) > 0$ such that

(2.2.4)
$$\det A(u) \le C_1(\alpha) \quad \text{for all } u \in \mathcal{M}^{\alpha}.$$

From (2.2.3) and (2.2.4), there exists a $c = c(\alpha) > 0$ such that

$$0 < c \le \|\tilde{\xi}_2\|$$
 for all $u \in \mathcal{M}^{\alpha}$.

(iv) Since

(2.2.5)
$$P_{T_u\mathcal{M}^{\perp}}U_1 = \langle U_1,\xi_1\rangle\xi_1 + \langle U_1,\xi_3\rangle\xi_3,$$

where ξ_j are given in (iii), it follows that

$$||U_1||^2 \ge ||P_{T_u\mathcal{M}^{\perp}}U_1||^2 = \langle U_1, \xi_1 \rangle^2 + \langle U_1, \xi_3 \rangle^2 \ge \langle U_1, \xi_1 \rangle^2 = \frac{4\mu_1^2 ||u_1||_{L^4}^8}{||\nabla J_1(u)||^2}.$$

By (i), it follows that there exist $c_3(\alpha) > 0$ and $c_4(\alpha) > 0$ such that

(2.2.6)
$$c_3(\alpha) \le \|P_{T_u \mathcal{M}^{\perp}} U_1\| \le c_4(\alpha) \quad \text{for all } u \in \mathcal{M}^{\alpha}.$$

Similarly we have (2.2.6) for U_2 . Since (2.2.5) and

$$P_{T_u\mathcal{M}^{\perp}}U_2 = \langle U_2, \xi_1 \rangle \xi_1 + \langle U_2, \xi_3 \rangle \xi_3,$$

we have

$$\|P_{T_u\mathcal{M}^{\perp}}U_1\|^2 \|P_{T_u\mathcal{M}^{\perp}}U_2\|^2 - |\langle P_{T_u\mathcal{M}^{\perp}}U_1, P_{T_u\mathcal{M}^{\perp}}U_2\rangle|^2 = (\det A(u))^2.$$

By (2.2.3) and the boundedness of $(P_{T_u\mathcal{M}^{\perp}}U_j)$, for sufficiently small $\varepsilon_2(\alpha) > 0$, the conclusion of (iv) holds.

Corollary 2.2.10. (i) Let $u \in \mathcal{N}$ satisfy I'(u)h = 0 for all $h \in T_u \mathcal{N}$. Then I'(u) = 0.

(ii) Let $u \in \mathcal{M}$ satisfy I'(u)h = 0 for all $h \in T_u\mathcal{M}$. Then I'(u) = 0.

Remark 2.2.11. Similar results hold for \mathcal{N}_{∞} and \mathcal{M}_{∞} .

Proof. We only treat (ii) since (i) can be shown similarly. Let $u \in \mathcal{M}$ satisfy I'(u)h = 0for all $h \in T_u\mathcal{M}$. It is sufficient to prove I'(u) = 0 on $T_u\mathcal{M}^{\perp}$. Let $U_1 = (u_1, 0)$ and $U_2 = (0, u_2)$ as in the proof of Proposition 2.2.9. Then from the proof of Proposition 2.2.9 (iv), we see that $T_u\mathcal{M}^{\perp} = \text{span} \{P_{T_u\mathcal{M}^{\perp}}U_1, P_{T_u\mathcal{M}^{\perp}}U_2\}$. Hence we show $I'(u)P_{T_u\mathcal{M}^{\perp}}U_1 =$ $I'(u)P_{T_u\mathcal{M}^{\perp}}U_2 = 0$. However, this is easily follows from the fact that $I'(u)U_1 = I'(u)U_2 =$ 0 and I'(u)h = 0 for all $h \in T_u\mathcal{M}$. Therefore we have I'(u) = 0.

2.2.3 $(PS)_c$ sequence

First, we introduce important values to obtain a nontrivial solution of (2.1.2).

We define $b_{\mathcal{N}}, \hat{b}_{\mathcal{M}}, b_{\mathcal{N}_{\infty}}, \hat{b}_{\mathcal{M}_{\infty}}$ as follows.

$$b_{\mathcal{N}} := \inf_{u \in \mathcal{N}} I(u), \quad \hat{b}_{\mathcal{M}} := \inf_{u \in \mathcal{M}} I(u), \quad b_{\mathcal{N}_{\infty}} := \inf_{u \in \mathcal{N}_{\infty}} I_{\infty}(u), \quad \hat{b}_{\mathcal{M}_{\infty}} := \inf_{u \in \mathcal{M}_{\infty}} I_{\infty}(u).$$

Remark 2.2.12. By Remark 2.2.2, it follows that

$$0 < b_{\mathcal{N}} \le \hat{b}_{\mathcal{M}}, \quad 0 < b_{\mathcal{N}_{\infty}} \le \hat{b}_{\mathcal{M}_{\infty}}.$$

To obtain a solution of (2.1.2), we see that $b_{\mathcal{N}}$ or $\hat{b}_{\mathcal{M}}$ is attained. So it is important to see the behavior of the minimizing sequence on \mathcal{N} or \mathcal{M} .

Definition 2.2.13. Let $c \in \mathbf{R}$.

(i) A sequence $(u_n) \subset H$ is said to be a Palais–Smale sequence of I on H at level c (in short (PS)_{c. H} sequence), if it satisfies

$$I(u_n) \to c, \quad \|I'(u_n)\|_{H^*} \to 0,$$

where

$$||I'(u)||_{H^*} := \sup_{h \in H, ||h||=1} I'(u)h.$$

(ii) A sequence $(u_n) \subset \mathcal{N}$ is said to be a $(PS)_{c,\mathcal{N}}$ sequence of I, if it satisfies

$$I(u_n) \to c, \quad ||I'(u_n)||_{T_{u_n}\mathcal{N}^*} \to 0,$$

where

$$||I'(v)||_{T_v\mathcal{N}^*} := \sup_{h \in T_v\mathcal{N}, ||h||=1} I'(v)h.$$

(iii) Let $\beta < \sqrt{\mu_1 \mu_2}$. A sequence $(u_n) \subset \mathcal{M}$ is said to be a $(PS)_{c, \mathcal{M}}$ sequence of I on \mathcal{M} , if it satisfies

$$I(u_n) \to c, \quad ||I'(u_n)||_{T_{u_n}\mathcal{M}^*} \to 0,$$

where

$$||I'(w)||_{T_w\mathcal{M}^*} := \sup_{h \in T_w\mathcal{M}, ||h||=1} I'(w)h.$$

Next we see the relationships between a $(PS)_{c, H}$, $(PS)_{c, N}$ and $(PS)_{c, M}$ sequence.

Lemma 2.2.14. (i) Any $(PS)_{c, H}$ sequence (u_n) is a bounded sequence on H.

- (ii) $Any (PS)_{c, \mathcal{N}}$ sequence (u_n) is a $(PS)_{c, H}$ sequence.
- (iii) If $c < \alpha$ and $\beta \in (0, \beta_1(\alpha))$, then any $(PS)_{c, \mathcal{M}}$ sequence is a $(PS)_{c, H}$ sequence, where $\beta_1(\alpha)$ appeared in Proposition 2.2.9.

Proof. (i) Let (u_n) be a $(PS)_{c, H}$ sequence. Since $||I(u_n)||_{H^*} \to 0$, there exists an $n_1 \in \mathbb{N}$ such that

$$|I'(u_n)u_n| \le ||u_n|| \quad \text{for all } n \ge n_1.$$

On the other hand, it holds

$$I(u_n) = \frac{1}{2} \|u_n\|^2 - \frac{1}{4} (\mu_1 \|u_1\|_{L^4}^4 + 2\beta \|u_{n,1}u_{n,2}\|_{L^2}^2 + \mu_2 \|u_{n,2}\|_{L^4}^4),$$

$$I'(u_n)u_n = \|u_n\|^2 - (\mu_1 \|u_1\|_{L^4}^4 + 2\beta \|u_{n,1}u_{n,2}\|_{L^2}^2 + \mu_2 \|u_{n,2}\|_{L^4}^4),$$

which implies

$$I(u_n) - \frac{1}{4}I'(u_n)u_n = \frac{1}{4}||u_n||^2.$$

Thus we conclude that for sufficiently large n,

$$\frac{1}{4}||u_n||^2 \le ||u_n|| + c + o(1),$$

which implies that (u_n) is a bounded sequence.

(ii) By $||I'(u_n)||_{T_{u_n}\mathcal{N}^*} \to 0$ and $I(u_n) \to c$, it is sufficient to prove $||I'(u_n)||_{H^*} \to 0$. Since we may assume that $(u_n) \subset \mathcal{N}^{\alpha}$ for some $\alpha > 0$, (u_n) is a bounded sequence. By Lemma 2.2.5, $H = \text{span} \{\nabla J(u_n)\} \oplus T_{u_n}\mathcal{N}$. So we prove that $I'(u_n)\zeta_n \to 0$ where $\zeta_n = \nabla J(u_n)/||\nabla J(u_n)||$ and it is equivalent to

(2.2.7)
$$I'(u_n) \left[\frac{P_{T_{u_n} \mathcal{N}^\perp} u_n}{\|P_{T_{u_n} \mathcal{N}^\perp} u_n\|} \right] \to 0.$$

First, we prove that $I'(u_n)[P_{T_{u_n}\mathcal{N}^{\perp}}u_n] \to 0$. Since $I'(u_n)u_n = J(u_n) = 0$ and $u_n - P_{T_{u_n}\mathcal{N}^{\perp}}u_n \in T_{u_n}\mathcal{N}$, it follows that

$$|I'(u_n)[P_{T_{u_n}\mathcal{N}^{\perp}}u_n]| = |I'(u_n)u_n - I'(u_n)[u_n - P_{T_{u_n}\mathcal{N}^{\perp}}u_n]| = |I'(u_n)[u_n - P_{T_{u_n}\mathcal{N}^{\perp}}u_n]|$$

$$\leq ||I'(u_n)||_{T_{u_n}\mathcal{N}^*}||u_n - P_{T_{u_n}\mathcal{N}^{\perp}}u_n|| \to 0.$$

By Corollary 2.2.8, $(||P_{T_{u_n}\mathcal{N}^{\perp}}u_n||)$ is bounded below away from 0. Thus (2.2.7) holds. (iii) Let (u_n) be a $(PS)_{c, \mathcal{M}}$ sequence and $c < \alpha$. We remark that (u_n) is bounded in H and $(U_{n,j})$ also. As in (ii), by Lemma 2.2.5 and Proposition 2.2.9, we prove that

(2.2.8)
$$I'(u_n)\xi_{n,1} \to 0, \quad I'(u_n)\xi_{n,3} \to 0,$$

where $(\xi_{n,1})$ and $(\xi_{n,3})$ are given in the proof of Proposition 2.2.9. Since $I'(u_n)U_{n,1} = I'(u_n)U_{n,2} = 0$, $U_{n,j} - P_{T_{u_n}\mathcal{M}^{\perp}}U_{n,j} \in T_{u_n}\mathcal{M}$ and $\|I'(u_n)\|_{T_{u_n}\mathcal{M}} \to 0$, we have

$$|I'(u_n)[P_{T_{u_n}\mathcal{M}^{\perp}}U_{n,j}]| = |I'(u_n)U_{n,j} - I'(u_n)[U_{n,j} - P_{T_{u_n}\mathcal{M}^{\perp}}U_{n,j}]|$$

= $|I'(u_n)[U_{n,j} - P_{T_{u_n}\mathcal{M}^{\perp}}U_{n,j}]|$
 $\leq ||I'(u_n)||_{T_{u_n}\mathcal{M}^*}||U_{n,j} - P_{T_{u_n}\mathcal{M}^{\perp}}U_{n,j}|| \to 0.$

By Proposition 2.2.9, $\|P_{T_{u_n}\mathcal{M}^{\perp}}U_{n,j}\|$ are bounded below away from 0, it follows that

$$I'(u_n)\xi_{n,1} \to 0, \quad I'(u_n)\xi_{n,2} \to 0.$$

Using Proposition 2.2.9 again, it follows that $\|\xi_{n,2} - \langle \xi_{n,2}, \xi_{n,1} \rangle \xi_{n,1}\|$ is bounded below away from 0, which implies (2.2.8).

The following lemma tells us that we can obtain a $(PS)_{b_{\mathcal{N}}, H}$ sequence and a $(PS)_{\hat{b}_{\mathcal{M}}, H}$ sequence from the minimizing sequence, respectively.

Lemma 2.2.15. (i) For each $\beta > 0$, there exists a (PS)_{b_N, H} sequence.

(ii) Suppose that $\alpha > \hat{b}_{\mathcal{M}}$ for all $\beta \in (0, \sqrt{\mu_1 \mu_2})$. Then there exists a $0 < \tilde{\beta}(\alpha) \le \sqrt{\mu_1 \mu_2}$ such that if $\beta \in (0, \tilde{\beta}(\alpha))$, then there exists a $(PS)_{\hat{b}_{\mathcal{M}}, H}$ sequence.

Remark 2.2.16. We remark that $\hat{b}_{\mathcal{M}}$ depends on β . In Proposition 2.6.1, we will prove $\sup_{\beta \in [0,\infty)} \hat{b}_{\mathcal{M}} < \infty$. In particular, there exists an α which satisfies the assumption of Lemma 2.2.15 (ii).

We can prove Lemma 2.2.15 by applying Ekeland's variational principle (See Ekeland [35] and Mahwin and Willem [78]). So we omit the proof.

The following lemma is so-called Concentration Compactness Lemma (cf. Lions [72]). This lemma plays an important role in analysing a $(PS)_{c-H}$ sequence.

Lemma 2.2.17 (Concentration Compactness Lemma). Let (u_n) be a $(PS)_{c, H}$ sequence. Then there exist a subsequence (u_{n_k}) , an $\ell \in \mathbf{N}$, a critical point u_0 of I, critical points $\omega_i (1 \leq i \leq \ell)$ of I_{∞} , $(y_k^i) \subset \mathbf{R}^N (1 \leq i \leq \ell)$ satisfying the following:

(i)
$$|y_k^i| \to \infty \ (1 \le i \le \ell), \quad |y_k^i - y_k^j| \to \infty \ (i \ne j).$$

(ii)
$$\left\| u_{n_k} - u_0 - \sum_{i=1}^{\ell} \omega_i (x - y_k^i) \right\| \to 0.$$

(iii)
$$I(u_{n_k}) \to c = I(u_0) + \sum_{i=1}^{\ell} I_{\infty}(\omega_i).$$

See Bahri and Lions [8] and Jeanjean and Tanaka [56] for a proof of Lemma 2.2.17. Remark 2.2.18. If $\ell = 0$ in the above lemma, then u_{n_k} converges to u_0 strongly in H.

2.3 Semitrivial solutions

Here, we consider some properties of semitrivial solutions, i.e., the solution of a form $(u_1, 0)$ or $(0, u_2)$.

The functionals $u_1 \mapsto I(u_1, 0)$ and $u_2 \mapsto I(0, u_2)$ corresponds to

(2.3.1)
$$\begin{cases} -\Delta u_1 + V_1(x)u_1 = \mu_1 u_1^3 & \text{in} \quad \mathbf{R}^N, \\ u_1 \in H^1(\mathbf{R}^N), \end{cases}$$

(2.3.2)
$$\begin{cases} -\Delta u_2 + V_2(x)u_2 = \mu_2 u_2^3 & \text{in } \mathbf{R}^N, \\ u_2 \in H^1(\mathbf{R}^N). \end{cases}$$

We define d_1 , d_2 as the least energy of (2.3.1), (2.3.2), respectively:

$$d_1 := \inf_{(u_1,0)\in\mathcal{N}} I(u_1,0), \quad d_2 := \inf_{(0,u_2)\in\mathcal{N}} I(0,u_2).$$

Similarly, we set

$$d_{\infty,1} := \inf_{(u_1,0)\in\mathcal{N}_{\infty}} I_{\infty}(u_1,0), \quad d_{\infty,2} := \inf_{(0,u_2)\in\mathcal{N}_{\infty}} I_{\infty}(0,u_2)$$

Remark 2.3.1. By the definition of d_j , we have

$$b_{\mathcal{N}} \le \min\{d_1, d_2\}$$

If the inequality (2.3.3) is strict, we can see the critical point corresponding to $b_{\mathcal{N}}$ is nontrivial. We will see in section 2.5 that this is the case when β is large.

The following lemma shows that d_j is attained and (2.1.2) has a semitrivial solution.

Lemma 2.3.2. Suppose N = 1, 2, 3 and V_1, V_2 satisfy (2–V1)–(2–V3). Then,

- (i) The equation (2.3.1) (resp. (2.3.2)) has a least energy solution which is positive in R^N.
- (ii) It holds that $d_j \leq d_{\infty,j}$. Moreover if $V_j(x) \neq V_{\infty,j}$, then $d_j < d_{\infty,j}$.

A proof of Lemma 2.3.2 is standard, so we omit it. For example, see Willem [110].

2.4 Achievements of $b_{\mathcal{N}}, b_{\mathcal{N}_{\infty}}$

In this section, we prove that $b_{\mathcal{N}}$ and $b_{\mathcal{N}_{\infty}}$ are attained for each $\beta > 0$. These facts are useful to prove the existence of nontrivial solutions of (2.1.2) in section 2.5.

First we recall the following result.

Proposition 2.4.1 (Ambrosetti and Colorado [4], and Sirakov [94]). It holds that

(i) For each $\beta > 0$, $b_{\mathcal{N}_{\infty}}$ is attained.

(ii) There exists a $\beta_0 > 0$ such that if $\beta > \beta_0$, then $b_{\mathcal{N}_{\infty}}$ is attained by a nontrivial positive solution of (2.2.1).

This proposition is proved in Ambrosetti and Colorado [4] and Sirakov [94] in the case N = 2, 3. For reader's convenience, we shall give a proof of Theorem 2.4.1 (i) for the case N = 2, 3 as well as the case N = 1. To prove Proposition 2.4.1 (i), we need the Schwarz symmetrization. We denote u^* the Schwarz symmetrization of u:

$$u^* = (u_1^*, u_2^*).$$

It is well-known that the Schwarz symmetrization satisfies the following: (See Lieb and Loss [64])

$$\|u_j^*\|_{L^4} = \|u_j\|_{L^4}, \quad \|\nabla u_j^*\|_{L^2} \le \|\nabla u_j\|_{L^2}, \quad \|u_1^*u_2^*\|_{L^2} \ge \|u_1u_2\|_{L^2}.$$

Proof of Proposition 4.1. We consider two cases.

Case 1: N = 2, 3.

Suppose that $(u_n) \subset \mathcal{N}_{\infty}$ satisfies $I_{\infty}(u_n) \to b_{\mathcal{N}_{\infty}}$. Then Proposition 2.2.4 implies that (u_n) is a bounded sequence. By the above properties of u^* , (u_n^*) is also a bounded sequence. Let $H_r^1(\mathbf{R}^N)$ be the space of radially symmetric functions in $H^1(\mathbf{R}^N)$. Since the embedding $H_r^1(\mathbf{R}^N) \hookrightarrow L^4(\mathbf{R}^N)$ is compact, there exists a subsequence (write still (u_n)) such that

$$u_n^* \rightarrow u_0$$
 weakly in $H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N)$,
 $u_n^* \rightarrow u_0$ strongly in $L^4(\mathbf{R}^N) \times L^4(\mathbf{R}^N)$.

Then it follows that

$$\begin{aligned} \|u_0\|_{\infty}^2 &\leq \liminf_{n \to \infty} \|u_n^*\|_{\infty}^2 \leq \liminf_{n \to \infty} \|u_n\|_{\infty}^2 \\ &= \liminf_{n \to \infty} (\mu_1 \|u_{n,1}\|_{L^4}^4 + 2\beta \|u_{n,1}u_{n,2}\|_{L^2}^2 + \mu_2 \|u_{n,2}\|_{L^4}^4) \\ &\leq \liminf_{n \to \infty} (\mu_1 \|u_{n,1}^*\|_{L^4}^4 + 2\beta \|u_{n,1}^*u_{n,2}^*\|_{L^2}^2 + \mu_2 \|u_{n,2}^*\|_{L^4}^4) \\ &= \mu_1 \|u_{0,1}\|_{L^4}^4 + 2\beta \|u_{0,1}u_{0,2}\|_{L^2}^2 + \mu_2 \|u_{0,2}\|_{L^4}^4. \end{aligned}$$

Hence there exists a unique $\theta_0 \in (0, 1]$ such that $\theta_0 u_0 \in \mathcal{N}_{\infty}$. Thus we see

$$b_{\mathcal{N}_{\infty}} \leq \frac{\theta_0^2}{4} \|u_0\|_{\infty}^2 \leq \liminf_{n \to \infty} \frac{\theta_0^2}{4} \|u_n^*\|_{\infty}^2 \leq \liminf_{n \to \infty} \frac{\theta_0^2}{4} \|u_n\|_{\infty}^2 = \theta_0^2 b_{\mathcal{N}_{\infty}},$$

which implies $\theta_0 = 1$, $u_0 \in \mathcal{N}_{\infty}$ and $I_{\infty}(u_0) = b_{\mathcal{N}_{\infty}}$.

Case 2: N = 1

By Lemma 2.2.15, there exists a $(PS)_{b_{\mathcal{N}}, H}$ sequence. We denote it by (u_n) . Then Proposition 2.2.4 implies that (u_n) is bounded. Furthermore, by Lemma 2.2.17, there exist a subsequence (u_{n_k}) , an $\ell \in \mathbf{N}$, critical points $\omega_i (0 \leq i \leq \ell)$ of I_{∞} and sequences $(y_k^i) \subset \mathbf{R}^N (1 \leq i \leq \ell)$ such that

(2.4.1)
$$\begin{aligned} |y_k^i| \to \infty \quad \text{if } 1 \le i \le \ell, \qquad |y_k^i - y_k^j| \to \infty \quad \text{if } 1 \le i < j \le \ell, \\ \|u_{n_k} - \omega_0 - \sum_{i=1}^{\ell} \omega_i (x - y_k^i)\| \to 0, \quad I_\infty(u_{n_k}) \to \sum_{i=0}^{\ell} I_\infty(\omega_i). \end{aligned}$$

Moreover, we may assume that if $\ell \geq 1$, then $\omega_i \neq 0$ for all $1 \leq i \leq \ell$.

If $\omega_0 \neq 0$, then we can conclude $\ell = 0$ and $I_{\infty}(\omega_0) = b_{\mathcal{N}_{\infty}}$. Indeed, since $\omega_0 \neq 0$, $I_{\infty}(\omega_0) \geq b_{\mathcal{N}_{\infty}}$ holds. Since it holds that $I_{\infty}(\omega_i) \geq b_{\mathcal{N}_{\infty}}$ for $1 \leq i \leq \ell$, it follows from (2.4.1) that $\ell = 0$ and $I_{\infty}(\omega_0) = b_{\mathcal{N}_{\infty}}$.

If $\omega_0 \equiv 0$, then we infer that $\ell = 1$. In fact, if $\ell = 0$, then from (2.4.1) we have $u_{n_k} \to 0$ strongly in H. This contradicts Proposition 2.2.4 (ii). On the other hand, if $\ell \geq 2$, then from (2.4.1) we have $\liminf_{k\to\infty} I_{\infty}(u_{n_k}) \geq 2b_{\mathcal{N}_{\infty}} > b_{\mathcal{N}_{\infty}}$. This is a contradiction and we obtain $\ell = 1$. We set $v_k(x) = u_{n_k}(x + y_k^1)$. Then we have $I_{\infty}(v_k) = I_{\infty}(u_{n_k}), v_k \in \mathcal{N}_{\infty}$ and $v_k \to \omega_1$ strongly in H. Hence we obtain $I_{\infty}(\omega_1) = b_{\mathcal{N}_{\infty}}$, which completes a proof.

Next we prove that $b_{\mathcal{N}}$ is attained.

Proposition 2.4.2. For each $\beta > 0$, $b_{\mathcal{N}}$ is attained.

Proof. First, we prove the inequality $b_{\mathcal{N}} \leq b_{\mathcal{N}_{\infty}}$. By Proposition 2.4.1, there exists a $u_{\infty} \in \mathcal{N}_{\infty}$ such that $I_{\infty}(u_{\infty}) = b_{\mathcal{N}_{\infty}}$. With the assumption of $V_j(x)$ we obtain

$$||u_{\infty}||^{2} \leq ||u_{\infty}||_{\infty}^{2} = \mu_{1}||u_{\infty,1}||_{L^{4}}^{4} + 2\beta ||u_{\infty,1}u_{\infty,2}||_{L^{2}}^{2} + \mu_{2}||u_{\infty,2}||_{L^{4}}^{4},$$

which implies that there exists a $\theta_{\infty} \in (0, 1]$ such that $\theta_{\infty} u_{\infty} \in \mathcal{N}$. Then it follows that

(2.4.2)
$$b_{\mathcal{N}} \le I(\theta_{\infty}u_{\infty}) = \frac{\theta_{\infty}^2}{4} ||u_{\infty}||^2 \le \frac{1}{4} ||u_{\infty}||_{\infty}^2 = I_{\infty}(u_{\infty}) = b_{\mathcal{N}_{\infty}}.$$

Thus we obtain $b_{\mathcal{N}} \leq b_{\mathcal{N}_{\infty}}$.

Next we consider two cases: $b_{\mathcal{N}} = b_{\mathcal{N}_{\infty}}$ and $b_{\mathcal{N}} < b_{\mathcal{N}_{\infty}}$.

If $b_{\mathcal{N}} = b_{\mathcal{N}_{\infty}}$ takes place, then by (2.4.2), we have $\theta_{\infty} = 1$. This implies that $u_{\infty} \in \mathcal{N}$ and $I(u_{\infty}) = b_{\mathcal{N}}$. This is our conclusion.

If $b_{\mathcal{N}} < b_{\mathcal{N}_{\infty}}$ takes place, then by Lemma 2.2.15, there exists a $(\text{PS})_{b_{\mathcal{N}, H}}$ sequence (u_n) . By Lemma 2.2.17, there exist subsequence $(u_{n_k}), \ell \in \mathbf{N}, u_0$ with $I'(u_0) = 0, \omega_i \neq 0$ with $I'_{\infty}(\omega_i) = 0$ and $(y_k^i) \subset \mathbf{R}^N$ such that

$$\left\| u_{n_k} - u_0 - \sum_{i=1}^{\ell} \omega_i (x - y_k^i) \right\| \to 0, \quad I(u_{n_k}) \to b_{\mathcal{N}} = I(u_0) + \sum_{i=1}^{\ell} I_{\infty}(\omega_i).$$

Since $\omega_i \neq (0,0)$, we have $b_{\mathcal{N}_{\infty}} \leq I_{\infty}(\omega_i)$. By $b_{\mathcal{N}} < b_{\mathcal{N}_{\infty}}$, it follows that $\ell = 0$, which implies

$$u_{n_k} \to u_0$$
 strongly in H

This shows that $u_0 \in \mathcal{N}$ and $I(u_0) = b_{\mathcal{N}}$.

Remark 2.4.3. We consider the situation $b_{\mathcal{N}} = b_{\mathcal{N}_{\infty}}$ more precisely. We deal with the two cases. (a) $b_{\mathcal{N}_{\infty}}$ is attained by nontrivial functions u_0 . In this case, we can show that both of $V_j(x)$ are constant functions. (b) $b_{\mathcal{N}_{\infty}}$ is attained by semitrivial functions u_0 . We may assume that $u_0 = (u_1, 0)$. Then we can show that $V_1(x)$ is a constant function. Moreover, we can prove the equality $b_{\mathcal{N}} = b_{\mathcal{N}_{\infty}} = d_{\infty,1} = d_1$.

2.5 Proof of Theorem 1.1 (ii) (when β is large)

In this section, we prove the existence of a nontrivial positive solution of (2.1.2) when β is large. By Proposition 2.4.2, there exists a $u_0 = (u_{0,1}, u_{0,2}) \in \mathcal{N}$ such that $I(u_0) = b_{\mathcal{N}}$. Moreover, Corollary 2.2.10 implies that u_0 is a solution of (2.1.2). Hence we need to prove $u_{0,1}, u_{0,2} \neq 0$.

Following Ambrosetti and Colorado [4], let us define constants which are related to the stability of semitrivial solutions on \mathcal{N} .

Definition 2.5.1. We define $\hat{\beta}_1$ and $\hat{\beta}_2$ as follows:

$$\hat{\beta}_1 := \inf_{(u_1,0)\in S_1} \inf_{\varphi_2 \in H^1(\mathbf{R}^N) \setminus \{0\}} \frac{\|\varphi_2\|_2^2}{\int_{\mathbf{R}^N} u_1^2 \varphi_2^2 dx}, \quad \hat{\beta}_2 := \inf_{(0,u_2)\in S_2} \inf_{\varphi_1 \in H^1(\mathbf{R}^N) \setminus \{0\}} \frac{\|\varphi_1\|_1^2}{\int_{\mathbf{R}^N} u_2^2 \varphi_1^2 dx}$$

Here, S_1 and S_2 are defined by

$$S_1 := \{ (u_1, 0) \in \mathcal{N} : I(u_1, 0) = d_1 \}, \quad S_2 := \{ (0, u_2) \in \mathcal{N} : I(0, u_2) = d_2 \}.$$

A main result in this section is the following:

Theorem 2.5.2. If $\beta > \max{\{\hat{\beta}_1, \hat{\beta}_2\}}$, then both components of any minimizer of I on \mathcal{N} are not zero, i.e.,

$$I(u_0) = b_{\mathcal{N}}, \quad u_0 \in \mathcal{N} \quad \Rightarrow \quad u_{0,1}, \ u_{0,2} \not\equiv 0.$$

Proof. It suffices to prove $b_{\mathcal{N}} < \min\{d_1, d_2\}$. Since $\beta > \max\{\hat{\beta}_1, \hat{\beta}_2\}$, there exist $(u_1, 0) \in S_1, (0, u_2) \in S_2, \varphi_1, \varphi_2 \in H^1(\mathbf{R}^N)$ such that

$$\frac{\|\varphi_1\|_1^2}{\int_{\mathbf{R}^N} u_2^2 \varphi_1^2 \, dx} < \beta, \quad \frac{\|\varphi_2\|_2^2}{\int_{\mathbf{R}^N} u_1^2 \varphi_2^2 \, dx} < \beta.$$

We remark that $\{0\} \times H^1(\mathbf{R}^N) \subset T_{(u_1,0)}\mathcal{N}$ and $H^1(\mathbf{R}^N) \times \{0\} \subset T_{(0,u_2)}\mathcal{N}$. In fact, for each $\psi_1, \psi_2 \in H^1(\mathbf{R}^N)$, we have

$$J'(u_1, 0)[(0, \psi_2)] = 0, \quad J'(0, u_2)[(\psi_1, 0)] = 0.$$

Thus, it holds that $\{0\} \times H^1(\mathbf{R}^N) \subset T_{(u_1,0)}\mathcal{N}$ and $H^1(\mathbf{R}^N) \times \{0\} \subset T_{(0,u_2)}\mathcal{N}$ by Lemma 2.2.5.

Next let $\gamma_1, \gamma_2 \in C^2((-\varepsilon, \varepsilon), \mathcal{N})$ satisfy

$$\gamma_1(0) = (u_1, 0), \quad \gamma'_1(0) = (0, \varphi_2), \quad \gamma_2(0) = (0, u_2), \quad \gamma'_2(0) = (\varphi_1, 0).$$

By the Taylor expansion of $I(\gamma_j(t))$ and $I'(u_1, 0) = I'(0, u_2) = 0$, we obtain

$$I(\gamma_j(t)) = I(\gamma_j(0)) + \frac{1}{2}I''(\gamma_j(0))[\gamma'_j(0), \gamma'_j(0)]t^2 + o(t^2).$$

Since

$$I''(u_1,0)[(0,\varphi_2),(0,\varphi_2)] = \|\varphi_2\|_2^2 - \beta \int_{\mathbf{R}^N} u_1^2 \varphi_2^2 \, dx < 0$$

$$I''(0,u_2)[(\varphi_1,0),(\varphi_1,0)] < 0,$$

it follows that for sufficiently small t > 0

$$I(\gamma_j(t)) - I(\gamma_j(0)) < 0.$$

Thus we have $b_{\mathcal{N}} < \min\{d_1, d_2\}$.

Next, we give a proof of Theorem 2.1.1 (ii).

Proof of Theorem 2.1.1 (ii). By Theorem 2.5.2, there exists a u_0 such that $b_{\mathcal{N}} = I(u_0)$, $u_{0,1} \neq 0, u_{0,2} \neq 0$. By Remark 2.2.3, we have

$$|u_0| = (|u_{0,1}|, |u_{0,2}|) \in \mathcal{N}, \quad b_{\mathcal{N}} = I(u_0) = I(|u_0|),$$

which implies that $|u_0|$ is also a minimizer of I on \mathcal{N} . Thus we may assume that $u_{0,1} \geq 0, u_{0,1} \neq 0, u_{0,2} \geq 0, u_{0,2} \neq 0$. Moreover, it hold that $I'(u_0) = 0$ and $u_{0,1}, u_{0,2} > 0$ by Corollary 2.2.10 and the maximum principle.

2.6 Proofs of Theorem 2.1.1 (i) and Theorem 2.1.2. (when β is small)

2.6.1 Proof of Theorem 2.1.1(i).

The aim of this subsection is to prove the existence of a nontrivial positive solution of (2.1.2) when β is small.

The following two propositions give some estimates of $\hat{b}_{\mathcal{M}}$.

Proposition 2.6.1. For each $\beta > 0$,

- (i) $\hat{b}_{\mathcal{M}} < \min\{d_1 + d_{\infty,2}, d_{\infty,1} + d_2\}.$
- (ii) $\hat{b}_{\mathcal{M}_{\infty}} < d_{\infty,1} + d_{\infty,2}$.

Remark 2.6.2. $\hat{b}_{\mathcal{M}}$ depends on β but $d_1, d_2, d_{\infty,1}, d_{\infty,2}$ are independent of β .

Proposition 2.6.3. There exists a $\tilde{\beta}_1 > 0$ such that for each $\beta \in (0, \tilde{\beta}_1)$

$$b_{\mathcal{M}} < b_{\mathcal{M}_{\infty}}.$$

Proofs of Propositions 2.6.1 and 2.6.3 will be given in subsection 2.6.2.

Theorem 2.6.4. There exists a $\tilde{\beta}_2 > 0$ such that for each $\beta \in (0, \tilde{\beta}_2)$, $\hat{b}_{\mathcal{M}}$ is attained.

Proof of Theorem 2.6.4. Set $\alpha_0 = \min\{d_1 + d_{\infty,2}, d_{\infty,1} + d_2\}$. By Proposition 2.6.1, $\mathcal{M}^{\alpha_0} \neq \emptyset$ for all $\beta \in (0, \sqrt{\mu_1 \mu_2})$. By Proposition 2.2.9, there exist $\hat{\beta}_0 > 0$ and $\delta_1 > 0$ such that for each $u \in \mathcal{M}^{\alpha_0}$ and $\beta \in (0, \hat{\beta}_0)$,

$$(2.6.1) ||u_1||_1 \ge \delta_1, ||u_2||_2 \ge \delta_1.$$

Suppose $0 < \beta < \min\{\tilde{\beta}_1, \hat{\beta}_0\}$. Then we remark that there exists a $(PS)_{\hat{b}_{\mathcal{M}}, H}$ sequence (u_n) by Lemmas 2.2.14 and 2.2.15. Then by Lemma 2.2.17, we have

(2.6.2)
$$\left\| u_n - u_0 - \sum_{i=1}^{\ell} \omega_i (x - y_n^i) \right\| \to 0,$$

(2.6.3)
$$I(u_n) \to \hat{b}_{\mathcal{M}} = I(u_0) + \sum_{i=1}^{\iota} I_{\infty}(\omega_i).$$

We shall show that $u_0 = (u_{0,1}, u_{0,2}), u_{0,1} \neq 0, u_{0,2} \neq 0$ and $\ell = 0$. We divide our argument into three steps.

Step 1. $u_0 \not\equiv (0,0)$.

We prove indirectly and we assume that $u_0 \equiv (0,0)$. By (2.6.3), it follows that

$$\hat{b}_{\mathcal{M}} = \sum_{i=1}^{\ell} I_{\infty}(\omega_i).$$

By $\hat{b}_{\mathcal{M}} > 0$, we obtain $\ell \neq 0$. Since $\hat{b}_{\mathcal{M}} < \hat{b}_{\mathcal{M}_{\infty}}$, we conclude that one of the components of ω_i equals 0. Moreover if $\ell \geq 2$, we have

(2.6.4)
$$\omega_{i,1} \equiv 0 \quad (1 \le i \le \ell) \quad \text{or} \quad \omega_{i,2} \equiv 0 \quad (1 \le i \le \ell).$$

Otherwise, we have $\hat{b}_{\mathcal{M}} \geq d_{\infty,1} + d_{\infty,2}$, which contradicts Proposition 2.6.1.

Suppose that $\omega_{i,1} \equiv 0$ $(1 \leq i \leq \ell)$. By (2.6.2), we obtain $||u_{n,1}||_1 \to 0$, which contradicts (2.6.1). In a similar way, $\omega_{i,2} \equiv 0$ $(1 \leq i \leq \ell)$ does not take place. This implies that $u_0 \neq (0,0)$.

Step 2. $u_0 \notin (H^1(\mathbf{R}^N) \times \{0\}) \cup (\{0\} \times H^1(\mathbf{R}^N)).$

We prove indirectly and we assume that $u_0 \in H^1(\mathbf{R}^N) \times \{0\}$. By (2.6.3) we have

$$\hat{b}_{\mathcal{M}} = I(u_0) + \sum_{i=1}^{\ell} I_{\infty}(\omega_i).$$

Since $\hat{b}_{\mathcal{M}} < \hat{b}_{\mathcal{M}_{\infty}}$, one of the components of ω_i is equal to 0 for $1 \leq i \leq \ell$. Since $\hat{b}_{\mathcal{M}} < d_1 + d_{\infty,2}$ and $d_1 \leq I(u_0)$, we have

(2.6.5)
$$\omega_{i,2} \equiv 0 \quad \text{for} \quad 1 \le i \le \ell.$$

From (2.6.2) and (2.6.5), it follows that $||u_{n,2}||_2 \to 0$, which contradicts (2.6.1). So, we conclude $u_0 \notin H^1(\mathbf{R}^N) \times \{0\}$. In a similar way, we can prove that $u_0 \notin \{0\} \times H^1(\mathbf{R}^N)$.

Step 3. Conclusion.

Now we complete a proof of Theorem 2.6.4. By Steps 1 and 2, it follows that $u_{0,1}, u_{0,2} \neq 0$. Since $\hat{b}_{\mathcal{M}} \leq I(u_0)$ and $I_{\infty}(\omega_i) > 0$, we have $\ell = 0$. By Remark 2.2.18, (u_n) converges to u_0 strongly in H, so $I(u_0) = \inf_{\mathcal{M}} I = \hat{b}_{\mathcal{M}}$.

We give a proof of Theorem 2.1.1 (i).

Proof of Theorem 2.1.1 (i). As in the proof of Theorem 2.1.1 (ii), we obtain a nontrivial positive solution of (2.1.2) by Theorem 2.6.4, Corollary 2.2.10 and the maximum principle.

2.6.2 Proofs of Propositions 2.6.1 and 2.6.3.

Before proving Propositions 2.6.1 and 2.6.3, we state a useful lemma. For $u \in H, u_1 \neq 0, u_2 \neq 0$, we set

$$f_u(s_1, s_2) := I(\sqrt{s_1}u_1, \sqrt{s_2}u_2)$$

= $\frac{s_1}{2} \|u_1\|_1^2 + \frac{s_2}{2} \|u_2\|_2^2 - \frac{s_1^2}{4} \mu_1 \|u_1\|_{L^4}^4 - \frac{s_1s_2}{2} \beta \|u_1u_2\|_{L^2}^2 - \frac{s_2^2}{4} \mu_2 \|u_2\|_{L^4}^4.$

Lemma 2.6.5. Let $u \in H$, $u_1 \neq 0, u_2 \neq 0$. Then the following hold.

- (i) Let $0 \leq \beta < \sqrt{\mu_1 \mu_2}$. Then $f_u(s_1, s_2)$ is strictly concave in $[0, \infty) \times [0, \infty)$.
- (ii) Let $u \in \mathcal{M}$ and $0 \leq \beta < \sqrt{\mu_1 \mu_2}$. Then (1,1) is a unique maximum point of $f_u(s_1, s_2)$. Namely, it follows

$$I(u) = f_u(1,1) = \max_{[0,\infty) \times [0,\infty)} I(\sqrt{s_1}u_1, \sqrt{s_2}u_2)$$

(iii) Let $\beta \ge 0$ and $(s_{0,1}, s_{0,2}) \in (0, \infty) \times (0, \infty)$ be a maximum point of $f_u(s_1, s_2)$. Then $(\sqrt{s_{0,1}}u_1, \sqrt{s_{0,2}}u_2) \in \mathcal{M}.$

Remark 2.6.6. Similar results hold for I_{∞} and \mathcal{M}_{∞} .

Proof. This lemma is proved in Lin and Wei [65], however, for reader's convenience, we give a proof.

(i) Differentiating $f_u(s_1, s_2)$, we have

$$(2.6.6) \qquad \begin{aligned} \frac{\partial f_u}{\partial s_1} &= \frac{1}{2} \|u_1\|_1^2 - \frac{s_1}{2} \mu_1 \|u_1\|_{L^4}^4 - \frac{s_2}{2} \beta \|u_1 u_2\|_{L^2}^2, \\ \frac{\partial f_u}{\partial s_2} &= \frac{1}{2} \|u_2\|_2^2 - \frac{s_1}{2} \beta \|u_1 u_2\|_{L^2}^2 - \frac{s_2}{2} \mu_2 \|u_2\|_{L^4}^4, \\ \frac{\partial^2 f_u}{\partial s_j^2} &= -\frac{1}{2} \mu_j \|u_j\|_{L^4}^4 \ (j = 1, 2), \qquad \frac{\partial^2 f_u}{\partial s_1 \partial s_2} &= -\frac{1}{2} \beta \|u_1 u_2\|_{L^2}^2. \end{aligned}$$

Since $0 \leq \beta < \sqrt{\mu_1 \mu_2}$, the matrix

$$\begin{pmatrix} \frac{\partial^2 f_u}{\partial s_1^2}(s_1, s_2) & \frac{\partial^2 f_u}{\partial s_1 \partial s_2}(s_1, s_2) \\ \frac{\partial^2 f_u}{\partial s_1 \partial s_2}(s_1, s_2) & \frac{\partial^2 f_u}{\partial s_2^2}(s_1, s_2) \end{pmatrix} = \frac{1}{2} \begin{pmatrix} -\mu_1 \|u_1\|_{L^4}^4 & -\beta \|u_1 u_2\|_{L^2}^2 \\ -\beta \|u_1 u_2\|_{L^2}^2 & -\mu_2 \|u_2\|_{L^4}^4 \end{pmatrix}$$

is negative definite. Thus $f_u(s_1, s_2)$ is strictly concave in $[0, \infty) \times [0, \infty)$. (ii) Suppose $u \in \mathcal{M}$. By (2.6.6) and $\beta \in [0, \sqrt{\mu_1 \mu_2})$, we have

$$\nabla f_u(s_1, s_2) = (0, 0) \iff (s_1, s_2) = (1, 1).$$

Since $f_u(s_1, s_2)$ is strictly concave, (1, 1) is an unique maximum point and

$$I(u) = f_u(1,1) = \max_{[0,\infty) \times [0,\infty)} I(\sqrt{s_1}u_1, \sqrt{s_2}u_2).$$

(iii) Suppose $(s_{0,1}, s_{0,2}) \in (0, \infty) \times (0, \infty)$ is a maximum point of $f_u(s_1, s_2)$. Since $\nabla f_u(s_{0,1}, s_{0,2}) = (0, 0)$, we have

$$s_{0,1} \|u_1\|_1^2 = s_{0,1}^2 \mu_1 \|u_1\|_{L^4}^4 + s_{0,1} s_{0,2} \beta \|u_1 u_2\|_{L^2}^2,$$

$$s_{0,2} \|u_2\|_2^2 = s_{0,1} s_{0,2} \beta \|u_1 u_2\|_{L^2}^2 + s_{0,2}^2 \mu_2 \|u_2\|_{L^4}^4.$$

Thus this implies $(\sqrt{s_{0,1}}u_{0,1}, \sqrt{s_{0,2}}u_{0,2}) \in \mathcal{M}.$

First, we prove Proposition 2.6.1.

Proof of Proposition 2.6.1. We only prove $\hat{b}_{\mathcal{M}} < d_1 + d_{\infty,2}$ since we can prove other inequality in a similar way. By Lemma 2.3.2, we suppose that $(\varphi_{0,1}, 0) \in \mathcal{N}, (0, \varphi_{\infty,2}) \in \mathcal{N}_{\infty}$ satisfy

$$I(\varphi_{0,1}, 0) = d_1, \quad I_{\infty}(0, \varphi_{\infty,2}) = d_{\infty,2}, \quad \varphi_{0,1} > 0, \quad \varphi_{\infty,2} > 0.$$

We remark that for a $k \in \mathbf{N}$, it follows $\|\varphi_{0,1}(x)\varphi_{\infty,2}(x-ke_1)\|_{L^2}^2 \to 0$ as $k \to \infty$ where $e_1 = (1, 0, \ldots, 0)$. Thus we have

$$g_k(s_1, s_2) := I(\sqrt{s_1}\varphi_{0,1}(x), \sqrt{s_2}\varphi_{\infty,2}(x - ke_1)) \to g(s_1, s_2) \text{ in } C^2_{\text{loc}}((0, \infty)^2)$$

where

$$g(s_1, s_2) := \frac{s_1}{2} \|\varphi_{0,1}\|_1^2 - \frac{s_1^2}{4} \mu_1 \|\varphi_{0,1}\|_{L^4}^4 + \frac{s_2}{2} \|\varphi_{\infty,2}\|_{\infty,2}^2 - \frac{s_2^2}{4} \|\varphi_{\infty,2}\|_{L^4}^4.$$

Since $g(s_1, s_2)$ has a unique maximum point (1, 1) and $g_k(s_1, s_2) \leq g(s_1, s_2)$, $g_k(s_1, s_2)$ has a maximum point $(s_{k,1}, s_{k,2}) \in (0, \infty) \times (0, \infty)$ for a sufficiently large k. By Lemma 2.6.5 we have $(\sqrt{s_{k,1}}\varphi_{0,1}(x), \sqrt{s_{k,2}}\varphi_{\infty,2}(x-ke_1)) \in \mathcal{M}$.

Thus we have

$$\begin{split} \hat{b}_{\mathcal{M}} &\leq I(\sqrt{s_{k,1}}\varphi_{0,1}, \sqrt{s_{k,2}}\varphi_{\infty,2}(x-ke_{1})) \\ &= \frac{1}{2}s_{k,1}\|\varphi_{0,1}\|_{1}^{2} + \frac{1}{2}s_{k,2}\|\varphi_{\infty,2}(x-ke_{1})\|_{2}^{2} - \frac{1}{4}s_{k,1}^{2}\mu_{1}\|\varphi_{0,1}\|_{L^{4}}^{4} \\ &\quad - \frac{1}{2}\beta s_{k,1}s_{k,2}\|\varphi_{0,1}\varphi_{\infty,2}(x-ke_{1})\|_{L^{2}}^{2} - \frac{1}{4}s_{k,2}^{2}\mu_{2}\|\varphi_{\infty,2}\|_{L^{4}}^{4} \\ &\leq \frac{1}{2}s_{k,1}\|\varphi_{0,1}\|_{1}^{2} + \frac{1}{2}s_{k,2}\|\varphi_{\infty,2}\|_{\infty,2}^{2} - \frac{1}{4}s_{k,1}^{2}\mu_{1}\|\varphi_{0,1}\|_{L^{4}}^{4} \\ &\quad - \frac{1}{2}\beta s_{k,1}s_{k,2}\|\varphi_{0,1}\varphi_{\infty,2}(x-ke_{1})\|_{L^{2}}^{2} - \frac{1}{4}s_{k,2}^{2}\mu_{2}\|\varphi_{\infty,2}\|_{L^{4}}^{4} \\ &= d_{1} + d_{\infty,2} + \frac{1}{2}(s_{k,1}-1)\|\varphi_{0,1}\|_{1}^{2} + \frac{1}{2}(s_{k,2}-1)\|\varphi_{\infty,2}\|_{\infty,2}^{2} \\ &\quad + \frac{\mu_{1}}{4}(1-s_{k,1}^{2})\|\varphi_{0,1}\|_{L^{4}}^{4} + \frac{\mu_{2}}{4}(1-s_{k,2}^{2})\|\varphi_{\infty,2}\|_{L^{4}}^{4} - \frac{1}{2}\beta s_{k,1}s_{k,2}\|\varphi_{0,1}\varphi_{\infty,2}(x-ke_{1})\|_{L^{2}}^{2}. \end{split}$$

Since

 $\|\varphi_{0,1}\|_1^2 = \mu_1 \|\varphi_{0,1}\|_{L^4}^4, \quad \|\varphi_{\infty,2}\|_{\infty,2} = \mu_2 \|\varphi_{\infty,2}\|_{L^4}^4,$

we obtain

$$\begin{aligned} &\frac{1}{2}(s_{k,1}-1)\|\varphi_{0,1}\|_{1}^{2} + \frac{\mu_{1}}{4}(1-s_{k,1}^{2})\|\varphi_{0,1}\|_{L^{4}}^{4} = \frac{\|\varphi_{0,1}\|_{1}^{2}}{4}(-s_{k,1}^{2}+2s_{k,1}-1) \\ &= -\frac{\|\varphi_{0,1}\|_{1}^{2}}{4}(s_{k,1}-1)^{2} \leq 0, \\ &\frac{1}{2}(s_{k,2}-1)\|\varphi_{\infty,2}\|_{\infty,2}^{2} + \frac{\mu_{2}}{4}(1-s_{k,2}^{2})\|\varphi_{\infty,2}\|_{L^{4}}^{4} \leq 0. \end{aligned}$$

Moreover, since $\varphi_{0,1}$, $\varphi_{\infty,2} > 0$, it follows that $\|\varphi_{0,1}\varphi_{\infty,2}(x-ke_1)\|_{L^2}^2 > 0$. Hence we have

$$\hat{b}_{\mathcal{M}} < d_1 + d_{\infty,2}.$$

The following lemma is related to the existence of minimizer for $b_{\mathcal{M}_{\infty}}$, which is due to Lin and Wei [65], and Sirakov [94] in the case N = 2, 3.

Lemma 2.6.7 (Lin and Wei [65], and Sirakov [94]). There exists a $\bar{\beta} \in (0, \sqrt{\mu_1 \mu_2}]$ such that if $\beta \in (0, \bar{\beta})$, then $\hat{b}_{\mathcal{M}_{\infty}}$ is attained by a nontrivial positive solution $\omega = (\omega_1, \omega_2)$ of (2.2.1).

Proof. We only consider the case N = 1 since the other case is proved in [65] and [94]. We use the arguments in the proof of Theorem 2.6.4. Set $\alpha_0 = d_{\infty,1} + d_{\infty,2}$. Since $\hat{b}_{\mathcal{M}_{\infty}} < \alpha_0$ by Proposition 2.6.1, $\mathcal{M}^{\alpha_0} \neq \emptyset$ for all $\beta \in (0, \sqrt{\mu_1 \mu_2})$. By Proposition 2.2.9, there exist $\hat{\beta}_0 > 0$ and $\delta_1 > 0$ such that for each $u \in \mathcal{M}^{\alpha_0}$ and $\beta \in (0, \hat{\beta}_0)$,

$$(2.6.7) ||u_1||_1 \ge \delta_1, ||u_2||_2 \ge \delta_1.$$

Suppose $0 < \beta < \min\{\tilde{\beta}_1, \hat{\beta}_0\}$ and let (u_n) be a $(PS)_{\hat{b}_{\mathcal{M}_{\infty}}, H}$ sequence. Then by Lemma 2.2.17, we have

(2.6.8)
$$\left\| u_n - u_0 - \sum_{i=1}^{\ell} \omega_i (x - y_n^i) \right\| \to 0, \quad I_{\infty}(u_n) \to \hat{b}_{\mathcal{M}_{\infty}} = I_{\infty}(u_0) + \sum_{i=1}^{\ell} I_{\infty}(\omega_i).$$

We shall show that $u_0 = (u_{0,1}, u_{0,2}), u_{0,1} \neq 0, u_{0,2} \neq 0$ and $\ell = 0$.

First we prove $\ell \leq 1$ indirectly and assume $\ell \geq 2$. Since $I_{\infty}(\omega_i) > 0$ for all $1 \leq i \leq \ell$, we may suppose $\omega_{i,1} \equiv 0$ or $\omega_{i,2} \equiv 0$ for all $1 \leq i \leq \ell$. Otherwise, we have a contradiction: $\lim_{n\to\infty} I_{\infty}(u_n) > \hat{b}_{\mathcal{M}_{\infty}}$. Suppose $\omega_{i,1} \equiv 0$ for all $1 \leq i \leq \ell$. In this case, it follows from (2.6.8) that $||u_{n,1}||_1 \to 0$, which contradicts (2.6.7). In the other case, we can lead a contradiction. Therefore we obtain $\ell \leq 1$.

If $\ell = 0$, then since $u_n \to u_0$ strongly in H, it follows from (2.6.7) and (2.6.8) that $u_0 \in \mathcal{M}_{\infty}$ and $I_{\infty}(u_0) = \hat{b}_{\mathcal{M}_{\infty}}$. This is our conclusion.

If $\ell = 1$ and $\omega_{1,1} \equiv 0$, then we can infer that $u_{0,2} \neq 0$. In fact, if $u_{0,2} \equiv 0$, then from (2.6.8), it holds that $||u_{n,2}||_2 \to 0$. This is a contradiction. Thus we have $u_{0,2} \neq 0$. However, in this case, from (2.6.8) we obtain $\hat{b}_{\mathcal{M}_{\infty}} \geq d_{\infty,1} + d_{\infty,2}$, which contradicts to $\hat{b}_{\mathcal{M}_{\infty}} < d_{\infty,1} + d_{\infty,2}$. In a similar way, the case $\ell = 1$ and $\omega_{1,2} \equiv 0$ does not take place.

Hence we have $\ell = 1$ and $\omega_{1,j} \neq 0$ for j = 1, 2. Since $I_{\infty}(\omega_1) \geq \hat{b}_{\mathcal{M}_{\infty}}$, we obtain $u_0 \equiv 0$ and $I_{\infty}(\omega_1) = \hat{b}_{\mathcal{M}_{\infty}}$, which completes a proof.

Now we prove Proposition 2.6.3.

Proof of Proposition 2.6.3. Set $\tilde{\beta}_1 := \bar{\beta}$ where $\bar{\beta}$ is given in Lemma 2.6.7. By Lemma 2.6.7, there exists an $\omega \in \mathcal{M}_{\infty}$ such that $I_{\infty}(\omega) = \hat{b}_{\mathcal{M}_{\infty}}$ and $\omega_j > 0$ in \mathbb{R}^N . By Lemma 2.6.5 a function

$$h(s_1, s_2) := I_{\infty}(\sqrt{s_1}\omega_1, \sqrt{s_2}\omega_2)$$

has a unique maximum point (1,1). Let $h_k(s_1, s_2) := I(\omega_1(x - ke_1), \omega_2(x - ke_1))$. Since $h_k(s_1, s_2) \le h(s_1, s_2)$ and

$$h_k(s_1, s_2) = I(\omega_1(x - ke_1), \omega_2(x - ke_1)) \to h(s_1, s_2)$$
 in $C^2_{\text{loc}}((0, \infty)^2)$

the function $h_k(s_1, s_2)$ has a maximum point $(s_{k,1}, s_{k,2}) \in (0, \infty) \times (0, \infty)$ for a sufficiently large k. By Lemma 2.6.5, we have

$$(\sqrt{s_{k,1}}\omega_1(x-ke_1),\sqrt{s_{k,2}}\omega_2(x-ke_1)) \in \mathcal{M}.$$

Since we can suppose that $V_1 \not\equiv \text{const.}$ or $V_2 \not\equiv \text{const.}$, by Lemma 2.6.5 again, we have

$$\begin{split} \hat{b}_{\mathcal{M}} &\leq I(\sqrt{s_{k,1}}\omega_1(x-ke_1), \sqrt{s_{k,2}}\omega_2(x-ke_1)) \\ &< I_{\infty}(\sqrt{s_{k,1}}\omega_1(x-ke_1), \sqrt{s_{k,2}}\omega_2(x-ke_1)) = I_{\infty}(\sqrt{s_{k,1}}\omega_1(x), \sqrt{s_{k,2}}\omega_2(x)) \\ &\leq \max_{(s_1,s_2)\in[0,\infty)\times[0,\infty)} I_{\infty}(\sqrt{s_1}\omega_1(x), \sqrt{s_2}\omega_2(x)) = \hat{b}_{\mathcal{M}_{\infty}}, \end{split}$$

which completes the proof.

2.6.3 Proof of Theorem 2.1.2.

In this subsection, we prove Theorem 2.1.2. When $\beta > 0$ is large, in other words, Theorem 2.1.2(ii) follows from the construction of a positive solution of (2.1.2). So we only prove (i). A main result in this section is the following.

Proposition 2.6.8. For each sufficiently small $\beta > 0$, it holds

$$(2.6.9) b_{\mathcal{N}} < \hat{b}_{\mathcal{M}}.$$

We remark that (2.6.9) shows that the minimizer of $\inf_{\mathcal{N}} I$ is a semitrivial solution and a proof of Theorem 2.1.2 easily follows.

Proof of Proposition 2.6.8. We prove (2.6.9) indirectly. So we assume that there exists a sequence (β_n) such that $\beta_n \to 0$ and $\hat{b}_{\mathcal{M}_n} = b_{\mathcal{N}_n}$, where

$$I_{n}(u) := \frac{1}{2} \|u\|^{2} - \frac{1}{4} \int_{\mathbf{R}^{N}} \mu_{1} u_{1}^{4} + 2\beta_{n} u_{1}^{2} u_{2}^{2} + \mu_{2} u_{2}^{4} dx,$$

$$\mathcal{N}_{n} := \{ u \in H : u \neq 0, \ I_{n}'(u)u = 0 \},$$

$$\mathcal{M}_{n} := \{ u \in H : u_{1}, u_{2} \neq 0, \ I_{n}'(u)(u_{1}, 0) = I_{n}'(u)(0, u_{2}) = 0 \},$$

$$b_{\mathcal{N}_{n}} := \inf_{u \in \mathcal{N}_{n}} I_{n}(u), \quad b_{\mathcal{M}_{n}} := \inf_{u \in \mathcal{M}_{n}} I_{n}(u).$$

By Theorem 2.6.4, there exists a $(u_n) \subset \mathcal{M}_n$ such that $I_n(u_n) = \hat{b}_{\mathcal{M}_n} = b_{\mathcal{N}_n}$. It is obvious that (u_n) is a bounded sequence. So we assume that $u_n \rightharpoonup u_0$ weakly in H. Since

$$\begin{cases} -\Delta u_{n,1} + V_1(x)u_{n,1} = \mu_1 u_{n,1}^3 + \beta_n u_{n,1} u_{n,2}^2 & \text{in } \mathbf{R}^N, \\ -\Delta u_{n,2} + V_2(x)u_{n,2} = \beta_n u_{n,1}^2 u_{n,2} + \mu_2 u_{n,2}^3 & \text{in } \mathbf{R}^N, \end{cases}$$

we have

(2.6.10)
$$\begin{cases} -\Delta u_{0,1} + V_1(x)u_{0,1} = \mu_1 u_{0,1}^3 & \text{in } \mathbf{R}^N, \\ -\Delta u_{0,2} + V_2(x)u_{0,2} = \mu_2 u_{0,2}^3 & \text{in } \mathbf{R}^N. \end{cases}$$

We prove the following claim.

Claim $u_{0,1} \equiv 0$ or $u_{0,2} \equiv 0$.

Proof of Claim. We assume that $u_{0,1} \neq 0$ and $u_{0,2} \neq 0$. From (2.6.10), we have $d_1 + d_2 \leq I_0(u_0)$. On the other hand, since $I_n(u_n) = ||u_n||^2/4$ and $u_n \rightharpoonup u_0$ weakly in H, it follows that

$$I_0(u_0) \le \liminf_{n \to \infty} I_n(u_n) = \liminf_{n \to \infty} b_{\mathcal{N}_n} \le \min\{d_1, d_2\}$$

This is a contradiction, hence $u_{0,1} \equiv 0$ or $u_{0,2} \equiv 0$.

Suppose that $u_{0,2} \equiv 0$. By Proposition 2.2.9, there exists a $\delta_1 > 0$ such that $||u_{n,j}||_{L^4} \ge \delta_1$ (j = 1, 2). Developing a concentration–compactness type argument, we can find a sequence $(y_n) \subset \mathbf{R}^N$ such that

 $|y_n| \to \infty$, $||u_{n,2}||_{L^4(Q+y_n)} \to c > 0$, $u_{n,2}(x+y_n) \rightharpoonup \omega_2$ weakly in $H^1(\mathbf{R}^N)$,

where $Q = [0, 1]^N$. Moreover ω_2 satisfies that $\omega_2 \neq 0$ and

$$-\Delta\omega_2 + V_{\infty,2}\omega_2 = \mu_2\omega_2^3.$$

Since

$$I_n(u_n) = \frac{1}{4} \int_{\mathbf{R}^N} \mu_1 u_{n,1}^4 + 2\beta_n u_{n,1}^2 u_{n,2}^2 + \mu_2 u_{n,2}^4 \, dx \ge \frac{\mu_1}{4} \int_{\mathbf{R}^N} u_{n,1}^4 \, dx + \frac{\mu_2}{4} \int_{\mathbf{R}^N} u_{n,2}^4(x+y_n) \, dx,$$

we have

$$I_{\infty,0}(0,\omega_2) = \frac{\mu_2}{4} \int_{\mathbf{R}^N} \omega_2^4 dx < \frac{\mu_1}{4} \delta_1^4 + \liminf_{n \to \infty} \frac{\mu_2}{4} \int_{\mathbf{R}^N} u_{n,2}^4 (x+y_n) dx$$

$$\leq \liminf_{n \to \infty} I_n(u_n) = \lim_{n \to \infty} b_{\mathcal{N}_n} \leq \min\{d_1, d_2\},$$

which implies that $d_{\infty,2} < \min\{d_1, d_2\} \le d_2$. This is a contradiction. The situation $u_{0,1} \equiv 0$ can be treated similarly. Thus we have (2.6.9).

2.7 Proof of Theorem 2.1.4.

In this section, we prove Theorem 2.1.4 and follow the idea in Tanaka [99] (cf. Wang [103]).

Proof of Theorem 2.1.4. We prove indirectly and we assume that (2.1.2) has a positive solution u. Since $V_j(x) \in C^1(\mathbf{R}^N) \cap L^{\infty}(\mathbf{R}^N)$, we remark $u_j \in H^2(\mathbf{R}^N)$. Without loss of generality we may assume that $\nu = e_1 = (1, 0, ..., 0)$. Since $I'(u) \left[\left(\frac{\partial u_1}{\partial x_1}, \frac{\partial u_2}{\partial x_1} \right) \right] = 0$, we have

(2.7.1)
$$\sum_{j=1}^{2} \left\langle u_{j}, \frac{\partial u_{j}}{\partial x_{1}} \right\rangle_{j} = \sum_{j=1}^{2} \int_{\mathbf{R}^{N}} \mu_{j} u_{j}^{3} \frac{\partial u_{j}}{\partial x_{1}} dx + \beta \int_{\mathbf{R}^{N}} u_{1} u_{2}^{2} \frac{\partial u_{1}}{\partial x_{1}} + u_{1}^{2} u_{2} \frac{\partial u_{2}}{\partial x_{1}} dx.$$

Here, we claim that

(2.7.2)
$$\int_{\mathbf{R}^N} \nabla u_j \cdot \nabla \left(\frac{\partial u_j}{\partial x_1}\right) \, dx = 0, \quad \int_{\mathbf{R}^N} \mu_j u_j^3 \frac{\partial u_j}{\partial x_1} \, dx = 0,$$

(2.7.3)
$$\int_{\mathbf{R}^N} u_1 u_2^2 \frac{\partial u_1}{\partial x_1} + u_1^2 u_2 \frac{\partial u_2}{\partial x_1} dx = 0,$$

(2.7.4)
$$\int_{\mathbf{R}^N} V_j(x) u_j \frac{\partial u_j}{\partial x_1} dx = -\frac{1}{2} \int_{\mathbf{R}^N} \frac{\partial V_j}{\partial x_1} u_j^2 dx.$$

Assuming (2.7.2)–(2.7.4). It follows from (2.7.1) that

$$-\frac{1}{2}\sum_{j=1}^{2}\int_{\mathbf{R}^{N}}\frac{\partial V_{j}}{\partial x_{1}}u_{j}^{2}\,dx=0.$$

By (2–V3'), (2–V4') and $u_j > 0$, this is a contradiction, so (2.1.2) has no positive solution.

Next we show (2.7.2)–(2.7.4). We only prove (2.7.3) since proofs of other cases are similar. For $\varphi_1, \varphi_2 \in C_0^{\infty}(\mathbf{R}^N)$, we have

$$\varphi_1 \varphi_2^2 \frac{\partial \varphi_1}{\partial x_1} + \varphi_1^2 \varphi_2 \frac{\partial \varphi_2}{\partial x_1} = \frac{1}{2} \frac{\partial}{\partial x_1} \left(\varphi_1^2 \varphi_2^2 \right).$$

Thus

$$\int_{\mathbf{R}^{N}} \varphi_{1} \varphi_{2}^{2} \frac{\partial \varphi_{1}}{\partial x_{1}} + \varphi_{1}^{2} \varphi_{2} \frac{\partial \varphi_{2}}{\partial x_{1}} dx = \int_{\mathbf{R}^{N}} \frac{\partial}{\partial x_{1}} \left(\varphi_{1}^{2} \varphi_{2}^{2}\right) dx$$
$$= \int_{\mathbf{R}^{N-1}} \int_{-\infty}^{\infty} \frac{\partial}{\partial x_{1}} \left(\varphi_{1}^{2} \varphi_{2}^{2}\right) dx_{1} dx' = 0.$$

Since $C_0^{\infty}(\mathbf{R}^N)$ is dense in $H^2(\mathbf{R}^N)$ and the functional

$$(u_1, u_2) \mapsto \int_{\mathbf{R}^N} u_1 u_2^2 \frac{\partial u_1}{\partial x_1} + u_1^2 u_2 \frac{\partial u_2}{\partial x_1} \, dx : H^2(\mathbf{R}^N) \times H^2(\mathbf{R}^N) \to \mathbf{R}$$

is continuous, (2.7.3) holds.

г		
н		
н		
L		

Chapter 3

Uniqueness of nontrivial positive solutions

3.1 Introduction

In this chapter, we consider the uniqueness of nontrivial positive solution of (CNLS) with $\varepsilon = 1$, namely,

in \mathbf{R}^N ,

(3.1.1)
$$-\Delta u_1 + V_1(x)u_1 = \mu_1 u_1^3 + \beta u_1 u_2^2 \qquad \text{in } \mathbf{R}^N,$$

(3.1.2)
$$-\Delta u_2 + V_2(x)u_2 = \beta u_1^2 u_2 + \mu_2 u_2^3 \qquad \text{in } \mathbf{R}^N,$$

(3.1.3)
$$u_1(x), u_2(x) \to 0 \text{ as } |x| \to \infty,$$

$$(3.1.4) u_1(x), u_2(x) > 0$$

where $\mu_1, \mu_2, \beta > 0$ and N = 1, 2, 3.

Recently the existence of nontrivial positive solutions has been studied extensively in [4, 31, 50, 75, 65, 94, 105]. In particular, the case where $V_1(x)$, $V_2(x)$ are positive and independent of x is well studied and it is shown in [4, 31, 75, 65, 94] that there exist positive constants $\beta_2 \geq \beta_1 > 0$ such that for $\beta \in [0, \beta_1) \cup (\beta_2, \infty)$, (3.1.1)–(3.1.4) has a nontrivial positive solution. And it has been extended to x-dependent situations in [50, 105]. However the uniqueness of nontrivial positive solutions is not studied and the main purpose of this chapter is to establish the uniqueness for small $\beta > 0$.

First we consider the uniqueness of nontrivial positive solutions in the constant coefficient case:

(3.1.5)
$$\begin{cases} -\Delta u_1 + V_1 u_1 = \mu_1 u_1^3 + \beta u_1 u_2^2 & \text{in } \mathbf{R}^N, \\ -\Delta u_2 + V_2 u_2 = \beta u_1^2 u_2 + \mu_2 u_2^3 & \text{in } \mathbf{R}^N, \\ u_1(x), u_2(x) > 0 & \text{in } \mathbf{R}^N, \\ u_1(x), u_2(x) \to 0 \text{ as } |x| \to \infty. \end{cases}$$

Here V_1, V_2 are positive constants and N = 1, 2, 3.

Now we state our result in the constant coefficient case.

Theorem 3.1.1. Suppose that N = 1, 2, 3. Then there exists a $\beta_0 > 0$ such that if $\beta \in (0, \beta_0)$, then nontrivial positive solutions of (3.1.5) are unique up to translation.

Remark 3.1.2. Wei and Yao [107] also obtains similar results concerning (3.1.5).

Next we deal with the variable coefficient case. Here we assume that N = 2, 3 and $V_j(x)$ satisfies the following condition:

(3–V1) For $j = 1, 2, V_j \in C^2(\mathbf{R}^N), V_j(x) = V_j(|x|)$ and $\lim_{|x|\to\infty} V_j(x) > 0$.

(3–V2) For j = 1, 2,

$$\inf\left\{\int_{\mathbf{R}^N} |\nabla u|^2 + V_j(x)u^2 dx: \|u\|_{H^1(\mathbf{R}^N)} = 1\right\} > 0.$$

- (3–V3) For j = 1, 2 and $r \ge 0, V'_j(r) \ge 0$.
- (3-V4) There exist C > 0 and M > 0 such that $|V_j(r)| \le C(1+r)^M$ for j = 1, 2 and $r \ge 0$.
- (3–V5) When N = 3, the function

$$H_j(r) := \frac{4}{3}r^2 V_j(r) + r^3 V'(r) - \frac{4}{27}$$

has a unique simple zero in $(0, \infty)$.

As in Remark 1.1.5, the function $V_j(r) = r^{\alpha}$ ($\alpha \geq 2$) satisfies (3–V1)–(3–V5). Therefore, we restrict ourselves in the following function space.

Definition 3.1.3. For V_1 and V_2 satisfying (3–V1)–(3–V5), we define $\mathcal{H}_{V_1,V_2,r} \subset H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$ as follows:

$$\mathcal{H}_{V_1,V_2,r} := \left\{ u \in H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N) : u(x) = u(|x|), \ \int_{\mathbf{R}^N} V_j(x) u_j^2 dx < \infty \text{ for } j = 1,2 \right\}.$$

Under (3-V1)-(3-V5), we have the following uniqueness result.

Theorem 3.1.4. Suppose N = 2, 3 and $V_j(x)$ satisfies (3-V1)-(3-V5). Then there exists a $\beta_0 > 0$ such that for $\beta \in (0, \beta_0)$ (3.1.1)-(3.1.4) has a unique nontrivial positive solution in $\mathcal{H}_{V_1,V_2,r}$.

Remark 3.1.5. (i) When N = 2, 3, by the result of Busca and Sirakov [20], it is known that any nontrivial positive solution of (3.1.1)-(3.1.4) is radially symmetric and monotone decreasing under (3–V1) and (3–V3), and radially symmetric with respect to some point $x_0 \in \mathbf{R}^N$ if $V_j(x)$ is independent of x and positive.

(ii) We remark that the argument in [20] also works for N = 1 after suitable modification. However, it is not clearly stated in [20], we give the symmetry and monotonicity result for N = 1 in section 3.4 for the sake of readers.

(iii) The uniqueness of nontrivial positive solution of (3.1.1)-(3.1.4) does not hold for some $\beta > 0$. Indeed, the following example is given in Sirakov [94], and Montefusco, Pellacci and Squassina [79]; let $V_1(x) \equiv V_2(x) \equiv 1$, $\mu_1 = \mu_2 = \beta = 1$ and ω_1 be a positive solution of

$$-\Delta u_1 + u_1 = u_1^3 \quad \text{in } \mathbf{R}^N.$$

Then $u(x) = (u_1(x), u_2(x)) = (\omega_1(x) \cos \theta, \omega_1(x) \sin \theta)$ is a nontrivial positive solution of (3.1.1)–(3.1.4) for any $\theta \in (0, \pi/2)$.

We say that a nontrivial positive solution $u = (u_1, u_2) \in \mathcal{H}_{V_1, V_2, r}$ of (3.1.1)–(3.1.4) is nondegenerate in the function space $\mathcal{H}_{V_1, V_2, r}$ if and only if the linearized system

$$\begin{cases} -\Delta w_1 + V_1(x)w_1 = 3\mu_1 u_1^2 w_1 + \beta (u_2^2 w_1 + 2u_1 u_2 w_2) & \text{in } \mathbf{R}^N, \\ -\Delta w_2 + V_2(x)w_2 = \beta (2u_1 u_2 w_1 + u_1^2 w_2) + 3\mu_2 u_2^2 w_2 & \text{in } \mathbf{R}^N, \\ (w_1, w_2) \in \mathcal{H}_{V_1, V_2, r} \end{cases}$$

has only trivial solution $w_1 \equiv w_2 \equiv 0$.

As to the nondegeneracy of our unique radial positive solution, we have

Theorem 3.1.6. The unique nontrivial radial positive solution of (3.1.1)–(3.1.4) is nondegenerate in $\mathcal{H}_{V_1,V_2,r}$ for $\beta \in (0, \beta_0)$.

To prove our Theorems 3.1.1, 3.1.4 and 3.1.6, a priori estimates and uniform exponential decays of nontrivial radial positive solutions are important. In section 3.2, we will obtain a priori L^{∞} estimates and uniform exponential decays of

 $\mathcal{S}_{\bar{\beta}} := \{ u \in \mathcal{H}_{V_1, V_2, r} : u \text{ is a nontrivial positive solution of } (3.1.1) - (3.1.4)$ for some $\beta \in [0, \bar{\beta}] \}.$

Here Liouville type result (Lemma 3.2.1) and monotonicity of solutions play roles. In section 3.3, we prove Theorems 3.1.1, 3.1.4 and 3.1.6. The behavior of nontrivial positive solutions as $\beta \to 0$ is a key of our argument.

3.2 A priori bounds for nontrivial positive solutions

In this section, we assume that $1 \le N \le 3$ and $V_j(x)$ satisfies (3-V1)-(3-V5).

3.2.1 L^{∞} bound of nontrivial positive solutions

First, we introduce an inner product in $\mathcal{H}_{V_1,V_2,r}$: for $u, v \in \mathcal{H}_{V_1,V_2,r}$, we define $\langle \cdot, \cdot \rangle_{\mathcal{H}_{V_1,V_2,r}}$ by

$$\langle u, v \rangle_{\mathcal{H}_{V_1, V_2, r}} := \int_{\mathbf{R}^N} \nabla u_1 \cdot \nabla v_1 + V_1(x) u_1 v_1 + \nabla u_2 \cdot \nabla v_2 + V_2(x) u_2 v_2 dx.$$

Then it is easily seen that $\mathcal{H}_{V_1,V_2,r}$ is a Hilbert space with $\langle \cdot, \cdot \rangle_{\mathcal{H}_{V_1,V_2,r}}$ and we denote its norm by $\| \cdot \|_{\mathcal{H}_{V_1,V_2,r}}$. Next, for $\bar{\beta} > 0$ we define $\mathcal{S}_{\bar{\beta}}$ as follows:

$$\mathcal{S}_{\bar{\beta}} := \{ u \in \mathcal{H}_{V_1, V_2, r} : u \text{ is a nontrivial positive solution of } (3.1.1) - (3.1.4) \\ \text{for some } \beta \in [0, \bar{\beta}],$$

The following lemma is essential to prove uniform L^{∞} estimates and uniform exponential decays of $u \in S_{\bar{\beta}}$.

Lemma 3.2.1. Let $1 < \alpha \leq 3$ if N = 3 and $1 < \alpha < \infty$ if N = 1, 2. Then there is no positive function such that

(3.2.1)
$$-\Delta u \ge |u|^{\alpha - 1} u \text{ in } \mathbf{R}^N.$$

Proof. When N = 3, we refer to Theorem 8.4 in Quittner and Souplet [90]. When N = 2, the conclusion of Lemma 3.2.1 follows from Liouville's theorem. For example, see Protter and Weinberger [88]. When N = 1, Lemma 3.2.1 can be shown easily.

The following proposition gives us a uniform a priori L^{∞} -bound of $S_{\bar{\beta}}$.

Proposition 3.2.2. For any $\bar{\beta} > 0$ there exists an $M_{\bar{\beta}} > 0$ such that the following inequality

$$||u_1||_{L^{\infty}} + ||u_2||_{L^{\infty}} \le M_{\bar{\beta}}$$

holds for all $u \in S_{\bar{\beta}}$.

Proof. We prove indirectly and assume that there exist $(u_k) \subset S_{\bar{\beta}}$ and (β_k) such that $\beta_k \to \beta_0$ and $||u_{k,1}||_{L^{\infty}} + ||u_{k,2}||_{L^{\infty}} \to \infty$. Furthermore, without loss of generality, we may assume $||u_{k,2}||_{L^{\infty}} \leq ||u_{k,1}||_{L^{\infty}} \to \infty$. For j = 1, 2, we set

$$\eta_k := \frac{1}{\|u_{k,1}\|_{L^{\infty}}}, \ v_{k,j}(x) := \eta_k u_{k,j}(\eta_k x).$$

By Remark 3.1.5, (u_k) is radially symmetric and monotone decreasing, and we have $v_{k,1}(0) = 1$ and $||v_{k,j}||_{L^{\infty}} \leq 1$ (j = 1, 2). Moreover, since u_k is a solution of (3.1.1)–(3.1.2), v_k satisfies

$$\begin{cases} -\Delta v_{k,1} + \eta_k^2 V_1(\eta_k x) v_{k,1} = \mu_1 v_{k,1}^3 + \beta_k v_{k,1} v_{k,2}^2, \\ -\Delta v_{k,2} + \eta_k^2 V_2(\eta_k x) v_{k,2} = \beta_k v_{k,1}^2 v_{k,2} + \mu_2 v_{k,2}^3. \end{cases}$$

By the standard elliptic argument, if necessary take a subsequence, it holds that $v_{k,j} \to v_{0,j}$ in $C^2_{\text{loc}}(\mathbf{R}^N)$ where $v_{0,j}$ is a nonnegative solution of

$$\begin{cases} -\Delta v_{0,1} = \mu_1 v_{0,1}^3 + \beta_0 v_{0,1} v_{0,2}^2, \\ -\Delta v_{0,2} = \beta_0 v_{0,1}^2 v_{0,2} + \mu_2 v_{0,2}^3. \end{cases}$$

Since $v_{0,1}(0) = 1$ and the maximum principle, $v_{0,1} > 0$ in \mathbb{R}^N . So $v_{0,1}(x)$ satisfies the following differential inequality:

$$-\Delta v_{0,1} \ge \mu_1 v_{0,1}^3$$
 in \mathbf{R}^N

This contradicts Lemma 3.2.1, so the conclusion of Proposition 3.2.2 holds.

3.2.2 Uniform exponential decay estimates in $S_{\bar{\beta}}$.

As pointed in Remark 3.1.5, any nontrivial positive solution of (3.1.1)-(3.1.4) is a radially symmetric function. So we rewrite (3.1.1)-(3.1.2) as follows:

(3.2.2)
$$\begin{cases} -u_1''(r) - \frac{N-1}{r}u_1'(r) + V_1(r)u_1(r) = \mu_1 u_1^3(r) + \beta u_1(r)u_2^2(r), \\ -u_2''(r) - \frac{N-1}{r}u_2'(r) + V_2(r)u_2(r) = \beta u_1^2(r)u_2(r) + \mu_2 u_2^3(r). \end{cases}$$

The following proposition gives us a uniform exponential decay of $S_{\bar{\beta}}$.

Proposition 3.2.3. For any $\bar{\beta} > 0$ there exist $C_1 = C_1(\bar{\beta})$ and $C_2 = C_2(\bar{\beta}) > 0$ such that

$$|u_1(r)| + |u_2(r)| \le C_1 \exp(-C_2 r), \quad |u_1'(r)| + |u_2'(r)| \le C_1 \exp(-C_2 r)$$

for all $u \in S_{\bar{\beta}}$.

Proof. The proof is divided into 2 steps.

Step 1 Let $\beta \geq 0$ and $u(r) \in \mathcal{H}_{V_1,V_2,r}$ be a nontrivial positive solution of (3.1.1)–(3.1.4). There exist $C_3 = C_3(u), C_4 = C_4(u) > 0$ such that

$$(3.2.3) |u_1(r)| + |u_2(r)| \le C_3 \exp(-C_4 r), |u_1'(r)| + |u_2'(r)| \le C_3 \exp(-C_4 r).$$

Let $\beta \geq 0$ and $u \in \mathcal{H}_{V_1,V_2,r}$ be a nontrivial positive solution of (3.2.2). We prove that $u_1(r)$ and $u'_1(r)$ decay exponentially. We follow arguments in Tanaka [99]. Since $u_j(r) \to 0$ as $r \to \infty$ and (3–V1) holds, there exists an $r_0 \geq 1$ such that for $r \geq r_0$, we have

(3.2.4)
$$u_1(r), \ u_2(r) \le 1, \quad 0 < \frac{V_1(r_0)}{2} \le V_1(r) - \mu_1 u_1^2(r) - \beta u_2^2(r).$$

Thus it holds

(3.2.5)
$$0 = -u_1''(r) - \frac{N-1}{r}u_1'(r) + \left(V_1(r) - \mu_1^2 u_1^2(r) - \beta u_2^2(r)\right)u_1(r)$$
$$\geq -u_1''(r) - \frac{N-1}{r}u_1'(r) + \frac{V_1(r_0)}{2}u_1(r) \quad \text{for } r \geq r_0.$$

Let $\delta > 0$ satisfy $\max{\{\delta^2, (N-1)\delta\}} < V_1(r_0)/4$ and for any $R > r_0$ we set

$$\varphi_{R,\delta}(r) := \exp(-\delta(r-r_0)) + \exp(-\delta(R-r)).$$

Then we have

(3.2.6)
$$-\varphi_{R,\delta}''(r) - \frac{N-1}{r}\varphi_{R,\delta}'(r) + \frac{V_1(0)}{2}\varphi_{R,\delta}(r) \ge 0 \quad \text{for all } r \in [r_0, R].$$

Since $\varphi_{R,\delta}(r_0) \ge 1$, $\varphi_{R,\delta}(R) \ge 1$ and (3.2.4)–(3.2.6) hold, by the comparison theorem, it follows

(3.2.7)
$$u_1(r) \le \varphi_{R,\delta}(r) \quad \text{for all } r \in [r_0, R].$$

Since $R > r_0$ is arbitrary, let $R \to \infty$, then we obtain

$$u_1(r) \le \exp(-\delta(r-r_0))$$
 for all $r \ge r_0$.

Thus $u_1(r)$ has an exponential decay. In a similar way, we can show that $u_2(r)$ has an exponential decay.

Next we will show that $u'_1(r)$ has an exponential decay. We follow the arguments in Berestycki and Lions [15]. By (3.2.2), it holds that

(3.2.8)
$$(r^{N-1}u_1'(r))' = r^{N-1}u_1(r)(V_1(r) - \mu_1 u_1^2(r) - \beta u_2^2(r)).$$

Hence the function $(r^{N-1}u'_1(r))'$ has an exponential decay, which implies that the limit $\lim_{r\to\infty} r^{N-1}u'_1(r)$ exists. Since u_1 has an exponential decay, it holds $r^{N-1}u'_1(r) \to 0$ as $r \to \infty$. Otherwise, u_1 does not have an exponential decay. Therefore we obtain

$$r^{N-1}u_1'(r) = -\int_r^\infty (s^{N-1}u_1'(s))'ds,$$

which implies that $u'_1(r)$ has an exponential decay. In a similar way, it follows that $u'_2(r)$ has an exponential decay.

Next we prove that for any $\overline{\beta} > 0$ we can choose C_3 , C_4 in (3.2.3) uniformly with respect to $u \in S_{\overline{\beta}}$, that is, we complete the proof of Proposition 3.2.3.

Step 2 Conclusion.

We prove that there exist $C_3 = C_3(\bar{\beta}), C_4 = C_4(\bar{\beta}) > 0$ such that

(3.2.9)
$$|u_1(r)| + |u_2(r)| \le C_3 \exp(-C_4 r)$$

for all $u \in S_{\bar{\beta}}$. From (3.2.8) and (3.2.9) it follows that

$$|u_1'(r)| + |u_2'(r)| \le C_3 \exp(-C_4 r)$$

for all $u \in S_{\bar{\beta}}$.

We prove (3.2.9) indirectly and assume that there exist $(u_k) \subset S_{\bar{\beta}}$ and (β_k) such that $\beta_k \to \beta_0$ and

(3.2.10)
$$u_{k,1}(r_k) + u_{k,2}(r_k) > k \exp(-r_k/k) \text{ for some } r_k \ge 0.$$

Since (u_k) satisfies (3.2.2) with $\beta = \beta_k$ and $||u_k||_{L^{\infty}}$ is bounded by Proposition 3.2.2, we can also assume that $u_{k,j} \to u_{0,j}$ in $C^2_{\text{loc}}(\mathbf{R}^N)$, where u_0 is a nonnegative solution of

(3.2.11)
$$\begin{cases} -u_{0,1}''(r) - \frac{N-1}{r}u_{0,1}'(r) + V_1(r)u_{0,1}(r) = \mu_1 u_{0,1}^3(r) + \beta_0 u_{0,1}(r)u_{0,2}^2(r), \\ -u_{0,2}''(r) - \frac{N-1}{r}u_{0,2}'(r) + V_2(r)u_{0,2}(r) = \beta_0 u_{0,1}^2(r)u_{0,2}(r) + \mu_2 u_{0,2}^3(r). \end{cases}$$

We set

$$E_k(r) := \sum_{j=1}^2 \left(\frac{1}{2} (u'_{k,j}(r))^2 + \frac{\mu_j}{4} u^4_{k,j} - \frac{1}{2} V_j(r) u^2_{k,j}(r) \right) + \frac{\beta_k}{2} u^2_{k,1}(r) u^2_{k,2}(r),$$

$$E_0(r) := \sum_{j=1}^2 \left(\frac{1}{2} (u'_{0,j}(r))^2 + \frac{\mu_j}{4} u^4_{0,j} - \frac{1}{2} V_j(r) u^2_{0,j}(r) \right) + \frac{\beta_0}{2} u^2_{0,1}(r) u^2_{0,2}(r).$$

Since $u_k \to u_0$ in $C^2_{\text{loc}}(\mathbf{R}^N)$, $E_k \to E_0$ in $C^1_{\text{loc}}(\mathbf{R}^N)$. By (3–V3) and (3.2.2) with $\beta = \beta_k$, we have

$$E'_{k}(r) = \sum_{j=1}^{2} \left(u''_{k,j}(r) + \mu_{j} u^{3}_{k,j}(r) - V_{j}(r) u_{k,j}(r) \right) u'_{k,j}(r) - \sum_{j=1}^{2} \frac{1}{2} V'_{j}(r) u^{2}_{k,j}(r) + \beta_{k} u_{k,1} u^{2}_{k,2}(r) u'_{k,1}(r) + \beta_{k} u^{2}_{k,1}(r) u_{k,2}(r) u'_{k,2}(r) = -\frac{N-1}{r} \sum_{j=1}^{2} (u'_{k,1}(r))^{2} - \sum_{j=1}^{2} \frac{1}{2} V'_{j}(r) u^{2}_{k,j}(r) \le 0.$$

Since u_k and u'_k have an exponential decay for each $k \ge 1$ by Step 1, it holds that $\lim_{r\to\infty} E_k(r) = 0$. By the monotonicity of E_k , it follows that $E_k(r) \ge 0$ for $r \ge 0$, which implies that $E_0(r) \ge 0$ for $r \ge 0$.

By [20], $u_k(r)$ is a decreasing function when N = 2, 3. We remark that this result holds even when N = 1. See Theorem 3.4.1 in section 3.4. Thus $u_0(r)$ is also a nonincreasing function. So we set

(3.2.12)
$$\lim_{r \to \infty} u_{0,j}(r) =: u_{0,\infty,j} \ge 0 \quad (j = 1, 2).$$

and next we claim

$$u_{0,\infty,1} = u_{0,\infty,2} = 0.$$

Since $E_0(r)$ and $u_{0,j}(r)$ are bounded in $[0, \infty)$, $u'_0(r)$ is also bounded in $[0, \infty)$. Furthermore, $u'_{0,j}(r)$ is bounded since $E_0(r)$ and $u_{0,j}(r)$ converge as $r \to \infty$.

Here we consider 3 cases.

Case 1 $V_{j,\infty} := \lim_{r \to \infty} V_j(r) \in (0,\infty)$ for j = 1, 2.

By (3.2.11), $u_{0,\infty,1}$ and $u_{0,\infty,2}$ satisfy

(3.2.13)
$$\begin{aligned} u_{0,\infty,1}(\mu_1 u_{0,\infty,1}^2 + \beta_0 u_{0,\infty,2}^2 - V_{1,\infty}) &= 0, \\ u_{0,\infty,2}(\beta_0 u_{0,\infty,1}^2 + \mu_2 u_{0,\infty,2}^2 - V_{2,\infty}) &= 0. \end{aligned}$$

Then we have

$$(u_{0,\infty,1}, u_{0,\infty,2}) \in \left\{ (0,0), \left(\sqrt{\frac{V_{1,\infty}}{\mu_1}}, 0\right), \left(0, \sqrt{\frac{V_{2,\infty}}{\mu_2}}\right), (u_1^*, u_2^*) \right\}$$

where (u_1^*, u_2^*) satisfies

$$V_{1,\infty} = \mu_1 (u_1^*)^2 + \beta_0 (u_2^*)^2$$

$$V_{2,\infty} = \beta_0 (u_1^*)^2 + \mu_2 (u_2^*)^2.$$

If $(u_{0,\infty,1}, u_{0,\infty,2}) \neq (0,0)$, then it is easily seen that $\lim_{r \to \infty} E_0(r) < 0$, which contradicts to $E_0(r) \ge 0$. Thus $u_{0,\infty,j} = 0$ (j = 1, 2).

Case 2 $V_{j,\infty} = \infty$ for j = 1, 2.

In this case, $u_{0,\infty,j} = 0$ follows easily. In fact, Since $E_0(r) \ge 0$ for all $r \ge 0$, it must be $u_{0,\infty,j} = 0$ for j = 1, 2.

Case 3 $V_{i_0,\infty} = \infty$ and $V_{j_0,\infty} < \infty$. Here $\{i_0, j_0\} = \{1, 2\}$.

Without loss of generality, we can assume $i_0 = 1$ and $j_0 = 2$. As in Case 2, we may infer that $u_{0,\infty,1} = 0$. Furthermore, by (3.2.11), it follows that

$$u_{0,\infty,2}(\mu_2 u_{0,\infty,2}^2 - V_{2,\infty}) = 0.$$

Now we suppose that $u_{0,\infty,2} = \sqrt{V_{2,\infty}/\mu_2}$. Since $V_1(r) \to \infty$ as $r \to \infty$ and $u_{0,2}$ is bounded, we can choose $r_1 > 0$ such that

$$0 < \inf_{r \ge r_1} \left\{ V_1(r) - \mu_1 u_{0,1}^2(r) - \beta u_{0,2}^2(r) \right\}.$$

Hence, as in Step 1, it holds that there exist $C_1, C_2 > 0$ such that

$$u_{0,1}(r) \le C_1 \exp(-C_2 r) \quad \text{for all } r \ge 0$$

By (3-V4), we obtain

$$\lim_{r \to \infty} E_0(r) = -\frac{V_{\infty,2}^2}{4\mu_2} < 0,$$

which contradicts that $E_0(r) \ge 0$ for all $r \ge 0$. Therefore we have $u_{0,\infty,2} = 0$.

By $\lim_{r\to\infty} u_{0,j}(r) = 0$, as in Step 1, we can prove that there exist $\bar{C}_1, \bar{C}_2 > 0$ such that

 $|u_{0,1}(r)| + |u_{0,2}(r)| \le \bar{C}_1 \exp(-\bar{C}_2 r).$

Moreover, since $u_{k,j}(r)$ is a decreasing function and $u_{k,j} \to u_{0,j}$ in $C^2_{\text{loc}}(\mathbf{R}^N)$, there exist $r_2 > 0, \ k_0 \in \mathbf{N}$ such that

$$\sup_{r \ge r_2} \{ u_{k,1}(r) + u_{k,2}(r) \} \le 1 \qquad \text{for all } k \text{ with } k \ge k_0, \\ 0 < \inf_{r \ge r_2} \{ V_1(r) - \mu_1 u_{k,1}^2(r) - \beta_k u_{k,2}^2(r) \} \quad \text{for all } k \text{ with } k \ge k_0.$$

Hence, we can show that there exist \bar{C}_3 , $\bar{C}_4 > 0$ such that

$$|u_{k,1}(r)| + |u_{k,2}(r)| \le \bar{C}_3 \exp(-\bar{C}_4 r)$$

for all $k \ge 1$ and $r \ge 0$. However this is a contradiction to (3.2.10) and (3.2.9) holds. \Box

As a corollary to Proposition 3.2.3, we have

Corollary 3.2.4. For any $\bar{\beta} > 0$, $S_{\bar{\beta}}$ is bounded in $\mathcal{H}_{V_1,V_2,r}$. Moreover it has the following compactness property : for any sequence $(u_k) \subset S_{\bar{\beta}}$, there exists a strongly convergent subsequence (u_{k_m}) in $\mathcal{H}_{V_1,V_2,r}$.

Proof. By Proposition 3.2.3 and (3–V4), it is clear that (u_k) is bounded in $\mathcal{H}_{V_1,V_2,r}$ and for any $\varepsilon > 0$ there exists an $R_{\varepsilon} > 0$ such that

$$\sum_{j=1}^{2} \int_{|x| \ge R_{\varepsilon}} |\nabla u_{k,j}(x)|^2 + V_j(x) u_{k,j}^2(x) dx < \varepsilon \quad \text{for } k \ge 1.$$

Moreover, since (u_k) is bounded in $\mathcal{H}_{V_1,V_2,r}$, there exists a subsequence (u_{k_m}) such that $u_{k_m} \rightharpoonup u_0$ weakly in $\mathcal{H}_{V_1,V_2,r}$. Since (u_k) satisfies (3.2.2) and is bounded in L^{∞} , we can assume $u_{k_m} \rightarrow u_0$ in $C^2_{\text{loc}}(\mathbf{R}^N)$. Therefore we have $u_{k_m} \rightarrow u_0$ strongly in $\mathcal{H}_{V_1,V_2,r}$.

3.3 Proof of Theorems 3.1.1, 3.1.4 and 3.1.6

In this section, we assume that $V_j(x)$ is a positive constant function if N = 1 and $V_j(x)$ satisfies (3–V1)–(3–V5) if N = 2, 3.

3.3.1 Nondegeneracy of solutions when $\beta = 0$

In this subsection, we consider the following limit equations as $\beta \to 0$ in (3.1.1)–(3.1.4):

(3.3.1)
$$\begin{cases} -\Delta v_j + V_j(x)v_j = \mu_j v_j^3 & \text{in } \mathbf{R}^N, \\ v_j \in \mathcal{H}_{V_j,r}, v_j > 0. \end{cases}$$

Here

$$\mathcal{H}_{V_j,r} := \left\{ u_j \in H^1(\mathbf{R}^N) : u_j(x) = u_j(|x|), \int_{\mathbf{R}^N} V_j(x) u_j^2 dx < \infty \right\}.$$

Definition 3.3.1. A solution ω_j of (3.3.1) is *nondegenerate* in $\mathcal{H}_{V_j,r}$ if the following equation has only a trivial solution $\psi_j \equiv 0$:

$$\begin{cases} -\Delta \psi_j + V_j(x)\psi_j = 3\mu_j \omega_j^2 \psi_j & \text{in } \mathbf{R}^N, \\ \psi_j \in \mathcal{H}_{V_j,r}. \end{cases}$$

The following proposition is due to Byeon and Ohshita [25], and Kabeya and Tanaka [57].

Proposition 3.3.2. If $V_j(x)$ satisfies (3–V1)–(3–V5) and N = 2, 3, then (3.3.1) has a unique radial positive solution $\omega_j(x) \in \mathcal{H}_{V_j,r}$. Moreover, $\omega_j(x)$ is nondegenerate in $\mathcal{H}_{V_j,r}$.

The following proposition is well-known. See Willem [110] and Kwong [60].

Proposition 3.3.3. Suppose that $1 \leq N \leq 3$, and $V_j(x)$ is positive and independent of x. Then (3.3.1) has a unique positive solution $\omega_j(x)$. Moreover, $\omega_j(x)$ is nondegenerate in $H_r^1(\mathbf{R}^N)$.

By Propositions 3.3.2 and 3.3.3, we have

Corollary 3.3.4. Under the assumptions in Theorem 3.1.1 or 3.1.4, (3.1.1)–(3.1.4) with $\beta = 0$ has a unique nontrivial positive solution $\omega(x) := (\omega_1(x), \omega_2(x))$ in $\mathcal{H}_{V_1, V_2, r}$ and $\omega(x)$ is nondegenerate in $\mathcal{H}_{V_1, V_2, r}$.

Remark 3.3.5. We set

$$\begin{split} I_j(\psi_j) &:= \frac{1}{2} \int_{\mathbf{R}^N} |\nabla \psi_j|^2 + V_j(x) \psi_j^2 dx - \frac{\mu_j}{4} \int_{\mathbf{R}^N} \psi_j^4 dx, \\ I_0(\psi) &:= I_0(\psi_1, \psi_2) = I_1(\psi_1) + I_2(\psi_2). \end{split}$$

Then solutions $\omega_j \in \mathcal{H}_{V_j,r}$ of (3.3.1) and $\omega(x)$ of (3.1.1)–(3.1.4) with $\beta = 0$ are nondegenerate in $\mathcal{H}_{V_j,r}$ and $\mathcal{H}_{V_1,V_2,r}$ if and only if the mapping

$$\psi_j \mapsto I_j''(\omega_j)[\psi_j, \cdot] : \mathcal{H}_{V_j, r} \to (\mathcal{H}_{V_j, r})^*, \psi \mapsto I_0''(\omega)[\psi, \cdot] : (\mathcal{H}_{V_1, V_2, r})^2 \to (\mathcal{H}_{V_1, V_2, r})^*$$

are invertible.

3.3.2 Proof of Theorems 3.1.1, 3.1.4 and 3.1.6

We prove Theorems 3.1.1, 3.1.4 and 3.1.6.

Proof of Theorems 3.1.1, 3.1.4 and 3.1.6. Let $(u_{\beta}) = (u_{\beta,1}, u_{\beta,2})$ be a nontrivial positive solution of (3.1.1)–(3.1.4). By Corollary 3.2.4, there exist (u_k) and (β_k) such that $u_k \to u_0$ in $\mathcal{H}_{V_1,V_2,r}$ and $\beta_k \to 0$. Since (u_k) satisfies (3.1.1)–(3.1.4) with $\beta = \beta_k$, we have

$$||u_{k,1}||^2_{\mathcal{H}_{V_1,r}} = \mu_1 ||u_{k,1}||^4_{L^4} + \beta_k ||u_{k,1}u_{k,2}||^2_{L^2}$$

Using Sobolev's embedding, Hölder's inequality and (3–V2), it holds that

$$||u_{k,1}||_{\mathcal{H}_{V_1,r}} \le C_1(||u_{k,1}||_{\mathcal{H}_{V_1,r}}^4 + \beta_k ||u_{k,1}||_{\mathcal{H}_{V_1,r}}^2 ||u_{k,2}||_{\mathcal{H}_{V_2,r}}^2).$$

Since $\beta_k \to 0$ and (u_k) is bounded in $\mathcal{H}_{V_1,V_2,r}$, there exists a $C_2 > 0$ such that

$$C_2 \leq ||u_{k,1}||_{\mathcal{H}_{V_1,r}}$$
 for all $k \in \mathbf{N}$.

Similar arguments lead that

$$0 < C_3 \le \|u_{k,2}\|_{\mathcal{H}_{V_2,r}} \quad \text{for all } k \in \mathbf{N}.$$

Since $u_{k,j} \to u_{0,j}$ in $C^2_{\text{loc}}(\mathbf{R}^N)$ and $\mathcal{H}_{V_1,V_2,r}$, we deduce that $u_{0,j} \ge 0$ and $u_{0,j} \not\equiv 0$ for j = 1, 2. By the maximum principle, we have $u_{0,j}(x) > 0$ in \mathbf{R}^N . By Propositions 3.3.2 and 3.3.3, we have $u_{0,j} = \omega_j$. This implies that

$$u_{\beta} \to \omega = (\omega_1, \omega_2)$$
 strongly in $\mathcal{H}_{V_1, V_2, r}$ as $\beta \to 0$.

Thus, for any $\varepsilon > 0$ there exists a $\tilde{\beta} = \tilde{\beta}(\varepsilon) > 0$ such that for any $\beta \in [0, \tilde{\beta}]$ and any nontrivial positive solution $u \in \mathcal{H}_{V_1, V_2, r}$ of (3.1.1)–(3.1.4) satisfies

$$\|u_j - \omega_j\|_{\mathcal{H}_{V_1, V_2, r}} < \varepsilon.$$

Set

$$I_{\beta}(u) := I_{1}(u_{1}) + I_{2}(u_{2}) - \frac{\beta}{2} \int_{\mathbf{R}^{N}} u_{1}^{2} u_{2}^{2} dx,$$

$$\Phi(\beta, u) := I_{\beta}'(u) : \ \mathbf{R} \times \mathcal{H}_{V_{1}, V_{2}, r} \to (\mathcal{H}_{V_{1}, V_{2}, r})^{*}.$$

Then it is clear that $\Phi(0,\omega) = 0$. Moreover, by Corollary 3.3.4 and Remark 3.3.5, $\Phi_u(0,\omega) = I_0''(\omega)$ is invertible. By the implicit function theorem, there exist $\beta_0 > 0$, $r_0 > 0$ and $\phi : (-\beta_0, \beta_0) \to B_{r_0}(\omega)$ such that

- 1. $\phi(0) = \omega,$
- 2. $\Phi(\beta, \phi(\beta)) = 0$ for all $\beta \in (-\beta_0, \beta_0)$,
- 3. For any $\beta \in (-\beta_0, \beta_0)$, $\Phi(\beta, u) = 0$ has a unique solution $u = \phi(\beta)$ in $B_{r_0}(\omega)$.
- 4. The mapping $\psi \mapsto I_0''(u)[\psi, \cdot] : \mathcal{H}_{V_1, V_2, r} \to (\mathcal{H}_{V_1, V_2, r})^*$ is invertible in $B_{r_0}(\omega)$.

By (3.3.2) and the above properties, we conclude that (3.1.1)–(3.1.4) has a unique radial nontrivial solution for any $\beta \in [0, \beta_0)$. The nondegeneracy of solution also follows from the above properties.

3.4 Symmetry and monotonicity of nontrivial positive solutions (N = 1)

In this section, we consider the symmetry and monotonicity of nontrivial positive solutions when N = 1. We consider the following equations:

(3.4.1)
$$\begin{cases} u_1'' + f_1(x, u_1(x), u_2(x)) = 0 & \text{in } \mathbf{R}, \\ u_2'' + f_2(x, u_1(x), u_2(x)) = 0 & \text{in } \mathbf{R}, \\ u_1(x), u_2(x) > 0 & \text{in } \mathbf{R}, \\ u_1(x), u_2(x) \to 0 & \text{as } |x| \to \infty. \end{cases}$$

Here we assume that f_i satisfies the following conditions:

(f0) $f_i(x, s_1, s_2) \in C^1(\mathbf{R} \times (0, \infty)^2)$ and $f_i(-x, s_1, s_2) = f_i(x, s_1, s_2)$ for any $x \in \mathbf{R}$, $(s_1, s_2) \in (0, \infty)^2$ and i = 1, 2.

(f1)
$$\frac{\partial f_i}{\partial x}(x, s_1, s_2) \le 0$$
 for all $x \ge 0$, $(s_1, s_2) \in (0, \infty)^2$ and $i = 1, 2$.

(f2)
$$\frac{\partial f_1}{\partial s_2}(x, s_1, s_2), \frac{\partial f_2}{\partial s_1}(x, s_1, s_2) \ge 0$$
 for all $x \in \mathbf{R}$, $(s_1, s_2) \in (0, \infty)^2$ and $i = 1, 2$.

(f3) There exist $R_1 > 0$, $\varepsilon_1 > 0$ and $\delta_1 > 0$ such that if $|x| \ge R_1$ and $s_1^2 + s_2^2 < \varepsilon_1^2$, then

$$\frac{\partial f_i}{\partial s_i}(x, s_1, s_2) < -\delta_1 \text{ for } i = 1, 2, \quad 0 < \frac{\partial f_i}{\partial s_j}(x, s_1, s_2) \le \delta_1 \text{ for } i \neq j.$$

In this section we will prove the following theorem:

Theorem 3.4.1. Suppose that f_i satisfies (f0)–(f3). Then any nontrivial solution of (3.4.1) is even and monotone decreasing.

Remark 3.4.2. We can apply Theorem 3.4.1 to (3.1.1)-(3.1.4) with N = 1 under (3-V1)-(3-V5).

In [20], they considered the symmetry and monotonicity of nontrivial positive solutions for $N \ge 2$. In the following, we show the argument in [20] still works for N = 1 after modification and give an outline of the proof of Theorem 3.4.1.

We use the moving plane method to prove the symmetry and monotonicity of nontrivial positive solutions of (3.4.1). Let $u(x) = (u_1(x), u_2(x))$ be a solution of (3.4.1) and we set

$$x^{\lambda} := 2\lambda - x,$$

$$U_i^{\lambda}(x) := u_i(x^{\lambda}) - u_i(x) = u_i(2\lambda - x) - u_i(x)$$

for $\lambda \geq 0$ and $x \geq \lambda$.

Outline of the proof of Theorem 3.4.1. First, we observe that $U_i^{\lambda}(x)$ satisfies

(3.4.2)
$$(U_i^{\lambda})'' + \sum_{k=1}^2 c_{ik}(x) U_k^{\lambda} \le 0 \text{ in } (\lambda, \infty),$$

where

$$c_{ik}(x) := \int_0^1 \frac{\partial f_i}{\partial u_k} (|x| + t(|x^{\lambda}| - |x|), u(x) + tU^{\lambda}(x)) dt,$$

$$U^{\lambda}(x) := (U_1^{\lambda}(x), U_2^{\lambda}(x)).$$

These inequalities are corresponding to the inequality (13) in [20] and we can show (3.4.2) in a similar way to [20].

Next we define

$$A := \{\lambda \ge 0 : U_i^{\mu} \ge 0 \text{ in } (\mu, \infty) \text{ for all } \mu \ge \lambda, \ i = 1, 2\},\$$

and a key step of the proof of Theorem 3.4.1 is to show $A \neq \emptyset$.

Here we need to modify the argument in [20]. In [20], they introduced the auxiliary function $\bar{U}^{\lambda}(x) := U^{\lambda}(x)/g(x)$, where

$$g(x) := \begin{cases} |x|^{-(N-2)/2} + 1 & \text{if } N \ge 3, \\ \log(\log(|x| + 27)) & \text{if } N = 2. \end{cases}$$

When N = 1, g(x) is not given in [20].

Under our assumption (f3), which is stronger than those in [20], we can show there exist $R_2 \ge R_1$ and $\lambda_1 > R_2$ such that

(3.4.3)
$$u_{1}^{2}(x) + u_{2}^{2}(x) < \varepsilon_{1} \quad \text{for } |x| \ge R_{2}, \\ \max_{[2\lambda - R_{2}, 2\lambda + R_{2}]} u_{i} < \min_{[-R_{2}, R_{2}]} u_{i} \quad \text{for } \lambda \ge \lambda_{1}, \\ c_{11}(x), c_{22}(x) < -\delta_{1}, \quad 0 < c_{12}(x), c_{21}(x) \le \delta_{1} \\ \text{for } \lambda \ge \lambda_{1} \text{ and } x, x^{\lambda} \notin [-R_{2}, R_{2}].$$

From (3.4.3), we have

(3.4.4)
$$\det \begin{pmatrix} c_{11}(x) & c_{12}(x) \\ c_{21}(y) & c_{22}(y) \end{pmatrix} > 0 \quad \text{for } \lambda \ge \lambda_1 \text{ and } x, x^{\lambda}, y, y^{\lambda} \notin [-R_2, R_2].$$

We will see that (3.4.4) enables us to show $A \neq \emptyset$ without introducing $\overline{U}^{\lambda}(x)$.

Indeed, we show $\lambda_1 \in A$ to prove $A \neq \emptyset$. We show indirectly and assume $\lambda_1 \notin A$. Then there exist a $\lambda_2 \geq \lambda_1$ and $x_1, x_2 \in [\lambda_2, \infty)$ and one of the following three cases takes place:

(i)
$$U_1^{\lambda_2}(x_1) = \min_{[\lambda_2,\infty)} U_1^{\lambda_2} < 0, \quad \min_{[\lambda_2,\infty)} U_2^{\lambda_2} = 0,$$

(ii) $\min_{[\lambda_2,\infty)} U_1^{\lambda_2} = 0, \quad U_2^{\lambda_2}(x_2) = \min_{[\lambda_2,\infty)} U_2^{\lambda_2} < 0,$
(iii) $U_1^{\lambda_2}(x_1) = \min_{[\lambda_2,\infty)} U_1^{\lambda_2} < 0, \quad U_2^{\lambda_2}(x_2) = \min_{[\lambda_2,\infty)} U_2^{\lambda_2} < 0.$

We consider the case (i). Since x_1 is a minimum point of $U_1^{\lambda_2}$ in (λ_2, ∞) and the inequality (3.4.2), $U_2^{\lambda_2}(x_1)c_{12}(x_1) \ge 0$ hold, we have

(3.4.5)
$$c_{11}(x_1)U_1^{\lambda_2}(x_1) \le 0.$$

On the other hand, by (3.4.3), $U_1^{\lambda_2}(x_1) < 0$ implies that $x_1, x_1^{\lambda_2} \notin [-R_2, R_2]$ and $c_{11}(x_1) < -\delta_1$, which contradicts (3.4.5). Thus the case (i) never occurs. Similarly, the case (ii) cannot occur.

In the case (iii), Since x_i is a minimum point of $U_i^{\lambda_2}$, it follows

$$\begin{pmatrix} c_{11}(x_1) & c_{12}(x_1) \\ c_{21}(x_2) & c_{22}(x_2) \end{pmatrix} \begin{pmatrix} U_1^{\lambda_2}(x_1) \\ U_2^{\lambda_2}(x_2) \end{pmatrix} = \begin{pmatrix} \xi_1 \\ \xi_2 \end{pmatrix} \quad \text{where } \xi_1, \xi_2 \le 0.$$

By (3.4.3), we have

(3.4.6)
$$x_1, x_2, x_1^{\lambda_2}, x_2^{\lambda_2} \notin [-R_2, R_2], \quad c_{ii}(x_i) < -\delta_1, \quad 0 \le c_{ij}(x_i) < \delta_1 \quad (j \ne i).$$

By (3.4.4) and (3.4.6), we have

$$U_1^{\lambda_2}(x_1) = \frac{c_{22}(x_2)\xi_1 - c_{12}(x_1)\xi_2}{c_{11}(x_1)c_{22}(x_2) - c_{12}(x_1)c_{21}(x_2)} \ge 0.$$

However, this is a contradiction. Thus $\lambda_1 \in A$.

Therefore $A \neq \emptyset$ and $\lambda_* = \inf A$ is well-defined. Using $U^{\lambda_*}(x)$, we can show the symmetry and monotonicity of u(x) along the argument in [20] after slight modification.

Chapter 4

Existence of concentration solutions

Introduction 4.1

In this chapter we consider a singular perturbation problem for (CNLS), namely

- $-\varepsilon^{2}\Delta v_{1} + V_{1}(x)v_{1} = \mu_{1}v_{1}^{3} + \beta v_{1}v_{2}^{2} \quad \text{in } \mathbf{R}^{N}, \\ -\varepsilon^{2}\Delta v_{2} + V_{2}(x)v_{2} = \beta v_{1}^{2}v_{2} + \mu_{2}v_{2}^{3} \quad \text{in } \mathbf{R}^{N},$ (4.1.1)
- (4.1.2)
- $v_1(x), v_2(x) > 0$ in \mathbf{R}^N , (4.1.3)
- $v_1(x), v_2(x) \in H^1(\mathbf{R}^N),$ (4.1.4)

where $N = 2, 3, \mu_1, \mu_2 > 0, \beta \in \mathbf{R}$ are constants, $V_1(x), V_2(x) : \mathbf{R}^N \to \mathbf{R}$ are bounded continuous positive functions, and $\varepsilon > 0$ is a small perturbation parameter.

One of the difficulties in the study of (4.1.1)-(4.1.4) is that it has semitrivial solutions of type $(v_1(x), 0)$ or $(0, v_2(x))$, where $v_1(x)$ or $v_2(x)$ solves

$$-\varepsilon^2 \Delta v_i + V_i(x)v_i = \mu_i v_i^3 \quad \text{in } \mathbf{R}^N.$$

We call solutions $(v_1(x), v_2(x))$ with $v_1(x) \neq 0$ and $v_2(x) \neq 0$ by nontrivial solutions.

(4.1.1)–(4.1.4) is studied in Lin and Wei [68], Pomponio [87], Montefusco, Pellacci and Squassina [79] and G. Wei [105], [106]. In [68], Lin and Wei studied (4.1.1)–(4.1.4) by analyzing least energy nontrivial solutions. They studied both of attractive interaction (i.e., $\beta > 0$) and repulsive interaction (i.e., $\beta < 0$). Especially, when $\beta > 0$, they showed the existence of a least energy nontrivial solution for a small $\varepsilon > 0$ under suitable conditions on the behavior of $V_1(x)$ and $V_2(x)$ as $|x| \to \infty$. Moreover they showed that if

$$\inf_{P \in \mathbf{R}^N} m(P) < \inf_{P_1 \in \mathbf{R}^N} e_1(P_1) + \inf_{P_2 \in \mathbf{R}^N} e_2(P_2)$$

(see (4.1.12) and (4.1.18) for notation), then both components of the least energy nontrivial solution $(v_{\varepsilon 1}(x), v_{\varepsilon 2}(x))$ concentrate to the some point P_0 satisfying $m(P_0) =$ $\inf_{P \in \mathbf{R}^N} m(P)$ as $\varepsilon \to 0$ after taking a subsequence. See Remark 4.1.6 below. We also refer to Lin and Wei [66] for study of a singularly perturbed system of nonlinear Schrödinger equations in a bounded domain.

In [79], Montefusco, Pellacci and Squassina studied the case $\beta > 0$. They consider concentration of solutions around the local minimum (possibly degenerate) point of the potentials; they assume that $z \in \mathbf{R}^N$ and r > 0 satisfy

$$\min_{|x-z| \le r} V_i(x) < \min_{|x-z|=r} V_i(x) \quad (i = 1, 2),$$

and they showed for small $\varepsilon > 0$ (4.1.1)–(4.1.4) has a nonzero solution $(v_{\varepsilon_1}(x), v_{\varepsilon_2}(x))$ such that $v_{\varepsilon_1}(x) + v_{\varepsilon_2}(x)$ has exactly one global maximum point in $\{x; |x - z| < r\}$. However, when $\beta > 0$ is small, one component of $(v_{\varepsilon_1}(x), v_{\varepsilon_2}(x))$ converges to 0 as $\varepsilon \to 0$ (see Theorem 2.1 (ii) in [79]). We also refer to [68, 87, 105] for the study of (4.1.1)–(4.1.4) when $\beta < 0$.

We consider the case where the interaction parameter β is positive and the aim in this chapter is to construct a family of solutions of (4.1.1)–(4.1.4) which concentrates to a nontrivial positive solution.

In the study of (4.1.1)–(4.1.4), the following constant coefficient problem plays an important role:

(4.1.5)
$$-\Delta u_1 + V_1 u_1 = \mu_1 u_1^3 + \beta u_1 u_2^2 \quad \text{in } \mathbf{R}^N,$$

(4.1.6)
$$-\Delta u_2 + V_2 u_2 = \beta u_1^2 u_2 + \mu_2 u_2^3 \quad \text{in } \mathbf{R}^N$$

(4.1.7)
$$u_1(x), \ u_2(x) > 0 \quad \text{in } \mathbf{R}^N,$$

(4.1.8)
$$u_1(x), \ u_2(x) \in H^1(\mathbf{R}^N),$$

where V_1 , $V_2 > 0$ are positive constants. We remark that (4.1.5)-(4.1.8) appears as a limit problem after a suitable rescaling. There are many works on the existence of nontrivial positive solutions of (4.1.5)-(4.1.8). See [4, 10, 12, 13, 29, 31, 47, 49, 50, 65, 75, 94, 107, 108]. Sign and size of β are important in the study of (4.1.5)-(4.1.8) and various situations are studied in the above papers.

Here we consider the case where the interaction β is positive and relatively small and treat the existence of a nontrivial radially symmetric positive solution, which is characterized as a critical point of

$$J_{(V_1,V_2)}(u_1,u_2) = \int_{\mathbf{R}^N} \frac{1}{2} (|\nabla u_1|^2 + V_1 u_1^2 + |\nabla u_2|^2 + V_2 u_2^2) - \frac{1}{4} (\mu_1 u_1^4 + 2\beta u_1^2 u_2^2 + \mu_2 u_2^4) dx$$

: $H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N) \to \mathbf{R}.$

Here $H_r^1(\mathbf{R}^N) := \{ u \in H^1(\mathbf{R}^N) : u(x) = u(|x|) \}$. We assume

$$(4.1.10) 0 < \beta < \sqrt{\mu_1 \mu_2}$$

and

Condition (AC) Let $\hat{\omega}_i(x) \in H^1_r(\mathbf{R}^N)$ be the least energy solution of $-\Delta \hat{\omega}_i + V_i \hat{\omega}_i = \mu_i \hat{\omega}_i^3$, in \mathbf{R}^N . Then the operators $-\Delta + V_1 - \beta \hat{\omega}_2^2$, $-\Delta + V_2 - \beta \hat{\omega}_1^2$ are positive definite on $H^1_r(\mathbf{R}^N)$, that is,

$$\int_{\mathbf{R}^N} |\nabla \varphi|^2 + (V_1 - \beta \hat{\omega}_2^2) \varphi^2 \, dx, \quad \int_{\mathbf{R}^N} |\nabla \varphi|^2 + (V_2 - \beta \hat{\omega}_1^2) \varphi^2 \, dx > 0$$

$$x) \in H^1(\mathbf{R}^N) \setminus \{0\}$$

for all $\varphi(x) \in H^1_r(\mathbf{R}^N) \setminus \{0\}.$

We will show the following existence result in Section 4.2.

Proposition 4.1.1. Assume (4.1.10) and (AC). Then (4.1.5)–(4.1.8) has a nontrivial radially symmetric positive solution $(u_{01}, u_{02}) \in H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$, which can be characterized as

 $(4.1.11) J_{(V_1,V_2)}(u_{01}, u_{02}) = b(V_1, V_2),$

where

$$b(V_1, V_2) := \inf_{\substack{(u_1, u_2) \in \mathcal{M}_r(V_1, V_2) \\ (u_1, u_2) \in \mathcal{M}_r(V_1, V_2)}} J_{(V_1, V_2)}(u_1, u_2),$$

$$\mathcal{M}_r(V_1, V_2) := \{(u_1, u_2) \in H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N) : u_1 \neq 0, u_2 \neq 0, \\ J'_{(V_1, V_2)}(u_1, u_2)(u_1, 0) = 0, J'_{(V_1, V_2)}(u_1, u_2)(0, u_2) = 0\}.$$

Moreover, suppose that there exists a set $A \subset (0, \infty) \times (0, \infty)$ such that (AC) holds for all $(V_1, V_2) \in A$. Then

- (i) $(V_1, V_2) \mapsto b(V_1, V_2); A \to \mathbf{R}$ is continuous.
- (ii) $V_1 \mapsto b(V_1, V_2)$ (resp. $V_2 \mapsto b(V_1, V_2)$) is strictly increasing for a fixed V_2 (resp. V_1).

Remark 4.1.2. (i) In [20], Busca and Sirakov showed that when $\beta > 0$, any nontrivial positive solution of (4.1.5)–(4.1.8) is radially symmetric with respect to some point $P_0 \in \mathbf{R}^N$.

(ii) Condition (AC) is introduced in Ambrosetti and Colorado [4] and they showed the existence of nontrivial positive solutions. We give another proof of their existence result as well as the characterization (4.1.11) and some additional compactness properties. We also refer to Lin and Wei [65], Sirakov [94] for the existence of least energy nontrivial solutions.

For (4.1.1)-(4.1.4), we assume the following

Assumption (4–A1) There exists a set $A = [a_{10}, a_{11}] \times [a_{20}, a_{21}] \subset (0, \infty) \times (0, \infty)$ with the following properties:

- (i) For any $(V_1, V_2) \in A$, the constant coefficient problem (4.1.5)–(4.1.6) satisfies the condition (AC).
- (ii) $(V_1(P), V_2(P)) \in A$ for all $P \in \mathbf{R}^N$.

We set

(4.1.12)
$$m(P) := b(V_1(P), V_2(P)) : \mathbf{R}^N \to \mathbf{R}.$$

As the second assumption, we assume

Assumption (4–A2) There exists a bounded open set $\Lambda \subset \mathbf{R}^N$ such that

(4.1.13)
$$\inf_{P \in \Lambda} m(P) < \inf_{P \in \partial \Lambda} m(P).$$

We set

$$m_0 := \inf_{P \in \Lambda} m(P),$$

$$K := \{P \in \Lambda : m(P) = m_0\}$$

Now we can state our main result.

Theorem 4.1.3. Suppose that (4.1.10) and Assumptions (4–A1), (4–A2) hold. Then there exists an $\varepsilon_0 > 0$ such that (4.1.1)–(4.1.4) has a family of nontrivial positive solutions $(v_{1\varepsilon}(x), v_{2\varepsilon}(x))_{0 < \varepsilon < \varepsilon_0}$ satisfying the following properties: after taking a subsequence $\varepsilon_j \to 0$ there exists a sequence $(P_{\varepsilon_j}) \subset \Lambda$ such that

$$(4.1.14) P_{\varepsilon_i} \to P_0 \in K,$$

(4.1.15)
$$(v_{1\varepsilon_j}(\varepsilon_j x + P_{\varepsilon_j}), v_{2\varepsilon_j}(\varepsilon_j x + P_{\varepsilon_j})) \to (w_1(x), w_2(x))$$

strongly in $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$

Here $(w_1(x), w_2(x))$ is a nontrivial positive solution of the limit problem:

$$\begin{cases} -\Delta w_1 + V_1(P_0)w_1 = \mu_1 w_1^3 + \beta w_1 w_2^2 & \text{in } \mathbf{R}^N, \\ -\Delta w_2 + V_2(P_0)w_2 = \beta w_1^2 w_2 + \mu_2 w_2^3 & \text{in } \mathbf{R}^N \end{cases}$$

and it satisfies $J_{(V_1(P_0),V_2(P_0))}(w_1,w_2) = m_0$.

Remark 4.1.4. If we assume $V_1(x)$, $V_2(x) \in C^1(\mathbf{R}^N)$ in addition to the assumptions of Theorem 4.1.3, we have

(4.1.16)
$$K \subset \{P \in \Lambda; \lambda_1 \nabla V_1(P) + \lambda_2 \nabla V_2(P) = 0 \text{ for some } \lambda_1, \lambda_2 > 0\}.$$

See Lemma 4.2.9 in section 4.2 for a proof of (4.1.16).

Remark 4.1.5. (i) We remark that $b(V_1, V_2)$ and m(P) also depend on β . We write dependence on β explicitly and use notation $b_{\beta}(V_1, V_2)$, $m_{\beta}(P)$ in this remark. We also remark that if (4.1.10) and (AC), (4–A1) hold for $\beta = \beta_0 > 0$, then they also hold for all $\beta \in (0, \beta_0]$. Concerning a behavior of $b_{\beta}(V_1, V_2)$ as $\beta \to 0$, we have

(4.1.17)
$$b_{\beta}(V_1, V_2) \to e_1(V_1) + e_2(V_2) \text{ as } \beta \to 0,$$

where

$$e_i(V_i) = \frac{V_i^{(4-N)/2}}{\mu_i} e_0$$

is the least energy level for $-\Delta u + V_i u = \mu_i u^3$. Here $e_0 > 0$ is the least energy level for $-\Delta \omega + \omega = \omega^3$, that is, $e_0 = \int_{\mathbf{R}^N} \frac{1}{2} (|\nabla \omega_0|^2 + \omega_0^2) - \frac{1}{4} \omega_0^4 dx$ where $\omega_0(x)$ is the unique radial positive solution of $-\Delta \omega + \omega = \omega^3$. See Lemma 4.2.10 in section 4.2 for a proof of (4.1.17).

In particular, from Proposition 4.1.1, (4–A1) and Remark 4.2.11, we have

$$m_{\beta}(P) \to e_1(P) + e_2(P)$$
 in $C_{\text{loc}}(\mathbf{R}^N)$ as $\beta \to 0$,
where

(4.1.18)
$$e_i(P) = e_i(V_i(P)) = \frac{V_i(P)^{(4-N)/2}}{\mu_i} e_0.$$

Thus if a bounded open set $\Lambda \subset \mathbf{R}^N$ satisfies

(4.1.19)
$$\inf_{P \in \Lambda} \left(e_1(P) + e_2(P) \right) < \inf_{P \in \partial \Lambda} \left(e_1(P) + e_2(P) \right),$$

then (4–A2) also holds for $\beta > 0$ small.

(ii) We can easily construct an example of $V_1(x)$, $V_2(x)$ and Λ , where $V_i(x)$ (i = 1, 2) has no critical points in Λ but (4.1.19) holds. Thus Theorem 4.1.3 can be applied for $\beta > 0$ small to find a concentrating family of solutions in Λ .

(iii) When $\beta = 0$, there is no interaction between two equations. Then concentration points must be critical points of $V_i(x)$'s. See Wang [103] and Wang and Zeng [104]. Thus, even if a bounded open set $\Lambda \subset \mathbf{R}^N$ satisfies (4.1.19), there does not exist a family of concentrating solutions in Λ in general and thus the positivity of β is necessary to find a concentrating solution in Λ .

Remark 4.1.6. Under suitable conditions on the behavior of $V_1(x)$ and $V_2(x)$ as $|x| \to \infty$, Lin and Wei [68] showed the existence of a least energy nontrivial solution $U_{\varepsilon}(x) = (u_{1\varepsilon}(x), u_{2\varepsilon}(x))$ which can be characterized as

$$I_{\varepsilon}(U_{\varepsilon}) = \inf_{(u_1, u_2) \in \mathcal{M}_{\varepsilon}} I_{\varepsilon}(u_1, u_2).$$

Here $I_{\varepsilon}(U)$ is a functional and $\mathcal{M}_{\varepsilon}$ is a Nehari type manifold corresponding to (4.1.1)–(4.1.4). See section 4.3 (especially (4.3.5) and (4.3.8)). They showed

$$I_{\varepsilon}(U_{\varepsilon}) \to \min\left\{\inf_{P \in \mathbf{R}^{N}} m(P), \inf_{P_{1} \in \mathbf{R}^{N}} e_{1}(P_{1}) + \inf_{P_{2} \in \mathbf{R}^{N}} e_{2}(P_{2})\right\}$$

and if

$$\inf_{P \in \mathbf{R}^N} m(P) > \inf_{P_1 \in \mathbf{R}^N} e_1(P_1) + \inf_{P_2 \in \mathbf{R}^N} e_2(P_2),$$

then, after taking a subsequence $u_{1\varepsilon}(x)$ and $u_{2\varepsilon}(x)$ concentrate to different points Q_1 and Q_2 in general. Here Q_i satisfies $V_i(Q_i) = \inf_{P \in \mathbf{R}^N} V_i(P)$. Thus, even if $P_0 \in \mathbf{R}^N$ is a global minimizer of m(P), i.e., $m(P_0) = \inf_{P \in \mathbf{R}^N} m(P)$, the minimizer $U_{\varepsilon}(x) = (u_{1\varepsilon}(x), u_{2\varepsilon}(x))$ of I_{ε} in $\mathcal{M}_{\varepsilon}$ does not have the desired behavior (4.1.14)–(4.1.15) in general. We remark that in a singular perturbation problem for a nonlinear Schrödinger equation: $-\varepsilon^2 \Delta u + V(x)u = u^3$ in \mathbf{R}^N , the situation is simpler and we can find a family of solutions concentrating to a global minimum of V(x) via global minimization of the functional on the Nehari manifold $\mathcal{N} := \{u \in H^1(\mathbf{R}^N) \setminus \{0\} : \int_{\mathbf{R}^N} |\nabla u|^2 + V(\varepsilon x)u^2 dx = \int_{\mathbf{R}^N} |u|^4 dx\}.$

The following sections are devoted to proofs of our Proposition 4.1.1 and Theorem 4.1.3. As stated in Remark 4.1.6, one of the difficulties in proving Theorem 4.1.3 is that the global minimization method on the Nehari type manifold $\mathcal{M}_{\varepsilon}$ does not work even for a global minimizer P_0 of m(P). Another difficulty is that uniqueness and nondegeneracy of

solutions of the limit equation (4.1.5)-(4.1.8) are not known and thus classical Liapunov-Schmidt reduction approach seems to be difficult to apply in our setting. To overcome these difficulties, we use an idea from Byeon and Jeanjean [22, 23] (c.f. [24]). In [22, 23], Byeon and Jeanjean developed a new variational approach to find a localized positive solution for nonlinear Schrödinger equations with a wide class of nonlinearities. We also refer to [2, 32, 33, 34, 38, 42, 56, 63, 93, 103, 104] and references therein for preceding results on nonlinear Schrödinger equations. Here, adapting the idea in [22, 23] on the Nehari type manifold $\mathcal{M}_{\varepsilon}$ and developing new estimates, we find a subset of $\mathcal{M}_{\varepsilon}$ in which the corresponding functional has a local minimizer with the desired property (4.1.14)– (4.1.15).

4.2 Constant coefficient problems

In this section we study the existence of a nontrivial radially symmetric positive solution of (4.1.5)-(4.1.8).

The main purpose of this section is to give a proof of Proposition 4.1.1 as well as additional compactness properties.

4.2.1 Preliminaries

In the following sections, we denote for $D \subset \mathbf{R}^N$

$$\begin{aligned} \|u\|_{L^{p}(D)} &:= \left(\int_{D} |u|^{p} \, dx\right)^{1/p} & \text{for } u \in L^{p}(D), \\ \|u\|_{L^{\infty}(D)} &:= \text{ess sup}_{x \in D} |u(x)| & \text{for } u \in L^{\infty}(D), \\ \|u\|_{H^{1}(D)} &:= \sqrt{\|\nabla u\|_{L^{2}(D)}^{2} + \|u\|_{L^{2}(D)}^{2}} & \text{for } u \in H^{1}(D). \end{aligned}$$

When $D = \mathbf{R}^N$, we also use abbreviation: $\|\cdot\|_p = \|\cdot\|_{L^p(\mathbf{R}^N)}$ $(p \in [1,\infty]), \|\cdot\|_{H^1} = \|\cdot\|_{H^1(\mathbf{R}^N)}$. We also write

$$(u,v)_2 := \int_{\mathbf{R}^N} uv \, dx \quad \text{for } u, v \in L^2(\mathbf{R}^N).$$

For $u_1, u_2 \in H^1(\mathbf{R}^N)$ we write $U = (u_1, u_2)$ and

$$||U||_{H^1}^2 := ||u_1||_{H^1}^2 + ||u_2||_{H^1}^2.$$

For $(V_1, V_2) \in (0, \infty)^2$ and $U \in H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$, we define $J_{(V_1, V_2)}(U)$ as in (4.1.9), that is,

$$J_{(V_1,V_2)}(U) := \frac{1}{2} \|U\|_{H^1,(V_1,V_2)}^2 - \int_{\mathbf{R}^N} W(U) \, dx,$$

where

$$\begin{aligned} |U||_{H^{1},(V_{1},V_{2})}^{2} &:= \|\nabla u_{1}\|_{2}^{2} + V_{1}\|u_{1}\|_{2}^{2} + \|\nabla u_{2}\|_{2}^{2} + V_{2}\|u_{2}\|_{2}^{2} \\ W(\xi_{1},\xi_{2}) &:= \frac{1}{4}(\mu_{1}\xi_{1}^{4} + 2\beta\xi_{1}^{2}\xi_{2}^{2} + \mu_{2}\xi_{2}^{4}). \end{aligned}$$

We can easily see that $J_{(V_1,V_2)} \in C^2(H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N), \mathbf{R})$ and

$$\begin{aligned} J'_{(V_1,V_2)}(u_1,u_2)(h_1,h_2) \\ &= (\nabla u_1,\nabla h_1)_2 + V_1(u_1,h_1)_2 + (\nabla u_2,\nabla h_2)_2 + V_2(u_2,h_2)_2 \\ &- \int_{\mathbf{R}^N} \nabla W(u_1,u_2)(h_1,h_2) \, dx \\ &= \int_{\mathbf{R}^N} \nabla u_1 \nabla h_1 + V_1 u_1 h_1 + \nabla u_2 \nabla h_2 + V_2 u_2 h_2 \\ &- \mu_1 u_1^3 h_1 - \beta u_1 u_2^2 h_1 - \mu_2 u_2^3 h_2 - \beta u_1^2 u_2 h_2 \, dx \end{aligned}$$

for all (u_1, u_2) , $(h_1, h_2) \in H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$. Thus critical points of $J_{(V_1, V_2)}$ are radially symmetric solutions of (4.1.5)–(4.1.8).

We have the following lemma.

Lemma 4.2.1. $J_{(V_1,V_2)} \in C^2(H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N), \mathbf{R})$ satisfies the Palais-Smale compactness condition.

Proof. Lemma 4.2.1 follows from the compactness of the embedding $H^1_r(\mathbf{R}^N) \subset L^4(\mathbf{R}^N)$ and the fact that $\nabla W(U)U = 4W(U)$ in a rather standard way. \Box

One of the difficulties in the studying (4.1.5)-(4.1.8) is to distinguish nontrivial solutions from semitrivial solutions. We remark that (4.1.5)-(4.1.8) has 2 semitrivial solutions:

$$\hat{\Omega}_1(V_1; x) := (\hat{\omega}_1(V_1; x), 0) = \left(\sqrt{\frac{V_1}{\mu_1}} \omega_0(\sqrt{V_1}x), 0\right),$$
$$\hat{\Omega}_2(V_2; x) := (0, \hat{\omega}_2(V_2; x)) = \left(0, \sqrt{\frac{V_2}{\mu_2}} \omega_0(\sqrt{V_2}x)\right),$$

where $\omega_0(x)$ is the unique radially symmetric positive symmetric solution of $-\Delta u + u = u^3$ in $H_r^1(\mathbf{R}^N)$. We note that for i = 1, 2

$$J_{(V_1,V_2)}(\hat{\Omega}_i) = e_i(V_i) = \frac{V_i^{(4-N)/2}}{\mu_i} e_0,$$

where $e_0 = \frac{1}{2} \|\omega_0\|_{H^1}^2 - \frac{1}{4} \|\omega_0\|_4^4$.

4.2.2 Ambrosetti and Colorado's condition

In [4], Ambrosetti and Colorado introduced the condition (AC) and they showed the existence of nontrivial positive solutions through a mountain pass argument on the Nehari manifold:

$$\mathcal{N}_r(V_1, V_2) := \{ U \in H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N) : U \neq (0, 0), \ J'_{(V_1, V_2)}(U)U = 0 \}.$$

We give another proof to their existence result. Since estimates, which is uniform in (V_1, V_2) , are important for the study of the singular perturbation problem, we assume (4.1.10) and the following condition:

(C1) There exists a set $A = [a_{10}, a_{11}] \times [a_{20}, a_{21}] \subset (0, \infty)^2$ such that (AC) holds for $(V_1, V_2) \in A$.

The condition (AC) ensures the positivity of the following bilinear forms:

$$(4.2.1) \begin{aligned} J_{(V_1,V_2)}'(\hat{\Omega}_2)(h_1,0)(h_2,0) &= (\nabla h_1, \nabla h_2)_2 + V_1(h_1,h_2)_2 - \beta \int_{\mathbf{R}^N} \hat{\omega}_2^2 h_1 h_2 \, dx \\ &: (H_r^1(\mathbf{R}^N) \times \{0\})^2 \to \mathbf{R}, \\ J_{(V_1,V_2)}'(\hat{\Omega}_1)(0,h_1)(0,h_2) &= (\nabla h_1, \nabla h_2)_2 + V_2(h_1,h_2)_2 - \beta \int_{\mathbf{R}^N} \hat{\omega}_1^2 h_1 h_2 \, dx \\ &: (\{0\} \times H_r^1(\mathbf{R}^N))^2 \to \mathbf{R}. \end{aligned}$$

It is easily seen that $\mathcal{N}_r(V_1, V_2)$ is a Hilbert manifold with codimension 1 and the critical points of the constraint functional $J_{(V_1, V_2)}|_{\mathcal{N}_r(V_1, V_2)} : \mathcal{N}_r(V_1, V_2) \to \mathbf{R}$ are nonzero critical points of $J_{(V_1, V_2)}$.

Under the condition (C1) we have

Lemma 4.2.2. Assume (C1). Then there exist $\rho_0 > 0$ and $0 < r_0 < \min\{\|\hat{\Omega}_1(V_1; x)\|_{H^1,(V_1,V_2)}, \|\hat{\Omega}_2(V_2; x)\|_{H^1,(V_1,V_2)} : (V_1, V_2) \in A\}$ such that for each $(V_1, V_2) \in A$, the following hold:

(i)
$$J_{(V_1,V_2)}(U) \ge e_1(V_1) + \rho_0 \text{ for all } U \in \mathcal{N}_r(V_1,V_2) \cap \{ \|U - \hat{\Omega}_1\|_{H^1,(V_1,V_2)} = r_0 \}.$$

(ii)
$$J_{(V_1,V_2)}(U) \ge e_2(V_2) + \rho_0 \text{ for all } U \in \mathcal{N}_r(V_1,V_2) \cap \{ \|U - \hat{\Omega}_2\|_{H^1,(V_1,V_2)} = r_0 \}.$$

Proof. We prove just (i). (ii) can be shown in a similar way. We set

$$\mathcal{N}_{1,r}(V_1) := \{ u_1 \in H_r^1(\mathbf{R}^N) \setminus \{0\} : J'_{(V_1,V_2)}(u_1,0)(u_1,0) = 0 \}$$

= $\{ u_1 \in H_r^1(\mathbf{R}^N) \setminus \{0\} : \|\nabla u_1\|_2^2 + V_1 \|u_1\|_2^2 = \mu_1 \|u_1\|_4^4 \},$

which is the Nehari manifold for the scalar equation: $-\Delta u + V_1 u = \mu_1 u^3$. We remark that $\mathcal{N}_{1,r}(V_1) \times \{0\} \subset \mathcal{N}_r(V_1, V_2)$ and

$$J_{(V_1,V_2)}(\hat{\Omega}_1) = e_1(V_1) = \inf_{u_1 \in \mathcal{N}_{1,r}(V_1)} J_{(V_1,V_2)}(u_1,0).$$

Since $\hat{\omega}(V_1; x)$ is a nondegenerate critical point of $J_{(V_1, V_2)}|_{\mathcal{N}_{1,r}(V_1) \times \{0\}}$, the conclusion (i) follows from the positivity of the bilinear form (4.2.1).

Now we introduce our minimax method to find a nontrivial solution. We set

$$\Gamma := \{ \gamma(s,t) \in C([0,\infty)^2, H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N)) :$$

For some $u_{01}(x), u_{02}(x) \ge 0 \ (\not\equiv 0)$ and $R > 0, \ \gamma(s,t)$ satisfies
 $\gamma(s,0) = (\sqrt{s} u_{01}, 0) \quad \text{for } s \ge 0,$
 $\gamma(0,t) = (0, \sqrt{t} u_{02}) \quad \text{for } t \ge 0,$
 $\gamma(s,t) = (\sqrt{s} u_{01}, \sqrt{t} u_{02}) \quad \text{for } s^2 + t^2 \ge R^2 \}$

....

4

and

(4.2.2)
$$b(V_1, V_2) := \inf_{\gamma \in \Gamma} \sup_{(s,t) \in [0,\infty)^2} J_{(V_1, V_2)}(\gamma(s, t)).$$

Our main result in this subsection is

Proposition 4.2.3. Suppose (4.1.10) and (C1) hold. Then for any $(V_1, V_2) \in A$,

- (i) $\max\{e_1(V_1) + \rho_0, e_2(V_2) + \rho_0\} \le b(V_1, V_2) \le e_1(V_1) + e_2(V_2).$
- (ii) The value $b(V_1, V_2)$ is attained by a nontrivial radially symmetric positive solution of (4.1.5)-(4.1.8).

Proof. Setting $\gamma_0(s,t) := (\sqrt{s}\hat{\omega}_1(V_1;x), \sqrt{t}\hat{\omega}_2(V_2;x))$ and

$$J_{(V_1,V_2),0}(u_1,u_2) := \frac{1}{2} (\|\nabla u_1\|_2^2 + V_1\|u_1\|_2^2 + \|\nabla u_2\|_2^2 + V_2\|u_2\|_2^2) - \frac{1}{4} (\mu_1\|u_1\|_4^4 + \mu_2\|u_2\|_4^4),$$

we have

$$b(V_1, V_2) \le \sup_{(s,t)\in[0,\infty)^2} J_{(V_1, V_2)}(\gamma_0(s, t))$$

$$\le \sup_{(s,t)\in[0,\infty)^2} J_{(V_1, V_2), 0}(\gamma_0(s, t)) = e_1(V_1) + e_2(V_2).$$

Thus we obtain $b(V_1, V_2) \leq e_1(V_1) + e_2(V_2)$. To prove the remaining parts, we set

$$\begin{split} \widetilde{\Gamma}(V_1, V_2) &:= \{ \gamma(s, t) \in C([0, \infty)^2, H_r^1(\mathbf{R}^N) \times H_r^1(\mathbf{R}^N)); \\ \text{For some } R > 0, \, \gamma(s, t) \text{ satisfies} \\ \gamma(s, 0) &= (\sqrt{s} \, \hat{\omega}_1(V_1; x), 0) \quad \text{for } s \ge 0, \\ \gamma(0, t) &= (0, \sqrt{t} \, \hat{\omega}_2(V_2; x)) \quad \text{for } t \ge 0, \\ \gamma(s, t) &= (\sqrt{s} \, \hat{\omega}_1(V_1; x), \sqrt{t} \, \hat{\omega}_2(V_2; x)) \quad \text{for } s^2 + t^2 \ge R^2 \; \} \end{split}$$

It is clear that $\widetilde{\Gamma}(V_1, V_2) \subset \Gamma$. First we show

Step 1: $b(V_1, V_2) = \inf_{\gamma \in \widetilde{\Gamma}(V_1, V_2)} \sup_{(s,t) \in [0,\infty)^2} J_{(V_1, V_2)}(\gamma(s, t)).$

It suffices to show that for any $\gamma \in \Gamma$ there exists $\widetilde{\gamma} \in \widetilde{\Gamma}(V_1, V_2)$ such that

$$\sup_{(s,t)\in[0,\infty)^2} J_{(V_1,V_2)}(\tilde{\gamma}(s,t)) = \sup_{(s,t)\in[0,\infty)^2} J_{(V_1,V_2)}(\gamma(s,t)).$$

Let $\gamma(s,t) \in \Gamma$ be a given path and suppose $\gamma(s,t) = (\sqrt{s} u_{01}, \sqrt{t} u_{02})$ for $(s,t) \in (\mathbb{R} \times \{0\}) \cup (\{0\} \times \mathbb{R}) \cup \{(s,t); s^2 + t^2 \ge R^2\}$. Moreover, we may assume $u_{01} \in \mathcal{N}_{1,r}(V_1)$ and $u_{02} \in \mathcal{N}_{2,r}(V_2)$ without loss of generality. We remark that $\{\sqrt{s} u_{01}; s \in [0,\infty)\}$ and $\mathcal{N}_{1,r}(V_1)$ intersects at exactly one point u_{01} . Take a curve $\zeta_1(\tau) \in C([0,1], \mathcal{N}_{1,r}(V_1))$ such that

$$\begin{aligned} \zeta_1(0) &= \hat{\omega}_1, \ \zeta_1(1) = u_{01}, \ \zeta_1(\tau)(x) \ge 0 \quad \text{for all } \tau \in [0,1] \\ \tau &\mapsto J_{(V_1,V_2)}(\zeta_1(\tau),0) \text{ is nondecreasing.} \end{aligned}$$

Such a curve exists since $J_{(V_1,V_2)}(\hat{\omega}_1,0) = \inf_{u \in \mathcal{N}_{1,r}(V_1)} J_{(V_1,V_2)}(u,0)$ and there are no other nonnegative critical points on $\mathcal{N}_{1,r}(V_1)$ other than $\hat{\omega}_1$.

Similarly, we take a curve $\zeta_2(\tau) \in C([0,1], \mathcal{N}_{2,r}(V_2))$ joining $\hat{\omega}_2(V_2; x)$ and u_{02} . We set

$$\tilde{\gamma}(s,t) := \begin{cases} \gamma(s-1,t-1) & \text{if } s \ge 1 \text{ and } t \ge 1, \\ (\sqrt{s-t}\,\zeta_1(t),0) & \text{if } t \in [0,1) \text{ and } t \le s, \\ (0,\sqrt{t-s}\,\zeta_2(s)) & \text{if } s \in [0,1) \text{ and } s < t. \end{cases}$$

Then we can see that

$$\sup_{(s,t)\in[0,\infty)^2} J_{(V_1,V_2)}(\tilde{\gamma}(s,t)) = \sup_{(s,t)\in[0,\infty)^2} J_{(V_1,V_2)}(\gamma(s,t)).$$

For sufficiently large $\tilde{R} > R$, modifying $\tilde{\gamma}(s,t)$ in a suitable way, we can get $\tilde{\tilde{\gamma}}(s,t) \in \tilde{\Gamma}(V_1, V_2)$ with the desired properties.

Next we show

Step 2: For any $\gamma \in \widetilde{\Gamma}(V_1, V_2)$,

$$\gamma([0,\infty)^2) \cap \mathcal{N}_r(V_1,V_2) \cap \{ \|U - \hat{\Omega}_1(V_1)\|_{H^1,(V_1,V_2)} = r_0 \} \neq \emptyset.$$

where $r_0 > 0$ is given in Lemma 4.2.2.

For a given $\gamma(s,t) \in \widetilde{\Gamma}(V_1,V_2)$ we set

$$F(s,t) := (J'_{(V_1,V_2)}(\gamma(s,t))\gamma(s,t), \|\gamma(s,t) - \hat{\Omega}_1(V_1)\|^2_{H^1,(V_1,V_2)} - r_0^2).$$

We compute deg(F, $[0, R]^2$, (0, 0)) for sufficiently large R > 0. Set $G(s, t) = \nabla g(s, t)$ where $g(s, t) = -(s - 1)^2 + (t - \varepsilon)^2$ for sufficiently small $\varepsilon > 0$ and consider the following homotopy between F and G: For $\theta \in [0, 1]$,

$$F_{\theta}(s,t) = (1-\theta)F(s,t) + \theta G(s,t)$$

For sufficiently large R > 0 we can see that $F_{\theta}(s,t) \neq (0,0)$ for all $(s,t) \in \partial[0,R]^2$ and $\theta \in [0,1]$. Therefore we obtain $\deg(F,[0,R]^2,(0,0)) = \deg(G,[0,R]^2,(0,0)) = -1$. Thus there exists $(s_0,t_0) \in (0,R)^2$ such that $F(s_0,t_0) = (0,0)$. That is, $U_0 = \gamma(s_0,t_0)$ satisfies $U_0 \in \mathcal{N}_r(V_1,V_2)$ and $\|U_0 - \hat{\Omega}_1(V_1)\|_{H^1,(V_1,V_2)} = r_0$.

Step 3: $b(V_1, V_2) \ge \max\{e_1(V_1) + \rho_0, e_2(V_2) + \rho_0\}.$

By Step 2 and Lemma 4.2.2 (i), we have

$$\sup_{(s,t)\in[0,\infty)^2} J_{(V_1,V_2)}(\gamma(s,t)) \geq \inf_{\substack{U\in\mathcal{N}_r(V_1,V_2)\cap\{\|U-\hat{\Omega}_1\|_{H^1}=r_0\}}} J_{(V_1,V_2)}(U)$$

$$\geq e_1(V_1) + \rho_0$$

for all $\gamma \in \widetilde{\Gamma}(V_1, V_2)$. Thus by Step 1, we have $b(V_1, V_2) \ge e_1(V_1) + \rho_0$. In a similar way, we can show $b(V_1, V_2) \ge e_2(V_2) + \rho_0$. Therefore we get the conclusion of Step 3.

Step 4: The value $b(V_1, V_2)$ is attained by a nontrivial positive solution of (4.1.5)–(4.1.8).

Since $\sup_{s\geq 0} J_{(V_1,V_2)}(\gamma(s,0)) \leq e_1(V_1)$, $\sup_{t\geq 0} J_{(V_1,V_2)}(\gamma(0,t)) \leq e_2(V_2)$ for $\gamma \in \widetilde{\Gamma}(V_1,V_2)$, we can see from Step 3 that $\widetilde{\Gamma}(V_1,V_2)$ is stable under deformation. Thus, by Lemma 4.2.1, $b(V_1,V_2) = \inf_{\gamma \in \widetilde{\Gamma}(V_1,V_2)} \sup_{(s,t) \in [0,\infty)^2} J_{(V_1,V_2)}(\gamma(s,t))$ is achieved.

Next we show that $b(V_1, V_2)$ is attained by a nontrivial positive solution. For $\gamma(s, t) = (\gamma_1(s, t), \gamma_2(s, t))$, we set $|\gamma|(s, t) := (|\gamma_1(s, t)|, |\gamma_2(s, t)|)$. Since $J_{(V_1, V_2)}(\gamma(s, t)) = (|\gamma_1(s, t)|, |\gamma_2(s, t)|)$.

 $J_{(V_1,V_2)}(|\gamma|(s,t))$, we can conclude that there exists a nonnegative critical point corresponding to $b(V_1, V_2)$. By Step 3, we can see that the corresponding critical point is a nontrivial positive solution.

4.2.3 Minimizing property

The aim of this subsection is to give characterizations to $b(V_1, V_2)$ using the Nehari type manifolds. In what follows, we define $J_{(V_1, V_2)}(U)$ by (4.1.9) also for $U \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$.

We consider 2 type of manifolds:

$$\mathcal{M}_{r}(V_{1}, V_{2}) := \{ U = (u_{1}, u_{2}) \in H_{r}^{1}(\mathbf{R}^{N}) \times H_{r}^{1}(\mathbf{R}^{N}) : u_{1} \neq 0, u_{2} \neq 0, \\ J_{(V_{1}, V_{2})}'(u_{1}, u_{2})(u_{1}, 0) = 0, J_{(V_{1}, V_{2})}'(u_{1}, u_{2})(0, u_{2}) = 0 \}, \\ \mathcal{M}(V_{1}, V_{2}) := \{ U = (u_{1}, u_{2}) \in H^{1}(\mathbf{R}^{N}) \times H^{1}(\mathbf{R}^{N}) : u_{1} \neq 0, u_{2} \neq 0, \\ J_{(V_{1}, V_{2})}'(u_{1}, u_{2})(u_{1}, 0) = 0, J_{(V_{1}, V_{2})}'(u_{1}, u_{2})(0, u_{2}) = 0 \}.$$

Such a type of manifolds were introduced in Lin and Wei [65] and Sirakov [94] and they studied the existence of a minimizer of $J_{(V_1,V_2)}$ on $\mathcal{M}_r(V_1,V_2)$ and $\mathcal{M}(V_1,V_2)$.

We have

Lemma 4.2.4 (cf. Lin and Wei [65]). Assume (4.1.10). Then we have

- (i) The set $\mathcal{M}_r(V_1, V_2)$ (resp. $\mathcal{M}(V_1, V_2)$) is a Hilbert submanifold of $H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$ (resp. $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$) with codimension 2.
- (ii) For $(u_1, u_2) \in H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$ (resp. $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$) with $u_1 \neq 0, u_2 \neq 0$ and s, t > 0,

$$(\sqrt{s} u_1, \sqrt{t} u_2) \in \mathcal{M}_r(V_1, V_2) \ (resp. \ \mathcal{M}(V_1, V_2))$$

if and only if

(4.2.3)
$$\begin{cases} \mu_1 \|u_1\|_4^4 s + \beta \|u_1 u_2\|_2^2 t = \|\nabla u_1\|_2^2 + V_1 \|u_1\|_2^2 \\ \beta \|u_1 u_2\|_2^2 s + \mu_2 \|u_2\|_4^4 t = \|\nabla u_2\|_2^2 + V_2 \|u_2\|_2^2 \end{cases}$$

Proof. We show for $\mathcal{M}_r(V_1, V_2)$. We can show for $\mathcal{M}(V_1, V_2)$ in a similar way. (i) Set

$$F_1(u_1, u_2) := J'_{(V_1, V_2)}(u_1, u_2)(u_1, 0), \quad F_2(u_1, u_2) := J'_{(V_1, V_2)}(u_1, u_2)(0, u_2).$$

Then $\mathcal{M}_r(V_1, V_2) = \{(u_1, u_2) \in H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N) : u_1 \neq 0, u_2 \neq 0, F_1(u_1, u_2) = F_2(u_1, u_2) = 0\}$. For $U = (u_1, u_2) \in \mathcal{M}_r(V_1, V_2)$ we have

$$\begin{bmatrix} F_1'(u_1, u_2)(u_1, 0) & F_1'(u_1, u_2)(0, u_2) \\ F_2'(u_1, u_2)(u_1, 0) & F_2'(u_1, u_2)(0, u_2) \end{bmatrix} = -2 \begin{bmatrix} \mu_1 \| u_1 \|_4^4 & \beta \| u_1 u_2 \|_2^2 \\ \beta \| u_1 u_2 \|_2^2 & \mu_2 \| u_2 \|_4^4 \end{bmatrix}$$

and

(4.2.4)
$$\det \begin{bmatrix} \mu_1 \| u_1 \|_4^4 & \beta \| u_1 u_2 \|_2^2 \\ \beta \| u_1 u_2 \|_2^2 & \mu_2 \| u_2 \|_4^4 \end{bmatrix} = \mu_1 \mu_2 \| u_1 \|_4^4 \| u_2 \|_4^4 - \beta^2 \| u_1 u_2 \|_2^4 \\ = (\mu_1 \mu_2 - \beta^2) \| u_1 \|_4^4 \| u_2 \|_4^4 > 0.$$

Thus $F'_1(u_1, u_2)$ and $F'_2(u_1, u_2)$ are linearly independent for all $(u_1, u_2) \in \mathcal{M}_r(V_1, V_2)$, and $\mathcal{M}_r(V_1, V_2)$ is a submanifold of $H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$ with codimension 2. (ii) Since

$$F_1(\sqrt{s}\,u_1,\sqrt{t}\,u_2) = s(\|\nabla u_1\|_2^2 + V_1\|u_1\|_2^2 - \mu_1\|u_1\|_4^4 s - \beta\|u_1u_2\|_2^2 t),$$

$$F_2(\sqrt{s}\,u_1,\sqrt{t}\,u_2) = t(\|\nabla u_2\|_2^2 + V_2\|u_2\|_2^2 - \beta\|u_1u_2\|_2^2 s - \mu_2\|u_2\|_4^4 t),$$

we see that (ii) holds.

The aim of this subsection is to show the following proposition.

Proposition 4.2.5. Assume (4.1.10) and (C1). Then

(4.2.5)
$$b(V_1, V_2) = \inf_{U \in \mathcal{M}_r(V_1, V_2)} J_{(V_1, V_2)}(U) = \inf_{U \in \mathcal{M}(V_1, V_2)} J_{(V_1, V_2)}(U).$$

Since all nontrivial positive solutions of (4.1.5)-(4.1.8) lie in $\mathcal{M}_r(V_1, V_2)$ after a suitable shift by the result of Busca and Sirakov [20], we can see that the critical point corresponding to $b(V_1, V_2)$ has the least energy among nontrivial solutions. Thus we call the solution corresponding to $b(V_1, V_2)$ the least energy nontrivial solution. We also call $b(V_1, V_2)$ the least energy level for nontrivial solution.

Proof. First we remark that $\mathcal{M}_r(V_1, V_2) \subset \mathcal{M}(V_1, V_2)$ implies

(4.2.6)
$$\inf_{U \in \mathcal{M}_r(V_1, V_2)} J_{(V_1, V_2)}(U) \ge \inf_{U \in \mathcal{M}(V_1, V_2)} J_{(V_1, V_2)}(U)$$

For $U = (u_1, u_2) \in \mathcal{M}(V_1, V_2)$, we set $\gamma(s, t) = (\sqrt{s} u_1, \sqrt{t} u_2)$. Since

$$J_{(V_1,V_2)}(\gamma(s,t)) = \frac{1}{2} (\|\nabla u_1\|_2^2 + V_1\|u_1\|_2^2)s + \frac{1}{2} (\|\nabla u_2\|_2^2 + V_2\|u_2\|_2^2)t - \frac{1}{4} (\mu_1\|u_1\|_4^4 s^2 + 2\beta \|u_1u_2\|_2^2 st + \mu_2\|u_2\|_4^4 t^2)$$

and (4.2.3) holds with (s,t) = (1,1), we can see

$$\sup_{(s,t)\in[0,\infty)^2} J_{(V_1,V_2)}(\gamma(s,t)) = J_{(V_1,V_2)}(\gamma(1,1)) = J_{(V_1,V_2)}(U).$$

We also set $\gamma^*(s,t) = (\sqrt{s} u_1^*, \sqrt{t} u_2^*) \in \Gamma$, where u_1^* (resp. u_2^*) is the Schwarz symmetrization of u_1 (resp. u_2). We can easily see that

$$J_{(V_1,V_2)}(\gamma^*(s,t)) \le J_{(V_1,V_2)}(\gamma(s,t))$$
 for all $(s,t) \in [0,\infty)^2$.

Thus

$$b(V_1, V_2) \leq \sup_{(s,t) \in [0,\infty)^2} J_{(V_1, V_2)}(\gamma^*(s, t)) \leq \sup_{(s,t) \in [0,\infty)^2} J_{(V_1, V_2)}(\gamma(s, t))$$

= $J_{(V_1, V_2)}(U).$

Since $U \in \mathcal{M}(V_1, V_2)$ is arbitrary, we have

(4.2.7)
$$b(V_1, V_2) \le \inf_{U \in \mathcal{M}(V_1, V_2)} J_{(V_1, V_2)}(U).$$

On the other hand, by Proposition 4.2.3, $b(V_1, V_2)$ is achieved by a nontrivial positive solution. Denoting the corresponding nontrivial solution (u_{01}, u_{02}) , we have $(u_{01}, u_{02}) \in \mathcal{M}_r(V_1, V_2)$ and thus

(4.2.8)
$$b(V_1, V_2) = J_{(V_1, V_2)}(u_{01}, u_{02}) \ge \inf_{U \in \mathcal{M}_r(V_1, V_2)} J_{(V_1, V_2)}(U).$$

Therefore we get (4.2.5) from (4.2.6)-(4.2.8).

Remark 4.2.6. In [4], Ambrosetti and Colorado showed the existence of nontrivial positive solutions via a mountain pass argument in $\mathcal{N}_r(V_1, V_2)$. More precisely they showed the following minimax value is corresponding to a nontrivial solution.

$$b_{mp}(V_1, V_2) = \inf_{\gamma \in \Gamma_{mp}(V_1, V_2)} \max_{t \in [0, 1]} J_{(V_1, V_2)}(\gamma(t)).$$

Here $\Gamma_{mp}(V_1, V_2)$ is the class of continuous curves in $\mathcal{N}_r(V_1, V_2)$ which join $\hat{\Omega}_1(V_1; x)$ and $\hat{\Omega}_2(V_2; x)$. We remark that $b_{mp}(V_1, V_2) = b(V_1, V_2)$ holds.

To show this fact, for any $U = (u_1, u_2) \in \mathcal{M}_r(V_1, V_2)$ we need to find a path $\gamma(t) \in \Gamma_{mp}(V_1, V_2)$ such that

(4.2.9)
$$\max_{t \in [0,1]} J_{(V_1, V_2)}(\gamma(t)) = J_{(V_1, V_2)}(U).$$

To construct such a path, we set

$$\gamma(t) := r(t)((1-t)u_1, tu_2),$$

where r(t) > 0 is uniquely determined so that $\gamma(t) \in \mathcal{N}_r(V_1, V_2)$. We can easily see that $\gamma(0) \in H_r^1(\mathbf{R}^N) \times \{0\}, \, \gamma(\frac{1}{2}) = U, \, \gamma(1) \in \{0\} \times H_r^1(\mathbf{R}^N) \text{ and } J_{(V_1,V_2)}(\gamma(t)) \leq J_{(V_1,V_2)}(U) \text{ for all } t \in [0,1].$ Using paths $\zeta_1(\tau)$ and $\zeta_2(\tau)$ in the proof of Proposition 4.2.3, we can find a path $\gamma(t) \in \Gamma_{mp}(V_1, V_2)$ satisfying (4.2.9).

4.2.4 Some compactness properties

We use the following notation:

$$\begin{aligned} \mathcal{S}_{r}(V_{1},V_{2}) &:= \{ \Omega = (\omega_{1},\omega_{2}) \in H^{1}_{r}(\mathbf{R}^{N}) \times H^{1}_{r}(\mathbf{R}^{N}) : J'_{(V_{1},V_{2})}(\Omega) = 0, \\ \omega_{1}(x) > 0, \, \omega_{2}(x) > 0, \, J_{(V_{1},V_{2})}(\Omega) = b(V_{1},V_{2}) \} & \text{for } (V_{1},V_{2}) \in A, \\ \mathcal{S}_{r,A} &:= \{ ((V_{1},V_{2}),\Omega) \in \mathbf{R}^{2} \times H^{1}_{r}(\mathbf{R}^{N}) \times H^{1}_{r}(\mathbf{R}^{N}) : (V_{1},V_{2}) \in A, \quad \Omega \in \mathcal{S}_{r}(V_{1},V_{2}) \}. \end{aligned}$$

The sets $\mathcal{S}_r(V_1, V_2)$ and $\mathcal{S}_{r,A}$ have the following compactness properties.

Proposition 4.2.7. Assume (4.1.10) and (C1). Then

- (i) There exist C_0 , C_1 , C_2 , $C_3 > 0$ such that for all $((V_1, V_2), \Omega) \in \mathcal{S}_{r,A}$, it follows that
 - $(4.2.10) \qquad \|\Omega\|_{H^1} \le C_0,$
 - $(4.2.11) \|\omega_1\|_4, \|\omega_2\|_4 \ge C_1,$
 - (4.2.12) $\omega_1(x), \, \omega_2(x), |\nabla \omega_1(x)|, \, |\nabla \omega_2(x)| \le C_2 e^{-C_3|x|}$ for all $x \in \mathbf{R}^N$.

Here we write $\Omega = (\omega_1, \omega_2)$.

(ii) The set $S_{r,A}$ is compact in $\mathbf{R}^2 \times H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$. More precisely, any sequence $((V_{1j}, V_{2j}), \Omega_j)_{j=1}^{\infty} \subset S_{r,A}$ has a subsequence $((V_{1j_n}, V_{2j_n}), \Omega_{j_n})$ and $((V_{10}, V_{20}), \Omega_0) \in S_{r,A}$ such that

$$(V_{1j_n}, V_{2j_n}) \to (V_{10}, V_{20})$$
 and $\Omega_{j_n} \to \Omega_0$ strongly in $H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$

- (iii) The function $b(V_1, V_2) : A \to \mathbf{R}$ is continuous.
- (iv) The function $b(V_1, V_2)$ is strictly increasing in V_1 (resp. V_2) for fixed V_2 (resp. V_1).

Proof. First we prove (iv). By the definition (4.2.2) of $b(V_1, V_2)$, it is clear that $b(V_1, V_2)$ is nondecreasing in V_1 for fixed V_2 . Since $b(V_1, V_2)$ is achieved by $\gamma(s, t) = (\sqrt{su_1}, \sqrt{tu_2})$ with $(u_1, u_2) \in S_r(V_1, V_2)$, it is also easy to see $b(V_1, V_2)$ is strictly increasing in V_1 for fixed V_2 . In a similar way, we can see that $b(V_1, V_2)$ is strictly increasing in V_2 for fixed V_1 .

Next we show (i). By (iv) we have

$$b(a_{10}, a_{20}) \le b(V_1, V_2) \le b(a_{11}, a_{21})$$
 for all $(V_1, V_2) \in A$.

Since $\|\Omega\|_{H^1(V_1,V_2)}^2 = 4b(V_1,V_2)$ for $\Omega \in \mathcal{S}_r(V_1,V_2)$, we have (4.2.10) for some constant C_0 independent of $(V_1,V_2) \in A$. We recall that for some constant $C_N > 0$, it holds that

$$|u(x)| \le \frac{C_N}{|x|^{(N-1)/2}} ||u||_{H^1}$$
 for $u \in H^1_r(\mathbf{R}^N)$ and $|x| \ge 1$.

(For example, see Lemma A.II of Berestycki and Lions [15].) Thus $\Omega = (\omega_1, \omega_2) \in S_r(V_1, V_2)$ satisfies

$$\omega_i(x) \le \frac{C_N C_0}{|x|^{(N-1)/2}}.$$

Since Ω satisfies (4.1.5)–(4.1.6), we can see (4.2.12) holds for some constants $C_2, C_3 > 0$ independent of $(V_1, V_2) \in A$.

Next we show (4.2.11). We argue indirectly and we assume that there exists a sequence $((V_{1j}, V_{2j}), \Omega_j) \in \mathcal{S}_{r,A}$ such that $\Omega_j = (\omega_{1j}, \omega_{2j})$ satisfies

(4.2.13)
$$\|\omega_{2j}\|_4 \to 0.$$

(we can deal with the case $\|\omega_{1j}\|_4 \to 0$ in a similar way.) We may also assume $(V_{1j}, V_{2j}) \to$ $(V_{10}, V_{20}) \in A$. Since Ω_j solves (4.1.5)–(4.1.8) and the embedding $H^1_r(\mathbf{R}^N) \subset L^4(\mathbf{R}^N)$ is compact, we can see that Ω_j has a strongly convergent subsequence. Extracting a subsequence if necessary, we may assume $\Omega_j \to \Omega_0 = (\omega_{10}, \omega_{20})$. It is easily seen that Ω_0 is a critical point of $J_{(V_{10},V_{20})}(U)$ and

(4.2.14)
$$b(V_{1j}, V_{2j}) = J_{(V_{1j}, V_{2j})}(\Omega_j) \to J_{(V_{10}, V_{20})}(\Omega_0).$$

On the other hand, it follows from $J'_{(V_{1j},V_{2j})}(\omega_{1j},\omega_{2j})(0,\omega_{2j})=0$ that

$$\|\nabla\omega_{2j}\|_{2}^{2} + \|\omega_{2j}\|_{2}^{2} = \mu_{2}\|\omega_{2j}\|_{4}^{4} + \beta\|\omega_{1j}\omega_{2j}\|_{2}^{2} \to 0 \quad \text{as } j \to \infty$$

under the condition (4.2.13). That is, $\omega_{20} = 0$ and $\Omega_0 = (\omega_{10}, 0)$ is a semitrivial positive solution. Thus,

$$J_{(V_{10},V_{20})}(\Omega_0) \le e_1(V_{10}).$$

By (4.2.14), we have $\limsup_{i\to\infty} b(V_{1i}, V_{2i}) \leq e_1(V_{10})$, which is in contradiction with (i) of Proposition 4.2.3. Thus (4.2.11) holds.

Next we show (ii). Suppose $((V_{1j}, V_{2j}), \Omega_j) \in \mathcal{S}_{r,A}$ $(j = 1, 2, \cdots)$ and $(V_{1j}, V_{2j}) \rightarrow$ $(V_{10}, V_{20}) \in A$. We may also assume $\Omega_j \to \Omega_0$ strongly in $H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$ and (4.2.14) holds. Here, by (4.2.11), Ω_0 is a nontrivial positive solution of (4.1.5)–(4.1.8) and we have

$$(4.2.15) J_{(V_{10},V_{20})}(\Omega_0) \ge b(V_{10},V_{20}).$$

On the other hand, by the definition of $b(V_1, V_2)$, we can see that $b(V_1, V_2)$ is upper semicontinuous, that is,

(4.2.16)
$$\limsup_{j \to \infty} b(V_{1j}, V_{2j}) \le b(V_{10}, V_{20}).$$

Thus by (4.2.14), (4.2.15) and (4.2.16), we have $J_{(V_{10},V_{20})}(\Omega_0) = b(V_{10},V_{20})$. Thus $\Omega_0 \in$ $S_r(V_{10}, V_{20})$, that is, $((V_{10}, V_{20}), \Omega_0) \in S_{r,A}$.

(iii) also follows from the proof of (ii).

Corollary 4.2.8. For any $\delta > 0$ there exists $\rho_1(\delta) > 0$ such that if (V_1, V_2) and $(\tilde{V}_1, \tilde{V}_2) \in$ A satisfy

$$|(V_1, V_2) - (\tilde{V}_1, \tilde{V}_2)| < \rho_1(\delta)$$

then

dist
$$(\Omega, \mathcal{S}_r(\tilde{V}_1, \tilde{V}_2)) \left(= \inf_{\tilde{\Omega} \in \mathcal{S}_r(\tilde{V}_1, \tilde{V}_2)} \|\Omega - \tilde{\Omega}\|_{H^1} \right) < \delta$$

for any $\Omega \in \mathcal{S}_r(V_1, V_2)$.

Now Proposition 4.1.1 easily follows.

Proof of Proposition 4.1.1. Proposition 4.1.1 follows from Propositions 4.2.3, 4.2.5, 4.2.7. \Box

Finally in this section, we prove (4.1.16) and (4.1.17).

For (4.1.16), we study the behavior of $m(P) = b(V_1(P), V_2(P))$ in the setting of Remark 4.1.4. Clearly (4.1.16) holds from the following lemma.

Lemma 4.2.9. Suppose that $V_1(x)$, $V_2(x) \in C^1(\mathbb{R}^N)$ in addition to the assumption of Theorem 4.1.3. If m(P) takes a local minimum at $P_0 \in \mathbb{R}^N$, that is, for some r > 0

 $m(P) \ge m(P_0)$ for all $|P - P_0| \le r$,

then there exist λ_1 and $\lambda_2 > 0$ such that

$$\lambda_1 \nabla V_1(P_0) + \lambda_2 \nabla V_2(P_0) = 0.$$

Proof. It suffices to show that if $P_0 \in \mathbf{R}^N$ satisfies

(4.2.17)
$$\lambda_1 \nabla V_1(P_0) + \lambda_2 \nabla V_2(P_0) \neq 0 \quad \text{for all } \lambda_1, \lambda_2 > 0,$$

then m(P) does not take a local minimum at P_0 .

First we remark that (4.2.17) implies that at least one of $\nabla V_1(P_0)$, $\nabla V_2(P_0)$ is not 0. We consider the case $\nabla V_1(P_0) \neq 0$. In this case, (4.2.17) implies $\nabla V_2(P_0) \notin \{-\lambda \nabla V_1(P_0) : \lambda > 0\}$ and we can find a vector $h_0 \in \mathbf{R}^N \setminus \{0\}$ such that

(4.2.18)
$$\nabla V_1(P_0)h_0 < 0 \text{ and } \nabla V_2(P_0)h_0 \le 0.$$

We will show that

(4.2.19)
$$m(P_0 + \tau h_0) < m(P_0)$$
 for small $\tau > 0$

Let $(u_{01}, u_{02}) \in \mathcal{M}_r(V_1(P_0), V_2(P_0))$ be a nontrivial solution corresponding to $m(P_0) = b(V_1(P_0), V_2(P_0))$ and set $\gamma_0(s, t) := (\sqrt{su_{01}}, \sqrt{tu_{02}}) \in \Gamma$. Then we have

$$m(P_0) = b(V_1(P_0), V_2(P_0)) = \sup_{(s,t) \in [0,\infty)^2} J_{(V_1(P_0), V_2(P_0))}(\gamma_0(s,t)).$$

We also remark that $(s,t) \mapsto J_{(V_1(P_0),V_2(P_0))}(\gamma_0(s,t))$ takes its maximum only at (s,t) = (1,1). Thus there exist δ_0 and $\tau_0 > 0$ such that for $0 < \tau < \tau_0$

$$(s,t) \mapsto J_{(V_1(P_0 + \tau h_0), V_2(P_0 + \tau h_0))}(\gamma_0(s,t))$$

takes its maximum in $\{(s,t); |s-1| + |t-1| \le \delta_0\}$. Thus we have

$$\begin{split} & m(P + \tau h_0) - m(P_0) \\ & \leq \max_{|s-1|+|t-1| \le \delta_0} J_{(V_1(P_0 + \tau h_0), V_2(P_0 + \tau h_0))}(\gamma_0(s, t)) - \max_{|s-1|+|t-1| \le \delta_0} J_{(V_1(P_0), V_2(P_0))}(\gamma_0(s, t)) \\ & \leq \max_{|s-1|+|t-1| \le \delta_0} \left[J_{(V_1(P_0 + \tau h_0), V_2(P_0 + \tau h_0))}(\gamma_0(s, t)) - J_{(V_1(P_0), V_2(P_0))}(\gamma_0(s, t)) \right] \\ & = \frac{1}{2} \max_{|s-1|+|t-1| \le \delta_0} \left[(V_1(P_0 + \tau h_0) - V_1(P_0)) \|u_{01}\|_2^2 s + (V_2(P_0 + \tau h_0) - V_2(P_0)) \|u_{02}\|_2^2 t \right]. \end{split}$$

We remark that (4.2.18) implies

$$m(P + \tau h_0) - m(P_0) \le \frac{\tau}{2} \Big[(\nabla V_1(P_0)h_0 + o(1)) \|u_{01}\|_2^2 (1 - \delta_0) + o(1) \|u_{02}\|_2^2 (1 + \delta_0) \Big]$$

< 0 for small $\tau > 0$,

which implies (4.2.19) and m(P) does not take a local minimum at P_0 . The case where $\nabla V_2(P_0) \neq 0$ can be treated in a similar way.

Next we deal with (4.1.17). In the following lemma we write dependence on β explicitly and use notation $b_{\beta}(V_1, V_2)$ for $b(V_1, V_2)$ as in Remark 4.1.5

Lemma 4.2.10. As $\beta \to 0$, it follows that $b_{\beta}(V_1, V_2) \to e_1(V_1) + e_2(V_2)$.

Proof. We also use notation $J_{(V_1,V_2),\beta}$ for $J_{(V_1,V_2)}$. We remark that $b_{\beta}(V_1, V_2)$ and $J_{(V_1,V_2),\beta}$ are nonincreasing in β and if (4.1.10) and (AC) hold for $\beta_0 > 0$, then (4.1.10) and (AC) hold for all $\beta \in (0, \beta_0]$.

By Proposition 4.2.3, there exists a nontrivial positive solution $U_{\beta}(x) \in H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$ such that for each $\beta \in (0, \beta_0]$

$$(4.2.20) \quad J_{(V_1,V_2),\beta}(U_\beta) = b_\beta(V_1,V_2) \in [\max\{e_1(V_1) + \rho_0, e_2(V_2) + \rho_0\}, e_1(V_1) + e_2(V_2)].$$

Here $\rho_0 > 0$ is independent of β . As in the proof of Proposition 4.2.7, we can show that $(U_{\beta})_{\beta \in (0,\beta_0]}$ is bounded in $H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$ and, after extracting a subsequence $\beta_j \to 0$, $U_{\beta_j}(x)$ converges strongly to some critical point $U_0(x)$ of $J_{(V_1,V_2),0}(U)$. We remark that $J_{(V_1,V_2),0}(U)$ is corresponding to two equations without interaction:

$$-\Delta u_1 + V_1 u_1 = \mu_1 u_1^3 \text{ in } \mathbf{R}^N, -\Delta u_2 + V_2 u_2 = \mu_2 u_2^3 \text{ in } \mathbf{R}^N.$$

By (4.2.20),

$$\max\{e_1(V_1) + \rho_0, e_2(V_2) + \rho_0\} \le J_{(V_1, V_2), 0}(U_0).$$

Since $U_0(x)$ is nonnegative, from (4.2.11), we have $U_0(x) = (\hat{\omega}_1(V_1; x), \hat{\omega}_2(V_2; x))$ and $b_{\beta_j}(V_1, V_2) \to e_1(V_1) + e_2(V_2)$. Since the limit does not depend on the choice of subsequence β_j , we have the conclusion of Lemma 4.2.10.

Remark 4.2.11. We can also show that if $(V_{1j}, V_{2j}) \to (V_{10}, V_{20})$ where $V_{10}, V_{20} > 0$, then $b_{\beta}(V_{1j}, V_{2j}) \to e_1(V_{10}) + e_2(V_{20})$ as $\beta \to 0$.

4.3 Nehari manifolds and the Palais-Smale condition

4.3.1 A singular perturbation problem

In the following sections we study a singular perturbation problem (4.1.1)-(4.1.4). From now on we assume (A1) and (A2).

We set $u_i(x) = v_i(\varepsilon x)$ and we try to find a solution $U(x) = (u_1(x), u_2(x))$ of

(4.3.1) $-\Delta u_1 + V_1(\varepsilon x)u_1 = \mu_1 u_1^3 + \beta u_1 u_2^2 \text{ in } \mathbf{R}^N,$

(4.3.2)
$$-\Delta u_2 + V_2(\varepsilon x)u_2 = \beta u_1^2 u_2 + \mu_2 u_2^3 \text{ in } \mathbf{R}^N$$

- (4.3.3) $u_1(x), u_2(x) > 0 \text{ in } \mathbf{R}^N,$
- (4.3.4) $u_1(x), \ u_2(x) \in H^1(\mathbf{R}^N).$

A functional corresponding to (4.3.1)–(4.3.4) is

$$\begin{split} I_{\varepsilon}(U) &:= I_{\varepsilon}(u_{1}, u_{2}) \\ &= \frac{1}{2} (\|\nabla u_{1}\|_{2}^{2} + \int_{\mathbf{R}^{N}} V_{1}(\varepsilon x) u_{1}^{2} dx) + \frac{1}{2} (\|\nabla u_{2}\|_{2}^{2} + \int_{\mathbf{R}^{N}} V_{2}(\varepsilon x) u_{2}^{2} dx) \\ &\quad - \frac{1}{4} (\mu_{1}\|u_{1}\|_{4}^{4} + 2\beta \|u_{1}u_{2}\|_{2}^{2} + \mu_{2} \|u_{2}\|_{4}^{4}) \\ &\in C^{1}(H^{1}(\mathbf{R}^{N}) \times H^{1}(\mathbf{R}^{N}), \mathbf{R}). \end{split}$$

We use the following notation:

$$\begin{aligned} \|u_i\|_{H^{1},i,\varepsilon}^2 &:= \|\nabla u_i\|_2^2 + \int_{\mathbf{R}^N} V_i(\varepsilon x) u_i^2 \, dx \quad \text{for } i = 1, 2, \\ \|U\|_{H^{1},\varepsilon}^2 &:= \|u_1\|_{H^{1},1,\varepsilon}^2 + \|u_2\|_{H^{1},2,\varepsilon}^2. \end{aligned}$$

With this notation, we can write

(4.3.5)
$$I_{\varepsilon}(U) = \frac{1}{2} \|U\|_{H^{1},\varepsilon}^{2} - \frac{1}{4} \int_{\mathbf{R}^{N}} W(U) \, dx.$$

We remark that under (A1) there exist constants $a_1, a_2 > 0$

(4.3.6)
$$a_1 \|U\|_{H^1} \le \|U\|_{H^{1,\varepsilon}} \le a_2 \|U\|_{H^1}$$
 for all $U \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$.

For d > 0 we set

$$K^d := \{ x \in \mathbf{R}^N : \operatorname{dist}(x, K) \le d \}.$$

We remark that there exists $d_1 > 0$ such that

$$K^{d_1} \subset \Lambda.$$

Under the conditions (A1)–(A2), we introduce for $P \in \mathbf{R}^N$ and $\delta > 0$

$$S_{r,P} := S_r(V_1(P), V_2(P)),$$

$$S_{r,P}^{\delta} := \{\Omega_P + \Phi : \Omega_P \in S_{r,P}, \ \Phi \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N), \ \|\Phi\|_{H^1} < \delta\},$$

$$m(P) := b(V_1(P), V_2(P)).$$

It follows from Proposition 4.2.7 that $S_{r,P}$ $(P \in \mathbf{R}^N)$ and $\bigcup_{P \in K^d} S_{r,P}$ are compact in $H^1_r(\mathbf{R}^N) \times H^1_r(\mathbf{R}^N)$. We set for ε , d, $\delta > 0$,

$$\begin{aligned} \mathcal{S}_{\varepsilon,d} &:= \{ \Omega_P(x - P/\varepsilon) : P \in K^d, \ \Omega_P \in \mathcal{S}_{r,P} \}, \\ \mathcal{S}^{\delta}_{\varepsilon,d} &:= \{ \Omega_P(x - P/\varepsilon) + \Phi(x) : P \in K^d, \ \Omega_P \in \mathcal{S}_{r,P}, \\ \Phi \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N), \|\Phi\|_{H^1} < \delta \}. \end{aligned}$$

Such a type of sets are introduced in Byeon and Jeanjean [22, 23] (cf. Byeon, Jeanjean and Tanaka [24]) for nonlinear Schrödinger equations $-\varepsilon^2 \Delta u + V(x)u = f(u)$ and used successfully to construct a family of solutions, which concentrates at a local minimum of V(x) without assumptions of uniqueness or nondegeneracy of solutions of the limit problems.

Remark 4.3.1. In [22, 23, 24], they introduced a class of sets which is slightly different from our $S_{\varepsilon,d}^{\delta}$. Their class of sets is, in our setting,

$$\mathcal{X}_{\varepsilon,d} := \{ \Omega_P(x - \tilde{P}/\varepsilon) : P \in K, |P - \tilde{P}| \le d, \, \Omega_P \in \mathcal{S}_{r,P} \}, \\ \mathcal{X}^{\delta}_{\varepsilon,d} := \{ \Omega_P(x - \tilde{P}/\varepsilon) + \Phi(x) : \, \Omega_P(x - \tilde{P}/\varepsilon) \in \mathcal{X}_{\varepsilon,d}, \\ \Phi \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N), \, \|\Phi\|_{H^1} < \delta \}.$$

We remark that

(4.3.7)
$$\mathcal{X}_{\varepsilon,d} \subset \mathcal{S}^{\delta}_{\varepsilon,d}, \quad \mathcal{X}^{\delta}_{\varepsilon,d} \subset \mathcal{S}^{2\delta}_{\varepsilon,d}$$

for sufficiently small d. In fact, by Corollary 4.2.8, for any $\delta > 0$ there exists a $\rho_2(\delta) > 0$ such that

 $\mathcal{S}_{r,P} \subset \mathcal{S}_{r,\tilde{P}}^{\delta}$ for any $P, \ \tilde{P} \in \overline{\Lambda}$ with $|P - \tilde{P}| \leq \rho_2(\delta)$.

Thus (4.3.7) holds for $d \in (0, \rho_2(\delta)]$.

In [22, 23, 24], the desired solution is obtained through mountain pass arguments in $H^1(\mathbf{R}^N)$. Here we deal with the constraint problem $I_{\varepsilon}|_{\mathcal{M}_{\varepsilon}} : \mathcal{M}_{\varepsilon} \to \mathbf{R}$, where

(4.3.8)
$$\mathcal{M}_{\varepsilon} := \{ U = (u_1, u_2) \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N) : u_1 \neq 0, u_2 \neq 0, \\ I'_{\varepsilon}(U)(u_1, 0) = 0, I'_{\varepsilon}(U)(0, u_2) = 0 \},$$

and we try to find a critical point corresponding to a local minimum of $m: \Lambda \to \mathbf{R}$.

Here we fix $d_0 \in (0, d_1]$ arbitrary and, in what follows, we try to find a critical point of $I_{\varepsilon}(U)$ in $\mathcal{S}_{\varepsilon,d_0}^{\delta}$. We use the following abbreviation:

$$\mathcal{S}_{\varepsilon} = \mathcal{S}_{\varepsilon, d_0}$$
 and $\mathcal{S}_{\varepsilon}^{\delta} = \mathcal{S}_{\varepsilon, d_0}^{\delta}$.

We start with the following

Lemma 4.3.2. There exist ε_1 and $\delta_1 > 0$ which have the following properties:

- (i) If $\varepsilon \in (0, \varepsilon_1]$ and $U(x) \in S_{\varepsilon}^{\delta_1}$ satisfy $I'_{\varepsilon}(U) = 0$, then U(x) is a nontrivial positive solution of (4.3.1)–(4.3.4).
- (ii) If $P \in K^{d_0}$ and $U \in \mathcal{S}_{r,P}^{\delta_1}$ satisfy $J'_{(V_1(P),V_2(P))}(U) = 0$, then U(x) is a nontrivial positive solution of (4.1.5)–(4.1.8) with $V_1 = V_1(P)$ and $V_2 = V_2(P)$.

Proof. We give just an outline of a proof of (i).

It suffices to show that both components of U(x) are positive. Via the standard regularity argument for solutions of elliptic equations, we can show that for any $\nu > 0$ there exist ε_{ν} and $\delta_{\nu} > 0$ such that

$$\|\Phi\|_{\infty} < \nu$$

for all $\varepsilon \in (0, \varepsilon_{\nu}]$ and $U(x) = \Omega_P(x - P/\varepsilon) + \Phi(x - P/\varepsilon) \in \mathcal{S}^{\delta_{\nu}}_{\varepsilon}$ satisfying $I'_{\varepsilon}(U) = 0$.

For any R > 0 we choose $\nu > 0$ small so that the following inequality holds for all $\varepsilon \in (0, \varepsilon_{\nu}]$ and $U(x) = (u_1(x), u_2(x)) (= \Omega_P(x - P/\varepsilon) + \Phi(x - P/\varepsilon)) \in \mathcal{S}_{\varepsilon}^{\delta_{\nu}}$ satisfying $I'_{\varepsilon}(U) = 0$:

$$u_1(x), u_2(x) > 0 \text{ for } |x - P/\varepsilon| \le R.$$

In particular, we have $\operatorname{supp} u_{1-}$, $\operatorname{supp} u_{2-} \subset \{x \in \mathbf{R}^N : |x - P/\varepsilon| \geq R\}$. Setting $U_- = (u_{1-}, u_{2-})$, it follows from $I'_{\varepsilon}(U)U_- = 0$ that

(4.3.9)
$$\|U_{-}\|_{H^{1},\varepsilon}^{2} - \int_{|x-P/\varepsilon| \ge R} \nabla W(U) U_{-} dx = 0$$

When R > 0 is sufficiently large, $|\nabla W(U(x))| \ll 1$ for $|x - P/\varepsilon| \ge R$ and (4.3.9) implies $U_{-} \equiv 0$, that is, both components of U(x) are nonnegative.

By the maximal principle we have $u_1(x)$, $u_2(x) > 0$ for all $x \in \mathbf{R}^N$ and U(x) is a nontrivial positive solution. Setting $\varepsilon_1 = \varepsilon_{\nu}$, $\delta_1 = \delta_{\nu}$, we have (i).

In what follows, we always assume $\varepsilon \in (0, \varepsilon_1]$ and $\delta \in (0, \delta_1]$.

4.3.2 Nehari type manifold $\mathcal{M}_{\varepsilon}$ and a projection $\mathcal{P}_{\varepsilon}: \, \mathcal{S}_{\varepsilon}^{\delta} \to \mathcal{M}_{\varepsilon}$

We can show the following lemma as in Lemma 4.2.4.

Lemma 4.3.3. Assume (4.1.10). Then we have

- (i) The set $\mathcal{M}_{\varepsilon}$ is a submanifold of $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$ with codimension 2.
- (ii) For $(u_1, u_2) \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$ with $u_1 \neq 0, u_2 \neq 0$ and s, t > 0,

$$(\sqrt{s}\,u_1,\sqrt{t}\,u_2)\in\mathcal{M}_{\varepsilon}$$

holds if and only if

(4.3.10)
$$\begin{cases} \mu_1 \|u_1\|_4^4 s + \beta \|u_1 u_2\|_2^2 t = \|\nabla u_1\|_2^2 + \int_{\mathbf{R}^N} V_1(\varepsilon x) u_1^2 dx, \\ \beta \|u_1 u_2\|_2^2 s + \mu_2 \|u_2\|_4^4 t = \|\nabla u_2\|_2^2 + \int_{\mathbf{R}^N} V_2(\varepsilon x) u_2^2 dx. \end{cases}$$

For $U = (u_1, u_2) \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$ with $u_1 \neq 0$ and $u_2 \neq 0$, we set

$$A(U) = \begin{bmatrix} \mu_1 \| u_1 \|_4^4 & \beta \| u_1 u_2 \|_2^2 \\ \beta \| u_1 u_2 \|_2^2 & \mu_2 \| u_2 \|_4^4 \end{bmatrix}.$$

By (4.2.4), we remark that A(U) is invertible and the system (4.3.10) has a unique solution — we denote it $(s_{\varepsilon}(U), t_{\varepsilon}(U))$ —.

For $P \in \mathbf{R}^N$, we also consider

$$A(U) \begin{bmatrix} s \\ t \end{bmatrix} = \begin{bmatrix} \|\nabla u_1\|_2^2 + V_1(P)\|u_1\|_2^2 \\ \|\nabla u_2\|_2^2 + V_2(P)\|u_2\|_2^2 \end{bmatrix}$$

which is equivalent to $(\sqrt{s} u_1, \sqrt{t} u_2) \in \mathcal{M}(V_1(P), V_2(P))$. We denote the unique solution by $(s_P(U), t_P(U))$. We have the following

Lemma 4.3.4. For any $\nu \in (0, \frac{1}{2})$ there exist $\varepsilon_2 \in (0, \varepsilon_1)$ and $\delta_2 \in (0, \delta_1)$ such that for each $\varepsilon \in (0, \varepsilon_2]$ and $\delta \in (0, \delta_2]$

- (i) For all $U \in S_{\varepsilon}^{\delta}$, it holds that $s_{\varepsilon}(U), t_{\varepsilon}(U) \in (1 \nu, 1 + \nu)$.
- (ii) It holds that $s_P(U)$, $t_P(U) \in (1 \nu, 1 + \nu)$ for all $U(x) = \Omega_P(x P/\varepsilon) + \Phi(x)$ satisfying $P \in \overline{\Lambda}$, $\Omega_P \in \mathcal{S}_{r,P}$ and $\|\Phi\|_{H^1} < \delta$.

Proof. We can see that

$$\begin{bmatrix} s_{\varepsilon}(U) \\ t_{\varepsilon}(U) \end{bmatrix} = A(U)^{-1} \begin{bmatrix} \|u_1\|_{H^{1,1,\varepsilon}}^2 \\ \|u_2\|_{H^{1,2,\varepsilon}}^2 \end{bmatrix}, \\ \begin{bmatrix} s_P(U) \\ t_P(U) \end{bmatrix} = A(U)^{-1} \begin{bmatrix} \|\nabla u_1\|_2^2 + V_1(P)\|u_1\|_2^2 \\ \|\nabla u_2\|_2^2 + V_2(P)\|u_2\|_2^2 \end{bmatrix}.$$

Since $\Omega_P = (\omega_{P1}, \omega_{P2}) \in \mathcal{S}_{r,P} \subset \mathcal{M}_r(V_1(P), V_2(P))$, we have (4.2.3) with (s, t) = (1, 1), that is,

$$\begin{bmatrix} 1\\1 \end{bmatrix} = A(\Omega_P)^{-1} \begin{bmatrix} \|\nabla \omega_{P1}\|_2^2 + V_1(P)\|\omega_{P1}\|_2^2\\ \|\nabla \omega_{P2}\|_2^2 + V_2(P)\|\omega_{P2}\|_2^2 \end{bmatrix}.$$

We remark that

(4.3.11)

$$\|\omega_{Pi}(x - P/\varepsilon)\|_{H^{1},i,\varepsilon}^{2} = \|\nabla\omega_{Pi}\|_{2}^{2} + \int_{\mathbf{R}^{N}} V_{1}(\varepsilon x + P)\omega_{Pi}^{2} \, dx \to \|\nabla\omega_{Pi}\|_{2}^{2} + V_{i}(P)\|\omega_{Pi}\|_{2}^{2}$$

for i = 1, 2 as $\varepsilon \to 0$ uniformly in $P \in \overline{\Lambda}$ and $\Omega_P \in \mathcal{S}_{r,P}$.

Thus the conclusions (i) and (ii) follow from the continuity of $U \mapsto A(U)^{-1}$ in a neighborhood of $\mathcal{S}_{\varepsilon}$ and compactness of $\bigcup_{P \in \overline{\Lambda}} \mathcal{S}_{r,P}$.

By Lemma 4.3.4, we can see the projections

$$\mathcal{P}_{\varepsilon}: \mathcal{S}_{\varepsilon}^{\delta} \to \mathcal{M}_{\varepsilon}; U = (u_1, u_2) \mapsto (\sqrt{s_{\varepsilon}(U)} \, u_1, \sqrt{t_{\varepsilon}(U)} \, u_2), \\ \mathcal{P}_P: \{\Omega_P(x - P/\varepsilon) + \Phi(x): \Omega_P \in \mathcal{S}_{r,P}, \|\Phi\|_{H^1} < \delta\} \to \mathcal{M}(V_1(P), V_2(P)); \\ U = (u_1, u_2) \mapsto (\sqrt{s_P(U)} \, u_1, \sqrt{t_P(U)} \, u_2)$$

are well-defined and continuous for $\varepsilon \in (0, \varepsilon_2]$ and $\delta \in (0, \delta_2]$.

The projection $\mathcal{P}_{\varepsilon}$ has the following properties:

Lemma 4.3.5. There exist $\varepsilon_3 \in (0, \varepsilon_2)$, $\delta_3 \in (0, \delta_2)$, $L_0 > 0$ and a nondecreasing function $\rho_3(\varepsilon) \in C([0, \varepsilon_3], [0, \infty))$ satisfying $\rho_3(0) = 0$ such that

(i) For all $\varepsilon \in (0, \varepsilon_3]$, $P \in \overline{\Lambda}$ and $\Omega_P \in \mathcal{S}_{r,P}$,

$$\|\mathcal{P}_{\varepsilon}(\Omega_P(x-P/\varepsilon)) - \Omega_P(x-P/\varepsilon)\|_{H^1 \times H^1} \le \rho_3(\varepsilon)$$

(ii) For all $\varepsilon \in (0, \varepsilon_3]$ and $U, \tilde{U} \in \mathcal{S}_{\varepsilon}^{\delta_3}$,

(4.3.12)
$$\|\mathcal{P}_{\varepsilon}(U) - \mathcal{P}_{\varepsilon}(\tilde{U})\|_{H^{1} \times H^{1}} \leq L_{0} \|U - \tilde{U}\|_{H^{1} \times H^{1}}$$

(iii) For all $\varepsilon \in (0, \varepsilon_3]$ and $\delta \in (0, \delta_3]$, it holds that $\mathcal{S}^{\delta}_{\varepsilon} \cap \mathcal{M}_{\varepsilon} \subset \mathcal{P}_{\varepsilon}(\mathcal{S}^{\delta}_{\varepsilon}) \subset \mathcal{S}^{L_0\delta + \rho_3(\varepsilon)}_{\varepsilon}$.

Proof. Noting the convergence (4.3.11) is uniform in $P \in \overline{\Lambda}$ and $\Omega_P \in \mathcal{S}_{r,P}$, we have (i). Since $U \mapsto A(U)^{-1}$, $\|\nabla u_i\|_{H^1,i,\varepsilon}^2$, \cdots are Lipschitz continuous in $\mathcal{S}_{\varepsilon}^{\delta}$, (ii) holds. (iii) follows from (i) and (ii).

Choosing ε_3 , $\delta_3 > 0$ smaller if necessary, we may assume

(i) It follows that

(4.3.13)
$$\delta_3 \leq \frac{1}{4} \min\{\|\omega_1\|_{H^1}, \|\omega_2\|_{H^1} : (\omega_1, \omega_2) \in \bigcup_{P \in \overline{\Lambda}} \mathcal{S}_{r,P}\}.$$

(ii) For $a_1 > 0$ appeared in (4.3.6),

(4.3.14)
$$\int_{\mathbf{R}^N} \nabla W(U) U \, dx \le \frac{a_1^2}{2} \|U\|_{H^1}^2 \quad \text{for} \quad \|U\|_{H^1} \le 2\delta_3.$$

(iii) There exist $C_4, C_5, C_6 > 0$ such that for any $\varepsilon \in (0, \varepsilon_3], \delta \in (0, \delta_3]$ and $U \in \mathcal{S}^{\delta}_{\varepsilon}$

$$(4.3.15) ||U||_{H^1} \le C_4,$$

- $(4.3.16) ||A(U)||_{M(2,2)} \le C_5,$
- (4.3.17) $||A(U)^{-1}||_{M(2,2)} \le C_6,$

(4.3.18)
$$A(U)^{-1} \begin{bmatrix} \|u_1\|_{H^1,1,\varepsilon}^2 \\ \|u_2\|_{H^1,2,\varepsilon}^2 \end{bmatrix} \in \left\{ \begin{bmatrix} s \\ t \end{bmatrix} : s,t \in \begin{bmatrix} \frac{1}{2},\frac{3}{2} \end{bmatrix} \right\}.$$

Here we use notation: $||A||_{M(2,2)} := \sup_{\xi \in \mathbf{R}^2, |\xi|=1} |A\xi|$ for 2×2 matrix A.

4.3.3 The Palais-Smale condition in $\mathcal{S}^{\delta}_{\varepsilon} \cap \mathcal{M}_{\varepsilon}$

The behavior of the Palais-Smale sequence is important for the proof of our main result. Here we consider its behavior in the set $S_{\varepsilon}^{\delta} \cap \mathcal{M}_{\varepsilon}$.

For $U \in \mathcal{M}_{\varepsilon}$ we use the following notation:

$$\|I_{\varepsilon}'(U)\|_{(T_U\mathcal{M}_{\varepsilon})^*} := \sup_{\Phi \in T_U\mathcal{M}_{\varepsilon}, \|\Phi\|_{H^1} \le 1} |I_{\varepsilon}'(U)\Phi|.$$

Here $T_U \mathcal{M}_{\varepsilon}$ is the tangent space of $\mathcal{M}_{\varepsilon}$ at U:

$$T_U\mathcal{M}_{\varepsilon} := \{ \Phi \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N) : F_{1\varepsilon}'(U)\Phi = F_{2\varepsilon}'(U)\Phi = 0 \},\$$

where $F_{1\varepsilon}(u_1, u_2) = I'_{\varepsilon}(u_1, u_2)(u_1, 0), \ F_{2\varepsilon}(u_1, u_2) = I'_{\varepsilon}(u_1, u_2)(0, u_2).$ We show 2 types of the concentration-compactness results.

Proposition 4.3.6 Assume $\delta \in (0, \delta_{2}]$ Furthermore for a fixed $\varepsilon \in (0, \varepsilon_{2})$

Proposition 4.3.6. Assume $\delta \in (0, \delta_3]$. Furthermore, for a fixed $\varepsilon \in (0, \varepsilon_3]$, suppose a sequence $(U_j)_{j=1}^{\infty} \subset S_{\varepsilon}^{\delta} \cap \mathcal{M}_{\varepsilon}$ satisfies for some $c_0 \in \mathbf{R}$

$$\begin{split} &I_{\varepsilon}(U_{j}) \to c_{0}, \\ &\|I_{\varepsilon}'(U_{j})\|_{(T_{U_{j}}\mathcal{M}_{\varepsilon})^{*}} \to 0 \quad as \ j \to \infty \end{split}$$

Then U_j has a strongly convergent subsequence in $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$ and its limit U_0 is a critical point of $I_{\varepsilon}(U)$ which satisfies $I_{\varepsilon}(U_0) = c_0$ and $U_0(x) > 0$. **Proposition 4.3.7.** Assume $\delta \in (0, \delta_3]$. Furthermore suppose that sequences $(\varepsilon_j)_{j=1}^{\infty} \subset (0, \varepsilon_3]$ and $(U_j)_{j=1}^{\infty} \subset H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$ satisfy for some $c_0 \leq m_0$

- $(4.3.19) \varepsilon_j \to 0,$
- (4.3.20) $U_j \in \mathcal{S}^{\delta}_{\varepsilon_j} \cap \mathcal{M}_{\varepsilon_j},$
- (4.3.21) $I_{\varepsilon_j}(U_j) \to c_0(\leq m_0),$
- (4.3.22) $\|I'_{\varepsilon_j}(U_j)\|_{(T_{U_j}\mathcal{M}_{\varepsilon_j})^*} \to 0.$

Then $c_0 = m_0$ and there exists a subsequence — we still denote it by $j - (Q_j)_{j=1}^{\infty} \subset \mathbf{R}^N$, $Q_0 \in K$ and $\Omega_0 \in \mathcal{S}_{r,Q_0}$ such that

(4.3.23)
$$Q_j \to Q_0 \in K,$$

(4.3.24) $U_j(x + Q_j/\varepsilon_j) \to \Omega_0(x)$ strongly in $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$

In what follows we give a proof of Proposition 4.3.7 and an outline of a proof of Proposition 4.3.6.

Proof of Proposition 4.3.7. Suppose $(\varepsilon_j)_{j=1}^{\infty}$ and $(U_j)_{j=1}^{\infty}$ satisfy (4.3.19)–(4.3.22). **Step 1:** $\|I'_{\varepsilon_j}(U_j)\|_{H^{-1}} \to 0$ as $j \to \infty$.

We remark that

$$\|I_{\varepsilon}'(U)\|_{(T_{U}\mathcal{M}_{\varepsilon})^{*}} = \min_{c_{1},c_{2}\in\mathbf{R}}\|I_{\varepsilon}'(U) - c_{1}F_{1\varepsilon}'(U) - c_{2}F_{2\varepsilon}'(U)\|_{H^{-1}}.$$

Thus, under the assumption (4.3.22), there exists a sequence $(c_{1j}, c_{2j}) \in \mathbf{R}^2$ such that

$$||I'_{\varepsilon_j}(U_j) - c_{1j}F'_{1\varepsilon_j}(U_j) - c_{2j}F'_{2\varepsilon_j}(U_j)||_{H^{-1}} \to 0.$$

In particular, writing $U_j := (u_{1j}, u_{2j})$, we have

(4.3.25)
$$(I'_{\varepsilon_j}(U_j) - c_{1j}F'_{1\varepsilon_j}(U_j) - c_{2j}F'_{2\varepsilon_j}(U_j))(u_{1j}, 0) \to 0,$$

(4.3.26)
$$(I'_{\varepsilon_j}(U_j) - c_{1j}F'_{1\varepsilon_j}(U_j) - c_{2j}F'_{2\varepsilon_j}(U_j))(0, u_{2j}) \to 0.$$

For $U = (u_1, u_2) \in \mathcal{M}_{\varepsilon}$, we note that $I'_{\varepsilon}(U)(u_1, 0) = I'_{\varepsilon}(U)(0, u_2) = 0$ and

$$F_{1\varepsilon}'(U)(u_1,0) = -2\mu_1 \|u_1\|_4^4, \qquad F_{2\varepsilon}'(U)(u_1,0) = -2\beta \|u_1u_2\|_2^2, F_{1\varepsilon}'(U)(0,u_2) = -2\beta \|u_1u_2\|_2^2, \qquad F_{2\varepsilon}'(U)(0,u_2) = -2\mu_2 \|u_2\|_4^4.$$

Thus (4.3.25) - (4.3.26) implies

$$A(U_j) \begin{bmatrix} c_{1j} \\ c_{2j} \end{bmatrix} \to \begin{bmatrix} 0 \\ 0 \end{bmatrix}$$
 as $j \to \infty$.

By (4.3.17), $A(U)^{-1}$ is bounded in $\mathcal{S}^{\delta}_{\varepsilon}$. Thus we have $c_{1j}, c_{2j} \to 0$ and

$$\|I_{\varepsilon_j}'(U_j)\|_{H^{-1}} \le |c_{1j}| \|F_{1\varepsilon_j}'(U_j)\|_{H^{-1}} + |c_{2j}| \|F_{2\varepsilon_j}'(U_j)\|_{H^{-1}} + o(1) \to 0$$

as $j \to \infty$. Thus Step 1 is proved.

Since $U_j \in \mathcal{S}_{\varepsilon_j}^{\delta}$, we can write $U_j(x) = \Omega_{P_j}(x - P_j/\varepsilon_j) + \Phi_j(x - P_j/\varepsilon_j)$ with $P_j \in K^{d_0}$, $\Omega_{P_j} \in \mathcal{S}_{r,P_j}, \|\Phi_j\|_{H^1} < \delta$. We set

$$\tilde{U}_j(x) := U_j(x + P_j/\varepsilon_j) = \Omega_{P_j}(x) + \Phi_j(x) \in \mathcal{S}^{\delta}_{r,P_j}$$

and suppose

$$P_{j} \to P_{0} \in K^{d_{0}},$$

$$\Omega_{P_{j}}(x) \to \Omega_{P_{0}}(x) \in \mathcal{S}_{r,P_{0}} \quad \text{strongly in } H^{1}(\mathbf{R}^{N}) \times H^{1}(\mathbf{R}^{N}),$$

$$\Phi_{i}(x) \to \Phi_{0}(x) \qquad \text{weakly in } H^{1}(\mathbf{R}^{N}) \times H^{1}(\mathbf{R}^{N}).$$

Here we used the compactness of $S_{r,A}$. We also set $\tilde{U}_0(x) := \Omega_{P_0}(x) + \Phi_0(x)$ and $\Psi_j(x) := \tilde{U}_j(x) - \tilde{U}_0(x)$. It follows from (4.3.13) that $\tilde{U}_0 = (\tilde{u}_{01}, \tilde{u}_{02})$ satisfies $\tilde{u}_{01} \neq 0$, $\tilde{u}_{02} \neq 0$. We also have

(4.3.27)
$$\begin{split} &\lim_{j \to \infty} \|\Psi_j\|_{H^1} \le 2\delta, \\ &\Psi_j(x) \rightharpoonup 0 \qquad \text{weakly in } H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N), \\ &\tilde{U}_j(x) = \tilde{U}_0(x) + \Psi_j(x) \rightharpoonup \tilde{U}_0(x) \quad \text{weakly in } H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N). \end{split}$$

Next we show

Step 2: $\|\Psi_j\|_{H^1} \to 0$ as $j \to \infty$. In particular, $\tilde{U}_j(x) \to \tilde{U}_0(x)$ strongly in $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$.

Since $I'_{\varepsilon_i}(U_j) \to 0$, we have

$$I'_{\varepsilon_j}(U_j)\Psi_j(x-P_j/\varepsilon_j)\to 0.$$

Thus, writing $\Psi_j = (\psi_{1j}, \psi_{2j})$ and noting $\tilde{U}_j = \tilde{U}_0 + \Psi_j = (\tilde{u}_{10} + \psi_{1j}, \tilde{u}_{20} + \psi_{2j})$, we have

$$\int_{\mathbf{R}^N} \nabla(\tilde{u}_{10} + \psi_{1j}) \nabla \psi_{1j} + V_1(\varepsilon_j x + P_j) (\tilde{u}_{1j} + \psi_{1j}) \psi_{1j} dx$$
$$+ \int_{\mathbf{R}^N} \nabla(\tilde{u}_{20} + \psi_{2j}) \nabla \psi_{2j} + V_2(\varepsilon_j x + P_j) (\tilde{u}_{2j} + \psi_{2j}) \psi_{2j} dx$$
$$- \int_{\mathbf{R}^N} \nabla W (\tilde{U}_0 + \Psi_j) \Psi_j dx \to 0.$$

It follows from (4.3.27) that for i = 1, 2, $\int_{\mathbf{R}^N} \nabla \tilde{u}_{i0} \nabla \psi_{ij} dx$, $\int_{\mathbf{R}^N} V_i(\varepsilon_j x + P_j) \tilde{u}_{i0} \psi_{ij} dx$, $\int_{\mathbf{R}^N} \tilde{u}_{10}^3 \psi_{ij} dx$, $\int_{\mathbf{R}^N} \tilde{u}_{10}^2 \psi_{ij}^2 dx$, $\dots \to 0$. Thus,

$$\|\Psi_j\|_{H^1,\varepsilon_j}^2 - \int_{\mathbf{R}^N} \nabla W(\Psi_j) \Psi_j \, dx \to 0.$$

Thus by (4.3.6) and (4.3.14),

$$\frac{a_1^2}{2} \|\Psi_j\|_{H^1}^2 \le \|\Psi_j\|_{H^{1},\varepsilon_j}^2 - \int_{\mathbf{R}^N} \nabla W(\Psi_j)\Psi_j \, dx \to 0.$$

That is, the conclusion of Step 2 holds.

Next we show

Step 3: $P_0 \in K$ and $\tilde{U}_0(x - \ell_0) \in S_{r,P_0}$ for some $\ell_0 \in \mathbf{R}^N$.

We remark that Step 2 implies $\tilde{U}_j \to \tilde{U}_0$ strongly in $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$, from which it follows

(4.3.28)
$$J_{(V_1(P_0),V_2(P_0))}(\tilde{U}_0) = \lim_{j \to \infty} I_{\varepsilon_j}(\tilde{U}_j(x - P_j/\varepsilon_j)) = c_0 (\leq m_0),$$
$$J'_{(V_1(P_0),V_2(P_0))}(\tilde{U}_0) = 0.$$

Thus, by Lemma 4.3.2, $\tilde{U}_0(x) \in \overline{\mathcal{S}_{r,P_0}^{\delta}} \subset \mathcal{S}_{r,P_0}^{\delta_1}$ is a nontrivial positive critical point of $J_{(V_1(P_0),V_2(P_0))}(U)$. By the result of Busca and Sirakov [20], $\tilde{U}_0(x)$ is radially symmetric after a suitable shift, that is, $\tilde{U}_0(x - \ell_0) \in \mathcal{M}_r(V_1(P_0), V_2(P_0))$ for some $\ell_0 \in \mathbf{R}^N$.

On the other hand, we have by Proposition 4.2.5

$$J_{(V_1(P_0),V_2(P_0))}(U_0) \ge b(V_1(P_0),V_2(P_0)) = m(P_0).$$

Thus, by (4.3.28), $m(P_0) \leq c_0 \leq m_0$, which implies $c_0 = m_0$, $P_0 \in K$ and $\tilde{U}_0(x-\ell_0) \in \mathcal{S}_{r,P_0}$ for some $\ell_0 \in \mathbf{R}^N$.

Step 4: Conclusion

Setting $Q_j = P_j + \varepsilon_j \ell_0$, we have the conclusion of Proposition 4.3.7 with $Q_0 = P_0$, $\Omega_0(x) = \tilde{U}_0(x - \ell_0)$.

Proof of Proposition 4.3.6. As in Step 1 of the proof of Proposition 4.3.7, we can show $||I'_{\varepsilon}(U_j)||_{H^{-1}} \to 0$. We assume that $U_j \to U_0$ weakly in $H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$. As in Step 2, we can show that $U_j \to U_0$ strongly. A positivity of $U_0(x)$ follows from Lemma 4.3.2. \Box

As a corollary to Proposition 4.3.7, we have the following corollary, which will play an important role in the proof of our main Theorem 4.1.3.

Corollary 4.3.8. For any $\delta \in (0, \delta_3/4]$, there exist ν_0 , h_0 , $\varepsilon_0 > 0$ such that

$$\|I_{\varepsilon}'(U)\|_{(T_U\mathcal{M}_{\varepsilon})^*} \ge \nu_0$$

for all $\varepsilon \in (0, \varepsilon_0]$ and $U \in (\mathcal{S}^{4\delta}_{\varepsilon} \setminus \mathcal{S}^{\delta}_{\varepsilon}) \cap \mathcal{M}_{\varepsilon}$ satisfying $I_{\varepsilon}(U) \leq m_0 + h_0$.

Proof. We argue indirectly and assume that there exists $\bar{\delta} \in (0, \delta_3/4]$ which satisfies the following property:

For any $j \in \mathbf{N}$, there exist $\varepsilon_j \in (0, \frac{1}{i}]$ and $U_j \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$ such that

(4.3.29) $U_{j} \in (\mathcal{S}_{\varepsilon_{j}}^{4\bar{\delta}} \setminus \mathcal{S}_{\varepsilon_{j}}^{\bar{\delta}}) \cap \mathcal{M}_{\varepsilon_{j}},$ $I_{\varepsilon_{j}}(U_{j}) \leq m_{0} + \frac{1}{j},$ $\|I_{\varepsilon_{j}}'(U_{j})\|_{(T_{U}\mathcal{M}_{\varepsilon_{j}})^{*}} < \frac{1}{j}.$ Applying Proposition 4.3.7, there exist a sequence $(Q_j)_{j=1}^{\infty} \subset \mathbf{R}^N$, $Q_0 \in K$ and $\Omega_0 \in \mathcal{S}_{r,Q_0}$ such that

$$Q_j \to Q_0, \quad U_j(x + Q_j/\varepsilon_j) \to \Omega_0(x) \quad \text{strongly in } H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N).$$

Thus dist $(U_j, \mathcal{S}_{\varepsilon_j}) \to 0$, which is in contradiction to (4.3.29). Therefore Corollary 4.3.8 holds.

4.4 An estimate of $I_{\varepsilon}(U)$ on $\mathcal{S}_{\varepsilon}^{\delta} \cap \mathcal{M}_{\varepsilon}$

The aim of this section is to show the following estimate which is a key of the proof of our main Theorem 4.1.3.

Proposition 4.4.1. For sufficiently small $\delta > 0$,

$$\lim_{\varepsilon \to 0} \inf_{U \in \mathcal{S}_{\varepsilon}^{\delta} \cap \mathcal{M}_{\varepsilon}} I_{\varepsilon}(U) = m_0.$$

Here we explain our idea to get the above estimate. It is rather easy to show the upper estimate $\lim_{\varepsilon \to 0} \inf_{U \in S_{\varepsilon}^{\delta} \cap \mathcal{M}_{\varepsilon}} \leq m_0$. The harder part is to show the lower estimate. First we show any $U(x) \in S_{\varepsilon}^{\delta}$ can be approximated uniformly by a function of a form:

First we show any $U(x) \in S^{\delta}_{\varepsilon}$ can be approximated uniformly by a function of a form: $\Xi(x - P/\varepsilon) + \Phi(x - P/\varepsilon)$, where $P \in K^{d_0}$ and

$$(4.4.1) \qquad \qquad \operatorname{supp} \Xi(x) \subset B(0, R)$$

$$(4.4.2) \|\Phi\|_{H^1} < 4\delta.$$

(4.4.3)
$$\operatorname{supp} \Xi(x) \cap \operatorname{supp} \Phi(x) = \emptyset.$$

By (4.4.3),

(4.4.4)
$$I_{\varepsilon}(U) \sim I_{\varepsilon}(\Xi(x - P/\varepsilon) + \Phi(x - P/\varepsilon)) = I_{\varepsilon}(\Xi(x - P/\varepsilon)) + I_{\varepsilon}(\Phi(x - P/\varepsilon)).$$

By (4.4.2), we have $I_{\varepsilon}(\Phi(x - P/\varepsilon)) \ge 0$ for $\delta > 0$ small. Thus

(4.4.5)
$$I_{\varepsilon}(U) \sim I_{\varepsilon}(\Xi(x - P/\varepsilon) + \Phi(x - P/\varepsilon)) \ge I_{\varepsilon}(\Xi(x - P/\varepsilon)).$$

By (4.4.1), we also have

(4.4.6)
$$I_{\varepsilon}(\Xi(x - P/\varepsilon)) \to J_{(V_1(P), V_2(P))}(\Xi) \text{ as } \varepsilon \to 0.$$

(4.4.4)–(4.4.6) are useful to estimate $I_{\varepsilon}(U)$ from below. We remark that to prove our Proposition 4.4.1 we need to deal with $I_{\varepsilon}(U)$ on $\mathcal{M}_{\varepsilon}$ and develop more precise arguments which involve the projection $\mathcal{P}_{\varepsilon}: S_{\varepsilon}^{\delta} \to \mathcal{M}_{\varepsilon}$.

For a proof of Proposition 4.4.1, we choose $\delta > 0$ and $\varepsilon > 0$ small so that

$$8\delta < \delta_3, \quad 8L_0\delta + \rho_3(\varepsilon) < \delta_3$$

Under these conditions we have

$$\begin{aligned} \mathcal{P}_{\varepsilon} : \, \mathcal{S}_{\varepsilon}^{8\delta} &\to \mathcal{M}_{\varepsilon} \quad \text{is well defined for small } \varepsilon, \\ \mathcal{P}_{\varepsilon}(\mathcal{S}_{\varepsilon}^{8\delta}) \subset \mathcal{S}_{\varepsilon}^{8L_0\delta + \rho_3(\varepsilon)} \subset \mathcal{S}_{\varepsilon}^{\delta_3}. \end{aligned}$$

We also assume that $\delta > 0$ satisfies

(4.4.7)
$$a_1^2 - 4a_2^2 C_4^4 C_6^2 C_7 (4\delta)^2 > 0,$$

where C_4 , $C_6 > 0$ are constants appeared in (4.3.15), (4.3.17) and $C_7 > 0$ is a constant such that

(4.4.8)
$$\left\| \begin{bmatrix} \mu_1 \| \Phi_1 \|_4^4 & \beta \| \Phi_1 \Phi_2 \|_2^2 \\ \beta \| \Phi_1 \Phi_2 \|_2^2 & \mu_2 \| \Phi_2 \|_4^4 \end{bmatrix} \right\|_{M(2,2)} \le C_7 \| \Phi \|_{H^1}^4$$

for all $\Phi = (\Phi_1, \Phi_2) \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N).$

Proof of Proposition 4.4.1. Proof of Proposition 4.4.1 consists of several steps. First we show that any $U \in S_{\varepsilon}^{\delta}$ can be approximated by a function of form: $\Xi(x-P/\varepsilon)+\Phi(x-P/\varepsilon)$.

Step 1: For any $\nu > 0$ there exists $\overline{R}_{\delta,\nu} > 0$ independent of ε and $U \in S_{\varepsilon}^{\delta}$ such that the following property holds: for any $U(x) \in S_{\varepsilon}^{\delta}$ there exist $P \in K^{d_0}$ and $\Xi(x)$, $\Phi(x) \in H^1(\mathbf{R}^N) \times H^1(\mathbf{R}^N)$ such that

(4.4.9)
$$\Xi(x - P/\varepsilon) + \Phi(x - P/\varepsilon) \in \mathcal{S}_{\varepsilon}^{8\delta},$$

$$(4.4.10) \qquad \qquad \Xi(x) \in \mathcal{S}_{r,P}^{4\delta}, \ \|\Phi\|_{H^1} < 4\delta, \\ (4.4.11) \qquad \qquad \|U(x) - (\Xi(x - R/c) + \Phi(x - R/c))\|_{H^1}$$

(4.4.11)
$$\|U(x) - (\Xi(x - P/\varepsilon) + \Phi(x - P/\varepsilon))\|_{H^1} < \nu_{\pm}$$

- (4.4.12) $\operatorname{supp} \Xi \subset B(0, \bar{R}_{\delta, \nu}),$
- (4.4.13) $\operatorname{supp} \Phi(x) \cap \operatorname{supp} \Xi(x) = \emptyset.$

Moreover, there exists $C_8 > 0$ such that for $U(x) \in \mathcal{S}^{\delta}_{\varepsilon} \cap \mathcal{M}_{\varepsilon}$

(4.4.14)
$$I_{\varepsilon}(U) \ge I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi(x - P/\varepsilon) + \Phi(x - P/\varepsilon))) - C_{8}\nu.$$

We choose $\underline{R}_{\delta} > 0$ such that

(4.4.15)
$$\|\Omega\|_{H^1(|x|\geq \underline{R}_{\delta})} < \delta \quad \text{for all } \Omega \in \mathcal{S}_{r,A}$$

Let $n_{\nu} \in \mathbf{N}$ be an integer such that

(4.4.16)
$$\left(\frac{\nu}{2}\right)^2 n_{\nu} \ge 4\delta^2$$

and we set

$$\bar{R}_{\delta,\nu} := \underline{R}_{\delta} + 3n_{\nu}.$$

We show that $\bar{R}_{\delta,\nu}$ has the desired property.

In fact, let $U(x) = \Omega_P(x - P/\varepsilon) + \Psi(x - P/\varepsilon) \in \mathcal{S}^{\delta}_{\varepsilon}$, where $P \in K^{d_0}$, $\Omega_P \in \mathcal{S}_{r,P}$ and $\|\Psi\|_{H^1} < \delta$. We set $\tilde{U}(x) := U(x + P/\varepsilon) = \Omega_P(x) + \Psi(x)$. By (4.4.15), we have $\|\tilde{U}(x)\|_{H^1(|x|\geq R_{\delta})} = \|\Omega_P(x) + \Psi(x)\|_{H^1(|x|\geq R_{\delta})} < 2\delta$. We remark that

$$\sum_{j=0}^{n_{\nu}-1} \|\tilde{U}\|_{H^{1}(|x|\in[\underline{R}_{\delta}+3j,\underline{R}_{\delta}+3(j+1)])}^{2} \leq \|\tilde{U}\|_{H^{1}(|x|\geq\underline{R}_{\delta})}^{2} < 4\delta^{2}.$$

By (4.4.16), there exists $n \in \{0, 1, 2, \dots, n_{\nu} - 1\}$ such that

$$\|\tilde{U}\|_{H^1(|x|\in[\underline{R}_{\delta}+3n,\underline{R}_{\delta}+3(n+1)])} \leq \frac{\nu}{2}.$$

We choose two functions $\zeta_1(r)$, $\zeta_2(r)$ such that

$$\begin{aligned} \zeta_1(r) &= \begin{cases} 1 & \text{for } r \in [0, \underline{R}_{\delta} + 3n], \\ 0 & \text{for } r \in [\underline{R}_{\delta} + 3n + \frac{4}{3}, \infty), \end{cases} \\ \zeta_2(r) &= \begin{cases} 0 & \text{for } r \in [0, \underline{R}_{\delta} + 3n + \frac{5}{3}], \\ 1 & \text{for } r \in [\underline{R}_{\delta} + 3n + 3, \infty), \end{cases} \\ \zeta_1(r), \, \zeta_2(r) \in [0, 1], \, \zeta_1'(r) \in [-1, 0], \, \zeta_2'(r) \in [0, 1] \text{ for all } r \in \mathbf{R} \end{cases} \end{aligned}$$

and we set

$$\Xi(x) = \zeta_1(|x|)\tilde{U}(x), \ \Phi(x) = \zeta_2(|x|)\tilde{U}(x).$$

(4.4.12)–(4.4.13) clearly hold. We note that $\|\zeta_1\varphi\|_{H^1} \leq 2\|\varphi\|_{H^1(|x|\in[0,\underline{R}_{\delta}+3n+\frac{4}{3}])}$ for all $\varphi \in H^1$ and similar inequalities hold for $\|(1-\zeta_1)\varphi\|_{H^1}$, $\|\zeta_2\varphi\|_{H^1}$, $\|(1-\zeta_2)\varphi\|_{H^1}$.

Thus we have

$$\begin{split} \|\Xi - \Omega_P\|_{H^1} &= \|\zeta_1(\Omega_P + \Psi) - \Omega_P\|_{H^1} \le \|(\zeta_1 - 1)\Omega_P\|_{H^1} + \|\zeta_1\Psi\|_{H^1} \\ &\le 2\|\Omega_P\|_{H^1(|x|\ge \underline{R}_{\delta}+3n)} + 2\|\Psi\|_{H^1} < 4\delta, \\ \|\Phi\|_{H^1} &= \|\zeta_2(\Omega_P + \Psi)\|_{H^1} \le \|\zeta_2\Omega_P\|_{H^1} + \|\zeta_2\Psi\|_{H^1} < 4\delta, \\ \|\tilde{U} - (\Xi + \Phi)\|_{H^1} &= \|(1 - \zeta_1 - \zeta_2)\tilde{U}\|_{H^1} \\ &\le 2\|\tilde{U}\|_{H^1(|x|\in [\underline{R}_{\delta}+3n, \underline{R}_{\delta}+3(n+1)])} < \nu. \end{split}$$

Thus (4.4.9) - (4.4.11) hold.

Next we suppose $U \in \mathcal{S}_{\varepsilon}^{\delta} \cap \mathcal{M}_{\varepsilon}$. Then $\mathcal{P}_{\varepsilon}(U) = U$ and by (4.3.12)

$$\|U - \mathcal{P}_{\varepsilon}(\Xi + \Phi)\|_{H^1} = \|\mathcal{P}_{\varepsilon}(U) - \mathcal{P}_{\varepsilon}(\Xi + \Phi)\|_{H^1} \le L_0 \nu.$$

Thus we have

$$I_{\varepsilon}(U) \ge I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi + \Phi)) - CL_0\nu$$

and (4.4.14) holds.

Next we show a property corresponding (4.4.5).

Step 2: For any given $U \in \mathcal{S}^{\delta}_{\varepsilon} \cap \mathcal{M}_{\varepsilon}$, let $\Xi(x) \in \mathcal{S}^{4\delta}_{r,P}$ and $\Phi \in H^{1}(\mathbb{R}^{N}) \times H^{1}(\mathbb{R}^{N})$ $(\|\Phi\|_{H^{1}} < 4\delta)$ be given in Step 1. Then

$$I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi(x-P/\varepsilon))) \leq I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi(x-P/\varepsilon) + \Phi(x-P/\varepsilon))).$$

It suffices to show that

$$\theta \mapsto I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi(x - P/\varepsilon) + \theta \Phi(x - P/\varepsilon))) : [0, 1] \to \mathbf{R}$$

is a nondecreasing function. We write $\Xi = (\Xi_1, \Xi_2), \Phi = (\Phi_1, \Phi_2)$ and

$$\mathcal{P}_{\varepsilon}((\Xi + \theta \Phi)(x - P/\varepsilon)) = (\sqrt{\lambda_1(\theta)}(\Xi_1 + \theta \Phi_1)(x - P/\varepsilon), \sqrt{\lambda_2(\theta)}(\Xi_2 + \theta \Phi_2)(x - P/\varepsilon)),$$

where $(\lambda_1(\theta), \lambda_2(\theta))$ satisfies

(4.4.17)
$$A(\Xi + \theta \Phi) \begin{bmatrix} \lambda_1(\theta) \\ \lambda_2(\theta) \end{bmatrix} = \begin{bmatrix} G_{\varepsilon_1}(\theta) \\ G_{\varepsilon_2}(\theta) \end{bmatrix}.$$

Here for i = 1, 2

$$G_{\varepsilon i}(\theta) := \|\Xi_i + \theta \Phi_i\|_{H^{1}, i, \varepsilon}^2 = \|\Xi_i\|_{H^{1}, i, \varepsilon}^2 + \theta^2 \|\Phi_i\|_{H^{1}, i, \varepsilon}^2.$$

It follows from (4.4.17) that

$$\begin{bmatrix} \lambda_1(\theta) \\ \lambda_2(\theta) \end{bmatrix} = A(\Xi + \theta \Phi)^{-1} \begin{bmatrix} G_{\varepsilon_1}(\theta) \\ G_{\varepsilon_2}(\theta) \end{bmatrix}.$$

We also have

$$\begin{bmatrix} \lambda_1'(\theta) \\ \lambda_2'(\theta) \end{bmatrix} = A(\Xi + \theta\Phi)^{-1} \begin{bmatrix} G_{\varepsilon 1}'(\theta) \\ G_{\varepsilon 2}'(\theta) \end{bmatrix}$$
$$-A(\Xi + \theta\Phi)^{-1} \left(\frac{d}{d\theta} A(\Xi + \theta\Phi) \right) A(\Xi + \theta\Phi)^{-1} \begin{bmatrix} G_{\varepsilon 1}(\theta) \\ G_{\varepsilon 2}(\theta) \end{bmatrix}$$

.

Since $I'_{\varepsilon}(\mathcal{P}_{\varepsilon}((\Xi + \theta \Phi)(x - P/\varepsilon)))(\mathcal{P}_{\varepsilon}((\Xi + \theta \Phi)(x - P/\varepsilon))) = 0$, we have

$$I_{\varepsilon}(\mathcal{P}_{\varepsilon}((\Xi + \theta\Phi)(x - P/\varepsilon))) = \frac{1}{4} \|\mathcal{P}_{\varepsilon}((\Xi + \theta\Phi)(x - P/\varepsilon))\|_{H^{1},\varepsilon}^{2}$$
$$= \frac{1}{4} (\lambda_{1}(\theta)G_{\varepsilon 1}(\theta) + \lambda_{2}(\theta)G_{\varepsilon 2}(\theta)) = \frac{1}{4} \left(\begin{bmatrix} G_{\varepsilon 1}(\theta) \\ G_{\varepsilon 2}(\theta) \end{bmatrix}, \begin{bmatrix} \lambda_{1}(\theta) \\ \lambda_{2}(\theta) \end{bmatrix} \right).$$

Thus

$$\begin{aligned} 4\frac{d}{d\theta}I_{\varepsilon}(\mathcal{P}_{\varepsilon}((\Xi+\theta\Phi)(x-P/\varepsilon))) \\ &= \left(\begin{bmatrix} G_{\varepsilon1}(\theta) \\ G_{\varepsilon2}(\theta) \end{bmatrix}, \begin{bmatrix} \lambda_{1}'(\theta) \\ \lambda_{2}'(\theta) \end{bmatrix} \right) + \left(\begin{bmatrix} G_{\varepsilon_{1}}'(\theta) \\ G_{\varepsilon_{2}}'(\theta) \end{bmatrix}, \begin{bmatrix} \lambda_{1}(\theta) \\ \lambda_{2}(\theta) \end{bmatrix} \right) \\ &= \left(\begin{bmatrix} G_{\varepsilon1}(\theta) \\ G_{\varepsilon2}(\theta) \end{bmatrix}, A(\Xi+\theta\Phi)^{-1} \begin{bmatrix} G_{\varepsilon1}'(\theta) \\ G_{\varepsilon2}'(\theta) \end{bmatrix} \\ &- A(\Xi+\theta\Phi)^{-1} \left(\frac{d}{d\theta}A(\Xi+\theta\Phi) \right) A(\Xi+\theta\Phi)^{-1} \begin{bmatrix} G_{\varepsilon1}(\theta) \\ G_{\varepsilon2}(\theta) \end{bmatrix} \right) \\ &+ \left(\begin{bmatrix} G_{\varepsilon1}'(\theta) \\ G_{\varepsilon2}'(\theta) \end{bmatrix}, A(\Xi+\theta\Phi)^{-1} \begin{bmatrix} G_{\varepsilon1}(\theta) \\ G_{\varepsilon2}(\theta) \end{bmatrix} \right) \\ &= 2 \left(\begin{bmatrix} G_{\varepsilon1}'(\theta) \\ G_{\varepsilon2}'(\theta) \end{bmatrix}, A(\Xi+\theta\Phi)^{-1} \begin{bmatrix} G_{\varepsilon1}(\theta) \\ G_{\varepsilon2}(\theta) \end{bmatrix} \right) \\ &- \left(\begin{bmatrix} G_{\varepsilon1}(\theta) \\ G_{\varepsilon2}(\theta) \end{bmatrix}, A(\Xi+\theta\Phi)^{-1} \left(\frac{d}{d\theta}A(\Xi+\theta\Phi) \right) A(\Xi+\theta\Phi)^{-1} \begin{bmatrix} G_{\varepsilon1}(\theta) \\ G_{\varepsilon2}(\theta) \end{bmatrix} \right). \end{aligned}$$

By (4.4.13), we remark that

$$A(\Xi + \theta \Phi) = \begin{bmatrix} \mu_1(\|\Xi_1\|_4^4 + \theta^4\|\Phi_1\|_4^4) & \beta(\|\Xi_1\Xi_2\|_2^2 + \theta^4\|\Phi_1\Phi_2\|_2^2) \\ \beta(\|\Xi_1\Xi_2\|_2^2 + \theta^4\|\Phi_1\Phi_2\|_2^2) & \mu_2(\|\Xi_2\|_4^4 + \theta^4\|\Phi_2\|_4^4) \end{bmatrix}.$$

Thus

$$\frac{d}{d\theta}A(\Xi+\theta\Phi) = 4\theta^3 \begin{bmatrix} \mu_1 \|\Phi_1\|_4^4 & \beta \|\Phi_1\Phi_2\|_2^2 \\ \beta \|\Phi_1\Phi_2\|_2^2 & \mu_2 \|\Phi_2\|_4^4 \end{bmatrix}.$$

By (4.4.8), we have

(4.4.19)
$$\left\| \frac{d}{d\theta} A(\Xi + \theta \Phi) \right\|_{M(2,2)} \le 4C_7 \theta^3 \|\Phi\|_{H^1}^4.$$

Since $\Xi + \theta \Phi \in \mathcal{S}^{8\delta}_{\varepsilon}$, we have by (4.3.18) that $\lambda_1(\theta), \lambda_2(\theta) \in [\frac{1}{2}, \frac{3}{2}]$ for all $\theta \in [0, 1]$. Thus

(4.4.20)
$$2\left(\begin{bmatrix} G'_{\varepsilon_1}(\theta)\\G'_{\varepsilon_2}(\theta)\end{bmatrix}, A(\Xi+\theta\Phi)^{-1}\begin{bmatrix} G_{\varepsilon_1}(\theta)\\G_{\varepsilon_2}(\theta)\end{bmatrix}\right)$$
$$=2\left(\lambda_1(\theta)G'_{\varepsilon_1}(\theta)+\lambda_2(\theta)G'_{\varepsilon_2}(\theta)\right)\geq G'_{\varepsilon_1}(\theta)+G'_{\varepsilon_2}(\theta)$$
$$\geq 2\theta\left(\|\Phi_1\|^2_{H^1,1,\varepsilon}+\|\Phi_2\|^2_{H^1,2,\varepsilon}\right)=2\theta\|\Phi\|^2_{H^1,\varepsilon}\geq 2a_1^2\theta\|\Phi\|^2_{H^1}.$$

By (4.3.15) we remark that

$$\left| \begin{bmatrix} G_{\varepsilon 1}(\theta) \\ G_{\varepsilon 2}(\theta) \end{bmatrix} \right| \le \sqrt{2} \|\Xi + \theta\Phi\|_{H^{1},\varepsilon}^{2} \le \sqrt{2}a_{2}^{2}\|\Xi + \theta\Phi\|_{H^{1}}^{2} \le \sqrt{2}a_{2}^{2}C_{4}^{2}.$$

Thus, by (4.4.18)-(4.4.20) and (4.3.16)-(4.3.17), we have

$$4\frac{d}{d\theta}I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi+\theta\Phi)) \geq 2a_{1}^{2}\theta\|\Phi\|_{H^{1}}^{2} - (\sqrt{2}a_{2}^{2}C_{4}^{2})^{2}C_{6}^{2} \cdot 4C_{7}\theta^{3}\|\Phi\|_{H^{1}}^{4}.$$

$$= 2\theta\|\Phi\|_{H^{1}}^{2} \left(a_{1}^{2} - 4a_{2}^{2}C_{4}^{4}C_{6}^{2}C_{7}\theta^{2}\|\Phi\|_{H^{1}}^{2}\right).$$

By (4.4.7),

$$\frac{d}{d\theta}I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi+\theta\Phi)) \ge 0 \quad \text{for } \|\Phi\|_{H^{1}} \le 4\delta,$$

which implies the conclusion of Step 2.

The following step shows a property corresponding to (4.4.6). **Step 3:** For any $\nu > 0$ there exists an $\varepsilon_4 > 0$ such that

$$\left| I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi(x - P/\varepsilon))) - J_{(V_1(P), V_2(P))}(\mathcal{P}_P(\Xi)) \right| < \nu$$

for any $\varepsilon \in (0, \varepsilon_4]$ and $\Xi(x) \in \mathcal{S}_{r,P}^{4\delta}$ with $\operatorname{supp} \Xi \subset B(0, \overline{R}_{\delta,\nu})$.

We have

$$\left| \int_{\mathbf{R}^N} V_i(\varepsilon x + P) \Xi_i(x)^2 \, dx - V_i(P) \|\Xi_i\|_2^2 \right| \le \max_{|x| \le \bar{R}_{\delta,\nu}} |V_i(\varepsilon x + P) - V_i(P)| \, \|\Xi_i\|_2^2$$

 $\to 0 \quad \text{as } \varepsilon \to 0 \text{ uniformly in } P \text{ and } \Xi \in \mathcal{S}_{r,P}^{4\delta} \text{ with } \operatorname{supp} \Xi \subset B(0, \bar{R}_{\delta,\nu}).$

Thus,

$$\|\Xi_i(x - P/\varepsilon)\|_{H^1, i, \varepsilon}^2 \to \|\nabla \Xi_i\|_2^2 + V_i(P)\|\Xi_i\|_2^2$$

uniformly. From the definition of $\mathcal{P}_{\varepsilon}$, \mathcal{P}_{P} , the conclusion of Step 3 holds.

Step 4: Conclusion

By Step 3, we have for $\varepsilon \in (0, \varepsilon_4]$

$$I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi(x-P/\varepsilon))) \ge J_{(V_1(P,V_2(P))}(\mathcal{P}_P(\Xi)) - \nu \ge m(P) - \nu.$$

Thus by Steps 1–2, we have for any $U(x) \in \mathcal{S}^{\delta}_{\varepsilon}$

$$I_{\varepsilon}(U) \ge I_{\varepsilon}(\mathcal{P}_{\varepsilon}((\Xi + \Phi)(x - P/\varepsilon))) - C_{8}\nu \ge I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Xi(x - P/\varepsilon)) - C_{8}\nu)$$
$$\ge m(P) - (C_{8} + 1)\nu \ge \inf_{P \in K^{d_{0}}} m(P) - (C_{8} + 1)\nu$$
$$= m_{0} - (C_{8} + 1)\nu.$$

Since $U \in \mathcal{S}^{\delta}_{\varepsilon} \cap \mathcal{M}_{\varepsilon}$ and $\nu > 0$ are arbitrary, we have

(4.4.21)
$$\liminf_{\varepsilon \to 0} \inf_{U \in \mathcal{S}^{\delta}_{\varepsilon} \cap \mathcal{M}_{\varepsilon}} I_{\varepsilon}(U) \ge m_0.$$

On the other hand, we have for $P \in K$ and $\Omega_P \in \mathcal{S}_{r,P}$

- (i) $\mathcal{P}_{\varepsilon}(\Omega_P(x-P/\varepsilon)) \in \mathcal{S}_{\varepsilon}^{\delta} \cap \mathcal{M}_{\varepsilon}$ for small ε .
- (ii) $I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Omega_P(x-P/\varepsilon))) \to J_{(V_1(P),V_2(P))}(\Omega_P) = m(P) = m_0.$

Thus we have

$$\limsup_{\varepsilon \to 0} \inf_{U \in \mathcal{S}_{\varepsilon}^{\delta} \cap \mathcal{M}_{\varepsilon}} I_{\varepsilon}(U) \leq \lim_{\varepsilon \to 0} I_{\varepsilon}(\mathcal{P}_{\varepsilon}(\Omega_{P}(x - P/\varepsilon))) = m_{0}$$

Together with (4.4.21), we complete the proof of Proposition 4.4.1.

Corollary 4.3.8 and Proposition 4.4.1 imply the following:

Proposition 4.4.2. Choose $\delta > 0$ small so that Corollary 4.3.8 and Proposition 4.4.1 hold. Then there exist $\nu_1 > 0$ and $\varepsilon_5 > 0$ such that

$$(4.4.22) I_{\varepsilon}(U) \ge m_0 + \nu_1$$

for all $\varepsilon \in (0, \varepsilon_5]$ and $U \in (\mathcal{S}^{3\delta}_{\varepsilon} \setminus \mathcal{S}^{2\delta}_{\varepsilon}) \cap \mathcal{M}_{\varepsilon}$.

Proof. By Corollary 4.3.8, there exist ν_0 and $h_0 > 0$ such that for small $\varepsilon > 0$

$$(4.4.23) \quad \|I_{\varepsilon}'(U)\|_{(T_{U}\mathcal{M}_{\varepsilon})^{*}} \geq \nu_{0} \quad \text{for any } U \in (\mathcal{S}_{\varepsilon}^{4\delta} \setminus \mathcal{S}_{\varepsilon}^{\delta}) \cap \mathcal{M}_{\varepsilon} \text{ with } I_{\varepsilon}(U) \leq m_{0} + h_{0}.$$

By Proposition 4.4.1, we have $\lim_{\varepsilon \to 0} \inf_{U \in S_{\varepsilon}^{4\delta} \cap \mathcal{M}_{\varepsilon}}(U) = m_0$. Thus for sufficiently small $\varepsilon > 0$, it holds that

(4.4.24)
$$\inf_{U \in \mathcal{S}_{\varepsilon}^{4\delta} \cap \mathcal{M}_{\varepsilon}} I_{\varepsilon}(U) \ge m_0 - \frac{1}{4}\nu_0 \delta.$$

We will show that (4.4.22) holds with $\nu_1 = \min\{h_0, \frac{1}{4}\nu_0\delta\}$.

By (4.4.23), we can construct a pseudo-gradient vector field $\mathcal{V}(U)$ on $(\mathcal{S}^{4\delta}_{\varepsilon} \setminus \mathcal{S}^{\delta}_{\varepsilon}) \cap \mathcal{M}_{\varepsilon} \cap$ $\{U: I_{\varepsilon}(U) \leq m_0 + h_0\}$ such that for all $U \in (\mathcal{S}_{\varepsilon}^{4\delta} \setminus \mathcal{S}_{\varepsilon}^{\delta}) \cap \mathcal{M}_{\varepsilon}$ with $I_{\varepsilon}(U) \leq m_0 + h_0$, it follows that

$$\mathcal{V}(U) \in T_U \mathcal{M}_{\varepsilon}, \\ \|\mathcal{V}(U)\|_{H^1 \times H^1} \le 1, \\ I'_{\varepsilon}(U) \mathcal{V}(U) \ge \frac{1}{2} \nu_0.$$

We consider the following ODE in $\mathcal{M}_{\varepsilon}$:

(4.4.25)
$$\frac{d\eta}{dt} = -\mathcal{V}(\eta), \quad \eta(0) = U_0.$$

We can easily see that

$$\begin{split} \left\| \frac{d\eta}{dt}(t) \right\|_{H^1 \times H^1} &\leq 1, \\ \frac{d}{dt} I_{\varepsilon}(\eta(t)) &= -I_{\varepsilon}'(\eta) \mathcal{V}(\eta) \leq -\frac{1}{2} \nu_0. \end{split}$$

as long as $\eta(t) \in (\mathcal{S}_{\varepsilon}^{4\delta} \setminus \mathcal{S}_{\varepsilon}^{\delta}) \cap \mathcal{M}_{\varepsilon}$ and $I_{\varepsilon}(\eta(t)) \leq m_0 + h_0$. Now let $U_0 \in (\mathcal{S}_{\varepsilon}^{3\delta} \setminus \mathcal{S}_{\varepsilon}^{2\delta}) \cap \mathcal{M}_{\varepsilon}$ and we show (4.4.22) holds for U_0 . Suppose that $I_{\varepsilon}(U_0) \leq m_0 + h_0$ otherwise (4.4.22) holds. We consider the solution $\eta(t)$ of (4.4.25). By the above argument we have

(4.4.26)
$$\eta(t) \in (\mathcal{S}^{4\delta}_{\varepsilon} \setminus \mathcal{S}^{\delta}_{\varepsilon}) \cap \mathcal{M}_{\varepsilon} \quad \text{for } t \in [0, \delta],$$

(4.4.27)
$$I_{\varepsilon}(\eta(\delta)) \le I_{\varepsilon}(U_0) - \frac{1}{2}\nu_0\delta.$$

It follows from (4.4.26)-(4.4.27) that

$$I_{\varepsilon}(U_0) \ge I_{\varepsilon}(\eta(\delta)) + \frac{1}{2}\nu_0 \delta \ge \inf_{U \in \mathcal{S}_{\varepsilon}^{4\delta} \cap \mathcal{M}_{\varepsilon}} I_{\varepsilon}(U) + \frac{1}{2}\nu_0 \delta.$$

By (4.4.24), we have

$$I_{\varepsilon}(U_0) \ge m_0 + \frac{1}{4}\nu_0\delta.$$

Since $U_0 \in (\mathcal{S}_{\varepsilon}^{3\delta} \setminus \mathcal{S}_{\varepsilon}^{2\delta}) \cap \mathcal{M}_{\varepsilon}$ is arbitrary, we have (4.4.22).

End of the proof of Theorem 4.1.3. We fix $\delta > 0$ small so that Propositions 4.4.1 and 4.4.2 hold and let $\varepsilon_5 > 0$ be given in Proposition 4.4.2. Then we have for $\varepsilon \in (0, \varepsilon_5]$

$$\inf_{U \in (\mathcal{S}^{3\delta}_{\varepsilon} \setminus \mathcal{S}^{2\delta}_{\varepsilon}) \cap \mathcal{M}_{\varepsilon}} I_{\varepsilon}(U) \ge m_0 + \nu_1$$

On the other hand, by Proposition 4.4.1, it follows that

(4.4.28)
$$\lim_{\varepsilon \to 0} \inf_{U \in \mathcal{S}_{\varepsilon}^{3\delta} \cap \mathcal{M}_{\varepsilon}} I_{\varepsilon}(U) = m_0.$$

Since $I_{\varepsilon}|_{\mathcal{M}_{\varepsilon}}$ satisfies the Palais-Smale condition in $\mathcal{S}_{\varepsilon}^{3\delta} \cap \mathcal{M}_{\varepsilon}$ by Proposition 4.3.6, there exists a minimizer $U_{\varepsilon}(x) \in \mathcal{S}_{\varepsilon}^{2\delta} \cap \mathcal{M}_{\varepsilon}$ via Ekeland principle. $U_{\varepsilon}(x)$ satisfies

$$I_{\varepsilon}(U_{\varepsilon}) = \inf_{U \in \mathcal{S}_{\varepsilon}^{3\delta} \cap \mathcal{M}_{\varepsilon}} I_{\varepsilon}(U),$$

$$I_{\varepsilon}'(U_{\varepsilon}) = 0.$$

By Lemma 4.3.2 (i), $U_{\varepsilon}(x)$ is a nontrivial positive solution of (4.3.1)–(4.3.4). Since $I_{\varepsilon}(U_{\varepsilon}) \to m_0$ by (4.4.28), Proposition 4.3.7 implies the existence of a subsequence $\varepsilon_j \to 0$ and $(Q_j)_{j=1}^{\infty} \subset \mathbf{R}^N$, $Q_0 \in K$ and $\Omega_0 \in S_{r,Q_0}$ that satisfies (4.3.23)–(4.3.24). Thus, denoting $U_{\varepsilon}(x) := (u_{1\varepsilon}(x), u_{2\varepsilon}(x))$ and setting $v_{1\varepsilon}(x) := u_{1\varepsilon}(x/\varepsilon), v_{2\varepsilon}(x) := u_{2\varepsilon}(x/\varepsilon)$ and $P_{\varepsilon_j} := Q_j$, $P_0 := Q_0$, we have (4.1.14)–(4.1.15). Thus the proof of Theorem 4.1.3 is completed. \Box

Part II Nonlinear scalar field equations

Chapter 5

Existence of positive and infinitely many solutions: homogeneous case

5.1 Introduction

In this chapter, we study (NSF) with $\Omega = \mathbf{R}^N$ and g(r, s) = g(s). Namely, we consider the existence of radially symmetric solutions of the following nonlinear scalar field equations:

(5.1.1)
$$-\Delta u = g(u) \quad \text{in } \mathbf{R}^N,$$

$$(5.1.2) u \in H^1(\mathbf{R}^N).$$

Here $N \ge 2$ and $g: \mathbf{R} \to \mathbf{R}$ is a continuous function. This type of problem appears in many models in mathematical physics etc. and almost necessary and sufficient conditions for the existence of nontrivial solutions are obtained by Berestycki and Lions [15, 16] for $N \ge 3$ and Berestycki, Gallouët and Kavian [14] for N = 2. See also Strauss [95] and Coleman, Glaser and Martin [28] for earlier works.

In [14, 15, 16], they assume:

(5–g0) The function $g \in C(\mathbf{R}, \mathbf{R})$ and g is odd: $g(-\xi) = -g(\xi)$. (5–g1) For $N \ge 3$,

$$\limsup_{\xi \to \infty} \frac{g(\xi)}{\xi^{(N+2)/(N-2)}} \le 0.$$

For N = 2,

$$\limsup_{\xi \to \infty} \frac{g(\xi)}{e^{\alpha \xi^2}} \le 0 \quad \text{for any } \alpha > 0.$$

(5–g2) For $N \ge 3$,

(5.1.3)
$$-\infty < \liminf_{\xi \to 0} \frac{g(\xi)}{\xi} \le \limsup_{\xi \to 0} \frac{g(\xi)}{\xi} < 0.$$

For N = 2,

(5.1.4)
$$-\infty < \lim_{\xi \to 0} \frac{g(\xi)}{\xi} < 0.$$

(5–g3) There exists a $\zeta_0 > 0$ such that $G(\zeta_0) > 0$, where $G(\xi) \equiv \int_0^{\xi} g(\tau) d\tau$.

Under the above conditions, they show the existence of a radially symmetric positive solution and infinitely many radially symmetric (possibly sign changing) solutions.

Remark 5.1.1. For the existence of a positive solution, it is sufficient to assume (5-g0)-(5-g3) just for $\xi > 0$. Namely we assume

(5-g0')
$$g \in C([0,\infty)), g(0) = 0$$

and (5–g1), (5–g3) and (5–g2) just for a limit as $\xi \to +0$.

Remark 5.1.2. (a) We refer to Berestycki and Lions [17] (see also Section 11, Chapter II of Struwe [98]) for the study of *zero mass case*, where $N \ge 3$. In particular, they assume

$$\limsup_{\xi \to 0} \frac{G(\xi)}{|\xi|^{2N/(N-2)}} \le 0$$

instead of (5–g2) and they show the existence of infinitely many solutions in $\mathcal{D}^{1,2}(\mathbf{R}^N)$. (b) For the study of the existence of at least one solution, especially the existence of a least energy solution, we also refer to Brezis and Lieb [18], in which they study the system of equations

(5.1.5)
$$-\Delta u_i = g^i(u) \quad \text{in } \mathbf{R}^d, \ i = 1, \dots, n,$$

where $d \ge 2$ with $u : \mathbf{R}^d \to \mathbf{R}^n$ and $g^i(u) = \partial G/\partial u_i$. Here we call $u = (u_1, \ldots, u_n)$ a *least* energy solution of (5.1.5) if u satisfies

(5.1.6)
$$J(u) = \inf\{J(v) : v \text{ is a nontrivial solution of (5.1.5)} \}$$

where

$$J(v) = \frac{1}{2} \sum_{i=1}^{n} \int_{\mathbf{R}^{d}} |\nabla v_{i}(x)|^{2} dx - \int_{\mathbf{R}^{d}} G(v(x)) dx.$$

Under suitable conditions on G (which differ between d = 2 and $d \ge 3$), they prove that (5.1.5) admits a nontrivial solution u which satisfies (5.1.6). We also refer to Brüning [19] for a generalization when d = 2.

(5-g0)-(5-g3) are natural conditions for the existence of solutions. However we can see a difference between cases $N \ge 3$ and N = 2 in the condition (5-g2). We remark that when N = 2, the existence of a limit $\lim_{\xi\to 0} g(\xi)/\xi \in (-\infty, 0)$ is used essentially to show that the Palais–Smale compactness condition for the corresponding functional under suitable constraint ([14]).

The aim of this chapter is to extend the result of [14] slightly and we prove the existence of radially symmetric positive solution and infinitely many radially symmetric solutions under the conditions (5-g0), (5-g1), (5-g3) and (5.1.3) (not (5.1.4)).

We also remark that in [14, 15, 16] (cf. [18, 19]), they constructed solutions of (5.1.1)–(5.1.2) through constraint problems in the space of radially symmetric functions:

• find critical points of

(5.1.7)
$$\left\{ \int_{\mathbf{R}^N} |\nabla u|^2 dx : \int_{\mathbf{R}^N} G(u) dx = 1 \right\} \quad (N \ge 3),$$

or

• find critical points of

(5.1.8)
$$\left\{ \int_{\mathbf{R}^2} |\nabla u|^2 dx : \int_{\mathbf{R}^2} G(u) dx = 0, \ \int_{\mathbf{R}^2} u^2 dx = 1 \right\} \quad (N=2)$$

In fact, if v(x) is a critical point of (5.1.7) or (5.1.8), then for a suitable $\lambda > 0$, $u(x) = v(x/\lambda)$ is a solution of (5.1.1)–(5.1.2). On the other hand, solutions of (5.1.1)–(5.1.2) are also characterized as critical points of the functional $I \in C^1(H_r^1(\mathbf{R}^N), \mathbf{R})$ defined by

$$I(u) := \frac{1}{2} \int_{\mathbf{R}^N} |\nabla u|^2 dx - \int_{\mathbf{R}^N} G(u) dx$$

Here we denote by $H_r^1(\mathbf{R}^N)$ the space of radially symmetric H^1 -functions defined on \mathbf{R}^N . It is natural to ask whether it is possible to find critical points through the unconstraint functional I.

Our second aim is to give another proof of the results of [14, 15, 16] using mountain pass and symmetric mountain pass arguments to I.

Before stating our main result in this chapter, we prepare one notation. We say that a nontrivial solution u of (5.1.1)-(5.1.2) is a *least energy solution* if and only if u satisfies

 $I(u) = \inf\{I(v) : v \in H^1(\mathbf{R}^N) \text{ is a nontrivial solution of } (5.1.1)\}.$

Now we can state our main result.

Theorem 5.1.3. Assume $N \ge 2$, (5–g0), (5–g1), (5–g3) and

(g2')
$$-\infty < \liminf_{\xi \to 0} \frac{g(\xi)}{\xi} \le \limsup_{\xi \to 0} \frac{g(\xi)}{\xi} < 0.$$

Then (5.1.1)–(5.1.2) has a least energy positive solution and infinitely many radially symmetric (possibly sign changing) solutions, which are characterized by the mountain pass and symmetric mountain pass minimax arguments in $H_r^1(\mathbf{R}^N)$ (see (5.3.1)–(5.3.2) and (5.6.1)–(5.6.3) below).

Remark 5.1.4. (a) When $N \ge 3$, the existence of solutions of (5.1.1)–(5.1.2) is obtained in [15, 16], and we provide another proof and give a minimax characterization of infinitely many solutions using the functional I.

(b) When N = 2, our existence result extends the result of [14] slightly. Indeed, we show the existence under the condition (5–g2') not (5.1.4).

In Jeanjean and Tanaka [55], they give a mountain pass characterization to a least energy solution of (5.1.1)-(5.1.2) under the conditions (5-g0)-(5-g3). More precisely, let bbe the mountain pass minimax value for I and m the least energy level. To show b = m, we argued in [55] as follows: To show $b \leq m$, for any solution u(x) we constructed a path $\gamma \in$ $C([0,1], H_r^1(\mathbf{R}^N))$ such that $u \in \gamma([0,1]), \gamma(0) = 0, I(\gamma(1)) < 0$ and $\max_{t \in [0,1]} I(\gamma(t)) =$ I(u). To show $b \geq m$, the existence of a minimizer of the minimization problems (5.1.7) or (5.1.8) is essential and we relied on the arguments in [14, 15]. We will take mountain pass and symmetric mountain pass approaches to prove Theorem 5.1.3. In section 5.3, we will observe that I is an even functional with a mountain pass geometry and it is possible to define a mountain pass minimax values b_{mp} and symmetric mountain pass values b_n $(n \in \mathbf{N})$ for I. By Ekeland's principle, we can find a Palais–Smale sequence $(u_j)_{j=1}^{\infty} \subset H_r^1(\mathbf{R}^N)$ at levels b_{mp} and b_n , that is, $(u_j)_{j=1}^{\infty}$ satisfies

(5.1.9)
$$I(u_j) \to b_{mp} \quad (\text{or } b_n),$$

(5.1.10)
$$I'(u_j) \to 0 \quad \text{strongly in } (H^1_r(\mathbf{R}^N))^*$$

However one of the difficulty is a lack of the Palais–Smale compactness condition and it seems difficult to show the existence of strongly convergent subsequence merely under the conditions (5.1.9)–(5.1.10). A key of our argument is to find a Palais–Smale sequence with an extra property related to the Pohozaev Identity. We recall that if u is a critical point of I, then u satisfies

$$P(u) = 0, \quad \text{where} \quad P(u) := \frac{N-2}{2} \int_{\mathbf{R}^N} |\nabla u|^2 dx - N \int_{\mathbf{R}^N} G(u) dx.$$

The above equality is called the Pohozaev Identity. It is natural to ask the existence of a Palais–Smale sequence $(u_j)_{j=1}^{\infty}$ satisfying (5.1.9)–(5.1.10) and $P(u_j) \rightarrow 0$. For this purpose, in section 5.4, we introduce an auxiliary functional:

$$\tilde{I}(\theta, u) := \frac{e^{(N-2)\theta}}{2} \int_{\mathbf{R}^N} |\nabla u|^2 dx - e^{N\theta} \int_{\mathbf{R}^N} G(u) dx : \mathbf{R} \times H^1_r(\mathbf{R}^N) \to \mathbf{R}$$

We will find a Palais–Smale sequence (θ_j, u_j) in the augmented space $\mathbf{R} \times H^1_r(\mathbf{R}^N)$ satisfying

 $(5.1.11) \qquad \qquad \theta_j \to 0,$

(5.1.12)
$$\tilde{I}(\theta_j, u_j) \to b_{mp} \quad (\text{or } b_n),$$

(5.1.13) $\tilde{I}'(\theta_j, u_j) \to 0$ strongly in $(H^1_r(\mathbf{R}^N))^*$,

(5.1.14)
$$\frac{N-2}{2}e^{(N-2)\theta_j}\int_{\mathbf{R}^N}|\nabla u_j|^2dx - Ne^{N\theta_j}\int_{\mathbf{R}^N}G(u_j)dx \to 0.$$

Remark 5.1.5. We remark that this type of auxiliary functionals was first used in Jeanjean [54] for a nonlinear eigenvalue problem. It should be compared with a monotonicity method due to Struwe [96, 97] and Jeanjean [54]. We expect that this type of auxiliary functionals can be applied to other problems.

We remark that our auxiliary functional $I(\theta, u)$ satisfies

$$\begin{split} \tilde{I}(0,u) &= I(u), \\ \tilde{I}(\theta,u) &= I(u(e^{-\theta}x)) \quad \text{for all } \theta \in \mathbf{R} \text{ and } u \in H^1_r(\mathbf{R}^N). \end{split}$$

Properties (5.1.11)–(5.1.14) enable us to obtain the boundedness and the existence of strongly convergent subsequence of (u_j) .
5.2 Preliminaries

We will deal with the cases N = 2 and $N \ge 3$ in a unified way. In what follows we assume $N \ge 2$ and g satisfies (5–g0), (5–g1), (5–g2') and (5–g3).

5.2.1 Modification of g

To give a proof of Theorem 5.1.3, we modify the nonlinearity g. First we remark that we can assume

(g1')
when
$$N \ge 3$$
, $\lim_{\xi \to \infty} \frac{g(\xi)}{\xi^{(N+2)/(N-2)}} = 0$,
when $N = 2$, $\lim_{\xi \to \infty} \frac{g(\xi)}{e^{\alpha \xi^2}} = 0$ for any $\alpha > 0$

In fact, if g satisfies $g(\xi) > 0$ for $\xi \ge \zeta_0$, (5–g1') clearly follows from (5–g1). If there exists a $\zeta_1 > \zeta_0$ such that $g(\zeta_1) = 0$, we set

$$\tilde{g}(\xi) := \begin{cases} g(\xi) & \text{for } 0 \le \xi \le \zeta_1, \\ 0 & \text{for } \xi > \zeta_1, \\ -g(-\xi) & \text{for } \xi < 0. \end{cases}$$

Then \tilde{g} satisfies (5–g0), (5–g1'), (5–g2'), (5–g3) and all solutions of $-\Delta u = \tilde{g}(u)$ in \mathbf{R}^N which belong to $H^1(\mathbf{R}^N)$ satisfy $-\zeta_1 \leq u(x) \leq \zeta_1$ for all $x \in \mathbf{R}^N$, that is, u also solves (5.1.1). Thus we may replace g by \tilde{g} and assume (5–g1').

In what follows, we assume that g satisfies (5–g0), (5–g1'), (5–g2') and (5–g3). Next we set

$$m_0 := -\frac{1}{2} \limsup_{\xi \to 0} \frac{g(\xi)}{\xi} \in (0, \infty)$$

and rewrite (5.1.1) as

$$-\Delta u + m_0 u = m_0 u + g(u) \quad \text{in } \mathbf{R}^N.$$

We introduce $h \in C(\mathbf{R}, \mathbf{R})$ by

$$h(\xi) := \begin{cases} \max\{m_0\xi + g(\xi), 0\} & \text{for } \xi \ge 0, \\ -h(-\xi) & \text{for } \xi < 0. \end{cases}$$

Furthermore, we choose $p_0 \in (1, (N+2)/(N-2))$ if $N \ge 3, p_0 \in (1, \infty)$ if N = 2 and set

$$\overline{h}(\xi) := \begin{cases} \xi^{p_0} \sup_{0 < \tau \le \xi} \frac{h(\tau)}{\tau^{p_0}} & \text{for } \xi > 0, \\ 0 & \text{for } \xi = 0, \\ -\overline{h}(-\xi) & \text{for } \xi < 0. \end{cases}$$

We also set

$$H(\xi) := \int_0^{\xi} h(\tau) d\tau, \quad \overline{H}(\xi) := \int_0^{\xi} \overline{h}(\tau) d\tau.$$

From the definition of h, h and m_0 , we have

Lemma 5.2.1. The following hold:

- (a) For all $\xi \ge 0$, $m_0\xi + g(\xi) \le h(\xi) \le \overline{h}(\xi)$.
- (b) For all $\xi \ge 0$, $h(\xi) \ge 0$ and $\overline{h}(\xi) \ge 0$.
- (c) There exists a $\delta_0 > 0$ such that $h(\xi) = \overline{h}(\xi) = 0$ for all $\xi \in [0, \delta_0]$.
- (d) There exists a $\xi_0 > 0$ such that $0 < h(\xi_0) \le \overline{h}(\xi_0)$.
- (e) The map $\xi \mapsto \overline{h}(\xi)/\xi^{p_0}$; $(0,\infty) \to \mathbf{R}$ is non-increasing.
- (f) The functions h, \overline{h} satisfy (5–g1').

Proof. (a), (b) and (e) follow from the definitions of h and \overline{h} .

(c) By the definition of m_0 , we can easily see that $\xi g(\xi) \leq -m_0 \xi^2$ in a neighborhood of $\xi = 0$. Thus (c) holds for small $\delta_0 > 0$.

(d) By (5–g3), there exists a $\xi_0 \in (0, \zeta_0)$ such that $g(\xi_0) > 0$. Thus $\overline{h}(\xi_0) \ge h(\xi_0) \ge m_0\xi_0 + g(\xi_0) > 0$ and (d) holds.

(f) It is easy to see that h satisfies (5–g1') and we will show (f) for \overline{h} . We consider the case $N \geq 3$ first. We remark that

$$\frac{\overline{h}(\xi)}{\xi^{(N+2)/(N-2)}} = \xi^{-((N+2)/(N-2)-p_0)} \sup_{0 < \tau \le \xi} \frac{h(\tau)}{\tau^{p_0}} = \sup_{0 < \tau \le \xi} \frac{h(\tau)}{\tau^{(N+2)/(N-2)}} \frac{\tau^{(N+2)/(N-2)-p_0}}{\xi^{(N+2)/(N-2)-p_0}}.$$

Since h satisfies (5–g1'), for any $\varepsilon > 0$ there exists a $\tau_{\varepsilon} > 0$ such that

$$\left|\frac{h(\tau)}{\tau^{(N+2)/(N-2)}}\right| < \varepsilon \quad \text{for all } \tau \ge \tau_{\varepsilon}.$$

Thus denoting $C_{\varepsilon} := \sup_{0 < \tau \leq \tau_{\varepsilon}} |h(\tau)/\tau^{(N+2)/(N-2)}|$, we have

$$\frac{\overline{h}(\xi)}{\xi^{(N+2)/(N-2)}} \leq \max\left\{ \sup_{0<\tau\leq\tau_{\varepsilon}} \left| \frac{h(\tau)}{\tau^{(N+2)/(N-2)}} \right| \frac{\tau_{\varepsilon}^{(N+2)/(N-2)-p_{0}}}{\xi^{(N+2)/(N-2)-p_{0}}}, \sup_{\tau_{\varepsilon}\leq\tau\leq\xi} \left| \frac{h(\tau)}{\tau^{(N+2)/(N-2)}} \right| \right\} \\
\leq \max\left\{ \frac{C_{\varepsilon}\tau_{\varepsilon}^{(N+2)/(N-2)-p_{0}}}{\xi^{(N+2)/(N-2)-p_{0}}}, \varepsilon \right\}.$$

Therefore we have

$$\limsup_{\xi \to \infty} \frac{h(\xi)}{\xi^{(N+2)/(N-2)}} \le \varepsilon.$$

Since $\varepsilon > 0$ is arbitrary, we have $\lim_{\xi \to \infty} \overline{h}(\xi) / \xi^{(N+2)/(N-2)} = 0$. Next we deal with the case N = 2. It suffices to show

(5.2.1)
$$\lim_{\xi \to \infty} \frac{h(\xi)}{\xi^{p_0} e^{\alpha \xi^2}} = 0 \quad \text{for any } \alpha > 0.$$

Since

$$\frac{\overline{h}(\xi)}{\xi^{p_0}e^{\alpha\xi^2}} = \frac{1}{e^{\alpha\xi^2}} \sup_{0 < \tau \le \xi} \frac{h(\tau)}{\tau^{p_0}} = \sup_{0 < \tau \le \xi} \frac{h(\tau)}{\tau^{p_0}e^{\alpha\tau^2}} \frac{e^{\alpha\tau^2}}{e^{\alpha\xi^2}}$$

and h satisfies $\lim_{\xi\to\infty} h(\xi)/\xi^{p_0}e^{\alpha\xi^2} = 0$, we can show (5.2.1) in a similar way.

Corollary 5.2.2. The following hold:

- (a) For all $\xi \in \mathbf{R}$, $m_0 \xi^2 / 2 + G(\xi) \le H(\xi) \le \overline{H}(\xi)$.
- (b) For all $\xi \in \mathbf{R}$, $H(\xi)$, $\overline{H}(\xi) \ge 0$.
- (c) There exists a $\delta_0 > 0$ such that $H(\xi) = \overline{H}(\xi) = 0$ for $|\xi| \le \delta_0$.
- (d) It holds that $\overline{H}(\zeta_0) m_0 \zeta_0^2/2 > 0.$
- (e) For all $\xi \in \mathbf{R}$, $0 \le (p_0 + 1)\overline{H}(\xi) \le \xi \overline{h}(\xi)$.
- (f) The functions H and \overline{H} satisfy

$$\lim_{|\xi| \to \infty} \frac{H(\xi)}{|\xi|^{2N/(N-2)}} = \lim_{|\xi| \to \infty} \frac{\overline{H}(\xi)}{|\xi|^{2N/(N-2)}} = 0 \quad \text{when } N \ge 3,$$
$$\lim_{|\xi| \to \infty} \frac{H(\xi)}{e^{\alpha\xi^2}} = \lim_{|\xi| \to \infty} \frac{\overline{H}(\xi)}{e^{\alpha\xi^2}} = 0 \quad \text{for any } \alpha > 0 \text{ when } N = 2.$$

Proof. (a)–(c) easily follow from (a)–(c) of Lemma 5.2.1.

By (a) and (5-g3), it follows that

$$\overline{H}(\zeta_0) \ge H(\zeta_0) \ge \frac{1}{2}m_0\zeta_0^2 + G(\zeta_0) > 0.$$

Thus (d) holds.

Since the map $\xi \mapsto \overline{h}(\xi)/\xi^{p_0}$; $(0,\infty) \to \mathbf{R}$ is nondecreasing, we have for $\xi > 0$

$$\begin{split} \xi \overline{h}(\xi) - (p_0 + 1) \overline{H}(\xi) &= \int_0^{\xi} \overline{h}(\xi) - (p_0 + 1) \overline{h}(\tau) d\tau = \int_0^{\xi} \xi^{p_0} \frac{\overline{h}(\xi)}{\xi^{p_0}} - (p_0 + 1) \tau^{p_0} \frac{\overline{h}(\tau)}{\tau^{p_0}} d\tau \\ &\geq \int_0^{\xi} \xi^{p_0} \frac{\overline{h}(\xi)}{\xi^{p_0}} - (p_0 + 1) \tau^{p_0} \frac{\overline{h}(\xi)}{\xi^{p_0}} d\tau = 0. \end{split}$$

Therefore (e) holds.

The statement (f) also follows from (f) of Lemma 5.2.1.

5.2.2 Fundamental properties of $H_r^1(\mathbf{R}^N)$

In what follows, we use the following notation: for $u \in H^1_r(\mathbf{R}^N)$ and $1 \le p < \infty$,

$$\|u\|_{p} := \left(\int_{\mathbf{R}^{N}} |u|^{p} dx\right)^{1/p}, \quad \|u\|_{\infty} := \underset{x \in \mathbf{R}^{N}}{\operatorname{ess sup}} |u(x)|, \\\|u\|_{H^{1}} := (\|\nabla u\|_{2}^{2} + m_{0}\|u\|_{2}^{2})^{1/2}.$$

We also write

$$(u,v)_2 := \int_{\mathbf{R}^N} uv dx, \quad (u,v)_{H^1} := \int_{\mathbf{R}^N} \nabla u \cdot \nabla v + m_0 uv dx.$$

We remark that $H_r^1(\mathbf{R}^N)$ equips the norm $\|\cdot\|_{H^1}$ and is a closed subspace of $H^1(\mathbf{R}^N)$. The following properties are well-known (see [1, 15]).

(i) For $N \ge 2$, there exists a $C_N > 0$ such that

(5.2.2) $|u(x)| \le C_N |x|^{-(N-1)/2} ||u||_{H^1}$ for $u \in H^1_r(\mathbf{R}^N)$ and $|x| \ge 1$.

(ii) The embedding $H_r^1(\mathbf{R}^N) \subset L^p(\mathbf{R}^N)$ is continuous for $2 \le p \le 2N/(N-2)$ if $N \ge 3$, 2 if <math>N = 2 and it is compact for $2 if <math>N \ge 3$, 2 if <math>N = 2.

(iii) Set $\Phi(s) := e^s - 1$. When N = 2, for any $\beta \in (0, 4\pi)$ there exists a $\tilde{C}_{\beta} < 0$ such that

(5.2.3)
$$\int_{\mathbf{R}^2} \Phi\left(\frac{\beta u^2}{\|\nabla u\|_2^2}\right) dx \le \tilde{C}_\beta \frac{\|u\|_2^2}{\|\nabla u\|_2^2} \quad \text{for all } u \in H^1(\mathbf{R}^2)/\{0\}.$$

(iv) In particular, for any M > 0

(5.2.4)
$$\int_{\mathbf{R}^2} \Phi\left(\frac{\beta u^2}{M^2}\right) dx \le \tilde{C}_\beta \frac{\|u\|_2^2}{M^2} \quad \text{for all } u \in H^1(\mathbf{R}^2) \text{ with } \|\nabla u\|_2 \le M.$$

In fact, if $\|\nabla u\|_2 \leq M$ holds, then we have

$$M^{2}\Phi\left(\frac{\beta u^{2}}{M^{2}}\right) = M^{2}\sum_{j=1}^{\infty}\frac{1}{j!}\left(\frac{\beta u^{2}}{M^{2}}\right)^{j} = \sum_{j=1}^{\infty}\frac{1}{j!}\frac{\beta^{j}u^{2j}}{M^{2j-2}} \leq \sum_{j=1}^{\infty}\frac{1}{j!}\frac{\beta^{j}u^{2j}}{\|\nabla u\|_{2}^{2j-2}}$$
$$= \|\nabla u\|_{2}^{2}\Phi\left(\frac{\beta u^{2}}{\|\nabla u\|_{2}^{2}}\right).$$

Thus (5.2.4) follows from (5.2.3) (see also Byeon, Jeanjean and Tanaka [24]).

Let $\delta_0 > 0$ be a number given in Lemma 5.2.1 (c) and Corollary 5.2.2 (c). By (5.2.2), for any M > 0 there exists an $R_M > 0$ such that

(5.2.5)
$$|u(x)| \le \delta_0 \quad \text{for all } |x| \ge R_M \text{ and } u \in H^1_r(\mathbf{R}^N) \text{ with } ||u||_{H^1} \le M.$$

In particular, it follows that from (5.2.5) that

(5.2.6)
$$h(u(x)), \ \overline{h}(u(x)), \ H(u(x)), \ \overline{H}(u(x)) = 0 \text{ for } |x| \ge R_M \text{ and } \|u\|_{H^1} \le M.$$

From (5.2.6) and the compactness of the embedding $H^1_r(\mathbf{R}^N) \to L^p(\mathbf{R}^N)$, we have

Lemma 5.2.3. Let $N \ge 2$ and suppose that $(u_j)_{j=1}^{\infty} \subset H_r^1(\mathbf{R}^N)$ converges to $u_0 \in H_r^1(\mathbf{R}^N)$ weakly in $H_r^1(\mathbf{R}^N)$. Then

(a)
$$\int_{\mathbf{R}^N} H(u_j) dx \to \int_{\mathbf{R}^N} H(u_0) dx \text{ and } \int_{\mathbf{R}^N} \overline{H}(u_j) dx \to \int_{\mathbf{R}^N} \overline{H}(u_0) dx.$$

(b) $h(u_j) \to h(u_0) \text{ and } \overline{h}(u_j) \to \overline{h}(u_0) \text{ strongly in } (H^1_r(\mathbf{R}^N))^*.$

Proof. We show only $h(u_j) \to h(u_0)$ strongly in $(H_r^1(\mathbf{R}^N))^*$ and deal with the case N = 2. Other cases can be treated similarly.

Suppose that $||u_j||_{H^1} \leq M$ for all $j \in \mathbb{N}$. By (5.2.4), we have

$$\int_{\mathbf{R}^N} \Phi\left(\frac{u_j^2}{M^2}\right) dx \le \frac{\tilde{C}_1}{M^2} \|u_j\|_2^2 \le \tilde{C}_1$$

Since h satisfies (5–g1'), for any $\varepsilon > 0$ there exists an $\ell_{\varepsilon} (\geq \delta_0 > 0)$ such that

$$|h(\xi)| \le \varepsilon \Phi\left(\frac{\xi^2}{2M^2}\right) \quad \text{for } |\xi| \ge \ell_{\varepsilon}.$$

We set

$$\tilde{h}(\xi) := \begin{cases} h(\xi) & \text{for } |\xi| \le \ell_{\varepsilon}, \\ h(\ell_{\varepsilon}) & \text{for } \xi > \ell_{\varepsilon}, \\ -h(\ell_{\varepsilon}) & \text{for } \xi < -\ell_{\varepsilon}. \end{cases}$$

Then we have

$$|h(\xi) - \tilde{h}(\xi)| \le 2\varepsilon \Phi\left(\frac{\xi^2}{2M^2}\right)$$
 for all $\xi \in \mathbf{R}$.

Since the embedding $H_r^1(\mathbf{R}^N) \subset L^2(|x| \leq R_M)$ is compact, we have $u_j \to u_0$ strongly in $L^2(|x| \leq R_M)$, which implies

$$\tilde{h}(u_j) \to \tilde{h}(u_0)$$
 strongly in $L^2(|x| \le R_M)$.

Thus, by (5.2.6) and the definition of \tilde{h} , we have $\tilde{h}(u_j(x)) = 0$ for $|x| \ge R_M$ and

$$\|\tilde{h}(u_j) - \tilde{h}(u_0)\|_2 \to 0 \text{ as } j \to \infty.$$

On the other hand,

$$\|h(u_j) - \tilde{h}(u_j)\|_2^2 \le 4\varepsilon^2 \int_{\mathbf{R}^2} \Phi\left(\frac{u_j^2}{2M^2}\right)^2 dx \le 4\varepsilon^2 \int_{\mathbf{R}^2} \Phi\left(\frac{u_j^2}{M^2}\right) dx \le 4\varepsilon^2 \tilde{C}_1.$$

Here we used the fact that $\Phi(s/2)^2 \leq \Phi(s)$ for all $s \geq 0$. Similarly we also have $||h(u_0) - \tilde{h}(u_0)||_2^2 \leq 4\varepsilon^2 \tilde{C}_1$. Thus

$$\|h(u_j) - h(u_0)\|_2 \le \|h(u_j) - \tilde{h}(u_j)\|_2 + \|\tilde{h}(u_j) - \tilde{h}(u_0)\|_2 + \|\tilde{h}(u_0) - h(u_0)\|_2$$
$$\le \|\tilde{h}(u_j) - \tilde{h}(u_0)\|_2 + 4\varepsilon \sqrt{\tilde{C}_1} \to 4\varepsilon \sqrt{\tilde{C}_1} \quad \text{as } j \to \infty.$$

Since $\varepsilon > 0$ is arbitrary, we have $||h(u_j) - h(u_0)||_2 \to 0$. We remark that $H^1_r(\mathbf{R}^N) \subset L^2(\mathbf{R}^N)$ implies $L^2(\mathbf{R}^N) \subset (H^1_r(\mathbf{R}^N))^*$ and thus $h(u_j) \to h(u_0)$ strongly in $(H^1_r(\mathbf{R}^N))^*$.

5.2.3 A comparison functional J

We define two functionals I and $J; H^1_r(\mathbf{R}^N) \to \mathbf{R}$ by

$$\begin{split} I(u) &:= \frac{1}{2} \|\nabla u\|_2^2 - \int_{\mathbf{R}^N} G(u) dx = \frac{1}{2} \|u\|_{H^1}^2 - \int_{\mathbf{R}^N} \frac{1}{2} m_0 u^2 + G(u) dx, \\ J(u) &:= \frac{1}{2} \|u\|_{H^1}^2 - \int_{\mathbf{R}^N} \overline{H}(u) dx. \end{split}$$

Critical points of I are solutions of our original problem (5.1.1)–(5.1.2) and critical points of J are solutions of the following equation: $-\Delta u + m_0 u = \overline{h}(u)$ in \mathbb{R}^N . For I and J, we have the following lemma.

Lemma 5.2.4. The following hold:

(a) The functionals $I, J \in C^1(H^1_r(\mathbf{R}^N), \mathbf{R})$ and for all $u, \varphi \in H^1_r(\mathbf{R}^N)$, it holds that

$$I'(u)\varphi = (u,\varphi)_{H^1} - \int_{\mathbf{R}^N} m_0 u\varphi + g(u)\varphi dx$$
$$J'(u)\varphi = (u,\varphi)_{H^1} - \int_{\mathbf{R}^N} \overline{h}(u)\varphi dx.$$

- (b) For all $u \in H^1_r(\mathbf{R}^N)$, $I(u) \ge J(u)$.
- (c) There exist $r_0 > 0$ and $\rho_0 > 0$ such that

$$I(u), J(u) \ge 0 \quad \text{for each } u \in H^1_r(\mathbf{R}^N) \text{ with } \|u\|_{H^1} \le r_0,$$

$$I(u), J(u) \ge \rho_0 \quad \text{for each } u \in H^1_r(\mathbf{R}^N) \text{ with } \|u\|_{H^1} = r_0.$$

(d) For any $n \in \mathbf{N}$, there exists an odd continuous mapping $\gamma_{0n} : S^{n-1} := \{ \sigma = (\sigma_1, \ldots, \sigma_n) \in \mathbf{R}^n; |\sigma| = 1 \} \rightarrow H^1_r(\mathbf{R}^N)$ such that

$$J(\gamma_{0n}(\sigma)) \le I(\gamma_{0n}(\sigma)) < 0$$
 for all $\sigma \in S^{n-1}$.

Proof. The statement (a) follows from (5–g1') and (5–g2'), and (b) follows from (a) of Corollary 5.2.2.

(c) By Corollary 5.2.2, for any $\varepsilon > 0$ there exists a C_{ε} such that $\overline{H}(\xi) \leq C_{\varepsilon}|\xi|^{p_0} + \varepsilon \Psi(\xi)$ for all $\xi \in \mathbf{R}$ where p appearing in the definition of \overline{h} and $\Psi(\xi) = |\xi|^{2N/(N-2)}$ if $N \geq 3$, $\Psi(\xi) = e^{\xi^2} - 1$ if N = 2. Thus by the Sobolev's inequality and (5.2.4), for all $u \in H_r^1(\mathbf{R}^N)$ with $||u||_{H^1} \leq 1$, we have

$$\int_{\mathbf{R}^{N}} \overline{H}(u) dx \le C_{\varepsilon} \|u\|_{H^{1}}^{p_{0}} + \varepsilon \int_{\mathbf{R}^{N}} \Psi(u) dx = \begin{cases} C_{\varepsilon} \|u\|_{H^{1}}^{p_{0}} + C_{1}\varepsilon \|u\|_{2}^{2} & \text{if } N = 2, \\ C_{\varepsilon} \|u\|_{H^{1}}^{p_{0}} + C\varepsilon \|u\|_{2N/(N-2)}^{2N/(N-2)} & \text{if } N \ge 3. \end{cases}$$

Therefore, for $||u||_{H^1} \leq 1$, it follows that

$$I(u) \ge J(u) \ge \begin{cases} \frac{1}{2} \|u\|_{H^{1}}^{2} - C_{\varepsilon} \|u\|_{H^{1}}^{p_{0}} - C_{1}\varepsilon \|u\|_{2}^{2} & \text{if } N = 2, \\ \frac{1}{2} \|u\|_{H^{1}}^{2} - C_{\varepsilon} \|u\|_{H^{1}}^{p_{0}} - C\varepsilon \|u\|_{2N/(N-2)}^{2N/(N-2)} & \text{if } N \ge 3. \end{cases}$$

Thus choosing $\varepsilon > 0$ and $r_0 > 0$ small enough, (c) holds.

(d) Since \overline{h} is an odd function and satisfies $\overline{H}(\zeta_0) - m_0 \zeta_0^2/2 \ge G(\zeta_0) > 0$, we can argue as in Theorem 10 of [16] and find for any $n \in \mathbb{N}$ an odd continuous mapping $\pi_n : S^{n-1} \to H^1_r(\mathbb{R}^N)$ such that

$$0 \notin \pi_n(S^{n-1}), \quad \int_{\mathbf{R}^N} G(\pi_n(\sigma)) dx \ge 1 \quad \text{for all } \sigma \in S^{n-1}.$$

For $\ell \geq 1$, set

$$\gamma_{0n}(\sigma)(x) := \pi_n(\sigma)(x/\ell) : S^{n-1} \to H^1_r(\mathbf{R}^N).$$

Then

$$I(\gamma_{0n}(\sigma)) = \frac{\ell^{N-2}}{2} \|\nabla \pi_n(\sigma)\|_2^2 - \ell^N \int_{\mathbf{R}^N} G(\pi_n(\sigma)) dx \le \frac{\ell^{N-2}}{2} \|\nabla \pi_n(\sigma)\|_2^2 - \ell^N.$$

Thus for sufficiently large $\ell = \ell_n \ge 1$, γ_{0n} has the desired property.

By the above lemma, I and J have symmetric mountain pass geometry and we can define symmetric mountain pass values. We will give them in section 5.3.

One of the virtue of our comparison functional J is the following:

Lemma 5.2.5. The functional J satisfies the Palais–Smale compactness condition.

Proof. Since \overline{h} satisfies the global Ambrosetti–Rabinowitz condition (see Corollary 5.2.2 (e)), we can easily verify the Palais–Smale condition. Indeed, let $(u_j)_{j=1}^{\infty} \subset H_r^1(\mathbf{R}^N)$ be a sequence satisfying

$$(5.2.7) J(u_j) \to b,$$

(5.2.8)
$$||J'(u_j)||_{(H^1_r(\mathbf{R}^N))^*} \to 0$$

From Corollary 5.2.2 (e), we have (5.2.9)

$$J(u_j) - \frac{1}{p_0 + 1} J'(u_j) u_j = \left(\frac{1}{2} - \frac{1}{p_0 + 1}\right) \|u_j\|_{H^1}^2 - \int_{\mathbf{R}^N} \overline{H}(u_j) - \frac{1}{p_0 + 1} \overline{h}(u_j) u_j dx$$
$$\geq \left(\frac{1}{2} - \frac{1}{p_0 + 1}\right) \|u_j\|_{H^1}^2.$$

Thus we can get the boundedness of $(u_j)_{j=1}^{\infty}$ in $H_r^1(\mathbf{R}^N)$ from (5.2.7)–(5.2.9) and extract a subsequence such that $u_{j_k} \rightharpoonup u_0$ weakly in $H_r^1(\mathbf{R}^N)$. By Lemma 5.2.3 (b), we have $\overline{h}(u_{j_k}) \rightarrow \overline{h}(u_0)$ strongly in $(H_r^1(\mathbf{R}^N))^*$, thus by (5.2.8), u_{j_k} converges to u_0 strongly in $H_r^1(\mathbf{R}^N)$, which completes the proof.

5.3 Minimax arguments

By Lemma 5.2.4, I and J have a symmetric mountain pass geometry and we can define mountain pass and symmetric mountain pass values. Here we follow Rabinowitz [92] essentially and set for $n \in \mathbb{N}$

(5.3.1)
$$b_n := \inf_{\gamma \in \Gamma_n} \max_{\sigma \in D_n} I(\gamma(\sigma)), \quad c_n := \inf_{\gamma \in \Gamma_n} \max_{\sigma \in D_n} J(\gamma(\sigma)).$$

Here $D_n := \{ \sigma \in \mathbf{R}^n : |\sigma| \le 1 \}$ and a family of mappings Γ_n is defined by

(5.3.2)
$$\Gamma_n := \{ \gamma \in C(D_n, H^1_r(\mathbf{R}^N)) : \gamma_n \text{ is odd and } \gamma_n(\sigma) = \gamma_{0n}(\sigma) \text{ on } \sigma \in \partial D_n \},$$

where $\gamma_{0n}: \partial D_n = S^{n-1} \to H^1_r(\mathbf{R}^N)$ is given in Lemma 5.2.4. We remark that

$$\gamma(\sigma) := \begin{cases} |\sigma|\gamma_{0n} \left(\frac{\sigma}{|\sigma|}\right) & \text{for } \sigma \in D_n \setminus \{0\} \\ 0 & \text{for } \sigma = 0, \end{cases}$$

belongs to Γ_n and $\Gamma_n \neq \emptyset$ for all $n \in \mathbf{N}$.

Remark 5.3.1. We can define mountain pass values b_{mp} , c_{mp} for I, J by

(5.3.3)
$$b_{mp} := \inf_{\gamma \in \Gamma_{mp}} \max_{0 \le t \le 1} I(\gamma(t)), \quad c_{mp} := \inf_{\gamma \in \Gamma_{mp}} \max_{0 \le t \le 1} J(\gamma(t)),$$

where $\Gamma_{mp} := \{\gamma \in C([0,1], H_r^1(\mathbf{R}^N)) : \gamma(0) = 0, \gamma(1) = e_0\}$ and $e_0 \in H_r^1(\mathbf{R}^N)$ is chosen so that $I(e_0) < 0$. We will show in section 5.6 that b_{mp} and c_{mp} do not depend on the choice of e_0 (see Lemma 5.6.1). Thus, recalling $S^0 = \{\pm 1\}$ and choosing $e_0 = \gamma_{01}(1)$, we can see $b_{mp} = b_1, c_{mp} = c_1$. We will also show that b_{mp} is corresponding to a positive least energy solution of (5.1.1)-(5.1.2) in section 5.6.

We can easily see that $\gamma(D_n) \cap \{u \in H^1_r(\mathbf{R}^N) : ||u||_{H^1} = r_0\} \neq \emptyset$ for all $\gamma \in \Gamma_n$. Thus it follows from Lemma 5.2.4 (b) and (c) that

$$(5.3.4) b_n \ge c_n \ge \rho_0 > 0.$$

Moreover, we have

Lemma 5.3.2. The following hold:

- (a) The value c_n is a critical value of J for all $n \in \mathbf{N}$.
- (b) As $n \to \infty$, $c_n \to \infty$.

Proof. (a) By Lemma 5.2.5, J satisfies the Palais–Smale condition. Thus (a) holds (see for example [92]).

(b) By Theorem 7.1.1 in chapter 7, we can see that $c_n \to \infty$ as $n \to \infty$.

By (5.3.4) and Lemma 5.3.2, the minimax values b_n satisfy

$$b_n > 0 \ (n \in \mathbf{N}), \quad b_n \to \infty \text{ as } n \to \infty$$

In the following sections we will see that the value b_n is critical value of I for all $n \in \mathbb{N}$.

5.4 Functional $I(\theta, u)$

It seems difficult to show the Palais–Smale compactness condition for I directly and it is a main difficulty in showing that b_n is a critical value of I.

As stated in section 5.1, we introduce an auxiliary functional $\tilde{I} \in C^1(\mathbf{R} \times H^1_r(\mathbf{R}^N), \mathbf{R})$ by

$$\tilde{I}(\theta, u) := \frac{1}{2} e^{(N-2)\theta} \int_{\mathbf{R}^N} |\nabla u|^2 dx - e^{N\theta} \int_{\mathbf{R}^N} G(u) dx.$$

The functional \tilde{I} is introduced based on the scaling properties of $\|\nabla u\|_2^2$ and $\int_{\mathbf{R}^N} G(u) dx$, and has the following properties:

(5.4.1)
$$\tilde{I}(0,u) = I(u),$$

(5.4.2)
$$\tilde{I}(\theta, u) = I(u(e^{-\theta}x)) \text{ for all } \theta \in \mathbf{R} \text{ and } u \in H^1_r(\mathbf{R}^N).$$

We equip a standard product norm $\|(\theta, u)\|_{\mathbf{R}\times H^1} \equiv (\theta^2 + \|u\|_{H^1}^2)^{1/2}$ to $\mathbf{R} \times H_r^1(\mathbf{R}^N)$.

We define a minimax value \tilde{b}_n for \tilde{I} by

$$b_{n} := \inf_{\tilde{\gamma} \in \tilde{\Gamma}_{n}} \max_{\sigma \in D_{n}} I(\tilde{\gamma}(\sigma)),$$

$$\tilde{\Gamma}_{n} := \{ \tilde{\gamma} \in C(D_{n}, \mathbf{R} \times H^{1}_{r}(\mathbf{R}^{N})) : \tilde{\gamma}(\sigma) = (\theta(\sigma), \eta(\sigma)) \text{ satisfies} \\ (\theta(-\sigma), \eta(-\sigma)) = (\theta(\sigma), -\eta(\sigma)) \text{ for all } \sigma \in D_{n}, \\ (\theta(\sigma), \eta(\sigma)) = (0, \gamma_{0n}(\sigma)) \text{ for all } \sigma \in \partial D_{n} \}.$$

Then we have

Lemma 5.4.1. For each $n \in \mathbf{N}$, $\tilde{b}_n = b_n$ holds.

Proof. For any $\gamma \in \Gamma_n$, we can see that $(0, \gamma) \in \tilde{\Gamma}_n$ and we may regard $\Gamma_n \subset \tilde{\Gamma}_n$. Thus by the definitions of b_n, \tilde{b}_n and (5.4.1), we have $\tilde{b}_n \leq b_n$. Next, for any given $\tilde{\gamma}(\sigma) = (\theta(\sigma), \eta(\sigma)) \in \tilde{\Gamma}_n$, we set $\gamma(\sigma) := \eta(\sigma)(e^{-\theta(\sigma)}x)$. We can

Next, for any given $\tilde{\gamma}(\sigma) = (\theta(\sigma), \eta(\sigma)) \in \Gamma_n$, we set $\gamma(\sigma) := \eta(\sigma)(e^{-\theta(\sigma)}x)$. We can verify that $\gamma \in \Gamma_n$ and by (5.4.2), $I(\gamma(\sigma)) = \tilde{I}(\tilde{\gamma}(\sigma))$ for all $\sigma \in D_n$. Thus we also have $\tilde{b}_n \geq b_n$.

As a virtue of $\tilde{I}(\theta, u)$, we can obtain a Palais–Smale sequence $(\theta_j, u_j)_{j=1}^{\infty}$ in the augmented space $\mathbf{R} \times H_r^1(\mathbf{R}^N)$ with an additional property (d) in Proposition 5.4.2 below. Namely we have:

Proposition 5.4.2. For any $n \in \mathbf{N}$, there exists a sequence $(\theta_j, u_j) \subset \mathbf{R} \times H^1_r(\mathbf{R}^N)$ such that :

(a) $\theta_j \to 0$. (b) $\tilde{I}(\theta_j, u_j) \to b_n$. (c) $\tilde{I}'(\theta_j, u_j) \to 0$ strongly in $(H_r^1(\mathbf{R}^N))^*$. (d) $\frac{\partial}{\partial \theta} \tilde{I}(\theta_j, u_j) \to 0$. To prove Proposition 5.4.2, we need the following lemma, which is a version of Ekeland's principle. We use the following notation:

$$D\tilde{I}(\theta, u) := \left(\frac{\partial \tilde{I}}{\partial \theta}(\theta, u), \tilde{I}'(\theta, u)\right),$$

$$dist_{\mathbf{R} \times H^1_r(\mathbf{R}^N)}((\theta, u), A) := \inf_{(\tau, v) \in A} (|\theta - \tau|^2 + ||u - v||^2_{H^1})^{1/2} \quad \text{for} \quad A \subset \mathbf{R} \times H^1_r(\mathbf{R}^N).$$

Lemma 5.4.3. Let $n \in \mathbf{N}$ and $\varepsilon > 0$. Suppose $\tilde{\gamma} \in \tilde{\Gamma}_n$ satisfies

$$\max_{\sigma \in D_n} \tilde{I}(\tilde{\gamma}(\sigma)) \le \tilde{b}_n + \varepsilon.$$

Then there exists $(\theta, u) \in \mathbf{R} \times H^1_r(\mathbf{R}^N)$ such that:

- (a) dist_{\mathbf{R} \times H^1_r(\mathbf{R}^N)}((\theta, u), \tilde{\gamma}(D_n)) \le 2\sqrt{\varepsilon}.
- (b) $\tilde{I}(\theta, u) \in [\tilde{b}_n \varepsilon, \tilde{b}_n + \varepsilon].$
- (c) $\|D\tilde{I}(\theta, u)\|_{\mathbf{R}\times (H^1_r(\mathbf{R}^N))^*} \leq 2\sqrt{\varepsilon}.$

Proof. Since \tilde{I} satisfies

$$\tilde{I}(\theta, -u) = \tilde{I}(\theta, u) \text{ for all } (\theta, u) \in \mathbf{R} \times H^1_r(\mathbf{R}^N),$$

we can see that the family $\tilde{\Gamma}_n$ is stable under the pseudo-deformation flow generated by \tilde{I} . Moreover, since $\tilde{b}_n = b_n > 0$, $\max_{\sigma \in \partial D_n} \tilde{I}(0, \gamma_{0n}(\sigma)) < 0$, we can show Lemma 5.4.3 in a standard way.

Proof of Proposition 5.4.2. For any $j \in \mathbf{N}$, we can find a $\gamma_j \in \Gamma_n$ such that

$$\max_{\sigma \in D_n} I(\gamma_j(\sigma)) \le b_n + \frac{1}{j}.$$

Since $\tilde{b}_n = b_n$, $\tilde{\gamma}_j(\sigma) := (0, \gamma_j(\sigma)) \in \tilde{\Gamma}_n$ satisfies $\max_{\sigma \in D_n} \tilde{I}(\tilde{\gamma}_j(\sigma)) \leq \tilde{b}_n + 1/j$. Applying Lemma 5.4.3, we can find a (θ_j, u_j) such that

(5.4.3)
$$\operatorname{dist}_{\mathbf{R}\times H^1_r(\mathbf{R}^N)}((\theta_j, u_j), \tilde{\gamma}_j(D_n)) \leq \frac{2}{\sqrt{j}}$$

(5.4.4)
$$\tilde{I}(\theta_j, u_j) \in \left[b_n - \frac{1}{j}, b_n + \frac{1}{j}\right],$$

(5.4.5)
$$\|D\tilde{I}(\theta_j, u_j)\|_{\mathbf{R} \times H^1} \le \frac{2}{\sqrt{j}}$$

Since $\tilde{\gamma}(D_n) \subset \{0\} \times H^1_r(\mathbf{R}^N)$, (5.4.3) implies $|\theta_j| \leq 2/\sqrt{j}$, in particular, (a). Clearly (5.4.4) implies (b) and (5.4.5) implies (c) and (d). Thus the proof of Proposition 5.4.2 is completed.

In the following section, we consider the boundedness and compactness properties of the sequence $(\theta_j, u_j)_{j=1}^{\infty}$ satisfying (a)–(d) of Proposition 5.4.2.

5.5 Boundedness and compactness of (θ_j, u_j)

Let $(\theta_j, u_j) \subset \mathbf{R} \times H^1_r(\mathbf{R}^N)$ be a sequence given in Proposition 5.4.2. In particular, u_j satisfies (a)–(d) of Proposition 5.4.2. First we observe that (b) and (d) imply the following

$$\frac{1}{2}e^{(N-2)\theta_j} \|\nabla u_j\|_2^2 - e^{N\theta_j} \int_{\mathbf{R}^N} G(u_j) dx \to b_n,$$

$$\frac{N-2}{2}e^{(N-2)\theta_j} \|\nabla u_j\|_2^2 - Ne^{n\theta_j} \int_{\mathbf{R}^N} G(u_j) dx \to 0 \quad \text{as } j \to \infty.$$

Thus we have

$$(5.5.1) \|\nabla u_j\|_2^2 \to Nb_n,$$

(5.5.2)
$$\int_{\mathbf{R}^N} G(u_j) dx \to \frac{N-2}{2} b_n.$$

First we show the boundedness of (u_i) in $H^1_r(\mathbf{R}^N)$.

Proposition 5.5.1. Let (θ_j, u_j) be a sequence satisfying (a)–(d) of Proposition 5.4.2. Then (u_j) is bounded in $H^1_r(\mathbf{R}^N)$.

Proof. (cf. Proof of Proposition 5.5 of Jeanjean and Tanaka [56]). We set

$$\varepsilon_j := \|\tilde{I}'(\theta_j, u_j)\|_{(H^1_r(\mathbf{R}^N))^*}.$$

By Proposition 5.4.2 (c), we have $\varepsilon_j \to 0$ and for any $\psi \in H^1_r(\mathbf{R}^N)$,

$$|\tilde{I}'(\theta_j, u_j)\psi| \le \varepsilon_j \|\psi\|_{H^1},$$

that is,

(5.5.3)
$$\left| e^{(N-2)\theta_j} \int_{\mathbf{R}^N} \nabla u_j \cdot \nabla \psi dx - e^{N\theta_j} \int_{\mathbf{R}^N} g(u_j) \psi dx \right| \le \varepsilon_j \sqrt{\|\nabla \psi\|_2^2 + m_0 \|\psi\|_2^2}.$$

We argue indirectly and assume $||u_j||_2 \to \infty$. We remark that $||\nabla u_j||_2$ is bounded by (5.5.1). We set $t_j := ||u_j||_2^{-2/N} \to 0$ and $v_j(y) := u_j(y/t_j)$. Then we have

(5.5.4)
$$||v_j||_2 = 1$$
 and $||\nabla v_j||_2^2 = t_j^{N-2} ||\nabla u_j||_2^2$.

In particular, (v_j) is bounded in $H^1_r(\mathbf{R}^N)$ and we can extract a subsequence $v_j \rightharpoonup v_0$ weakly in $H^1_r(\mathbf{R}^N)$. First we claim:

Step 1: $v_0 = 0$.

Let $\varphi \in H^1_r(\mathbf{R}^N)$ be a function with compact support. Setting $\psi := \varphi(t_j x)$ in (5.5.3), we have

$$\left| e^{(N-2)\theta_j} t_j^{-(N-2)} (\nabla v_j, \nabla \varphi)_2 - e^{N\theta_j} t_n^{-N} \int_{\mathbf{R}^N} g(v_j) \varphi dy \right| \le \varepsilon_j \sqrt{t_j^{-(N-2)} \|\nabla \varphi\|_2^2 + m_0 t_j^{-N} \|\varphi\|_2^2}.$$

Multiplying t_i^N ,

$$\left| e^{(N-2)\theta_j} t_j^2(\nabla v_j, \nabla \varphi) - e^{N\theta_j} \int_{\mathbf{R}^N} g(v_j)\varphi dy \right| \le \varepsilon_j t_j^{N/2} \sqrt{t_j^2 \|\nabla \varphi\|_2^2 + m_0 \|\varphi\|_2^2} \to 0.$$

Thus $v_0 \in H^1_r(\mathbf{R}^N)$ satisfies

(5.5.5)
$$\int_{\mathbf{R}^N} g(v_0)\varphi dy = 0 \quad \text{for all } \varphi \in H^1_r(\mathbf{R}^N) \text{ with compact support,}$$

which implies $g(v_0) \equiv 0$. Since $\xi = 0$ is an isolated solution of $g(\xi) = 0$ by (5–g2'), it follows from (5.5.5) that $v_0 \equiv 0$.

Step 2: Conclusion

Next we set $\psi(x) := u_j(x)$ in (5.5.3). We have

$$\left| e^{(N-2)\theta_j} t_j^{-(N-2)} \|\nabla v_j\|_2^2 - e^{N\theta_j} t_j^{-N} \int_{\mathbf{R}^N} g(v_j) v_j dx \right| \le \varepsilon_j \sqrt{t_j^{-(N-2)}} \|\nabla v_j\|_2^2 + m_0 t_j^{-N} \|v_j\|_2^2.$$

Again, multiplying t_j^N , it follows that

$$\delta_j := e^{(N-2)\theta_j} t_j^2 \|\nabla v_j\|_2^2 - e^{N\theta_j} \int_{\mathbf{R}^N} g(v_j) v_j dx \to 0.$$

Thus

(5.5.6)
$$e^{(N-2)\theta_j} t_j^2 \|\nabla v_j\|_2^2 + m_0 e^{N\theta_j} \|v_j\|_2^2 = e^{N\theta_j} \int_{\mathbf{R}^N} m_0 v_j^2 + g(v_j) v_j dx + \delta_j$$
$$\leq e^{N\theta_j} \int_{\mathbf{R}^N} h(v_j) v_j dx + \delta_j.$$

Here we used Lemma 5.2.1 (a). Since $v_j \rightarrow 0$ weakly in $H^1_r(\mathbf{R}^N)$, Lemma 5.2.3 (b) implies $\int_{\mathbf{R}^N} h(v_j) v_j dx \rightarrow 0$. Thus (5.5.6) implies $||v_j||_2 \rightarrow 0$, which is in contradiction to (5.5.4). Therefore (u_j) is bounded in $H^1_r(\mathbf{R}^N)$.

Remark 5.5.2. When $N \ge 3$, we can prove Proposition 5.5.1 in a direct way. Indeed, by the definition of h, we have for some constant C > 0

$$|h(\xi)| \le C|\xi|^{(N+2)/(N-2)}$$
 for all $\xi \in \mathbf{R}$

It follows from $\varepsilon_j = \|\tilde{I}'(\theta_j, u_j)\|_{(H^1_r(\mathbf{R}^N))^*} \to 0$ that $|\tilde{I}'(\theta_j, u_j)u_j| \le \varepsilon_j \|u_j\|_{H^1}$. Thus

(5.5.7)

$$e^{(N-2)\theta_{j}} \|\nabla u_{j}\|_{2}^{2} + m_{0}e^{N\theta_{j}}\|u_{j}\|_{2}^{2} \leq e^{N\theta_{j}} \int_{\mathbf{R}^{N}} m_{0}u_{j}^{2} + g(u_{j})u_{j}dx + \varepsilon_{j}\|u_{j}\|_{H^{1}}$$

$$\leq e^{N\theta_{j}} \int_{\mathbf{R}^{N}} h(u_{j})u_{j}dx + \varepsilon_{j}\|u_{j}\|_{H^{1}}$$

$$\leq Ce^{N\theta_{j}}\|u_{j}\|_{2N/(N-2)}^{2N/(N-2)} + \varepsilon_{j}\|u_{j}\|_{H^{1}}.$$

Since $\|\nabla u_j\|_2$ is bounded, we can observe that $\|u_j\|_{2N/(N-2)}$ is also bounded. Thus (5.5.7) implies the boundedness of $\|u_j\|_2$, that is, (u_j) is bounded in $H_r^1(\mathbf{R}^N)$.

Lastly in this section, we prove that (u_j) has a strongly convergent subsequence in $H^1_r(\mathbf{R}^N)$.

Proposition 5.5.3. Let (θ_j, u_j) be a sequence satisfying (a)–(d) of Proposition 5.4.2. Then (θ_j, u_j) has a strongly convergent subsequence in $\mathbf{R} \times H_r^1(\mathbf{R}^N)$.

Proof. It suffices to prove (u_j) has a strongly convergent subsequence in $H_r^1(\mathbf{R}^N)$. By Proposition 5.5.1, (u_j) is bounded in $H_r^1(\mathbf{R}^N)$ and we may assume $u_j \rightharpoonup u_0$ weakly in $H_r^1(\mathbf{R}^N)$ as $j \rightarrow \infty$.

It follows from Proposition 5.4.2 (c) that $\tilde{I}'(\theta_j, u_j)\varphi \to 0$ as $j \to \infty$ for any $\varphi \in H^1_r(\mathbf{R}^N)$, that is,

(5.5.8)
$$\int_{\mathbf{R}^N} e^{(N-2)\theta_j} \nabla u_j \cdot \nabla \varphi - e^{N\theta_j} g(u_j) \varphi dx \to 0 \quad \text{as } j \to \infty.$$

Thus u_0 satisfies $\int_{\mathbf{R}^N} \nabla u_0 \cdot \nabla \varphi - g(u_0) \varphi dx = 0$ for all $\varphi \in H^1_r(\mathbf{R}^N)$ and u_0 is a solution of (5.1.1)–(5.1.2). In particular, we have $\|\nabla u_0\|_2^2 - \int_{\mathbf{R}^N} g(u_0) u_0 dx = 0$, that is,

(5.5.9)
$$\|u_0\|_{H^1}^2 - \int_{\mathbf{R}^N} m_0 u_0^2 + g(u_0) u_0 dx = 0.$$

Setting $\varphi := u_j$ in (5.5.8), we have $e^{(N-2)\theta_j} \|\nabla u_j\|_2^2 - e^{N\theta_j} \int_{\mathbf{R}^N} g(u_j) u_j dx \to 0$. Thus

$$(5.5.10) \qquad e^{(N-2)\theta_j} \|\nabla u_j\|_2^2 + m_0 e^{N\theta_j} \|u_j\|_2^2 = e^{N\theta_j} \int_{\mathbf{R}^N} m_0 u_j^2 + g(u_j) u_j dx + o(1)$$
$$= e^{N\theta_j} \int_{\mathbf{R}^N} h(u_j) u_j dx - e^{N\theta_j} \int_{\mathbf{R}^N} h(u_j) u_j - m_0 u_j^2 - g(u_j) u_j dx + o(1)$$
$$= e^{N\theta_j} (\mathbf{I}) - e^{N\theta_j} (\mathbf{II}) + o(1) \quad \text{as } j \to \infty.$$

By Lemma 5.2.3 (b), we have

(5.5.11)
$$(\mathbf{I}) \to \int_{\mathbf{R}^N} h(u_0) u_0 dx$$

On the other hand, by Lemma 5.2.1 (a), we have

$$h(u_j(x))u_j(x) - m_0 u_j(x)^2 - g(u_j(x))u_j(x) \ge 0 \quad \text{for all } j \in \mathbf{N} \text{ and } x \in \mathbf{R}.$$

Thus by Fatou's lemma,

(5.5.12)
$$\liminf_{j \to \infty} (\mathrm{II}) \ge \int_{\mathbf{R}^N} h(u_0) u_0 - m_0 u_0^2 - g(u_0) u_0 dx.$$

It follows from (5.5.10)-(5.5.12) that

$$\limsup_{j \to \infty} \|u_j\|_{H^1}^2 = \limsup_{j \to \infty} \left(e^{(N-2)\theta_j} \|\nabla u_j\|_2^2 + m_0 e^{N\theta_j} \|u_j\|_2^2 \right) \le \int_{\mathbf{R}^N} m_0 u_0^2 + g(u_0) u_0 dx.$$

Thus by (5.5.9), we have $\limsup_{j\to\infty} \|u_j\|_{H^1} \le \|u_0\|_{H^1}$, which implies $u_j \to u_0$ strongly in $H^1_r(\mathbf{R}^N)$.

Now we can prove

Theorem 5.5.4. Assume $N \ge 2$ and (5-g0), (5-g1'), (5-g2'), (5-g3). Then $b_n(n \in \mathbf{N})$ defined in (5.3.1)–(5.3.2) is a critical value of I. That is , for any $n \in \mathbf{N}$, there exists a critical point $u_{0n} \in H_r^1(\mathbf{R}^N)$, which is a solution of (5.1.1)–(5.1.2), such that

(5.5.13)
$$I(u_{0n}) = b_n, \quad I'(u_{0n}) = 0.$$

Proof. Let (θ_j, u_j) be a sequence obtained in Proposition 5.4.2. By Proposition 5.5.3, we may assume $u_j \to u_{0n}$ strongly in $H^1_r(\mathbf{R}^N)$. Then u_{0n} satisfies

$$I(0, u_{0n}) = b_n, \quad I'(0, u_{0n}) = 0,$$

that is nothing but (5.5.13). Thus b_n is a critical value of I which completes the proof. \Box

5.6 Least energy solutions

In this section, we show that a mountain pass value b_{mp} is corresponding to a positive solution of (5.1.1)–(5.1.2), which has the least energy among all nontrivial solutions.

We start with the following lemma.

Lemma 5.6.1. Suppose $N \ge 2$ and assume (5–g0), (5–g1'), (5–g2') and (5–g3). Let $O = \{u \in H^1_r(\mathbf{R}^N) : I(u) < 0\}$. Then the set O is arcwise connected.

We will give a proof of Lemma 5.6.1 in the section 5.7. By Lemma 5.6.1, we can easily see that the mountain pass value b_{mp} given in (5.3.3) does not depend on the end point e_0 and we may write

(5.6.1) $b_{mp} = \inf_{\gamma \in \Gamma_{mp}} \max_{0 \le t \le 1} I(\gamma(t)),$

(5.6.2)
$$\Gamma_{mp} := \{ \gamma \in C([0,1], H^1_r(\mathbf{R}^N)) : \gamma(0) = 0, \ I(\gamma(1)) < 0 \}.$$

This fact is also used in Remark 5.3.1.

Remark 5.6.2. Lemma 5.6.1 is also obtained in Byeon [21] (but with a different proof). We learned [21] from Professor J. Byeon and the referee after a submission of [48].

Our main result in this section is the following.

Theorem 5.6.3. Suppose $N \ge 2$ and assume (5–g0), (5–g1'), (5–g2'), (5–g3). Then for b_{mp} defined in (5.6.1)–(5.6.2) it holds that:

(a) There exists a positive solution u_0 of (5.1.1)-(5.1.2) such that

(5.6.3)
$$I(u_0) = b_{mp}$$
.

- (b) For any nontrivial solution v of (5.1.1)-(5.1.2), we have
 - $(5.6.4) b_{mp} \le I(v),$

that is, u_0 is a least energy solution of (5.1.1)–(5.1.2) and the value b_{mp} is the least energy level.

Proof. (a) We argue as in previous sections and for any $\gamma_j \in \Gamma_{mp}$ satisfying

(5.6.5)
$$\max_{0 \le t \le 1} I(\gamma_j(t)) \le b_{mp} + \frac{1}{j}$$

we can find a $(\theta_j, u_j) \in \mathbf{R} \times H^1_r(\mathbf{R}^N)$ such that

(5.6.6)
$$\operatorname{dist}_{\mathbf{R} \times H^1_r(\mathbf{R}^N)}((\theta_j, u_j), \{0\} \times \gamma_j([0, 1])) \le \frac{2}{\sqrt{j}},$$

(5.6.7) $u_j \to u_0$ strongly in $H^1_r(\mathbf{R}^N)$.

Here u_0 is a critical point of I satisfying $I(u_0) = b_{mp}$. Since I(u) = I(|u|) for all $u \in H^1_r(\mathbf{R}^N)$, we may assume $\gamma_j \in \Gamma_{mp}$ in (5.6.5) satisfies

$$\gamma_i(t)(x) \ge 0$$
 for all $t \in [0,1]$ and $x \in \mathbf{R}^N$.

Then it follows from (5.6.6) that

$$||(u_j)_-||_{H^1} \le \operatorname{dist}_{\mathbf{R} \times H^1_r(\mathbf{R}^N)}((\theta_j, u_j), \{0\} \times \gamma_j([0, 1])) \to 0,$$

where $u_{-}(x) = \max\{0, -u(x)\}$. Thus we have $(u_0)_{-} = 0$ and by the maximum principle, $u_0(x) > 0$ in \mathbf{R}^N , and (a) is proved.

(b) To see (5.6.4), we can use argument in [55] and for any given nontrivial solution $v \in H^1_r(\mathbf{R}^N)$, we can construct a path $\gamma \in \Gamma_{mp}$ such that

$$v \in \gamma([0,1]), \quad \max_{0 \le t \le 1} I(\gamma(t)) = I(v).$$

Thus we have (b) and the proof of Theorem 5.6.3 is completed.

5.7 Proof of Lemma 5.6.1

The aim of this section is to give a proof of Lemma 5.6.1. We will show that for any $u_0, u_1 \in O$, there exists a continuous path γ in O joining u_0 and u_1 .

In this section, we write r = |x| and we identify u(r) and a radially symmetric function u(x) = u(|x|). We set for $R \ge 1$, $t \ge 0$,

$$\eta(R,t;r) := \begin{cases} 0 & \text{if } r \in [0,R], \\ \zeta_0(r-R) & \text{if } r \in [R,R+1], \\ \zeta_0 & \text{if } r \in [R+1,R+1+t], \\ \zeta_0(R+2+t-r) & \text{if } r \in [R+1+t,R+2+t], \\ 0 & \text{if } r \in [R+2+t,\infty). \end{cases}$$

Here $\zeta_0 > 0$ is given in (5–g3). In particular, we have $G(\zeta_0) > 0$.

We will see that $\eta(R,T;r) \in O$ for large R, T and there exist continuous curves joining u_i (i = 0, 1) and $\eta(R, T; r)$ in O. Clearly this proves our Lemma 5.6.1.

We start with the following lemma.

Lemma 5.7.1. There exist $R_0 \ge 1$ and $C_0, C_1 > 0$ which do not depend on R and t such that:

- (a) For all (R, t) with $t \ge R \ge R_0$, $I(\eta(R, t; r)) \le -C_0 G(\zeta_0) t^N$.
- (b) For all $R \ge R_0$, $\sup_{0 \le t \le \infty} I(\eta(R, t; r)) \le C_1 R^{N-1}$.
- (c) For all $R \ge R_0$, $\max_{0 \le s \le 1} I(s\eta(R, 0; r)) \le C_1 R^{N-1}$.

Proof. For $R \ge 1, t \ge 0$, a direct computation gives us

$$\begin{split} &I(\eta(R,t;r)) \\ = &\omega_{N-1} \left(\int_{R}^{R+1} + \int_{R+1}^{R+1+t} + \int_{R+1+t}^{R+2+t} \right) \left(\frac{1}{2} |\eta_r(R,t;r)|^2 - G(\eta(R,t;r)) \right) r^{N-1} dr \\ \leq &\frac{\omega_{N-1}}{N} B((R+1)^N - R^N + (R+2+t)^N - (R+1+t)^N) \\ &- \frac{\omega_{N-1}}{N} G(\zeta_0) ((R+1+t)^N - (R+1)^N), \end{split}$$

where ω_{N-1} is the surface area of the unit sphere in \mathbf{R}^N and B is defined by

(5.7.1)
$$B = \frac{1}{2}\zeta_0^2 + \max_{\xi \in [0,\zeta_0]} |G(\xi)|.$$

We remark for $R\geq 1$ and $t\geq 0$

$$(R+1)^{N} - R^{N} = {}_{N}C_{1}R^{N-1} + {}_{N}C_{2}R^{N-2} + \ldots + {}_{N}C_{N} \leq ({}_{N}C_{1} + \ldots + {}_{N}C_{N})R^{N-1}$$

= $(2^{N} - 1)R^{N-1}$,
 $(R+2+t)^{N} - (R+1+t)^{N} \leq (2^{N} - 1)(R+1+t)^{N-1} \leq 2^{N-1}(2^{N} - 1)(R+t)^{N-1}$,
 $(R+1+t)^{N} - (R+1)^{N} \geq t^{N}$.

Thus there exists a constant $C_2 > 0$ independent of $R \ge 1, t \ge 0$ such that

(5.7.2)
$$I(\eta(R,t;r)) \le C_2(R^{N-1} + (R+t)^{N-1}) - \frac{\omega_{N-1}}{N}G(\zeta_0)t^N$$

(a)–(c) follow from (5.7.2). Indeed, if $t \ge R$, it follows from (5.7.2) that

$$I(\eta(R,t;r)) \le C_2(t^{N-1} + (2t)^{N-1}) - \frac{\omega_{N-1}}{N}G(\zeta_0)t^N$$

Thus for sufficiently large $R_0 \ge 1$, (a) holds.

By (a), for each $R \ge R_0$, we have $\sup_{0 \le t < \infty} I(\eta(R, t; r)) = \max_{0 \le t \le R} I(\eta(R, t; r))$. From (5.7.2), we have

$$I(\eta(R,t;r)) \le C_2(R^{N-1} + (2R)^{N-1})$$
 for $t \in [0,R]$.

Thus we have (b).

For (c), recalling (5.7.1), we have

$$\begin{split} I(s\eta(R,0;r)) &\leq \omega_{N-1} \int_{R}^{R+2} \left(\frac{1}{2} |s\eta_r(R,0;r)|^2 - G(s\eta(R,0;r)) \right) r^{N-1} dr \\ &\leq \frac{\omega_{N-1}}{N} B\left((R+2)^N - R^N \right) \quad \text{for } s \in [0,1]. \end{split}$$

Thus, choosing $C_1 > 0$ larger if necessary, we get (c).

Now, suppose $u_0, u_1 \in O$ and we try to join u_0 and u_1 through $\eta(R_1, T_1; r)$ $(T_1 \ge R_1 \gg 1)$ in O. We remark that we may assume that u_0, u_1 have compact supports and

$$\operatorname{supp} u_0, \operatorname{supp} u_1 \subset [0, L_0] \text{ for some constant } L_0 > 0.$$

We consider the following curves:

$$\begin{split} \gamma_1 &: [L_0, R_1] \to H^1_r(\mathbf{R}^N), \quad R \mapsto u_0(L_0 r/R), \\ \gamma_2 &: [0, 1] \to H^1_r(\mathbf{R}^N), \qquad s \mapsto u_0(L_0 r/R_1) + s\eta(R_1, 0; r), \\ \gamma_3 &: [0, T_1] \to H^1_r(\mathbf{R}^N), \qquad t \mapsto u_0(L_0 r/R_1) + \eta(R_1, t; r), \\ \gamma_4 &: [0, 1] \to H^1_r(\mathbf{R}^N), \qquad s \mapsto (1 - s)u_0(L_0 r/R_1) + \eta(R_1, T_1; r). \end{split}$$

Joining these curves, we get the desired path joining u_0 and $\eta(R_1, T_1; r)$. We need to show with suitable choices of R_1, T_1 , our path is included in O.

Lemma 5.7.2. It holds that

- (a) For all $R \in [L_0, \infty)$, $I(u_0(L_0r/R)) < 0$.
- (b) There exists an $R_1 \ge R_0$ such that

$$(5.7.3) I(u_0(L_0r/R_1) + s\eta(R_1, 0; r)) < 0 for all s \in [0, 1],$$

(5.7.4) $I(u_0(L_0r/R_1) + \eta(R_1, t; r)) < 0 \text{ for all } t \in [0, \infty).$

(c) There exists a $T_1 \ge R_1$ such that

(5.7.5)
$$I((1-s)u_0(L_0r/R_1) + \eta(R_1, T_1; r)) < 0 \text{ for all } s \in [0, 1].$$

Proof. (a) Since $u_0 \in O$, we have $\int_{\mathbf{R}^N} G(u_0) dx > 0$ and we can see that the map $R \mapsto I(u_0(r/R)) : [1, \infty) \to H^1_r(\mathbf{R}^N)$ is strictly decreasing. Thus (a) holds.

(b) We mainly deal with (5.7.4). Suppose $R_1 \ge R_0$, where $R_0 \ge 1$ is given in Lemma 5.7.1. We remark

supp
$$u_0(L_0r/R_1) \subset [0, R_1]$$
, supp $\eta(R_1, t; r) \subset [R_1, R_1 + 2 + t]$.

Thus for all $t \ge 0, R_1 \ge R_0$,

.

$$I(u_0(L_0r/R_1) + \eta(R_1, t; r)) = I(u_0(L_0r/R_1)) + I(\eta(R_1, t; r))$$

$$\leq \frac{1}{2} \left(\frac{R_1}{L_0}\right)^{N-2} \|\nabla u_0\|_2^2 - \left(\frac{R_1}{L_0}\right)^N \int_{\mathbf{R}^N} G(u_0) dx + C_1 R_1^{N-1}.$$

Here we used Lemma 5.7.1 (b). Thus for sufficiently large $R_1 \ge R_0$, we have (5.7.4). Using Lemma 5.7.1 (c), we also get (5.7.3).

(c) As in the proof of (b), for $T_1 \ge R_1$, we have from Lemma 5.7.1 (a)

$$I((1-s)u_0(L_0r/R_1) + \eta(R_1, T_1; r)) = I((1-s)u_0(L_0r/R_1)) + I(\eta(R_1, T_1; r))$$

$$\leq I((1-s)u_0(L_0r/R_1)) - C_0T_1^N.$$

Taking $T_1 \ge R_1$ large, we have (5.7.5).

Proof of Lemma 5.6.1. We choose $R_1 \ge R_0$ and $T_1 \ge R_1$ as in Lemma 5.7.2. We can see $\gamma_1([L_0, R_1]), \gamma_2([0, 1]), \gamma_3([0, T_1]), \gamma_4([0, 1]) \subset O$ and thus u_0 and $\eta(R_1, T_1; r)$ are connected by a continuous path in O. We can also join u_1 and $\eta(R_1, T_1; r)$ in O in a similar way. Thus Lemma 5.6.1 is proved.

Chapter 6

Existence of positive and infinitely many solutions: inhomogeneous case

6.1 Introduction

In this chapter, we are concerned with the following nonlinear scalar field equation:

(6.1.1)
$$\begin{cases} -\Delta u = g(|x|, u) & \text{in } \Omega, \\ u \in H^1(\Omega). \end{cases}$$

Here $\Omega \subset \mathbf{R}^N$ is either the whole space $\Omega = \mathbf{R}^N$ or the exterior domain of the ball $B_R(0)$ with radius R > 0 ($\Omega = \{x \in \mathbf{R}^N : |x| > R\}$) and the function $g(r, s) : [R, \infty) \times \mathbf{R} \to \mathbf{R}$ is continuous in both variables and odd with respect to $s \in \mathbf{R}$. In the case where Ω is the exterior domain, we consider (6.1.1) under the homogeneous Dirichlet or Neumann boundary condition:

(D)
$$u = 0 \text{ on } \partial \Omega$$

(N)
$$\frac{\partial u}{\partial \nu} = 0 \text{ on } \partial \Omega,$$

where ν is the outward normal vector of $\partial \Omega$. Namely, we consider the following equations:

$$\begin{aligned} &(P_{\mathbf{R}^N}) & -\Delta u = g(|x|, u) & \text{in } \mathbf{R}^N, \quad u \in H^1(\mathbf{R}^N). \\ &(P_D) & -\Delta u = g(|x|, u) & \text{in } \{|x| > R\}, \quad u = 0 \quad \text{on } |x| = R, \quad u \in H^1(\{|x| > R\}). \\ &(P_N) & -\Delta u = g(|x|, u) & \text{in } \{|x| > R\}, \quad \frac{\partial u}{\partial \nu} = 0 \quad \text{on } |x| = R, \quad u \in H^1(\{|x| > R\}) \end{aligned}$$

When $\Omega = \mathbf{R}^N$ and g(r,s) does not depend on r, that is g(r,s) = g(s), $(P_{\mathbf{R}^N})$ has been studied by many researchers. For example, we refer to [14, 15, 16, 18, 19, 48, 55, 95] and references therein.

On the other hand, when g(r, s) depends on r in a monotone decreasing way, Li and Li [61] and Li [62] studied $(P_{\mathbf{R}^N})$ and (P_D) . They showed the existence of a radial positive solution and infinitely many radial possibly sign changing solutions for a suitable class of nonlinearities (see Remark 6.2.3 for a precise statement). One of the aims in this chapter is to deal with the Neumann boundary problem (P_N) as well as $(P_{\mathbf{R}^N})$ and (P_D) , and give a generalization of the results of [61, 62]. Especially, we relax the conditions on the behavior of g(r, s) near s = 0. In [61, 62], they assumed $\lim_{s\to 0} g(r, s)/s = -1$ uniformly with respect to r (see Remark 6.2.3). However, our main results (Theorems 6.2.1 and 6.2.2 below) enable us to deal with the following case: $-\infty < \liminf_{s\to 0} \inf_{r\geq R} g(r, s)/s \le \limsup_{s\to 0} \sup_{r\geq R} g(r, s)/s < 0$. Therefore we can treat the following example: $-\Delta u = -(V(|x|) + a(|x|) \sin^2(1/u))u + b(|x|)f(u)$ in Ω where V(r), a(r), b(r) are monotone functions and f(s) is superlinear near s = 0.

Another aim of this chapter is to deal with nonlinear Schrödinger type problems without a monotonicity assumption on g(r, s) with respect to r. Namely, setting $g(r, s) := -V(r)s + \tilde{g}(s)$ in (6.1.1), we consider the following equation:

(6.1.2)
$$\begin{cases} -\Delta u + V(|x|)u = \tilde{g}(u) & \text{in } \Omega, \\ u \in H^1(\Omega). \end{cases}$$

When $\Omega = \mathbf{R}^N$, Azzollini and Pomponio [7] studied (6.1.2) and obtained the existence of at least one radial positive solution. We give an extension of their result to the exterior problems (P_D) and (P_N) . Moreover, we show the existence of infinitely many solutions. See Theorem 6.2.4 for a precise statement (see also Remark 6.2.5).

We will prove our theorems by variational methods and use the monotonicity method due to Struwe [96], and developed by Jeanjean [54] and Rabier [91]. With the monotonicity method, a newly developed Pohozaev type inequality (see Propositions 6.5.5 and 6.5.7) will play important roles in our argument.

This chapter is organized as follows. We state our main results in section 6.2. In section 6.3, we introduce an auxiliary functional J and prepare some lemmas. Proofs of lemmas in section 6.3 will be given in section 6.6. In section 6.4, we define minimax values based on the symmetric mountain pass arguments. Section 6.5 is devoted to prove Theorems 6.2.1, 6.2.2 and 6.2.4. In section 6.6, we prove some lemmas.

6.2 Statement of main results

In this section, we state our main results of this chapter.

6.2.1 Results for the equation (6.1.1)

First we consider the equation (6.1.1). We assume that $g(r, s) : [R, \infty) \times \mathbf{R} \to \mathbf{R}$ satisfies the following conditions. In what follows, we regard R = 0 if $\Omega = \mathbf{R}^N$.

(6-g1)
$$g \in C([R,\infty) \times \mathbf{R}, \mathbf{R})$$
 and $g(r, -s) = -g(r, s)$ for all $r \ge R$ and $s \in \mathbf{R}$.

(6-g2) If $R \le r_1 \le r_2 < \infty$ and $s \ge 0$, then $g(r_1, s) \le g(r_2, s)$.

(6-g3) As $r \to \infty$, $g(r,s) \to g_{\infty}(s)$ in $L^{\infty}_{\text{loc}}(\mathbf{R})$.

(6–g4) There exists an $m_1 > 0$ such that

$$\infty < \liminf_{s \to 0} \inf_{r \ge R} \frac{g(r, s)}{s} \le \limsup_{s \to 0} \sup_{r \ge R} \frac{g(r, s)}{s} \le -m_1$$

(6–g5) For $N \ge 3$,

$$\lim_{s \to \infty} \sup_{r > R} \frac{|g(r, s)|}{s^{2^* - 1}} = 0 \quad \text{where} \quad 2^* = 2N/(N - 2).$$

For N = 2,

$$\lim_{s \to \infty} \sup_{r \ge R} \frac{|g(r,s)|}{\exp(\alpha s^2)} = 0 \quad \text{for any } \alpha > 0.$$

(6–g6) There exist $\zeta_0 > 0$ and $R \ge R_0$ such that

$$\inf_{r \ge R_0} G(r, \zeta_0) > 0 \quad \text{where} \quad G(r, s) := \int_0^s g(r, \tau) d\tau$$

Except for (6–g3) and (6–g4), the above conditions are same to the ones in [61, 62]. As for (6–g4), this type of condition is used in [15, 16, 48, 95] when g(r, s) does not depend on r, i.e., g(r, s) = g(s) (cf. see also (\tilde{g} 2) below). We remark that in [61, 62], they suppose $\lim_{s\to 0} g(r, s)/s = -1$ uniformly with respect to r, which is stronger than (6–g4).

For the Neumann problem (P_N) , in addition to (6-g1)-(6-g6), we assume

 $(6-g7) -\infty < \inf_{s \in \mathbf{R}} G(R,s).$

Our main results are as follows. First we state a result for (P_N) .

Theorem 6.2.1. Suppose that $\Omega = \{|x| > R\}$ and (6–g1)–(6–g7) are satisfied. Then (P_N) has at least one radial positive solution and infinitely many radial possibly sign changing solutions.

For $(P_{\mathbf{R}^N})$ and (P_D) , we assume (6-g1)-(6-g6) and we do not need (6-g7).

Theorem 6.2.2. Suppose that $\Omega = \mathbf{R}^N$ (resp. $\Omega = \{|x| > R\}$) and (6–g1)–(6–g6) are satisfied. Then $(P_{\mathbf{R}^N})$ (resp. (P_D)) has at least one radial positive solution and infinitely many radial possibly sign changing solutions.

Remark 6.2.3. In [61, 62], in addition to (6–g1), (6–g2), (6–g4)–(6–g6), they suppose that the function g has a form g(r,s) = -s + f(r,s) where f(r,s) = o(1) uniformly with respect to r as $s \to 0$ (cf. (6–g4)). Under these conditions, they proved the existence of one radial positive solution and infinitely many radial possibly sign changing solutions to $(P_{\mathbf{R}^N})$ and (P_D) . However Theorem 6.2.2 enables us to deal with the following type of equations: $-\Delta u = -(V(|x|) + a(|x|)\sin^2(1/u))u + b(|x|)f(u)$ where V, a, b are monotone functions and f(s) is superlinear near s = 0.

6.2.2 Results for the equation (6.1.2)

Next we consider (6.1.2) for $N \ge 3$. We write $g(r,s) = -V(r)s + \tilde{g}(s)$ and assume the following conditions:

- ($\tilde{g}1$) $\tilde{g} \in C(\mathbf{R}, \mathbf{R})$ and $\tilde{g}(-s) = -\tilde{g}(s)$ for all $s \in \mathbf{R}$.
- (\tilde{g}_2) There exists an $m_1 > 0$ such that $-\infty < \liminf_{s \to 0} \frac{\tilde{g}(s)}{s} \le \limsup_{s \to 0} \frac{\tilde{g}(s)}{s} \le -\tilde{m}_1$.
- (§3) $\limsup_{s \to \infty} \frac{\tilde{g}(s)}{s^{2^* 1}} \le 0.$

(§4) There exists a $\tilde{\zeta}_0 > 0$ such that $\tilde{G}(\tilde{\zeta}_0) > 0$ where $\tilde{G}(s) := \int_0^s \tilde{g}(\tau) d\tau$.

(
$$ilde{g}5$$
) $-\infty < \inf_{s \in \mathbf{R}} \left(-\frac{1}{2} V(R) s^2 + \tilde{G}(s) \right).$

The conditions ($\tilde{g}1$)-($\tilde{g}4$) are same to the ones in [15, 16, 48]. The condition ($\tilde{g}5$) corresponds to (6–g7) above and is only needed for (P_N). For V, we assume the following:

(6–V1)
$$V \in C^1([R,\infty))$$
 and $V(r) \ge 0$ for all $r \ge R$.

(6–V2)
$$\lim V(r) = 0.$$

(6–V3) $||(x \cdot \nabla V(|x|))^+||_{L^{\frac{N}{2}}(|x|>R)} < 2S_N$ where

$$(x \cdot \nabla V(|x|))^{+} := \max\{0, x \cdot \nabla V(|x|)\} \text{ and } S_{N} := \inf_{u \in H^{1}(\mathbf{R}^{N}) \setminus \{0\}} \frac{\|\nabla u\|_{L^{2}(\mathbf{R}^{N})}^{2}}{\|u\|_{L^{2^{*}}(\mathbf{R}^{N})}^{2}}$$

When $\Omega = \mathbf{R}^N$, the above conditions ($\tilde{g}1$)–($\tilde{g}4$) and (6–V1)–(6–V3) are same to the ones in [7]. Next we give a remark about (6–V3). If $g(r,s) = -V(r)s + \tilde{g}(s)$ satisfies (6–g2), then we can see $x \cdot \nabla V(|x|) \leq 0$, which implies (6–V3). Therefore, we can relax the monotonicity condition (6–g2) by (6–V3) for the equation (6.1.2).

Now we state a result for (6.1.2).

Theorem 6.2.4. Suppose that $N \ge 3$ and $g(r,s) = -V(r)s + \tilde{g}(s)$ satisfies $(\tilde{g}1)-(\tilde{g}4)$ and (6-V1)-(6-V3). Then the following hold:

- (i) $(P_{\mathbf{R}^N})$ (resp. (P_D)) admits at least one radial positive solution and infinitely many possibly radial sign-changing solutions.
- (ii) Assume (§5) in addition to (§1)-(§4) and (6-V1)-(6-V3). Then (P_N) admits at least one radial positive solution and infinitely many possibly radial sign changing solutions.

Remark 6.2.5. In [7], they showed the existence of one radially symmetric positive solution to $(P_{\mathbf{R}^N})$ with $g(r,s) = -V(r)s + \tilde{g}(s)$ under the conditions (\tilde{g}_1)-(\tilde{g}_4) and (6-V1)-(6-V3).

In the following, we give an idea of proofs of Theorems 6.2.1, 6.2.2, 6.2.4.

We will prove Theorems 6.2.1, 6.2.2 and 6.2.4 by variational methods, and find critical points of

$$I(u) := \frac{1}{2} \int_{\Omega} |\nabla u|^2 dx - \int_{\Omega} G(|x|, u) dx.$$

One of difficulties is to show the boundedness of Palais–Smale (for short (PS)) sequences.

In [61, 62], they introduced the following parametrized functional in order to obtain bounded (PS) sequences: (cf. Remark 6.2.3)

$$\hat{I}_{\lambda}(u) := \frac{1}{2} \int_{\Omega} |\nabla u|^2 + u^2 dx - \int_{\Omega} F(|x|, u) dx - \lambda \int_{\Omega} q(|x|) B(u) dx, \qquad \lambda \in [0, 1].$$

Here $F(r,s) := \int_0^s f(r,t)dt$, and B(s) and q(r) are suitable penalty functions. The virtue of their penalty functions is that \hat{I}_{λ} satisfies the (PS) condition. However, the construction is rather complicated.

In our proofs, we consider another parametrized functional to obtain bounded (PS) sequences:

$$I_{\lambda}(u) := \frac{1}{2} \int_{\Omega} |\nabla u|^2 dx - \int_{\Omega} G(|x|, u) dx - \lambda \int_{\Omega} H(u) dx \qquad \lambda \in [0, 1]$$

Here H(s) is also a penalty function which is different from B(s) in \hat{I}_{λ} and we can construct the function H(s) in a simply way (see the definition of H(s) in section 6.3). To obtain critical points of I, we will apply the monotonicity method to I_{λ} . Here, we apply a version of Rabier [91] (see Propositions 6.5.1 and 6.5.2), and obtain sequences (λ_k) and (u_k) such that

$$\lambda_k \to 0, \quad -\Delta u_k = g(|x|, u_k) + \lambda_k h(u_k) \quad \text{in } \Omega,$$

where h(s) := H'(s). To show that (u_k) has a strongly convergent subsequence, we use the Pohozaev type inequality (6.5.2), (6.5.8), (6.5.9). Here we remark that in [61, 62] they used the Pohozaev Identity (for instance, see (6.5.7), (6.5.10), (6.5.11)) which includes the term $x \cdot \nabla G(|x|, u)$ and they need to approximate g(r, s) with a function of class C^1 in r. However, in this paper, we introduce a new Pohozaev type inequality, which enables us to argue without introducing approximations.

Our proofs can also be applied for the equation (6.1.2), namely $g(r, s) = -V(r)s + \tilde{g}(s)$ in (6.1.1). By virtue of our proofs of Theorems 6.2.1 and 6.2.2, we will be able to show that not only $(P_{\mathbf{R}^N})$ but also (P_D) and (P_N) admit at least one radial positive solution and infinitely many radial possibly sign changing solutions under the conditions $(\tilde{g}1)-(\tilde{g}4)$, (6-V1)-(6-V3) or $(\tilde{g}1)-(\tilde{g}5)$, (6-V1)-(6-V3).

6.3 Preliminaries

In this section, we introduce an auxiliary functional J and state some lemmas. A Proof of Lemma 6.3.2 will be given in section 6.6.

First, we remark that when we consider (P_D) or (P_N) under the assumptions of Theorems 6.2.1, 6.2.2 or 6.2.4 we may assume R = 1 without loss of generality. Indeed, set $\Omega = \{|x| > R\}, v(x) := u(Rx) \text{ and } g_R(r,s) := R^2 g(Rr,s).$ Then (6.1.1) is equivalent to the following equation:

$$-\Delta v = g_R(r, v)$$
 in $\{|x| > 1\}$.

Moreover, it is easily seen that g satisfies (6-g1)-(6-g7) in $\{|x| > R\}$ if and only if g_R satisfies (6-g1)-(6-g7) in $\{|x| > 1\}$. In the case where $g(r,s) = -V(r)s + \tilde{g}(s)$, set $V_R(r) := R^2V(Rr)$ and $\tilde{g}_R(s) := R^2\tilde{g}(s)$. Then it is also clear that V and \tilde{g} satisfy (6-V1)-(6-V3), $(\tilde{g}1)-(\tilde{g}5)$ in $\{|x| > R\}$ if and only if V_R and \tilde{g}_R satisfy (6-V1)-(6-V3), $(\tilde{g}1)-(\tilde{g}5)$ in $\{|x| > R\}$ if and only if V_R and \tilde{g}_R satisfy (6-V1)-(6-V3), $(\tilde{g}1)-(\tilde{g}5)$ in $\{|x| > 1\}$. Therefore to prove Theorems 6.2.1, 6.2.2 and 6.2.4, we may R = 1 without loss of generality.

Hereafter we mainly consider (P_N) and let $\Omega = \{x \in \mathbf{R}^N : |x| > 1\}$. Furthermore we assume the following condition in this section:

 (\mathcal{H}_1) The conditions (6–g1) and (6–g3)–(6–g5) are satisfied.

In order to obtain radial solutions, we consider the following function space:

 $E := H^1_r(\Omega) = \{ u \in H^1(\Omega) : u \text{ is a radial function } \}.$

The following properties hold (For (i) and (ii), see Berestycki and Lions [15], Strauss [95]):

(i) There exists a C > 0 such that for all $u \in E$ and $|x| \ge 1$,

(6.3.1)
$$|u(x)| \le C|x|^{-\frac{N-1}{2}} ||u||_{H^1(\Omega)}.$$

- (ii) The embedding $E \subset L^q(\Omega)$ is continuous for $2 \le q \le 2^*$ if $N \ge 3$ and $2 \le q < \infty$ if N = 2 and it is compact for $2 < q < 2^*$ if $N \ge 3$ and $2 < q < \infty$ if N = 2.
- (iii) For each $s \in (0, 1]$, we define the extension operator $T_s : H^1_r(\{|x| > s\}) \to H^1_r(\mathbf{R}^N)$ by

(6.3.2)
$$(T_s u)(x) := (T_s u)(|x|) = \begin{cases} u(|x|) & \text{if } |x| \ge s, \\ u(2s - |x|) & \text{if } |x| < s. \end{cases}$$

Then, for each $s \in (0, 1]$ and $u \in H^1_r(\{|x| > s\})$, it holds that

(6.3.3)
$$||T_s u||_{L^2(\mathbf{R}^N)} \le \sqrt{2} ||u||_{L^2(\{|x|>s\})}, \quad ||\nabla T_s u||_{L^2(\mathbf{R}^N)} \le \sqrt{2} ||\nabla u||_{L^2(\{|x|>s\})}.$$

Using (6.3.3), we have the following Sobolev inequality holds for $N \ge 3$:

 $(6.3.4) \quad \|u\|_{L^{2^*}(\{|x|>s\})} \le C \|\nabla u\|_{L^2(\{|x|>s\})} \quad \text{for all } u \in H^1_r(\{|x|>s\}), \ s \in (0,1].$

We define the following functional:

$$I(u) := \frac{1}{2} \int_{\Omega} |\nabla u|^2 dx - \int_{\Omega} G(|x|, u) dx : E \to \mathbf{R}.$$

We note that $I \in C^1(E, \mathbf{R})$ under the condition (\mathcal{H}_1) and the functional I corresponds to (P_N) . So, we will find critical points of I.

Following chapter 5, we prepare a penalty function to construct an auxiliary functional. For $s \ge 0$, we define f(s) and h(s) as follows:

$$f(s) := \max\left\{0, \frac{1}{2}m_1s + \sup_{r \ge 1} g(r, s)\right\}, \quad h(s) := s^p \sup_{0 < \tau \le s} \frac{f(\tau)}{\tau^p}$$

Here m_1 is a constant appearing in (6–g4) and p is a positive number satisfying $1 if <math>N \ge 3$ and 1 if <math>N = 2. Note that by (6–g3) and (6–g4), f and h are well–defined. We extend h as an odd function on \mathbf{R} and set

$$H(s) := \int_0^s h(t)dt$$

Then h and H have the following properties.

Lemma 6.3.1 (cf. Lemma 5.2.1 and Corollary 5.2.2 in Chapter 5). *The following properties hold*:

- (i) $h \in C(\mathbf{R}), 0 \le h(s)$ and h(-s) = -h(s) for all $s \in [0, \infty)$.
- (ii) There exists an $s_0 > 0$ such that h = H = 0 on $[-s_0, s_0]$.
- (iii) For all $s \in \mathbf{R}$, it follows that

$$\frac{1}{2}m_1s^2 + \sup_{r \ge 1} g(r,s)s \le h(s)s, \qquad \frac{1}{4}m_1s^2 + \sup_{r \ge 1} G(r,s) \le H(s).$$

(iv) It holds that

$$\begin{split} &\lim_{s\to\infty}\frac{h(s)}{\exp(\alpha s^2)}=0 \quad \text{for all } \alpha>0 \quad \text{if } N=2,\\ &\lim_{s\to\infty}\frac{h(s)}{s^{2^*-1}}=0 \quad \text{if } N\geq 3. \end{split}$$

(v) The function h satisfies a global Ambrosetti–Rabinowitz condition:

$$0 \le (p+1)H(s) \le h(s)s$$
 for all $s \in \mathbf{R}$.

Here p appears in the definition of h.

Since we can prove Lemma 6.3.1 as in Lemma 5.2.1 and Corollary 5.2.2 in Chapter 5, we omit a proof.

Next we rewrite the functional I as follows:

$$I(u) = \frac{1}{2} \|\nabla u\|_{L^2}^2 - \int_{\Omega} G(|x|, u) dx = \frac{1}{2} \|u\|_E^2 - \int_{\Omega} \frac{m_1}{4} u^2 + G(|x|, u) dx$$

where

$$||u||_E^2 := ||\nabla u||_{L^2}^2 + \frac{m_1}{2} ||u||_{L^2}^2.$$

We remark that $\|\cdot\|_E$ and the standard H^1 -norm are equivalent.

Next, we define a parametrized functional I_{λ} ($\lambda \in [0, 1]$) and an auxiliary functional J which gives us lower bounds of minimax values $b_n(\lambda)$ defined in section 6.4:

$$\begin{split} I_{\lambda}(u) &:= \frac{1}{2} \|u\|_{E}^{2} - \int_{\Omega} \frac{m_{1}}{4} u^{2} + G(|x|, u) + \lambda H(u) dx \in C^{1}(E, \mathbf{R}), \\ J(u) &:= \frac{1}{2} \|u\|_{E}^{2} - 2 \int_{\Omega} H(u) dx \in C^{1}(E, \mathbf{R}), \end{split}$$

Note that if $\lambda = 0$, then $I_0(u) = I(u)$. Furthermore, by Lemma 6.3.1, I_{λ} and J satisfy the following: for any $0 \le \lambda_1 \le \lambda_2 \le 1$ and $u \in E$,

(6.3.5)
$$J(u) \le I_1(u) \le I_{\lambda_2}(u) \le I_{\lambda_1}(u) \le I_0(u).$$

Now we state properties of I_{λ} and J. Similar properties are obtained in [7, 48]

Lemma 6.3.2. (cf. Lemma 3.5 in [7], Lemmas 5.2.3, 5.2.5, Proposition 5.5.3 in Chapter 5) Set $K(u) := \int_{\Omega} H(u) dx$. Then,

- (i) The maps $K: E \to \mathbf{R}$ and $K': E \to E^*$ are weakly continuous.
- (ii) Any bounded (PS) sequence $(u_k) \subset E$ for I_{λ} has a strongly convergent subsequence.
- (iii) The functional J satisfies the (PS) condition.

6.4 Minimax arguments

In this section, we define minimax values $b_n(\lambda)$ of I_{λ} based on the arguments of symmetric mountain pass theorem (cf. [48] and Rabinowitz [92]). In this section, we assume the following conditions:

 (\mathcal{H}_2) The conditions (6–g1) and (6–g3)–(6–g6) are satisfied.

First of all, we prove that I_{λ} and J have a symmetric mountain pass geometry under the condition (\mathcal{H}_2) . More precisely, we have

Lemma 6.4.1. The following hold:

(i) There exist $\delta > 0$ and $\rho > 0$ such that

 $0 < \delta \le J(u) \quad \text{for all } u \in E \text{ with } \|u\|_E = \rho, \\ 0 \le J(u) \quad \text{for all } u \in E \text{ with } \|u\|_E \le \rho.$

(ii) For each $n \in \mathbf{N}$, there exists an odd continuous map $\gamma_n : S^{n-1} \to H^1_{0,r}(\Omega)$ such that

$$I_0(\gamma_n(\sigma)) < 0$$
 for all $\sigma \in S^{n-1}$.

Here

$$S^{n-1} := \{ \sigma = (\sigma_1, \dots, \sigma_n) \in \mathbf{R}^n : |\sigma| = 1 \}, \quad H^1_{0,r}(\Omega) := \{ u \in E : u(1) = 0 \}.$$

Remark 6.4.2. By (6.3.5), we see that I_{λ} and J have a symmetric mountain pass geometry.

Proof. We only prove (i). (ii) will be proven in section 6.6.

First, we show for $N \ge 3$. By Lemma 6.3.1, there exists a C > 0 such that

$$H(s) \le C|s|^{2^*}$$
 for all $s \in \mathbf{R}$.

Using Sobolev's embedding, we obtain

$$J(u) \ge ||u||_E^2 - C||u||_{L^{2^*}(\Omega)}^{2^*} \ge ||u||_E^2(1 - C||u||_E^{2^{*-2}}).$$

Thus (i) holds for $N \geq 3$.

Next we consider the case N = 2. By Lemma 6.3.1, there exists a $C_1 > 0$ such that

$$H(s) \le C_1 \Phi(s^2/2)$$
 where $\Phi(s) := \exp(s) - 1 - s$.

By Lemma 6.6.2 (iii), we have

$$\int_{\Omega} H(u)dx \le C_2 \|u\|_E^4 \quad \text{for all } u \in E \text{ with } \|u\|_E \le 1$$

Thus it follows that if $||u||_E \leq 1$, then

$$J(u) \ge \|u\|_E^2 - C_2 \|u\|_E^4$$

which completes the proof of (i).

Next, we define minimax values of I_{λ} and J using mappings (γ_n) appearing in Lemma 6.4.1.

Definition 6.4.3. For each $n \in \mathbb{N}$ and $\lambda \in [0, 1]$, we define $b_n(\lambda)$ and c_n as follows:

$$b_n(\lambda) := \inf_{\gamma \in \Gamma_n} \max_{\sigma \in D_n} I_\lambda(\gamma(\sigma)), \quad c_n := \inf_{\gamma \in \Gamma_n} \max_{\sigma \in D_n} J(\gamma(\sigma)),$$

where

$$D_n := \{ \sigma \in \mathbf{R}^n : |\sigma| \le 1 \}, \quad \Gamma_n := \{ \gamma \in C(D_n, E) : \gamma \text{ is odd and } \gamma = \gamma_n \text{ on } S^{n-1} \}$$

The values $b_n(\lambda)$ and c_n have the following properties.

Lemma 6.4.4. The following properties hold:

- (i) $\Gamma_n \neq \emptyset$ for all $n \in \mathbf{N}$.
- (ii) For each $0 \le \lambda_1 \le \lambda_2 \le 1$, $b_n(\lambda_2) \le b_n(\lambda_1)$.
- (iii) For each $n \in \mathbf{N}$ and $\lambda \in [0, 1]$, it holds that $0 < \delta \le c_n \le b_n(\lambda)$ where δ appears in Lemma 6.4.1 (i).

Proof. (i) We define $\tilde{\gamma}_n$ as follows: for $\sigma \in D^n$, $\tilde{\gamma}_n(\sigma) := |\sigma|\gamma_n(\sigma/|\sigma|)$. Then $\tilde{\gamma}_n \in \Gamma_n$. (ii) By (6.3.5), (ii) holds. (iii) By (6.3.5) and (i), it holds $c_n \leq b_n(\lambda)$ for each $\lambda \in [0, 1]$. The property $\delta \leq c_n$ follows from the fact

$$\{u \in E : \|u\|_E = \rho\} \cap \gamma(D_n) \neq \emptyset \quad \text{for all } \gamma \in \Gamma_n$$

Since J satisfies the (PS) condition by Lemma 6.3.2, we can show the following lemma by Theorem 7.1.1 in Chapter 7.

Lemma 6.4.5. (c.f. Lemma 5.3.2 in Chapter 5) The following hold:

- (i) The values c_n is a critical value of J.
- (ii) As $n \to \infty$, $c_n \to \infty$.

6.5 proofs of Theorems 6.2.1, 6.2.2 and 6.2.4

In this section, we prove Theorems 6.2.1, 6.2.2 and 6.2.4 by using the monotonicity method and the Pohozaev type inequality (Propositions 6.5.5 and 6.5.7).

6.5.1 Monotonicity method

First, we will recall Rabier's result [91]. Let $(X, \|\cdot\|)$ be a Banach space and $\mathcal{A} : X \to \mathbf{R}$, $\mathcal{B} : [0,1] \times X \to \mathbf{R}$ be C^1 functionals and set $\mathcal{I}_{\lambda}(u) := \mathcal{A}(u) - \mathcal{B}(\lambda, u)$. We assume that \mathcal{A} and \mathcal{B} satisfy the following:

(BPS1) $\mathcal{B}(\cdot, u)$ is nondecreasing on [0, 1] for every $u \in X$.

(BPS2)
$$\lim_{\mathcal{B}(\lambda,u)\to\infty}\frac{\partial\mathcal{B}}{\partial\lambda}(\lambda,u)=\infty$$

(BPS3) $\lim_{\|u\|\to\infty} \mathcal{A}(u) = \infty.$

Moreover, we suppose that there exist $e_1, e_2 \in X$ such that

(BSP4) max{ $\mathcal{I}_{\lambda}(e_1), \mathcal{I}_{\lambda}(e_2)$ } < c_{λ} for all $\lambda \in [0, 1]$.

Here

$$\begin{split} c_{\lambda} &:= \inf_{\gamma \in \Gamma^*} \max_{0 \le t \le 1} \mathcal{I}_{\lambda}(\gamma(t)), \\ \Gamma^* &:= \{ \gamma \in C([0,1], X) \mid \gamma(0) = e_1, \ \gamma(1) = e_2 \}. \end{split}$$

Then the following proposition holds.

Proposition 6.5.1 (Rabier [91]). Under the conditions (BPS1)–(BPS4), for almost every $\lambda \in [0, 1], \mathcal{I}_{\lambda}$ has a bounded (PS) sequence at level c_{λ} .

We will apply the above proposition for the functional which satisfies the symmetric mountain pass structure. Assume the following conditions in addition to (BSP1)–(BSP3):

(BPS5) A(-u) = A(u) and $B(\lambda, -u) = B(\lambda, u)$ for all $u \in X$ and $\lambda \in [0, 1]$.

(BPS6) For each $n \in \mathbf{N}$, there exists a continuous odd map $\gamma_n^* : S^{n-1} \to H^1_r(\Omega)$ such that

$$\max_{\sigma \in S^{n-1}} \mathcal{I}_{\lambda}(\gamma_n^*(\sigma)) < d_n(\lambda)$$

Here

$$d_n(\lambda) := \inf_{\gamma \in \Gamma_n^*} \max_{\sigma \in D^n} \mathcal{I}_{\lambda}(\gamma(\sigma)),$$

$$\Gamma_n^* := \{ \gamma \in C(D_n, E) : \gamma \text{ is odd and } \gamma = \gamma_n^* \text{ on } S^{n-1} \}$$

The following proposition holds from the arguments in [91].

Proposition 6.5.2. Suppose (BPS1)–(BPS3), (BPS5)–(BPS6). Then, for almost every $\lambda \in [0, 1]$, there exists a bounded (PS) sequence of \mathcal{I}_{λ} at level $d_n(\lambda)$ for all $n \in \mathbf{N}$.

Next, we show that we can apply Proposition 6.5.2 for I_{λ} to obtain a bounded (PS) sequence of I_{λ} .

Lemma 6.5.3. Under the assumption (\mathcal{H}_2) , for almost every $\lambda \in [0, 1]$, I_{λ} has a bounded (PS) sequence at level $b_n(\lambda)$ for all $n \in \mathbf{N}$.

Proof. Set $X := E, \gamma_n^* := \gamma_n$,

$$\mathcal{A}(u) := \frac{1}{2} \|u\|_E^2, \quad \mathcal{B}(\lambda, u) := \int_{\Omega} \frac{m_1}{4} u^2 + G(|x|, u) + \lambda H(u) dx.$$

It is easily seen that (BPS1), (BPS3) and (BPS5) are satisfied. Moreover, by Lemmas 6.4.1 and 6.4.4, (BPS6) holds. As to (BPS2), by Lemma 6.3.1, we have

$$\mathcal{B}(\lambda, u) \le (1+\lambda) \int_{\Omega} H(u) dx.$$

On the other hand, it follows that

$$\frac{\partial \mathcal{B}}{\partial \lambda}(\lambda, u) = \int_{\Omega} H(u) dx,$$

which implies (BPS2). Then by Proposition 6.5.2, for almost every $\lambda \in [0, 1]$, I_{λ} has a bounded (PS) sequence at level $b_n(\lambda)$ for all $n \in \mathbb{N}$.

Combining Lemmas 6.3.2 and 6.5.3, we have the following:

Proposition 6.5.4. Suppose that (\mathcal{H}_2) is satisfied. Then for almost every $\lambda \in (0, 1]$, there is a critical point $u_{\lambda,n} \in E$ such that $I_{\lambda}(u_{\lambda,n}) = b_n(\lambda)$ for all $n \in \mathbf{N}$.

From Proposition 6.5.4, it follows that for each $n \in \mathbf{N}$, there exist $(\lambda_{n,k}) \subset [0,1]$, $(u_{n,k}) \subset E$ such that $\lambda_{n,k} \to 0$ and

(6.5.1)
$$I_{\lambda_{n,k}}(u_{n,k}) = b_n(\lambda_{n,k}), \quad I'_{\lambda_{n,k}}(u_{n,k}) = 0.$$

6.5.2 Pohozaev type inequality

To show that $(u_{n,k})$ in (6.5.1) is bounded, we introduce the following Pohozaev type inequality.

Proposition 6.5.5. Assume that the conditions (6–g1)–(6–g6) are satisfied. Let $u_N \in E$ be a solution of

$$-\Delta u = g(|x|, u) + \lambda h(u)$$
 in Ω , $\frac{\partial u_N}{\partial \nu} = 0$ on $\partial \Omega$

where ν is the outward normal vector of $\partial\Omega$. Then u_N satisfies the following:

(6.5.2)
$$\frac{N-2}{2} \|\nabla u_N\|_{L^2}^2 - N \int_{\Omega} \hat{G}_{\lambda}(|x|, u_N) dx \ge \int_{\partial \Omega} \hat{G}_{\lambda}(|x|, u_N) dS.$$

Here $\hat{G}_{\lambda}(|x|, s) := G(|x|, s) + \lambda H(s).$

Proof. Note that under the conditions (6-g1)-(6-g6), u_N has an exponential decay:

$$|u_N(r)| + |u'_N(r)| + |u''_N(r)| \le C_1 \exp(-C_2 r)$$
 for all $r \ge 1$.

Therefore $x \cdot \nabla u_N \in H^1(\Omega)$ and the curve $\eta(t) := u_N(tx) : [1, 2] \to H^1(\Omega)$ is of class C^1 . Since $I'_{\lambda}(u_N) = 0$, we have

(6.5.3)
$$\frac{d}{dt}I_{\lambda}(\eta(t))\big|_{t=1} = I_{\lambda}'(u_N(x))(x \cdot \nabla u_N(x)) = 0.$$

On the other hand, it holds that (6.5.4)

$$I_{\lambda}(\eta(t)) = \frac{t^{-N+2}}{2} \int_{|x| \ge t} |\nabla u_N(x)|^2 dx - t^{-N} \int_{|x| \ge t} G\left(\frac{|x|}{t}, u_N(|x|)\right) + \lambda H(u_N(|x|)) dx.$$

By (6-g2), it follows that (6.5.5)

$$I_{\lambda}(\eta(t)) \ge \hat{I}_{\lambda}(t) := \frac{t^{-N+2}}{2} \int_{|x|\ge t} |\nabla u_N(x)|^2 dx - t^{-N} \int_{|x|\ge t} G(|x|, u_N(|x|)) + \lambda H(u_N(|x|)) dx.$$

Noting that $I(\eta(1)) = \hat{I}_{\lambda}(1)$, from (6.5.5), we infer

(6.5.6)
$$\frac{I_{\lambda}(\eta(t)) - I_{\lambda}(\eta(1))}{t - 1} \ge \frac{\hat{I}_{\lambda}(t) - \hat{I}_{\lambda}(1)}{t - 1} \quad \text{for all } t \in (1, 2]$$

By (6.5.3),

$$\frac{I_{\lambda}(\eta(t)) - I_{\lambda}(\eta(1))}{t - 1} \to 0 \quad \text{as } t \to 1 + 0.$$

On the other hand, since $\partial u_N/\partial \nu = 0$ on $\partial \Omega$, it is easily seen that

$$\frac{\hat{I}_{\lambda}(t) - \hat{I}_{\lambda}(1)}{t - 1}$$

$$\rightarrow -\frac{N - 2}{2} \|\nabla u_N\|_{L^2}^2 + N \int_{\Omega} G(|x|, u_N) + \lambda H(u_N) dx + \int_{\partial \Omega} G(|x|, u_N) + \lambda H(u_N) dS$$

as $t \to 1 + 0$. Thus, from (6.5.6), we conclude that

$$\int_{\partial\Omega} G(|x|, u_N) + \lambda H(u_N) dS \le \frac{N-2}{2} \|\nabla u_N\|_{L^2}^2 - N \int_{\Omega} G(|x|, u_N) + \lambda H(u_N) dx.$$

Remark 6.5.6. If we suppose that g is of class C^1 with respect to r in addition to (6–g1)–(6–g6), then the Pohozaev Identity holds: (6.5.7)

$$\frac{N-2}{2} \|\nabla u_N\|_{L^2(\Omega)}^2 - N \int_{\Omega} \hat{G}_{\lambda}(|x|, u_N) dx = \int_{\Omega} x \cdot \nabla G(|x|, u_N) dx + \int_{\partial \Omega} \hat{G}_{\lambda}(|x|, u_N) dS.$$

Thus from (6.5.7) and (6-g2), we can see that (6.5.2) holds. Noting that the right hand side of (6.5.4) is differentiable with respect to t and combining (6.5.3), we can obtain (6.5.7). See also Lemma 1.4 in Chapter III of Struwe [98].

Here we also state the Pohozaev type inequality for $(P_{\mathbf{R}^N})$ and (P_D) .

Proposition 6.5.7. Assume that (6–g1)–(6–g6) are satisfied. Let $u_D \in H^1_{0,r}(\Omega)$ (resp. $u_{\mathbf{R}^N} \in H^1_r(\mathbf{R}^N)$) be a solution of

$$-\Delta u = g(|x|, u) + \lambda h(u) \quad \text{in } \Omega, \ u = 0 \text{ on } \partial \Omega \quad (\text{resp.} \ -\Delta u = g(|x|, u) + \lambda h(u) \quad \text{in } \mathbf{R}^N).$$

Then u_D (resp. $u_{\mathbf{R}^N} \in H^1_r(\mathbf{R}^N)$) satisfies the following:

following Pohozaev identity holds:

(6.5.8)
$$\frac{N-2}{2} \|\nabla u_D\|_{L^2(\Omega)}^2 - N \int_{\Omega} \hat{G}_{\lambda}(|x|, u_D) dx \ge \frac{1}{2} \int_{\partial \Omega} \left(\frac{\partial u_D}{\partial \nu}\right)^2 dS.$$

(6.5.9)
$$\left(\text{ resp. } \frac{N-2}{2} \|\nabla u_{\mathbf{R}^N}\|_{L^2(\mathbf{R}^N)}^2 - N \int_{\Omega} \hat{G}_{\lambda}(|x|, u_{\mathbf{R}^N}) dx \ge 0 \right).$$

Remark 6.5.8. As in Remark 6.5.6, if
$$g(r,s)$$
 is of class C^1 with respect to r , then the

$$(6.5.10) \frac{N-2}{2} \|\nabla u_D\|_{L^2}^2 - N \int_{\Omega} \hat{G}_{\lambda}(|x|, u_D) dx = \int_{\Omega} x \cdot \nabla G(|x|, u_D) dx + \frac{1}{2} \int_{\partial \Omega} \left(\frac{\partial u_D}{\partial \nu}\right)^2 dS,$$

$$(6.5.11) \left(\text{ resp. } \frac{N-2}{2} \|\nabla u_{\mathbf{R}^N}\|_{L^2(\mathbf{R}^N)}^2 - N \int_{\mathbf{R}^N} \hat{G}_{\lambda}(|x|, u_{\mathbf{R}^N}) dx = \int_{\mathbf{R}^N} x \cdot \nabla G(|x|, u_{\mathbf{R}^N}) dx \right).$$

By (6-g2), we can show (6.5.8) and (6.5.9) from (6.5.10) and (6.5.11).

Proof of Proposition 6.5.7. We only show for u_D since a proof for $u_{\mathbf{R}^N}$ is similar to the one of Proposition 6.5.5.

For the Dirichlet problem, critical points of $I_{\lambda} \in C^1(H_0^1(\Omega), \mathbf{R})$ corresponds to solutions. However, for technical reasons, we regard $I_{\lambda} \in C^1(H^1(\Omega), \mathbf{R})$ in this proof. We set $\tilde{\eta}(t) := u_D(tx) \in C^1([1,2], H^1(\Omega))$ and as in the proof of Proposition 6.5.5, we shall calculate

$$\frac{d}{dt}I_{\lambda}(\tilde{\eta}(t))\big|_{t=1}.$$

Since u_D satisfies $-\Delta u_D = g(|x|, u_D) + \lambda h(u_D)$ in Ω , $u_D = 0$ on $\partial \Omega$, using integration by parts, for any $\varphi \in H^1(\Omega) \cap C^1(\overline{\Omega})$, we have

$$I_{\lambda}'(u_D)\varphi = \int_{\Omega} \nabla u_D \cdot \nabla \varphi dx - \int_{\Omega} (g(|x|, u_D) + \lambda h(u_D))\varphi dx = -\int_{\partial \Omega} \nabla u_D \cdot x\varphi dS$$

Noting $\tilde{\eta}'(1) = x \cdot \nabla u_D(x) \in H^1(\Omega) \cap C^1(\overline{\Omega})$, it follows that

(6.5.12)
$$\frac{d}{dt}I_{\lambda}(\tilde{\eta}(t))\big|_{t=1} = -\int_{\partial\Omega} \left(\frac{\partial u_D}{\partial\nu}\right)^2 dS.$$

On the other hand, set

$$\tilde{I}_{\lambda}(t) := \frac{t^{-N+2}}{2} \int_{|x| \ge t} |\nabla u_D(x)|^2 dx - t^{-N} \int_{|x| \ge t} G(|x|, u_D) + \lambda H(u_D) dx,$$

then we have

(6.5.13)
$$\tilde{I}'_{\lambda}(1) = -\frac{N-2}{2} \|\nabla u_D\|_{L^2}^2 - \frac{1}{2} \int_{\partial\Omega} \left(\frac{\partial u_D}{\partial\nu}\right)^2 dS + N \int_{\Omega} G(|x|, u_D) + \lambda H(u_D) dx.$$

Since $I(\tilde{\eta}(t)) \geq \tilde{I}_{\lambda}(t)$ and $I(\tilde{\eta}(1)) = \tilde{I}_{\lambda}(1)$, by (6.5.12) and (6.5.13), it follows that

$$\frac{1}{2} \int_{\partial \Omega} \left(\frac{\partial u_D}{\partial \nu} \right)^2 dS \le \frac{N-2}{2} \| \nabla u_D \|_{L^2}^2 - N \int_{\Omega} G(|x|, u_D) + \lambda H(u_D) dx.$$

6.5.3 Proof of Theorem 6.2.1

Now we prove Theorem 6.2.1. Suppose that the conditions (6-g1)-(6-g7) are satisfied. Let $(u_{n,k})$ be a sequence satisfying (6.5.1) and set

$$b_{n,0} := \lim_{\lambda \to 0} b_n(\lambda) = \lim_{k \to \infty} I_{\lambda_{n,k}}(u_{n,k}) \in [b_n(1), b_n(0)]$$

Proposition 6.5.9. There exists a $C_n > 0$ such that $||u_{n,k}||_E \leq C_n$ for all $k \in \mathbf{N}$.

Proof. First, we prove that $(\nabla u_{n,k})_{k=1}^{\infty}$ is bounded in $L^2(\Omega)$. Since $I'_{\lambda_{n,k}}(u_{n,k}) = 0$, by Proposition 6.5.5, we have

(6.5.14)
$$-\int_{\Omega} G(|x|, u_{n,k}) + \lambda_{n,k} H(u_{n,k}) dx \ge -\frac{N-2}{2N} \|\nabla u_{n,k}\|_{L^{2}(\Omega)}^{2} \\ + \frac{1}{N} \int_{\partial\Omega} G(1, u_{n,k}) + \lambda_{n,k} H(u_{n,k}) dx.$$

From (6.5.14), we obtain

(6.5.15)
$$b_{n}(\lambda_{n,k}) = \frac{1}{2} \|\nabla u_{n,k}\|_{L^{2}(\Omega)}^{2} - \int_{\Omega} G(|x|, u_{n,k}) + \lambda_{n,k} H(u_{n,k}) dx$$
$$\geq \frac{1}{N} \|\nabla u_{n,k}\|_{L^{2}(\Omega)}^{2} + \frac{1}{N} \int_{\partial \Omega} G(1, u_{n,k}) + \lambda_{n,k} H(u_{n,k}) dx.$$

Noting that $H(s) \ge 0$ for all $s \in \mathbf{R}$, $\lim_{k\to\infty} b_n(\lambda_{n,k}) = b_{n,0} \le b_n(0)$ and (6–g7), we deduce from (6.5.15) that there exists a $C_n > 0$ such that $\|\nabla u_{n,k}\|_{L^2(\Omega)} \le C_n$ for all $k \in \mathbf{N}$.

Next, we show $||u_{n,k}||_E \leq C_n$ for all $k \in \mathbb{N}$. First, we consider the case $N \geq 3$. By Lemma 6.3.1, it holds

(6.5.16)
$$b_{n}(\lambda_{n,k}) = I_{\lambda_{n,k}}(u_{n,k}) = \frac{1}{2} \|u_{n,k}\|_{E}^{2} - \int_{\Omega} \frac{m_{1}}{4} u_{n,k}^{2} + G(|x|, u_{n,k}) + \lambda_{n,k} H(u_{n,k}) dx$$
$$\geq \frac{1}{2} \|u_{n,k}\|_{E}^{2} - (1 + \lambda_{n,k}) \int_{\Omega} H(u_{k}) dx \geq \frac{1}{2} \|u_{n,k}\|_{E}^{2} - C \|u_{n,k}\|_{L^{2^{*}}(\Omega)}^{2^{*}}.$$

From (6.3.4) and (6.5.16), it holds that

(6.5.17)
$$b_n(\lambda_{n,k}) \ge \frac{1}{2} \|u_{n,k}\|_E^2 - C \|\nabla u_{n,k}\|_{L^2(\Omega)}^{2^*}$$

Since $b_n(\lambda_{n,k})$ and $(\|\nabla u_{n,k}\|_{L^2(\Omega)})_{k=1}^{\infty}$ are bounded, taking C_n sufficiently large, $\|u_{n,k}\|_E \leq C_n$ follows from (6.5.17).

Next we consider the case N = 2. Following the arguments in [61] (cf. Proof of Proposition 5.5 in [56]), we prove indirectly. Assume that $r_k := ||u_{n,k}||_{L^2(\Omega)}^{-1} \to 0$. Set

$$v_k(x) := (T_{r_k} \tilde{v}_k)(x), \quad \tilde{v}_k(x) := u_{n,k} \left(\frac{x}{r_k}\right), \quad \Omega_k := \{x \in \mathbf{R}^N : |x| > r_k\},\$$

where T_{r_k} defined by (6.3.2). From $\|\nabla \tilde{v}_k\|_{L^2(\Omega_k)} = \|\nabla u_k\|_{L^2(\Omega)}, \|\tilde{v}_k\|_{L^2(\Omega_k)} = 1$ and (6.3.3), (v_k) is bounded in $H^1(\mathbf{R}^2)$. Therefore, we may assume

$$v_k \rightharpoonup v_0$$
 weakly in $H^1(\mathbf{R}^2)$ and $v_k(x) \rightarrow v_0(x)$ a.a. $x \in \mathbf{R}^2$.

Next, we show $v_0 = 0$. We remark that since $v_k(x) = \tilde{v}_k(x)$ in Ω_k , v_k satisfies

(6.5.18)
$$\begin{cases} -r_k^2 \Delta v_k = g\left(\frac{|x|}{r_k}, v_k\right) + \lambda_{n,k} h(v_k) & \text{in } \Omega_k; \\ v'_k(r_k) = 0. \end{cases}$$

By the boundedness of (v_k) in $H^1(\mathbf{R}^2)$, for any $\varphi \in C_0^{\infty}(\mathbf{R}^2)$ with $\operatorname{supp} \varphi \subset \mathbf{R}^2 \setminus \{0\}$, we can show

(6.5.19)
$$\int_{\Omega_k} h(v_k)\varphi dx \to \int_{\mathbf{R}^2} h(v_0)\varphi dx, \quad \int_{\Omega_k} g\left(\frac{|x|}{r_k}, v_k\right)\varphi dx \to \int_{\mathbf{R}^2} g_{\infty}(v_0)\varphi dx.$$

By (6.5.18) and (6.5.19), we obtain

$$\int_{\mathbf{R}^2} g_{\infty}(v_0) \varphi dx = 0 \quad \text{for any } \varphi \in C_0^{\infty}(\mathbf{R}^2) \text{ with supp } \varphi \subset \mathbf{R}^2 \setminus \{0\},$$

which implies

(6.5.20)
$$g_{\infty}(v_0(x)) = 0$$
 a.a. $x \in \mathbf{R}^2$

Since $v_0 \in H^1_r(\mathbf{R}^2) \subset C(\mathbf{R}^2 \setminus \{0\})$, (6–g4) and (6.5.20), we infer that $v_0 \equiv 0$. On the other hand, by (6.5.18), we have

(6.5.21)
$$r_k^2 \int_{\Omega_k} |\nabla v_k|^2 dx = \int_{\Omega_k} g\left(\frac{|x|}{r_k}, v_k\right) v_k + \lambda_{n,k} h(v_k) v_k dx$$

Therefore it follows from (6.5.21), $1 = \|\tilde{v}_k\|_{L^2(\Omega_k)} = \|v_k\|_{L^2(\Omega_k)}$ and Lemma 6.3.1 that

$$0 < \frac{m_1}{2} = \frac{m_1}{2} \|v_k\|_{L^2(\Omega_k)}^2 \le r_k^2 \|\nabla v_k\|_{L^2(\Omega_k)}^2 + \frac{m_1}{2} \|v_k\|_{L^2(\Omega_k)}^2$$
$$\le \int_{\Omega_k} \frac{m_1}{2} v_k^2 + g\left(\frac{|x|}{r_k}, v_k\right) v_k + \lambda_k h(v_k) v_k dx \le (1 + \lambda_{n,k}) \int_{\Omega_k} h(v_k) v_k dx.$$

Since $\lambda_{n,k} \leq 1$ and $h(s)s \geq 0$ for all $s \in \mathbf{R}$, we obtain

(6.5.22)
$$\frac{m_1}{2} \le 2 \int_{\mathbf{R}^2} h(v_k) v_k dx.$$

On the other hand, since $v_k \rightarrow 0$ weakly in $H^1(\mathbf{R}^2)$, by Lemma 6.3.2 (i), we have

$$\int_{\mathbf{R}^2} h(v_k) v_k dx \to 0.$$

This contradicts to (6.5.22), therefore it holds that $||u_{n,k}||_{L^2(\Omega)} \leq C_n$, which completes the proof.

By virtue of Proposition 6.5.9, we have

Corollary 6.5.10. The sequence $(u_{n,k})_{k=1}^{\infty}$ is a bounded (PS) sequence at level $b_{n,0}$ for I_0 . *Proof.* We remark that it holds that

$$|I_0(u_{n,k}) - I_{\lambda_{n,k}}(u_{n,k})| \le \lambda_{n,k} K(u_{n,k}), \quad |I'(u_{n,k})\varphi - I'_{\lambda_{n,k}}(u_{n,k})\varphi| \le \lambda_{n,k} \|K'(u_{n,k})\|_{E^*} \|\varphi\|_{E^*}.$$

By Lemma 6.3.2 and $\lambda_{n,k} \to 0$, we can prove $I_0(u_{n,k}) \to b_{n,0}$ and $I'_0(u_{n,k}) \to 0$ as $k \to \infty$. Thus $(u_{n,k})_{k=1}^{\infty}$ is a bounded (PS) sequence at level $b_{n,0}$ for I.

Now we complete a proof of Theorem 6.2.1.

Proof of Theorem 6.2.1. For each $n \in \mathbf{N}$, by Corollary 6.5.10, there exists a bounded sequence $(u_{n,k})_{k=1}^{\infty} \subset E$

$$I_0(u_{n,k}) \to b_{n,0}, \quad I'_0(u_{n,k}) \to 0 \quad \text{as } k \to \infty.$$

Thus by Lemma 6.3.2 (ii), there exists a $u_{n,0} \in E$ such that

$$I_0(u_{n,0}) = b_{n,0}, \quad I'_0(u_{n,0}) = 0.$$

On the other hand, by Lemmas 6.4.4 and 6.4.5, $b_{n,0} \to \infty$ as $n \to \infty$. Therefore we can show the existence of infinitely many radial solutions.

In order to obtain positive solutions, we modify q(r,s) as follows:

$$g_+(r,s) := \begin{cases} g(r,s) & \text{if } s \ge 0, \\ 0 & \text{if } s \le 0. \end{cases}$$

Then any nontrivial radial solution of

$$-\Delta u = g_+(|x|, u) \quad \text{in } \Omega, \quad u'(1) = 0$$

is positive on $\{|x| \ge 1\}$ by the maximum principle. Thus we will find a critical point of

$$I_{+}(u) := \frac{1}{2} \|\nabla u\|_{L^{2}}^{2} - \int_{\Omega} G_{+}(|x|, u) dx$$

We can prove that I_{+} has a mountain pass geometry as in Lemma 6.4.1. Moreover, using the monotonicity method as before, we can show that I_+ has a nontrivial critical point. Thus we complete a proof.

Outline of proof of Theorem 6.2.2 6.5.4

In this subsection, we give an outline of proof of Theorem 6.2.2. Throughout this subsection, we assume the conditions (6-g1)-(6-g6).

As in the Neumann case, we define the following functionals: for each $\lambda \in [0, 1]$,

$$I_{D,\lambda}(v) := \frac{1}{2} \|\nabla v\|_{L^{2}(\Omega)}^{2} - \int_{\Omega} G(|x|, v) + \lambda H(v) dx \in C^{1}(H_{0,r}^{1}(\Omega), \mathbf{R}),$$

$$I_{\mathbf{R}^{N},\lambda}(w) := \frac{1}{2} \|\nabla w\|_{L^{2}(\mathbf{R}^{N})}^{2} - \int_{\mathbf{R}^{N}} G(|x|, w) + \lambda H(w) dx \in C^{1}(H_{r}^{1}(\mathbf{R}^{N}), \mathbf{R}),$$

$$J_{\mathbf{R}^{N}}(w) := \frac{1}{2} \|\nabla w\|_{L^{2}(\mathbf{R}^{N})}^{2} - 2 \int_{\mathbf{R}^{N}} H(w) dx \in C^{1}(H_{r}^{1}(\mathbf{R}^{N}), \mathbf{R}).$$

Then, noting $H^1_{0,r}(\Omega) \subset H^1_r(\mathbf{R}^N)$, we can see that

$$J_{\mathbf{R}^N}(v) \le I_{D,\lambda}(v), \quad J_{\mathbf{R}^N}(w) \le I_{\mathbf{R}^N,\lambda}(w)$$

for all $\lambda \in [0,1]$, $v \in H_{0,r}^1(\Omega)$, $w \in H_r^1(\mathbf{R}^N)$. Furthermore $I_{D,\lambda}$, $I_{\mathbf{R}^N,\lambda}$ satisfy (6.3.5). Let $\gamma_n \in C(S^{n-1}, H_{0,r}^1(\Omega))$ appear in Lemma 6.4.1. Then $\gamma_n \in C(S^{n-1}, H_r^1(\mathbf{R}^N))$ and we can define minimax values for $I_{D,\lambda}$, $I_{\mathbf{R}^N,\lambda}$ and $J_{\mathbf{R}^N}$:

$$\begin{split} b_{n,D}(\lambda) &:= \inf_{\gamma \in \Gamma_{n,D}} \max_{\sigma \in D_n} I_{D,\lambda}(\gamma(\sigma)), \quad b_{n,\mathbf{R}^N}(\lambda) := \inf_{\gamma \in \Gamma_{n,\mathbf{R}^N}} \max_{\sigma \in D_n} I_{\mathbf{R}^N,\lambda}(\gamma(\sigma)), \\ c_{n,\mathbf{R}^N} &:= \inf_{\gamma \in \Gamma_{n,\mathbf{R}^N}} \max_{\sigma \in D_n} J_{\mathbf{R}^N}(\gamma(\sigma)), \end{split}$$

where

$$\Gamma_{n,D} := \{ \gamma \in C(D_n, H^1_{0,r}(\Omega)) : \gamma = \gamma_n \text{ on } S^{n-1} \},$$

$$\Gamma_{n,\mathbf{R}^N} := \{ \gamma \in C(D_n, H^1_r(\mathbf{R}^N)) : \gamma = \gamma_n \text{ on } S^{n-1} \}.$$

It is easily seen that all lemmas in sections 6.3 and 6.4 hold if we replace I_{λ} , J, $b_n(\lambda)$, c_n by $I_{D,\lambda}$, $I_{\mathbf{R}^N,\lambda}$, $b_{n,D}(\lambda)$, $b_{n,\mathbf{R}^N}(\lambda)$, c_{n,\mathbf{R}^N} . Moreover, we can apply the monotonicity method for $I_{D,\lambda}$ and $I_{\mathbf{R}^N,\lambda}$ (cf. Lemma 6.5.3). Therefore for each $n \in \mathbf{N}$ there are sequences $(\lambda_{n,k}) \subset [0,1], (v_{n,k}) \subset H_{0,r}^1(\Omega), (w_{n,k}) \subset H_r^1(\mathbf{R}^N)$ such that $\lambda_{n,k} \to 0$ and

$$I_{D,\lambda_{n,k}}(v_{n,k}) = b_{n,D}(\lambda_{n,k}), \qquad I'_{D,\lambda_{n,k}}(v_{n,k}) = 0,$$

$$I_{\mathbf{R}^{N},\lambda_{n,k}}(w_{n,k}) = b_{n,\mathbf{R}^{N}}(\lambda_{n,k}), \quad I'_{\mathbf{R}^{N},\lambda_{n,k}}(w_{n,k}) = 0.$$

As in the Neumann case, it is sufficient to show that $(v_{n,k})_{k=1}^{\infty}$ (resp. $(w_{n,k})_{k=1}^{\infty}$) is bounded in $H_{0,r}^1(\Omega)$ (resp. $H_r^1(\mathbf{R}^N)$). Using (6.5.8) and (6.5.9) instead of (6.5.2), it is easily seen that $(v_{n,k})_{k=1}^{\infty}$ (resp. $(w_{n,k})_{k=1}^{\infty}$) is bounded in $H_{0,r}^1(\Omega)$ (resp. $H_r^1(\mathbf{R}^N)$) in a similar way to the proof of Proposition 6.5.9.

The remaining part of proof of Theorem 6.2.2 is the same as the proof of Theorem 6.2.1, so we omit it.

6.5.5 Proof of Theorem 6.2.4

In this subsection, we prove Theorem 6.2.4 and let $g(r,s) = -V(r)s + \tilde{g}(s)$. We only consider (P_N) , since proofs in other cases are similar. As mentioned before, we can suppose $\Omega := \{x \in \mathbb{R}^N : |x| > 1\}$. Furthermore, as in [7, 15, 16, 48], instead of (§3), we can assume

(
$$\tilde{g}3'$$
)
$$\lim_{s \to \infty} \frac{\tilde{g}(s)}{s^{2^*-1}} = 0$$

Indeed, set

$$\tilde{\zeta}_1 := \inf \left\{ s \in [\tilde{\zeta}_0, \infty) : \tilde{g}(s) = 0 \right\}$$

where $\tilde{\zeta}_0 > 0$ appearing in ($\tilde{g}4$). If $\tilde{g}(s) > 0$ for all $s \geq \tilde{\zeta}_0$, then we set $\tilde{\zeta}_1 = \infty$. We define $\bar{g}(s)$ as follows:

$$\bar{g}(s) := \begin{cases} \tilde{g}(s) & \text{if } |s| \leq \tilde{\zeta}_1, \\ 0 & \text{if } |s| > \tilde{\zeta}_1. \end{cases}$$

Then \bar{g} satisfies ($\tilde{g}1$), ($\tilde{g}2$), ($\tilde{g}3'$) and ($\tilde{g}4$). Moreover, any solution of

(6.5.23)
$$-\Delta u + V(|x|)u = \bar{g}(u) \quad \text{in } \Omega, \quad \frac{\partial u}{\partial \nu} = 0 \quad \text{on } \partial \Omega,$$

satisfies $||u||_{L^{\infty}(\Omega)} \leq \tilde{\zeta}_1$ by the maximum principle. Therefore any solution of (6.5.23) satisfies (P_N) with $g(r,s) = -V(r)s + \tilde{g}(s)$, which implies that we can assume $(\tilde{g}3')$ instead of $(\tilde{g}3)$ without loss of generality.

As stated in the above, we prove Theorem 6.2.4 under

 (\mathcal{H}_3) $N \geq 3$, the conditions ($\tilde{g}1$), ($\tilde{g}2$), ($\tilde{g}3'$), ($\tilde{g}4$), ($\tilde{g}5$), (6–V1)–(6–V3) are satisfied.
Under the condition (\mathcal{H}_3) we will find infinitely many critical points of

$$\begin{split} \tilde{I}(u) &:= \frac{1}{2} \|\nabla u\|_{L^2(\Omega)}^2 + \frac{1}{2} \int_{\Omega} \left(V(|x|) + \frac{\tilde{m}_1}{2} \right) u^2 dx - \int_{\Omega_R} \frac{\tilde{m}_1}{4} u^2 + \tilde{G}(u) dx \\ &= \frac{1}{2} \|u\|^2 - \int_{\Omega} \frac{\tilde{m}_1}{4} u^2 + \tilde{G}(u) dx, \end{split}$$

where \tilde{m}_1 appears in (\tilde{g}_2) and $\tilde{G}(s) = \int_0^s \tilde{g}(t) dt$.

In this case, we can define $h \in C(\mathbf{R})$ satisfying Lemma 6.3.1. Thus we define an auxiliary functional \tilde{J} and parametrized functional \tilde{I}_{λ} for each $\lambda \in [0, 1]$. We note that all lemmas and propositions in section 6.4 hold for these functionals. Moreover, noting $V(r) \to 0$ as $r \to \infty$ and a proof of Proposition 6.6.1 in subsection 6.6.1, we can also prove that \tilde{I}_{λ} , \tilde{J} have a symmetric mountain pass structure and define $\tilde{b}_n(\lambda)$ and \tilde{c}_n as in Definition 6.4.3. Furthermore, we see that all lemmas in section 6.3 hold. By Proposition 6.5.2, for each $n \in \mathbf{N}$ there exist $(\tilde{\lambda}_{n,k})_{k=1}^{\infty}$ and $(\tilde{u}_{n,k})_{k=1}^{\infty} \subset H_r^1(\Omega)$ such that $\tilde{\lambda}_{n,k} \to 0$ as $k \to \infty$ and

$$\tilde{I}_{\tilde{\lambda}_{n,k}}(\tilde{u}_{n,k}) = \tilde{b}_n(\tilde{\lambda}_{n,k}), \quad \tilde{I}'_{\tilde{\lambda}_{n,k}}(\tilde{u}_{n,k}) = 0$$

Next, we show that $(\tilde{u}_{n,k})_{k=1}^{\infty}$ is bounded in $H^1_r(\Omega)$.

Lemma 6.5.11. There exists a $C_n > 0$ such that $\|\tilde{u}_{n,k}\| \leq C_n$ for all $k \geq 1$.

Proof. As in Proposition 6.5.9, firstly we show that $(\nabla \tilde{u}_{n,k})_{k=1}^{\infty}$ is bounded in $L^2(\Omega)$. By Remark 6.5.6, $\tilde{u}_{n,k}$ satisfies

$$\frac{1}{2} \int_{\Omega} V(|x|) \tilde{u}_{n,k}^2 dx - \int_{\Omega} \tilde{G}(\tilde{u}_{n,k}) + \tilde{\lambda}_{n,k} H(\tilde{u}_{n,k}) dx$$
$$= -\frac{1}{2^*} \|\nabla \tilde{u}_{n,k}\|_{L^2(\Omega)}^2 - \frac{1}{2N} \int_{\Omega} x \cdot \nabla V(|x|) \tilde{u}_{n,k}^2 dx$$
$$+ \int_{\partial \Omega} -\frac{1}{2} V(1) \tilde{u}_{n,k}^2 + \tilde{G}(\tilde{u}_{n,k}) + \tilde{\lambda}_{n,k} H(\tilde{u}_{n,k}) dS.$$

By ($\tilde{g}5$) and Hölder's inequality, there exists a C > 0 such that

$$\begin{split} \tilde{I}_{\tilde{\lambda}_{n,k}}(\tilde{u}_{n,k}) &= \frac{1}{2} \|\nabla \tilde{u}_{n,k}\|_{L^{2}(\Omega)}^{2} + \frac{1}{2} \int_{\Omega} V(|x|) \tilde{u}_{n,k}^{2} dx - \int_{\Omega} \tilde{G}(\tilde{u}_{n,k}) + \tilde{\lambda}_{n,k} H(\tilde{u}_{n,k}) dx \\ &= \frac{1}{N} \|\nabla \tilde{u}_{n,k}\|_{L^{2}(\Omega)}^{2} - \frac{1}{2N} \int_{\Omega} x \cdot \nabla V(|x|) \tilde{u}_{n,k}^{2} dx \\ &+ \int_{\partial \Omega} -\frac{1}{2} V(1) \hat{u}_{n,k}^{2} + \tilde{G}(\tilde{u}_{n,k}) + \tilde{\lambda}_{n,k} H(\tilde{u}_{n,k}) dS \\ &\geq \frac{1}{N} \|\nabla \tilde{u}_{n,k}\|_{L^{2}(\Omega)}^{2} - \frac{1}{2N} \|(x \cdot \nabla V(|x|))^{+}\|_{L^{\frac{N}{2}}(\Omega)} \|\tilde{u}_{n,k}\|_{L^{2*}(\Omega)}^{2} - C. \end{split}$$

We extend $\tilde{u}_{n,k}$ as follows:

$$\hat{u}_{n,k}(x) := \begin{cases} \tilde{u}_{n,k}(|x|) & \text{if } |x| \ge 1, \\ \tilde{u}_{n,k}(1) & \text{if } |x| < 1. \end{cases}$$

Then it is clear that $\hat{u}_{n,k} \in H^1_r(\mathbf{R}^N)$, $\|\nabla \hat{u}_{n,k}\|_{L^2(\mathbf{R}^N)} = \|\nabla \tilde{u}_{n,k}\|_{L^2(\Omega)}$ and $\|\tilde{u}_{n,k}\|_{L^{2^*}(\Omega)} \leq \|\hat{u}_{n,k}\|_{L^{2^*}(\mathbf{R}^N)}$. Furthermore, since $\|\hat{u}_{n,k}\|_{L^{2^*}(\mathbf{R}^N)}^2 \leq \|\nabla \hat{u}_{n,k}\|_{L^2(\mathbf{R}^N)}^2/S_N$ holds, we obtain $\|\tilde{u}_{n,k}\|_{L^{2^*}(\Omega)}^2 \leq \|\nabla \tilde{u}_{n,k}\|_{L^2(\Omega)}^2/S_N$. Here, from (6–V3), we can take an $\varepsilon_0 > 0$ such that

$$\left\| (x \cdot \nabla V(|x|))^+ \right\|_{L^{\frac{N}{2}}(\Omega)} < 2S_N - \varepsilon_0.$$

Then we have

$$\tilde{I}_{\tilde{\lambda}_{n,k}}(\tilde{u}_{n,k}) \ge \frac{1}{N} \|\nabla \tilde{u}_{n,k}\|_{L^{2}(\Omega)}^{2} - \frac{1}{N} \frac{2S_{N} - \varepsilon_{0}}{2S_{N}} \|\nabla \tilde{u}_{n,k}\|_{L^{2}(\Omega)}^{2} - C \ge \varepsilon_{1} \|\nabla \tilde{u}_{n,k}\|_{L^{2}(\Omega)}^{2} - C$$

for some $\varepsilon_1 > 0$. Thus there exists a $C_n > 0$ such that $\|\nabla \tilde{u}_{n,k}\|_{L^2(\Omega)} \leq C_n$ for all $k \in \mathbb{N}$.

Since a proof of the boundedness of $(\tilde{u}_{n,k})_{k=1}^{\infty}$ in $L^2(\Omega)$ is similar to the one of Proposition 6.5.9, we omit it.

Now we complete a proof of Theorem 6.2.4.

Proof of Theorem 6.2.4. From Lemma 6.5.11, we see that $(\tilde{u}_{n,k})_{k=1}^{\infty}$ is a bounded (PS) sequence for \tilde{I} as in Corollary 6.5.10. Therefore we can show the existence of infinitely many solutions as in Theorem 6.2.1. For the existence of at least one positive solution, we replace \tilde{I} by

$$\tilde{I}_{+}(u) = \frac{1}{2} \int_{\Omega} |\nabla u|^{2} + V(|x|)u^{2}dx - \int_{\Omega} \tilde{G}_{+}(u)dx$$

where $\tilde{G}_+(s) = \int_0^s \tilde{g}_+(\tau)\tau$. In this case, we can show the existence of nontrivial critical point in the similar way to the proof of Theorem 6.2.1.

6.6 Proofs of Proposition 6.4.1 (ii) and Lemma 6.3.2, and technical lemma

In this section, we prove Proposition 6.4.1 (ii) and Lemma 6.3.2. Moreover, we state a useful lemma. First, we give a proof of Proposition 6.4.1 (ii).

6.6.1 Proof of Proposition 6.4.1 (ii)

In this subsection, we prove the following proposition.

Proposition 6.6.1. Let $\Omega = \{x \in \mathbf{R}^N : |x| > 1\}$ and (\mathcal{H}_2) be satisfied. Then for each $n \in \mathbf{N}$, there exists a continuous odd map $\gamma_n : S^{n-1} \to H^1_{0,r}(\Omega)$ such that

$$I(\gamma_n(\sigma)) < 0$$
 for all $\sigma \in S^{n-1}$.

Before proving Proposition 6.6.1, we introduce some notations. First, we define $\underline{G}(s)$ for $s \ge 0$ as follows:

$$\underline{G}(s) := \inf_{r \ge R_0} G(r, s),$$

where R_0 appears in (6–g6). By (6–g3) and (6–g6), $\underline{G}(s)$ is well-defined and satisfies $\underline{G}(\zeta_0) > 0$. We also set

$$\underline{I}(u) := \frac{1}{2} \|\nabla u\|_{L^2(\mathbf{R}^N)}^2 - \int_{\mathbf{R}^N} \underline{G}(u) dx \in C(H^1_r(\mathbf{R}^N), \mathbf{R}).$$

Note that if $u \in H^1_r(\Omega)$ and $\operatorname{supp} u \subset \{|x| > R_0\}$, then $I(u) \leq \underline{I}(u)$. Therefore it is sufficient to prove that there exists a continuous odd map $\gamma_n : S^{n-1} \to H^1_r(\Omega)$ such that

(6.6.1) $\underline{I}(\gamma_n(\sigma)) < 0$, $\operatorname{supp} \gamma_n(\sigma) \subset \{|x| > R_0\}$ for all $\sigma \in S^{n-1}$.

Proof of Proposition 6.6.1. By the arguments of Theorem 10 in [16], for each $n \in \mathbf{N}$, there exists a $\pi_n \in C(S^{n-1}, H_r^1(\mathbf{R}^N))$ such that

$$\pi_n(-\sigma) = -\pi_n(\sigma), \quad \|\pi_n(\sigma)\|_{L^{\infty}(\mathbf{R}^N)} = \zeta_0, \quad \int_{\mathbf{R}^N} \underline{G}(\pi_n(\sigma)) dx \ge 1 \quad \text{for all } \sigma \in S^{n-1}.$$

We modify π_n to obtain γ_n satisfying the property (6.6.1). Let $\varphi \in C^{\infty}([0,\infty))$ be a cut-off function such that

$$0 \le \varphi(t) \le 1, \quad \varphi(t) = \begin{cases} 0 & \text{if } t \le 1, \\ 1 & \text{if } t \ge 2, \end{cases}$$

and set $\varphi_k(t) := \varphi(kt)$ and $\eta_k(\sigma)(x) := \varphi_k(|x|)\pi_n(\sigma)(x)$ for $k \in \mathbf{N}$. Then it holds $\operatorname{supp} \eta_k(\sigma) \subset \{|x| \ge 1/k\}$ for all $\sigma \in S^{n-1}$ and

$$\int_{\mathbf{R}^N} \underline{G}(\eta_k(\sigma)) dx \to \int_{\mathbf{R}^N} \underline{G}(\pi_n(\sigma)) dx \quad \text{as } k \to \infty \quad \text{uniformly w.r.t. } \sigma \in S^{n-1},$$

since $\pi_n(S^{n-1})$ is bounded in $L^{\infty}(\mathbf{R}^N)$. Therefore for a large $k_0 \in \mathbf{N}$, we have

(6.6.2)
$$\int_{\mathbf{R}^N} \underline{G}(\eta_{k_0}) dx \ge \frac{1}{2} \quad \text{for all } \sigma \in S^{n-1}$$

We consider $\eta_{k_0}(\sigma)(x/t)$ for $t \ge 1$. By (6.6.2), we see that $\operatorname{supp} \eta_{k_0}(\sigma)(\cdot/t) \subset \{|x| \ge t/k_0\}$ and

$$\underline{I}(\eta_{k_0}(\sigma)(\cdot/t)) = t^{N-2} \left(\frac{1}{2} \|\nabla \eta_{k_0}(\sigma)\|_{L^2(\mathbf{R}^N)}^2 - t^2 \int_{\mathbf{R}^N} \underline{G}(\eta_{k_0}) dx\right)$$
$$\leq t^{N-2} \left(\frac{1}{2} \|\nabla \eta_{k_0}(\sigma)\|_{L^2(\mathbf{R}^N)}^2 - \frac{t^2}{2}\right).$$

Since $\|\nabla \eta_{k_0}(\sigma)\|_{L^2(\mathbf{R}^N)}$ is uniformly bounded with respect to $\sigma \in S^{n-1}$, we can choose a $t_0 \geq 1$ satisfying $t_0/k_0 > R_0$ and

$$\underline{I}(\eta_{k_0}(\sigma)(\cdot/t_0)) < 0 \quad \text{for all } \sigma \in S^{n-1}.$$

Set $\gamma_n(\sigma)(x) := \eta_{k_0}(\sigma)(x/t_0)$, then γ_n satisfies (6.6.1). The oddness and continuity of γ_n follows from the ones of η_{k_0} , which completes a proof.

6.6.2 Proof of Lemma 6.3.2

Next we give a proof of Lemma 6.3.2.

Proof of Lemma 6.3.2. (i) First we show that K is weakly continuous. Let u_k satisfy $u_k \rightharpoonup u_0$ weakly in E. Without loss of generality, we may assume

$$u_k(x) \to u_0(x)$$
 a.a. $x \in \Omega$, $||u_k||_E \le M$.

Since (u_k) is bounded, by (6.3.1) and Lemma 6.3.1, there exists an $R_1 > 0$ such that if $|x| \ge R_1$, then $H(u_k(x)) = H(u_0(x)) = 0$ for all $k \ge 1$. Therefore, it is sufficient to show

$$\int_{\Omega \cap B_{R_1}} |H(u_k) - H(u_0)| dx \to 0.$$

We set $Q(s) := |s|^{2^*}$ $(N \ge 3)$, $Q(s) := \exp(s^2/(2M^2)) - 1 - s^2/(2M^2)$ (N = 2). Then by Lemma 6.3.1, for each $\varepsilon > 0$ there exists an $s_{\varepsilon} \ge 0$ such that if $|s| \ge s_{\varepsilon}$, then $H(s) \le \varepsilon Q(s)$. We define $\hat{H}(s)$ as follows:

$$\hat{H}(s) := \begin{cases} H(s) & \text{if } |s| \le s_{\varepsilon}, \\ H(s_{\varepsilon}) & \text{if } |s| > s_{\varepsilon}. \end{cases}$$

Since \hat{H} is bounded, it is easy to see that

$$\hat{H}(u_k) \to \hat{H}(u_0)$$
 in $L^1(\Omega \cap B_{R_1})$.

On the other hand, since $|\hat{H}(s) - H(s)| \leq \varepsilon Q(s)$ we have

$$\begin{split} &\int_{\Omega \cap B_{R_1}} |H(u_k) - H(u_0)| dx \\ \leq &\int_{\Omega \cap B_{R_1}} |H(u_k) - \hat{H}(u_k)| + |\hat{H}(u_k) - \hat{H}(u_0)| + |\hat{H}(u_0) - H(u_0)| dx \\ \leq &\varepsilon \int_{\Omega \cap B_{R_1}} Q(u_k) + Q(u_0) dx + \|\hat{H}(u_k) - \hat{H}(u_0)\|_{L^1(\Omega \cap B_{R_1})} \end{split}$$

Thus to prove the weak continuity of K, it is sufficient to prove

(6.6.3)
$$\sup_{k\geq 1} \int_{\Omega} Q(u_k) dx < \infty.$$

In the case $N \ge 3$, (6.6.3) follows from Sobolev's inequality and in the case N = 2, (6.6.3) holds by Lemma 6.6.2 (iii). Therefore K is weakly continuous.

Next we prove that $K'(u_k) \to K'(u_0)$ strongly in E^* . Since

$$K'(u_k)\varphi = \int_{\Omega} h(u_k)\varphi \, dx \quad \text{for all } \varphi \in E,$$

if we can show

(6.6.4)
$$h(u_k) \to h(u_0)$$
 strongly in $L^{p_N}(\Omega)$, $p_N := \begin{cases} 2 & \text{if } N = 2, \\ 2N/(N+2) & \text{if } N \ge 3, \end{cases}$

then $K'(u_k) \to K'(u_0)$ strongly in E^* .

We prove (6.6.4). As in the above, there exists an $R_1 \ge 1$ such that if $|x| \ge R_1$, then $h(u_k(x)) = h(u_0(x)) = 0$ for all $k \ge 1$. Therefore we only show

 $h(u_k) \to h(u_0)$ strongly in $L^{p_N}(\Omega \cap B_{R_1})$.

Set $Q(s) := \exp(s^2/(8M^2)) - 1 - s^2/(8M^2)$ $(N = 2), Q(s) := |s|^{(N+2)/(N-2)}$ $(N \ge 3)$. By Lemma 6.3.1, for each $\varepsilon > 0$, there exists an $s_{\varepsilon} \ge 0$ such that if $|s| \ge s_{\varepsilon}$, then $|h(s)| \le \varepsilon Q(s)$. Define $\hat{h}(s)$ as follows:

$$\hat{h}(s) := \begin{cases} h(s) & \text{if } |s| \le s_{\varepsilon}, \\ h(s_{\varepsilon}) & \text{if } s > s_{\varepsilon}, \\ h(-s_{\varepsilon}) & \text{if } s < -s_{\varepsilon}. \end{cases}$$

Then we have $\hat{h}(u_k) \to \hat{h}(u_0)$ strongly in $L^{p_N}(\Omega \cap B_{R_1})$. Therefore, to prove (6.6.4), it is sufficient to show

(6.6.5)
$$\sup_{k\geq 1} \int_{\Omega} Q(u_k)^{p_N} dx < \infty$$

In the case $N \ge 3$, by Sobolev's inequality and $p_N(2^* - 1) = 2^*$, (6.6.5) holds. In the case N = 2, we remark that $Q(s)^2 \le Q(2s)$ for all $s \in \mathbf{R}$. By Lemma 6.6.2 (iii), we have

$$\sup_{k \ge 1} \int_{\Omega} Q(u_k)^2 dx \le \sup_{k \ge 1} \int_{\Omega} Q(2u_k) dx \le C \sup_{k \ge 1} \|u_k\|_E^4 < \infty,$$

which implies (6.6.5). Therefore $K'(u_k) \to K'(u_0)$ strongly in E^* . (ii) Let $(u_k) \subset E$ be a (PS) sequence at level c for I_{λ} and $||u_k||_E \leq M$. Since (u_k) is bounded, there exist $u_0 \in E$ and subsequence (u_{k_ℓ}) such that

$$u_{k_{\ell}} \rightharpoonup u_0$$
 weakly in E , $u_{k_{\ell}}(x) \rightarrow u_0(x)$ a.a. $x \in \Omega$.

Let $\varphi \in C_{0,r}^{\infty}(\{|x| \ge 1\}) := \{\phi \in C^{\infty}(\{|x| \ge 1\}) : \phi(x) = \phi(|x|) \text{ and supp } \phi \text{ is compact}\}.$ Set $p_N := 2$ $(N = 2), p_N := 2N/(N+2)$ $(N \ge 3)$. Applying the similar arguments in the above, we can show

$$g(|x|, u_{k_{\ell}}) \to g(|x|, u_0(x)) \quad \text{strongly in } L^{p_N}(\Omega \cap B_{\hat{R}}),$$

$$h(u_{k_{\ell}}) \to h(u_0) \quad \text{strongly in } L^{p_N}(\Omega)$$

for all $\hat{R} > 1$. Therefore we obtain

(6.6.6)
$$\int_{\Omega} g(|x|, u_{k_{\ell}})\varphi dx \to \int_{\Omega} g(|x|, u_0)\varphi dx, \quad \int_{\Omega} h(u_{k_{\ell}})u_{k_{\ell}} dx \to \int_{\Omega} h(u_0)u_0 dx.$$

Noting that $I'_{\lambda}(u_{k_{\ell}}) \to 0$, by (6.6.6), we see that $I'_{\lambda}(u_0)\varphi = 0$ for all $\varphi \in C^{\infty}_{0,r}(\{|x| \ge 1\})$. Since $C^{\infty}_{0,r}(\{|x| \ge 1\})$ is dense in $E, I'_{\lambda}(u_0)u_0 = 0$ holds, that is,

(6.6.7)
$$\|u_0\|_E^2 = \int_{\Omega} \frac{m_1}{2} u_0^2 + g(|x|, u_0) u_0 + \lambda h(u_0) u_0 dx.$$

On the other hand, since $(u_{k_{\ell}})$ is bounded, we have $I'_{\lambda}(u_{k_{\ell}})u_{k_{\ell}} \to 0$, which implies

(6.6.8)
$$\|u_{k_{\ell}}\|_{E}^{2} - \int_{\Omega} \frac{m_{1}}{2} u_{k_{\ell}}^{2} + g(|x|, u_{k_{\ell}}) u_{k_{\ell}} + \lambda h(u_{k_{\ell}}) u_{k_{\ell}} dx \to 0.$$

Next, we rewrite

$$\int_{\Omega} \frac{m_1}{2} u_{k_{\ell}}^2 + g(|x|, u_{k_{\ell}}) u_{k_{\ell}} + \lambda h(u_{k_{\ell}}) u_{k_{\ell}} dx$$

=(1 + λ) $\int_{\Omega} h(u_{k_{\ell}}) u_{k_{\ell}} dx - \int_{\Omega} h(u_{k_{\ell}}) u_{k_{\ell}} - \frac{m_1}{2} u_{k_{\ell}}^2 - g(|x|, u_{k_{\ell}}) u_{k_{\ell}} dx.$

By Lemma 6.3.1 and Fatou's lemma, we have (6.6.9)

$$\liminf_{\ell \to \infty} \int_{\Omega} h(u_{k_{\ell}}) u_{k_{\ell}} - \frac{m_1}{2} u_{k_{\ell}}^2 - g(|x|, u_{k_{\ell}}) u_{k_{\ell}} dx \ge \int_{\Omega} h(u_0) u_0 - \frac{m_1}{2} u_0^2 - g(|x|, u_0) u_0 dx.$$

By (6.6.6)-(6.6.9), we obtain

$$\begin{split} \limsup_{\ell \to \infty} \|u_{k_{\ell}}\|_{E}^{2} &\leq (1+\lambda) \int_{\Omega} h(u_{0})u_{0}dx - \int_{\Omega} h(u_{0})u_{0} - \frac{m_{1}}{2}u_{0}^{2} - g(|x|, u_{0})u_{0}dx \\ &= \int_{\Omega} \frac{m_{1}}{2}u_{0}^{2} + g(|x|, u_{0})u_{0} + \lambda h(u_{0})u_{0}dx = \|u_{0}\|_{E}^{2}. \end{split}$$

Thus $u_{k_{\ell}}$ converges to u_0 strongly in E, which completes the proof.

(iii) Next we prove that J satisfies the (PS) condition. Let $(u_k) \subset E$ be a (PS) sequence at level c of J, i.e., $J(u_k) \to c$ and $J'(u_k) \to 0$ strongly in E^* . Since h(s) satisfies a global Ambrosetti–Rabinowitz condition, we can infer that (u_k) is bounded. Indeed, the boundedness of (u_k) in E comes from

$$J(u_k) - \frac{J'(u_k)u_k}{p+1} = \left(1 - \frac{2}{p+1}\right) \|u_k\|_E^2 - \int_{|x|>1} H(u_k) - \frac{1}{p+1}h(u_k)u_k dx$$
$$\ge \left(1 - \frac{2}{p+1}\right) \|u_k\|_E^2.$$

Thus we may assume that taking a subsequence if necessary,

$$u_k \rightharpoonup u_0$$
 weakly in E .

By (i), we have $K'(u_k) \to K'(u_0)$ strongly in E^* . Therefore by standard arguments we can conclude that (u_k) has a strongly convergent subsequence and this completes the proof.

6.6.3 A technical lemma

The following lemma is useful and we use it in proofs of Lemmas 6.3.2 and 6.4.1.

Lemma 6.6.2 (cf. Adachi–Tanaka [1], Byeon, Jeanjean and Tanaka [24], Ogawa [84]).

(i) Let $\Phi(s) := \exp(s) - 1 - s$ and $\beta \in (0, 4\pi)$. Then there exists a $\tilde{C}_{\beta} > 0$ such that

$$\int_{\mathbf{R}^2} \Phi\left(\beta \frac{u^2}{\|\nabla u\|_{L^2(\mathbf{R}^2)}^2}\right) dx \le \tilde{C}_\beta \frac{\|u\|_{L^4(\mathbf{R}^2)}^4}{\|\nabla u\|_{L^2(\mathbf{R}^2)}^4} \quad \text{for all } u \in H^1(\mathbf{R}^2) \setminus \{0\}.$$

(ii) For any M > 0 and $\beta \in (0, 4\pi)$, there exists a $\tilde{C}_{\beta,1} > 0$ such that

$$\int_{\mathbf{R}^2} \Phi\left(\frac{\beta u^2}{M^2}\right) dx \le \tilde{C}_{\beta,1} \frac{\|u\|_{L^4(\mathbf{R}^2)}^4}{M^4} \quad \text{for all } u \in H^1(\mathbf{R}^2) \text{ with } \|\nabla u\|_{L^2(\mathbf{R}^2)} \le M.$$

(iii) For any M > 0 and $\beta \in (0, 4\pi)$, there exists a $\tilde{C}_{\beta,2} > 0$ such that

$$\int_{\Omega} \Phi\left(\frac{\beta u^2}{2M^2}\right) dx \le \tilde{C}_{\beta,2} \frac{\|u\|_E^4}{M^4} \quad \text{for all } u \in H^1_r(\Omega) \text{ with } \|\nabla u\|_{L^2(\Omega)} \le M_r$$

where $\Omega := \{ x \in \mathbf{R}^2 : |x| > 1 \}.$

Proof. The inequality in (i) can be proven in the same way to [1]. (ii) is a direct consequence of (i). Indeed, since for each $x \in \mathbb{R}^2$ it follows that

$$M^{4}\Phi\left(\frac{\beta u^{2}(x)}{M^{2}}\right) = M^{4}\sum_{j=2}^{\infty} \frac{(\beta u^{2}(x))^{j}}{j!M^{2j}} = \sum_{j=2}^{\infty} \frac{(\beta u^{2}(x))^{j}}{j!M^{2j-4}}$$
$$\leq \sum_{j=2}^{\infty} \frac{(\beta u^{2}(x))^{j}}{j!\|\nabla u\|_{L^{2}}^{2j-4}} = \|\nabla u\|_{L^{2}}^{4}\Phi\left(\frac{\beta u^{2}(x)}{\|\nabla u\|_{L^{2}}^{2}}\right),$$

(ii) holds by (i). As to (iii), Using the operator T_1 (see (6.3.2)), by (ii) and Sobolev's inequality, we can easily obtain (iii).

Chapter 7

Appendix

7.1 Assumptions and main statement

The aim of this chapter is to prove that minimax values defined in Chapters 5 and 6 diverge to infinity.

In this chapter, we slightly generalize our settings. Namely, let E be a Banach space, $\|\cdot\|$ denote its norm and $I \in C^1(E, \mathbf{R})$ satisfy the Palais–Smale condition. Moreover, I satisfies the following conditions:

$$(I_0) I(-u) = I(u) for all u \in E.$$

There exist ρ and $\alpha > 0$ such that

(I₁)
$$I(u) \ge \alpha \quad \text{for all } u \in E \text{ with } ||u|| = \rho,$$
$$I(u) \ge 0 \quad \text{for all } u \in E \text{ with } ||u|| \le \rho.$$

(I₂) For each
$$n \in \mathbf{N}$$
 there exists an $h_n \in C(S^{n-1}, E)$ such that
 $h_n(-\sigma) = -h_n(\sigma), \quad I(h_n(\sigma)) < 0 \quad \text{for all } \sigma \in S^{n-1},$
where $S^{n-1} := \{x \in \mathbf{R}^n : |x| = 1\}.$

Next, we define minimax values of I: For each $n \in \mathbf{N}$, we set

(7.1.1)
$$c_n := \inf_{\gamma \in \tilde{\Gamma}_n} \max_{\sigma \in D_n} I(\gamma(\sigma)),$$

where $D_n := \{x \in \mathbf{R}^n : |x| \le 1\}$ and

$$\tilde{\Gamma}_n := \{ \gamma \in C(D_n, E) : \gamma(-\sigma) = -\gamma(\sigma) \text{ for all } \sigma \in D_n, \ \gamma(\sigma) = \gamma_n(\sigma) \text{ for all } \sigma \in S^{n-1} \}.$$

Note that the auxiliary functionals in Chapters 5 and 6 satisfy $(I_0)-(I_2)$. Moreover, the minimax values defined in Chapters 5 and 6 correspond to c_n in (7.1.1).

Now we will prove the following theorem:

Theorem 7.1.1. As $n \to \infty$, $c_n \to \infty$.

7.2 Proof of Theorem 7.1.1

We will prove Theorem 7.1.1 using similar arguments in Rabinowitz [92] and need some preparations.

Definition 7.2.1. For each $n, j \in \mathbf{N}$, we define $\mathcal{E}, \mathcal{E}_{\mathbf{R}^n}$ and Γ_j as follows:

$$\mathcal{E} := \{ A \subset E \setminus \{0\} : A \text{ is closed}, -A = A \},$$

$$\mathcal{E}_{\mathbf{R}^n} := \{ B \subset \mathbf{R}^n \setminus \{0\} : B \text{ is closed}, -B = B \},$$

$$\Gamma_j := \{ h(\overline{D_n \setminus Y}) : n \ge j, \ h \in \tilde{\Gamma}_n, \ Y \in \mathcal{E}_{\mathbf{R}^n}, \ \text{genus}\,(Y) \le n - j \}.$$

Here genus (Y) is Krasnoselski's genus.

First, we state properties of Γ_j .

Proposition 7.2.2. The following hold:

- (i) For all $j \in \mathbf{N}$, $\Gamma_j \neq \emptyset$.
- (ii) If $\varphi \in C(E, E)$ satisfies $\varphi(-u) = -\varphi(u)$ for all $u \in E$ and $\varphi = \mathrm{id}_E$ on $h_n(S^{n-1})$ for all $n \in \mathbf{N}$, then $\varphi : \Gamma_j \to \Gamma_j$ for each $j \in \mathbf{N}$.
- (iii) If $B \in \Gamma_j$, $Z \in \mathcal{E}$ and genus $(Z) \leq s < j$, then $\overline{B \setminus Z} \in \Gamma_{j-s}$.

Proof. (i) For each $n \in \mathbf{N}$, we set $\tilde{h}_n(\sigma) := |\sigma|h_n(\sigma/|\sigma|)$. Then it is easily seen $\tilde{h}_n \in \tilde{\Gamma}_n$, which implies $\Gamma_j \neq \emptyset$ for all $j \in \mathbf{N}$.

(ii) Let $B \in \Gamma_j$, $B = h(\overline{D_n \setminus Y})$, $h \in \tilde{\Gamma}_n$, $n \ge j$, $Y \in \mathcal{E}_{\mathbf{R}^n}$, genus $(Y) \le n - j$. Then the composite map $\varphi \circ h : D_n \to E$ is odd and continuous. Furthermore it holds that $\varphi \circ h = h_n$ on S^{n-1} . Thus $\varphi \circ h \in \tilde{\Gamma}_n$ and $\varphi(B) = (\varphi \circ h)(\overline{D_n \setminus Y}) \in \Gamma_j$.

(iii) Let $B = h(\overline{D_n \setminus Y}) \in \Gamma_j, Z \in \mathcal{E}$, genus $(Z) \leq s < j$. If the following equality

(7.2.1)
$$\overline{B\backslash Z} = h(\overline{D_n\backslash (Y \cup h^{-1}(Z))})$$

holds, then we can prove (iii). Indeed, we assume (7.2.1) and prove (iii). By $h^{-1}(Z) \in \mathcal{E}_{\mathbf{R}^n}$ and $Y \in \mathcal{E}_{\mathbf{R}^n}$, we see $Y \cup h^{-1}(Z) \in \mathcal{E}_{\mathbf{R}^n}$. By the mapping property and subadditivity of the genus, it follows that

(7.2.2)
$$genus (Y \cup h^{-1}(Z)) \le genus (Y) + genus (h^{-1}(Z)) \le genus (Y) + genus (Z) \le m - j + s = m - (j - s).$$

By (7.2.1) and (7.2.2), we can conclude $\overline{B\setminus Z} \in \Gamma_{j-s}$.

Now we prove (7.2.1). Let $b \in h(D_n \setminus (Y \cup h^{-1}(Z)))$. Then, we have

$$b \in h(D_n \backslash Y) \backslash Z \subset B \backslash Z \subset \overline{B \backslash Z}.$$

Hence $\overline{h(D_n \setminus (Y \cup h^{-1}(Z)))} \subset \overline{B \setminus Z}$.

On the other hand, let $b \in B \setminus Z$ and $w \in \overline{D_n \setminus Y}$ satisfy b = h(w). Since $b \notin Z$, $w \notin h^{-1}(Z)$ holds. Thus we have

$$w \in \left(\overline{D_n \setminus Y}\right) \setminus h^{-1}(Z) \subset \overline{D_n \setminus (Y \cup h^{-1}(Z))},$$

which implies $B \setminus Z \subset h(D_n \setminus (Y \cup h^{-1}(Z)))$. Therefore we can infer

$$\overline{B\backslash Z} \subset h(\overline{D_n\backslash (Y\cup h^{-1}(Z))})$$

This completes the proof.

Next we define another minimax values of I.

Definition 7.2.3. For each $j \in \mathbf{N}$, we define d_j by

$$d_j := \inf_{B \in \Gamma_j} \max_{u \in B} I(u).$$

Note that since genus $(\emptyset) = 0$, it is easily seen that $d_j \leq c_j$ for all $j \in \mathbf{N}$. Thus in order to prove Theorem 7.1.1, it is sufficient to show $d_j \to \infty$ as $j \to \infty$.

The following proposition shows that d_j is well-defined and $0 < \alpha \leq d_j$ for all $j \in \mathbf{N}$. Here $\alpha > 0$ appears in (I_1) .

Proposition 7.2.4. For each $j \in \mathbb{N}$ and $B \in \Gamma_j$, $B \cap \partial B_{\rho} \neq \emptyset$ holds.

Proof. Let $B = h(\overline{D_n \setminus Y})$, $n \ge j$, genus $(Y) \le n-j$. Set $\hat{\mathcal{O}} := \{x \in \text{Int}(D_n) : h(x) \in B_\rho\}$, then $0 \in \hat{\mathcal{O}}, -\hat{\mathcal{O}} = \hat{\mathcal{O}}$ and $\hat{\mathcal{O}}$ is open. Let \mathcal{O} denote a component of $\hat{\mathcal{O}}$ containing 0, then we have genus $(\partial \mathcal{O}) = n$. We claim

(7.2.3)
$$h(\partial \mathcal{O}) \subset \partial B_{\rho}.$$

Assuming (7.2.3) for the moment, set $W = \{x \in D_n : h(x) \in \partial B_\rho\}$. By (7.2.3), we can see $\partial \mathcal{O} \subset W$, which implies genus (W) = n. Therefore we have

genus
$$(W \setminus Y) \ge n - (n - j) = j.$$

and

genus
$$(h(\overline{W\setminus Y})) \ge \text{genus}(\overline{W\setminus Y}) \ge j > 0.$$

Thus it follows that $h(\overline{W\setminus Y}) \neq \emptyset$. On the other hand, since $h(\overline{W\setminus Y}) \subset B \cap \partial B_{\rho}$, it holds that $B \cap \partial B_{\rho} \neq \emptyset$.

Now we prove (7.2.3). By the property of h_n , it holds that I(u) < 0 for all $u \in h_n(\partial D_n)$. On the other hand, by (I_1) , we have $I(u) \ge 0$ for all $u \in \overline{B_\rho}$. Therefore we have $h_n(\partial D_n) \subset (\overline{B_\rho})^c$. Let $x \in \partial \mathcal{O}$. If $x \in D_n$ and $h(x) \in B_\rho$, then there exists an $r_0 > 0$ such that $h(y) \in B_\rho$ for all $y \in D_n$ satisfying $|y - x| < r_0$. However, this contradicts to $x \in \partial \mathcal{O}$. Therefore $x \in \partial D_n$ when $h(x) \in B_\rho$. In this case, since $h = h_n$ on ∂D_n and $h_n(\partial D_n) \cap B_\rho = \emptyset$, this is also a contradiction. This implies $h(\partial \mathcal{O}) \subset \partial B_\rho$.

Next, we will show that d_j is a critical value of I and estimate genus (K_{d_j}) where

$$K_c := \{ u \in E : I(u) = c, \ I'(u) = 0 \}.$$

Proposition 7.2.5.

(i) For each $j \in \mathbf{N}$, d_j is a critical value of I.

(ii) If there exist $p_1 < p_2 < ... < p_k$ such that $d := d_j = d_{j+p_1} = \cdots = d_{j+p_k}$, then genus $(K_d) \ge p_k + 1$.

Proof. (i) Since I satisfies the (PS) condition, it is standard to show that d_j is a critical value of I and we omit its proof.

(ii) First we note that $K_d \in \mathcal{E}$. Indeed, since I(0) = 0 and $d \ge \alpha > 0$ hold, we have $0 \notin K_d$, which implies $K_d \in \mathcal{E}$. Furthermore, K_d is compact since I satisfies the (PS) condition.

We shall prove indirectly and assume genus $(K_d) \leq p_k$. Then there exists a $\delta > 0$ such that genus $(N_{\delta}(K_d)) \leq p_k$ where $N_{\delta}(K_d) := K_d + B_{\delta}(0)$. By applying the Deformation Theorem, there exist $\varepsilon \in (0, \alpha/2)$ and $\eta \in C([0, 1] \times E, E)$ such that

(7.2.4) $\eta(1, -u) = -\eta(1, u) \text{ for all } u \in E \text{ and } \eta(1, A_{d+\varepsilon} \setminus \mathcal{O}) \subset A_{d-\varepsilon},$

where

$$\mathcal{O} := N_{\delta}(K_d), \quad A_c := \{ u \in E : I(u) \le c \}.$$

Here we choose a $B \in \Gamma_{j+p_k}$ satisfying $\max_{u \in B} I(u) < d - \varepsilon$. Then by Proposition 7.2.2 (iii), we obtain $\overline{B \setminus \mathcal{O}} \in \Gamma_j$. On the other hand, for each $n \in \mathbb{N}$ and $u \in h_n(\partial D_n)$, I(u) < 0 holds, which implies $\eta(1, \cdot) = \text{id on } h_n(\partial D_n)$. By Proposition 7.2.2 (ii), it follows that $\eta(1, \overline{B \setminus \mathcal{O}}) \in \Gamma_j$. However, by (7.2.4), we obtain

$$\max_{u \in \eta(1, \overline{B \setminus \mathcal{O}})} I(u) < d - \varepsilon,$$

which contradicts to the definition of d_j . Therefore genus $(K_d) \ge p_k + 1$ holds.

In the following proposition, we will prove that $d_j \to \infty$ as $j \to \infty$, which completes a proof of Theorem 7.1.1.

Proposition 7.2.6. As $j \to \infty$, $d_j \to \infty$.

Proof. We prove indirectly and assume that (d_j) is bounded. We may assume that there exists a subsequence (d_{j_k}) such that $d_{j_k} \to d_0$.

If there exists a sequence (k_m) such that $k_m \to \infty$ and $d_0 = d_{j_{k_1}} = \cdots = d_{j_{k_m}} = \cdots$, then we obtain genus $(K_{d_0}) = \infty$ by Proposition 7.2.5 (ii). However, since K_{d_0} is compact by the (PS) condition, genus $(K_{d_0}) < \infty$ holds. This is a contradiction.

Therefore we may suppose $d_0 \neq d_{j_k}$ for all $k \in \mathbf{N}$. For each $\xi > 0$, we set

$$K_{\xi} := \{ u \in E : d_0 - \xi \le I(u) \le d_0 + \xi, \ I'(u) = 0 \}.$$

By the (PS) condition, \tilde{K}_{ξ} is compact. We choose a $\delta > 0$ such that

$$s = \operatorname{genus}(K_{\xi}) = \operatorname{genus}(N_{\delta}(K_{\xi})), \quad \frac{\alpha}{2} < d_0 - \xi$$

Since d_{j_k} is a critical value of I by Proposition 7.2.5 (i), $K_{\xi} \neq \emptyset$ for each $\xi > 0$. This implies $s \ge 1$. Set $\bar{\varepsilon} := d_0 - \xi/2$ and $\mathcal{O} := N_{\delta}(K_{\xi})$. Applying the Deformation Theorem, we obtain $\varepsilon \in (0, \bar{\varepsilon})$ and $\eta \in C([0, 1] \times E, E)$ such that $\eta(1, A_{d_0+\varepsilon} \setminus \mathcal{O}) \subset A_{d_0-\varepsilon}$.

Next we take j_k, j_ℓ and $B \in \Gamma_{j_k+j_\ell}$ satisfying

$$s < j_{\ell}, \quad d_{j_k} \in (d_0 - \varepsilon, d_0 + \varepsilon), \quad \max_{u \in B} I(u) < d_0 + \varepsilon.$$

Since $s < j_{\ell}$, we have $\overline{B \setminus \mathcal{O}} \in \Gamma_{j_k}$ by Proposition 7.2.2 (iii). Furthermore, for each $n \in \mathbf{N}$, it follows that

$$I(u) \le 0$$
 on $h_n(\partial D_n)$, $0 < \frac{\alpha}{2} < d_0 - \bar{\varepsilon} < d_{j_k}$,

which implies $\eta(1, \cdot) = \text{id on } h_n(\partial D_n)$. Therefore $\eta(1, \overline{B \setminus \mathcal{O}}) \in \Gamma_{j_k}$.

On the other hand, it follow from the property of η that

$$d_{j_k} \le \max_{u \in \eta(1, \overline{B \setminus O})} I(u) \le d_0 - \varepsilon < d_{j_k}.$$

This is a contradiction and we obtain $d_j \to \infty$ as $j \to \infty$.

Proof of Theorem 7.1.1. As we note in the above, it holds $d_j \leq c_j$ for each $j \in \mathbf{N}$. By Proposition 7.2.6, we can see that $c_j \to \infty$ as $j \to \infty$, which completes a proof. \Box

Bibliography

- S. Adachi and K. Tanaka, Trudinger type inequalities in R^N and their best exponents, Proc. Amer. Math. Soc. 128 (2000), no. 7, 2051–2057.
- [2] A. Ambrosetti, M. Badiale, and S. Cingolani, Semiclassical states of nonlinear Schrödinger equations. Arch. Rational Mech. Anal. 140 (1997), no. 3, 285–300.
- [3] A. Ambrosetti and E. Colorado, Bound and ground states of coupled nonlinear Schrödinger equations. C. R. Math. Acad. Sci. Paris **342** (2006), no. 7, 453–458.
- [4] A. Ambrosetti and E. Colorado, Standing waves of some coupled nonlinear Schrödinger equations. J. Lond. Math. Soc. (2) **75** (2007), no. 1, 67–82.
- [5] A. Ambrosetti and A. Malchiodi, Nonlinear analysis and semilinear elliptic problems. Cambridge Studies in Advanced Mathematics, 104. Cambridge University Press, Cambridge, 2007.
- [6] A. Ambrosetti and P.H. Rabinowitz, Dual variational methods in critical point theory and applications. J. Functional Analysis 14 (1973), 349–381.
- [7] A. Azzollini and A. Pomponio, On the Schrödinger equation in \mathbb{R}^N under the effect of a general nonlinear term. *Indiana Univ. Math. J.* **58** (2009), no. 3, 1361–1378.
- [8] A. Bahri and P.-L. Lions, On the existence of a positive solution of semilinear elliptic equations in unbounded domains. Ann. Inst. H. Poincaré Anal. Non Linéaire 14 (1997), no. 3, 365–413.
- [9] W. Bao, Ground states and dynamics of multicomponent Bose–Einstein condensates. Multiscale Model. Simul. 2 (2004), no. 2, 210–236
- [10] T. Bartsch, N. Dancer, and Z.-Q. Wang, A Liouville theorem, a-priori bounds, and bifurcating branches of positive solutions for a nonlinear elliptic system. *Calc. Var. Partial Differential Equations* **37** (2010), no. 3-4, 345–361.
- [11] T. Bartsch and Z.-Q. Wang, Existence and multiplicity results for some superlinear elliptic problems on R^N. Comm. Partial Differential Equations 20 (1995), no. 9-10, 1725–1741.
- [12] T. Bartsch and Z.-Q. Wang, Note on ground states of nonlinear Schrödinger systems. J. Partial Differential Equations 19 (2006), no. 3, 200–207.

- [13] T. Bartsch, Z.-Q. Wang and J. Wei, Bound states for a coupled Schrödinger system. J. Fixed Point Theory Appl. 2 (2007), no. 2, 353–367.
- [14] H. Berestycki, T. Gallouët and O. Kavian, Équations de champs scalaires euclidiens non linéaires dans le plan. C. R. Acad. Sci. Paris Sér. I Math. 297 (1983), no. 5, 307–310.
- [15] H. Berestycki and P.-L. Lions, Nonlinear scalar field equations. I. Existence of a ground state. Arch. Rational Mech. Anal. 82 (1983), no. 4, 313–345.
- [16] H. Berestycki and P.-L. Lions, Nonlinear scalar field equations. II. Existence of infinitely many solutions. Arch. Rational Mech. Anal. 82 (1983), no. 4, 347–375.
- [17] H. Berestycki and P.-L. Lions, Existence d'états multiples dans des équations de champs scalaires non linéaires dans le cas de masse nulle. C. R. Acad. Sci. Paris Sér. I Math. 297 (1983), no. 4, 267–270.
- [18] H. Brezis and E.H. Lieb, Minimum action solutions of some vector field equations. Comm. Math. Phys. 96 (1984), no. 1, 97–113.
- [19] E. Brüning, A note on solutions of two-dimensional semilinear elliptic vector-field equations with strong nonlinearity. New methods and results in nonlinear field equations (Bielefeld, 1987), Lecture Notes in Phys. 347, 37–57, Springer, Berlin, 1989.
- [20] J. Busca and B. Sirakov, Symmetry results for semilinear elliptic systems in the whole space. J. Differential Equations 163 (2000), no. 1, 41–56.
- [21] J. Byeon, Singularly perturbed nonlinear Dirichlet problems with a general nonlinearity. Trans. Amer. Math. Soc. 362 (2010), no. 4, 1981–2001.
- [22] J. Byeon and L. Jeanjean, Standing waves for nonlinear Schrödinger equations with a general nonlinearity. Arch. Ration. Mech. Anal. 185 (2007), no. 2, 185–200 and Arch. Ration. Mech. Anal. 190 (2008), no. 3, 549–551.
- [23] J. Byeon and L. Jeanjean, Multi-peak standing waves for nonlinear Schrödinger equations with a general nonlinearity. *Discrete Contin. Dyn. Syst.* 19 (2007), no. 2, 255–269.
- [24] J. Byeon, L. Jeanjean and K. Tanaka, Standing waves for nonlinear Schrödinger equations with a general nonlinearity: one and two dimensional cases. *Comm. Partial Differential Equations* **33** (2008), no. 4-6, 1113–1136.
- [25] J. Byeon and Y. Oshita, Uniqueness of standing waves for nonlinear Schrödinger equations. Proc. Roy. Soc. Edinburgh Sect. A 138 (2008), no. 5, 975–987.
- [26] J. Chang and Z. Liu, Ground states of nonlinear Schrödinger systems. Proc. Amer. Math. Soc. 138 (2010), no. 2, 687–693.

- [27] S.-L. Chang, C.-S. Chien and B.-W. Jeng, Computing wave functions of nonlinear Schrödinger equations: a time-independent approach. J. Comput. Phys. 226 (2007), no. 1, 104–130.
- [28] S. Coleman, V. Glaser and A. Martin, Action minima among solutions to a class of Euclidean scalar field equations. *Comm. Math. Phys.* 58 (1978), no. 2, 211–221.
- [29] E.N. Dancer and J. Wei, Spike solutions in coupled nonlinear Schrödinger equations with attractive interaction. *Trans. Amer. Math. Soc.* **361** (2009), no. 3, 1189–1208.
- [30] E.N. Dancer, J. Wei and T. Weth, A priori bounds versus multiple existence of positive solutions for a nonlinear Schrödinger system. Ann. Inst. H. Poincaré Anal. Non Linéaire 27 (2010), no. 3, 953–969.
- [31] D.G. de Figueiredo and O. Lopes, Solitary waves for some nonlinear Schrödinger systems. Ann. Inst. H. Poincaré Anal. Non Linéaire 25 (2008), no. 1, 149–161.
- [32] M. del Pino and P.L. Felmer, Local mountain passes for semilinear elliptic problems in unbounded domains. *Calc. Var. Partial Differential Equations* 4 (1996), no. 2, 121–137.
- [33] M. del Pino and P.L. Felmer, Semi-classical states of nonlinear Schrödinger equations: a variational reduction method. *Math. Ann.* **324** (2002), no. 1, 1–32.
- [34] M. del Pino, M. Kowalczyk and J. Wei, Concentration on curves for nonlinear Schrödinger equations. Comm. Pure Appl. Math. 60 (2007), no. 1, 113–146.
- [35] I. Ekeland, Nonconvex minimization problems. Bull. Amer. Math. Soc. (N.S.) 1 (1979), no. 3, 443–474.
- [36] B.D. Ersy, C.H. Greene, J.P. Burke Jr. and J.L. Bohn, Hartree-Fock theory for double condensates. *Phys. Rev. Lett.* 78 (1997), 3594–3597.
- [37] L. Fanelli and E. Montefusco, On the blow-up threshold for weakly coupled nonlinear Schrödinger equations. J. Phys. A 40 (2007), no. 47, 14139–14150.
- [38] A. Floer and A. Weinstein, Nonspreading wave packets for the cubic Schrödinger equation with a bounded potential. J. Funct. Anal. 69 (1986), no. 3, 397–408.
- [39] P.K. Ghosh, Exact results on the dynamics of a multicomponent Bose-Einstein condensate. *Phys. Rev. A* 65 (2002), 053601.
- [40] B. Gidas, Symmetry properties and isolated singularities of positive solutions of nonlinear elliptic equations. Nonlinear partial differential equations in engineering and applied science (Proc. Conf., Univ. Rhode Island, Kingston, R.I., 1979), pp. 255–273, Lecture Notes in Pure and Appl. Math., 54, Dekker, New York, 1980.
- [41] B. Gidas and J. Spruck, Global and local behavior of positive solutions of nonlinear elliptic equations. Comm. Pure Appl. Math. 34 (1981), no. 4, 525–598.

- [42] C. Gui, Existence of multi-bump solutions for nonlinear Schrödinger equations via variational method. Comm. Partial Differential Equations 21 (1996), no. 5-6, 787–820.
- [43] H. Hajaiej, Symmetric ground state solutions of *m*-coupled nonlinear Schrödinger equations. Nonlinear Anal. 71 (2009), no. 10, 4696–4704.
- [44] D.S. Hall, M.R. Matthews, J.R. Ensher, C.E. Wieman and E.A. Cornell, Dynamics of Component Separation in a Binary Mixture of Bose-Einstein Condensates. *Phys. Rev. Lett.* 81 (1998), 1539–1542.
- [45] T.T. Hioe, Solitary waves for N coupled nonlinear Schrödinger equations. Phys. Rev. Lett. 82 (1999), 1152-1155.
- [46] N. Hirano, Multiple existence of nonradial positive solutions for a coupled nonlinear Schrödinger system. NoDEA Nonlinear Differential Equations Appl. 16 (2009), no. 2, 159–188.
- [47] N. Hirano and N. Shioji, Multiple existence of solutions for coupled nonlinear Schrödinger equations. Nonlinear Anal. 68 (2008), no. 12, 3845–3859.
- [48] J. Hirata, N. Ikoma and K. Tanaka, Nonlinear scalar field equations in R^N: mountain pass and symmetric mountain pass approaches. *Topol. Methods Nonlinear Anal.* 35 (2010), no. 2, 253–276.
- [49] N. Ikoma, Uniqueness of positive solutions for a nonlinear elliptic system. NoDEA Nonlinear Differential Equations Appl. 16 (2009), no. 5, 555–567.
- [50] N. Ikoma, Existence of standing waves for coupled nonlinear Schrödinger equations. Tokyo J. Math. 33 (2010), 89-116.
- [51] N. Ikoma and K. Tanaka, A local mountain pass type result for a system of nonlinear Schrödinger equations. *Calc. Var. Partial Differential Equations*, online fist.
- [52] N. Ikoma, On radial solutions of inhomogeneous nonlinear scalar field equations. preprint.
- [53] L. Jeanjean, Existence of solutions with prescribed norm for semilinear elliptic equations. *Nonlinear Anal.* 28 (1997), no. 10, 1633–1659.
- [54] L. Jeanjean, On the existence of bounded Palais-Smale sequences and application to a Landesman-Lazer-type problem set on R^N. Proc. Roy. Soc. Edinburgh Sect. A 129 (1999), no. 4, 787–809.
- [55] L. Jeanjean and K. Tanaka, A remark on least energy solutions in R^N. Proc. Amer. Math. Soc. 131 (2003), no. 8, 2399–2408.
- [56] L. Jeanjean and K. Tanaka, Singularly perturbed elliptic problems with superlinear or asymptotically linear nonlinearities. *Calc. Var. Partial Differential Equations* 21 (2004), no. 3, 287–318.

- [57] Y. Kabeya and K. Tanaka, Uniqueness of positive radial solutions of semilinear elliptic equations in \mathbb{R}^N and Séré's non-degeneracy condition. Comm. Partial Differential Equations 24 (1999), no. 3-4, 563–598.
- [58] Y.-C. Kuo, W.-W. Lin, S.-F. Shieh and W. Wang, A minimal energy tracking method for non-radially symmetric solutions of coupled nonlinear Schrödinger equations. J. Comput. Phys. 228 (2009), no. 21, 7941–7956.
- [59] Y.-C. Kuo, W.-W. Lin, S.-F. Shieh and W. Wang, A hyperplane-constrained continuation method for near singularity in coupled nonlinear Schrödinger equations. *Appl. Numer. Math.* **60** (2010), no. 5, 513–526.
- [60] M.K. Kwong, Uniqueness of positive solutions of $\Delta u u + u^p = 0$ in \mathbb{R}^n . Arch. Rational Mech. Anal. 105 (1989), no. 3, 243–266.
- [61] C. Li and Y.Y. Li, Nonautonomous nonlinear scalar field equations in \mathbb{R}^2 . J. Differential Equations 103 (1993), no. 2, 421–436.
- [62] Y.Y. Li, Nonautonomous nonlinear scalar field equations. Indiana Univ. Math. J. 39 (1990), no. 2, 283–301.
- [63] Y.Y. Li, On a singularly perturbed elliptic equation. Adv. Differential Equations 2 (1997), no. 6, 955–980.
- [64] E.H. Lieb and M. Loss, Analysis. Second edition. Graduate Studies in Mathematics 14. American Mathematical Society, Providence, RI, 2001.
- [65] T.-C. Lin and J. Wei, Ground state of N coupled nonlinear Schrödinger equations in \mathbb{R}^n , $n \leq 3$. Comm. Math. Phys. **255** (2005), no. 3, 629–653 and Comm. Math. Phys. **277** (2008), no. 2, 573–576.
- [66] T.-C. Lin and J. Wei, Spikes in two coupled nonlinear Schrödinger equations. Ann. Inst. H. Poincaré Anal. Non Linéaire 22 (2005), no. 4, 403–439.
- [67] T.-C. Lin and J. Wei, Solitary and self-similar solutions of two-component system of nonlinear Schrödinger equations. *Phys. D* 220 (2006), no. 2, 99–115.
- [68] T.-C. Lin and J. Wei, Spikes in two-component systems of nonlinear Schrödinger equations with trapping potentials. J. Differential Equations 229 (2006), no. 2, 538–569.
- [69] T.-C. Lin and J. Wei, Symbiotic bright solitary wave solutions of coupled nonlinear Schrödinger equations. *Nonlinearity* 19 (2006), no. 12, 2755–2773.
- [70] T.-C. Lin and J. Wei, Half-skyrmions and spike-vortex solutions of two-component nonlinear Schrödinger systems. J. Math. Phys. 48 (2007), no. 5, 053518.
- [71] T.-C. Lin and J. Wei, Orbital stability of bound states of semiclassical nonlinear Schrödinger equations with critical nonlinearity. SIAM J. Math. Anal. 40 (2008), no. 1, 365–381.

- [72] P.-L. Lions, The concentration-compactness principle in the calculus of variations. The locally compact case. I. Ann. Inst. H. Poincaré Anal. Non Linéaire 1 (1984), no. 2, 109–145.
- [73] Z. Liu, Phase separation of two-component Bose-Einstein condensates. J. Math. Phys. 50 (2009), no. 10, 102104.
- [74] Z. Liu and Z.-Q. Wang, Ground states and bound states of a nonlinear Schrödinger system. Adv. Nonlinear Stud. 10 (2010), no. 1, 175–193.
- [75] L.A. Maia, E. Montefusco and B. Pellacci, Positive solutions for a weakly coupled nonlinear Schrödinger system. J. Differential Equations 229 (2006), no. 2, 743–767.
- [76] L.A. Maia, E. Montefusco and B. Pellacci, Infinitely many nodal solutions for a weakly coupled nonlinear Schrödinger system. *Commun. Contemp. Math.* 10 (2008), no. 5, 651–669.
- [77] L.A. Maia, E. Montefusco and B. Pellacci, Orbital stability property for coupled nonlinear Schrödinger equations. Adv. Nonlinear Stud. 10 (2010), no. 3, 681–705.
- [78] J. Mawhin and M. Willem, Critical point theory and Hamiltonian systems. Applied Mathematical Sciences 74. Springer-Verlag, New York, 1989.
- [79] E. Montefusco, B. Pellacci and M. Squassina, Semiclassical states for weakly coupled nonlinear Schrödinger systems. J. Eur. Math. Soc. (JEMS) 10 (2008), no. 1, 47–71.
- [80] E. Montefusco, B. Pellacci and M. Squassina, Soliton dynamics for CNLS systems with potentials. Asymptot. Anal. 66 (2010), no. 2, 61–86.
- [81] Z.H. Musslimani and J. Yang, Transverse instability of strongly coupled dark-bright Manakov vector solitons. Optics Letters 26 (2001), 1981–1983.
- [82] B. Noris and M. Ramos, Existence and bounds of positive solutions for a nonlinear Schrödinger system. Proc. Amer. Math. Soc. 138 (2010), no. 5, 1681–1692.
- [83] G. K. Newboult, D.F. Parker and T.R. Faulkner, Coupled nonlinear Schrödinger equations arising in the study of monomode step-index optical fibers. J. Math. Phys. 30 (1989), no. 4, 930–936.
- [84] T. Ogawa, A proof of Trudinger's inequality and its application to nonlinear Schrödinger equations. Nonlinear Anal. 14 (1990), no. 9, 765–769.
- [85] M. Ohta, Stability of solitary waves for coupled nonlinear Schrödinger equations. Nonlinear Anal. 26 (1996), no. 5, 933–939.
- [86] A. Pastor, Orbital stability of periodic travelling waves for coupled nonlinear Schrödinger equations. *Electron. J. Differential Equations* 2010, No. 07, 1–19.
- [87] A. Pomponio, Coupled nonlinear Schrödinger systems with potentials. J. Differential Equations 227 (2006), no. 1, 258–281.

- [88] M.H. Protter and H.F. Weinberger, Maximum principles in differential equations. Corrected reprint of the 1967 original. Springer-Verlag, New York, 1984.
- [89] H. Pu and N.P. Bigelow, Properties of Two-Species Bose Condensates. Phys. Rev. Lett. 80 (1998), 1130–1133.
- [90] P. Quittner and P. Souplet, Superlinear parabolic problems. Blow-up, global existence and steady states. Birkhäuser Advanced Texts: Basler Lehrbücher. Birkhäuser Verlag, Basel, 2007.
- [91] P.J. Rabier, Bounded Palais-Smale sequences for functionals with a mountain pass geometry. Arch. Math. (Basel) 88 (2007), no. 2, 143–152.
- [92] P.H. Rabinowitz, Minimax methods in critical point theory with applications to differential equations. CBMS Regional Conference Series in Mathematics 65. Published for the Conference Board of the Mathematical Sciences, Washington, DC; by the American Mathematical Society, Providence, RI, 1986.
- [93] P.H. Rabinowitz, On a class of nonlinear Schrödinger equations. Z. Angew. Math. Phys. 43 (1992), no. 2, 270–291.
- [94] B. Sirakov, Least energy solitary waves for a system of nonlinear Schrödinger equations in Rⁿ. Comm. Math. Phys. 271 (2007), no. 1, 199–221.
- [95] W. A. Strauss, Existence of solitary waves in higher dimensions. Comm. Math. Phys. 55 (1977), no. 2, 149–162.
- [96] M. Struwe, The existence of surfaces of constant mean curvature with free boundaries. Acta Math. 160 (1988), no. 1-2, 19–64.
- [97] M. Struwe, Existence of periodic solutions of Hamiltonian systems on almost every energy surface. *Bol. Soc. Brasil. Mat.* (N.S.) **20** (1990), no. 2, 49–58.
- [98] M. Struwe, Variational methods. Applications to nonlinear partial differential equations and Hamiltonian systems. Fourth edition. Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge. A Series of Modern Surveys in Mathematics 34. Springer-Verlag, Berlin, 2008.
- [99] K. Tanaka, Introduction to variational problems. Iwanamishoten, 2008 (in Japanese).
- [100] S. Terracini and G. Verzini, Multipulse phases in k-mixtures of Bose-Einstein condensates. Arch. Ration. Mech. Anal. 194 (2009), no. 3, 717–741.
- [101] T. Utsumi, T. Aoki, J. Koga and M. Yamagiwa, Solutions of the 1D coupled nonlinear Schrödinger equations by the CIP-BS method. *Commun. Comput. Phys.* 1 (2006), 261–275.
- [102] M. Wadati, T. Iizuka and M. Hisakado, A Coupled Nonlinear Schrödinger Equation and Optical Solitons. J. Phys. Soc. Japan 61 (1992), no. 7, 2241–2245.

- [103] X. Wang, On concentration of positive bound states of nonlinear Schrödinger equations. Comm. Math. Phys. 153 (1993), no. 2, 229–244.
- [104] X. Wang and B. Zeng, On concentration of positive bound states of nonlinear Schrödinger equations with competing potential functions. SIAM J. Math. Anal. 28 (1997), no. 3, 633–655.
- [105] G.M. Wei, Existence and concentration of ground states of coupled nonlinear Schrödinger equations. J. Math. Anal. Appl. **332** (2007), no. 2, 846–862.
- [106] G.M. Wei, Existence and concentration of ground states of coupled nonlinear Schrödinger equations with bounded potentials. *Chin. Ann. Math. Ser. B* 29 (2008), no. 3, 247–264.
- [107] J. Wei and W. Yao, Note on uniqueness of positive solutions for some coupled nonlinear Schrödinger equations. *Methods and Applications of Analysis* (to appear).
- [108] J. Wei and T. Weth, Nonradial symmetric bound states for a system of coupled Schrödinger equations. Atti Accad. Naz. Lincei Cl. Sci. Fis. Mat. Natur. Rend. Lincei (9) Mat. Appl. 18 (2007), no. 3, 279–293.
- [109] J. Wei and T. Weth, Radial solutions and phase separation in a system of two coupled Schrödinger equations. Arch. Ration. Mech. Anal. 190 (2008), no. 1, 83– 106.
- [110] M. Willem, *Minimax theorems*. Progress in Nonlinear Differential Equations and their Applications 24. Birkhäuser Boston, Inc., Boston, MA, 1996.
- [111] V.E. Zakharov and A.B. Shabat, Exact theory of two-dimensional self-focusing and one-dimensional self-modulation of waves in nonlinear media. *Zh. Eksp. Teor. Fiz.* 61 (1971), 118–134, English translation in *J. Exp. Th. Phys.* 34 (1972), 62–69.

List of original papers

- 1. Uniqueness of positive solutions for a nonlinear elliptic system. NoDEA Nonlinear Differential Equations Appl. 16 (2009), no. 5, 555–567.
- Existence of standing waves for coupled nonlinear Schrödinger equations. Tokyo J. Math. 33 (2010), 89–116.
- (with J. Hirata and K. Tanaka) Nonlinear scalar field equations in R^N: mountain pass and symmetric mountain pass approaches. *Topol. Methods Nonlinear Anal.* 35 (2010), no. 2, 253–276.
- (with K. Tanaka) A local mountain pass type result for a system of nonlinear Schrödinger equations. *Calc. Var. Partial Differential Equations* 40 (2011), no. 3-4, 449–480.